Table of Contents

UNITED STATES SECURITIES AND EXCHANGE COMMISSION

Washington, D.C. 20549

FORM 10-Q

🖂 QUARTERLY REPORT PURSUANT TO SECTION 13 or 15 (d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the Quarterly Period Ended September 30, 2013

or

TRANSITION REPORT PURSUANT TO SECTION 13 OR 15 (d) OF THE SECURITIES EXCHANGE ACT OF 1934

For the transition period from

Commission File Number 001-03970

to



HARSCO CORPORATION

(Exact name of registrant as specified in its charter)

Delaware

(State or other jurisdiction of incorporation or organization)

350 Poplar Church Road, Camp Hill, Pennsylvania

(Address of principal executive offices)

Registrant's telephone number, including area code 717-763-7064

Indicate by check mark whether the registrant (1) has filed all reports required to be filed by Section 13 or 15 (d) of the Securities Exchange Act of 1934 during the preceding 12 months (or for such shorter period that the registrant was required to file such reports), and (2) has been subject to such filing requirements for the past 90 days. YES ⊠ NO o

Indicate by check mark whether the registrant has submitted electronically and posted on its corporate Web site, if any, every Interactive Data File required to be submitted and posted pursuant to Rule 405 of Regulation S-T (§ 232.405 of this chapter) during the preceding 12 months (or for such shorter period that the registrant was required to submit and post such files). YES 🛛 NO o

Indicate by check mark whether the registrant is a large accelerated filer, an accelerated filer, a non-accelerated filer, or a smaller reporting company. See definitions of "large accelerated filer," "accelerated filer" and "smaller reporting company" in Rule 12b-2 of the Exchange Act.

Large accelerated filer \boxtimes

Non-accelerated filer o (Do not check if a smaller reporting company)

Indicate by check mark whether the registrant is a shell company (as defined in Rule 12b-2 of the Exchange Act). YES o NO

Indicate the number of shares outstanding of each of the registrant's classes of common stock, as of the latest practicable date.

Class

Common stock, par value \$1.25 per share

Outstanding at November 1, 2013

80,666,667

17011

23-1483991

(I.R.S. employer identification number)

(Zip Code)

Accelerated filer o

Smaller reporting company o

HARSCO CORPORATION FORM 10-Q INDEX

PART I - FINANCIAL INFORMATION

<u>Item 1.</u>	Financial Statements	<u>3</u>
	Condensed Consolidated Balance Sheets (Unaudited)	<u>3</u>
	Condensed Consolidated Statements of Operations (Unaudited)	<u>4</u>
	Condensed Consolidated Statements of Comprehensive Income (Loss) (Unaudited)	<u>5</u>
	Condensed Consolidated Statements of Cash Flows (Unaudited)	<u>6</u>
	Condensed Consolidated Statements of Equity (Unaudited)	<u>7</u>
	Notes to Condensed Consolidated Financial Statements (Unaudited)	<u>8</u>
<u>Item 2.</u>	Management's Discussion and Analysis of Financial Condition and Results of Operations	<u>24</u>
<u>Item 3.</u>	Quantitative and Qualitative Disclosures About Market Risk	<u>37</u>
<u>Item 4.</u>	Controls and Procedures	<u>37</u>
PART II — OTHE	ER INFORMATION	
<u>Item 1.</u>	Legal Proceedings	<u>38</u>
<u>Item 1A.</u>	Risk Factors	<u>38</u>
<u>Item 6.</u>	Exhibits	<u>38</u>
SIGNATURES		<u>39</u>
EXHIBIT INDEX		<u>40</u>

Page

PART I - FINANCIAL INFORMATION

ITEM 1. FINANCIAL STATEMENTS

HARSCO CORPORATION

CONDENSED CONSOLIDATED BALANCE SHEETS (Unaudited)

(In thousands)	September 30 2013		December 31 2012	
ASSETS				
Current assets:				
Cash and cash equivalents	\$ 115,81	2 \$	95,250	
Trade accounts receivable, net	354,12	1	600,264	
Other receivables	29,62	7	39,836	
Inventories	176,79	2	236,512	
Assets held-for-sale	714,98	6	2,399	
Other current assets	74,23	8	92,182	
Total current assets	1,465,57	6	1,066,443	
Property, plant and equipment, net	706,61	8	1,266,225	
Goodwill	428,02	7	429,198	
Intangible assets, net	54,68	4	77,726	
Other assets	114,62	8	136,377	
Total assets	\$ 2,769,53	3\$	2,975,969	
LIABILITIES				
Current liabilities:				
Short-term borrowings	\$ 274,28	3 \$	8,560	
Current maturities of long-term debt	7,33		3,278	
Accounts payable	196,53		221,479	
Accrued compensation	53,78		94,398	
Income taxes payable	1,54		10,109	
Dividends payable	16,53		16,520	
Insurance liabilities	13,64	0	19,434	
Advances on contracts	28,10	8	47,696	
Liabilities of assets held-for-sale	218,04	6	_	
Other current liabilities	150,64		216,101	
Total current liabilities	960,45	1	637,575	
Long-term debt	787,56		957,428	
Deferred income taxes	8,33		18,880	
Insurance liabilities	42,00		63,248	
Retirement plan liabilities	314,59		385,062	
Other liabilities	51,24		52,152	
Total liabilities	2,164,20		2,114,345	
COMMITMENTS AND CONTINGENCIES			_,,0.0	
HARSCO CORPORATION STOCKHOLDERS' EQUITY				
Preferred stock	-	_	_	
Common stock	140,22	4	140,080	
Additional paid-in capital	158,41		152,645	
Accumulated other comprehensive loss	(429,66		(411,168)	
Retained earnings	1,423,20	-	1,675,490	
Treasury stock	(746,04		(745,205)	
Total Harsco Corporation stockholders' equity	546,13		811,842	
Noncontrolling interests	59,19		49,782	
Total equity	605,33		861,624	
Total liabilities and equity	\$ 2,769,53		2,975,969	
Lotal nuomato and equity	چ 2,709,33 	ب 	2,373,309	

See accompanying notes to unaudited condensed consolidated financial statements.

HARSCO CORPORATION

CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS (Unaudited)

		Three Months Ended September 30					nths Ended nber 30		
(In thousands, except per share amounts)		2013		2012		2013		2012	
Revenues from continuing operations:									
Service revenues	\$	571,595	\$	571,892	\$	1,707,658	\$	1,769,415	
Product revenues		168,450		184,891		507,518		510,278	
Total revenues		740,045		756,783		2,215,176		2,279,693	
Costs and expenses from continuing operations:			-						
Cost of services sold		450,048		453,304		1,353,749		1,406,727	
Cost of products sold		117,844		126,683		355,555		351,707	
Selling, general and administrative expenses		124,004		121,856		374,325		376,653	
Research and development expenses		3,077		2,162		7,457		6,908	
Long-lived asset impairment charge and transaction costs		253,677		_		253,677		_	
Other (income) expenses		(228)		2,383		2,158		65,351	
Total costs and expenses		948,422		706,388		2,346,921		2,207,346	
Operating income (loss) from continuing operations		(208,377)		50,395		(131,745)		72,347	
Interest income		388		1,449		1,624		3,005	
Interest expense		(12,815)		(11,609)		(37,413)		(36,041)	
Income (loss) from continuing operations before income taxes and equity income		(220,804)		40,235		(167,534)		39,311	
Income tax expense		(10,795)		(13,545)		(27,268)		(28,489)	
Equity in income of unconsolidated entities, net		434		182		1,015		479	
Income (loss) from continuing operations		(231,165)		26,872		(193,787)		11,301	
Discontinued operations:		(231,103)		20,072		(155,707)		11,501	
-		(6.40)		(52)		(2.1 <i>4</i> E)		(1 017)	
Loss on disposal of discontinued business Income tax benefit related to discontinued business		(640) 239		(52) 248		(2,145) 814		(1,217) 685	
						<u> </u>			
Income (loss) from discontinued operations		(401)		196		(1,331)		(532)	
Net income (loss)		(231,566)		27,068		(195,118)		10,769	
Less: Net income attributable to noncontrolling interests	<u>_</u>	(2,090)		(664)		(7,495)	<u>_</u>	(1,023)	
Net income (loss) attributable to Harsco Corporation	\$	(233,656)	\$	26,404	\$	(202,613)	\$	9,746	
Amounts attributable to Harsco Corporation common stockholders:									
Income (loss) from continuing operations, net of tax	\$	(233,255)	\$	26,208	\$	(201,282)	\$	10,278	
Income (loss) from discontinued operations, net of tax		(401)		196		(1,331)		(532)	
Net income (loss) attributable to Harsco Corporation common stockholders	\$	(233,656)	\$	26,404	\$	(202,613)	\$	9,746	
				00.050				00.000	
Weighted-average shares of common stock outstanding		80,775		80,659		80,747		80,623	
Basic earnings (loss) per common share attributable to Harsco Corporation co				0.22	¢	(0.40)	đ	0.10	
Continuing operations	\$	(2.89)	\$	0.32	\$	(2.49)	\$	0.13	
Discontinued operations						(0.02)		(0.01)	
Basic earnings (loss) per share attributable to Harsco Corporation common stockholders	\$	(2.89)	\$	0.33 _(a)	\$	(2.51)	\$	0.12	
Diluted weighted-average shares of common stock outstanding		80,775		80,910		80,747		80,863	
Diluted earnings (loss) per common share attributable to Harsco Corporation	comm	on stockhold	ers:						
Continuing operations	\$	(2.89)	\$	0.32	\$	(2.49)	\$	0.13	
Discontinued operations				—		(0.02)		(0.01)	
Diluted earnings (loss) per share attributable to Harsco Corporation common stockholders	\$	(2.89)	\$	0.33 _(a)	\$	(2.51)	\$	0.12	
		. ,		(u)					
Cash dividends declared per common share (a) Does not total due to rounding	\$	0.205	\$	0.205	\$	0.615	\$	0.615	
ומו בסכש הטר וטומו מתכ וט וטמותוווצ									

(a) Does not total due to rounding

See accompanying notes to unaudited condensed consolidated financial statements.

Table of Contents

HARSCO CORPORATION

CONDENSED CONSOLIDATED STATEMENTS OF COMPREHENSIVE INCOME (LOSS) (Unaudited)

	Three Months Ended September 30			
(In thousands)		2013	2012	
Net income (loss)	\$	(231,566)	\$ 27,068	
Other comprehensive income (loss):				
Foreign currency translation adjustments, net of deferred income taxes		12,678	38,812	
Net loss on cash flow hedging instruments, net of deferred income taxes of \$146 and \$242 in 2013 and 2012, respectively		(2,696)	(760)	
Pension liability adjustments, net of deferred income taxes of \$2,109 and \$1,514 in 2013 and 2012, respectively		(13,636)	(10,143)	
Unrealized gain on marketable securities, net of deferred income taxes of \$(8) and \$(1) in 2013 and 2012, respectively		13	1	
Total other comprehensive income (loss)		(3,641)	27,910	
Total comprehensive income (loss)		(235,207)	54,978	
Less: Comprehensive loss attributable to noncontrolling interests		2,968	1,265	
Comprehensive income (loss) attributable to Harsco Corporation	\$	(232,239)	\$ 56,243	
(In thousands)		Nine Mon Septen 2013	ıber 30	
(In thousands)	<u> </u>	Septen 2013	1ber 30 2012	
Net income (loss)	\$	Septen 2013	ıber 30	
Net income (loss) Other comprehensive income (loss):	\$	Septen 2013 (195,118)	aber 30 2012 \$ 10,769	
Net income (loss)	\$	Septen 2013	1ber 30 2012	
Net income (loss) Other comprehensive income (loss): Foreign currency translation adjustments, net of deferred income taxes Net gain (loss) on cash flow hedging instruments, net of deferred income taxes of \$915 and \$(653) in 2013	\$	Septen 2013 (195,118) (33,877)	2012 \$ 10,769 7,733 10,763	
Net income (loss) Other comprehensive income (loss): Foreign currency translation adjustments, net of deferred income taxes Net gain (loss) on cash flow hedging instruments, net of deferred income taxes of \$915 and \$(653) in 2013 and 2012, respectively Pension liability adjustments, net of deferred income taxes of \$(2,439) and \$189 in 2013 and 2012,	\$	Septen 2013 (195,118) (33,877) (2,160)	2012 \$ 10,769 7,733 1,941	
Net income (loss) Other comprehensive income (loss): Foreign currency translation adjustments, net of deferred income taxes Net gain (loss) on cash flow hedging instruments, net of deferred income taxes of \$915 and \$(653) in 2013 and 2012, respectively Pension liability adjustments, net of deferred income taxes of \$(2,439) and \$189 in 2013 and 2012, respectively Unrealized gain on marketable securities, net of deferred income taxes of \$(13) and \$(2) in 2013 and 2012,	\$	Septen 2013 (195,118) (33,877) (2,160) 17,587	2012 \$ 10,769 7,733 1,941 (1,294)	
Net income (loss) Other comprehensive income (loss): Foreign currency translation adjustments, net of deferred income taxes Net gain (loss) on cash flow hedging instruments, net of deferred income taxes of \$915 and \$(653) in 2013 and 2012, respectively Pension liability adjustments, net of deferred income taxes of \$(2,439) and \$189 in 2013 and 2012, respectively Unrealized gain on marketable securities, net of deferred income taxes of \$(13) and \$(2) in 2013 and 2012, respectively	\$	Septen 2013 (195,118) (33,877) (2,160) 17,587 21	2012 \$ 10,769 7,733 1,941 (1,294) 3	
Net income (loss) Other comprehensive income (loss): Foreign currency translation adjustments, net of deferred income taxes Net gain (loss) on cash flow hedging instruments, net of deferred income taxes of \$915 and \$(653) in 2013 and 2012, respectively Pension liability adjustments, net of deferred income taxes of \$(2,439) and \$189 in 2013 and 2012, respectively Unrealized gain on marketable securities, net of deferred income taxes of \$(13) and \$(2) in 2013 and 2012, respectively Total other comprehensive income (loss)	\$	Septen 2013 (195,118) (33,877) (2,160) 17,587 21 (18,429)	2012 \$ 10,769 7,733 1,941 (1,294) 3 8,383 1,983	

See accompanying notes to unaudited condensed consolidated financial statements.

HARSCO CORPORATION

CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS (Unaudited)

	Nine Month Septembe	
(In thousands)	2013	2012
Cash flows from operating activities:		
Net income (loss)	\$ (195,118) \$	10,769
Adjustments to reconcile net income (loss) to net cash provided by operating activities:		
Depreciation	178,136	188,182
Amortization	12,967	15,662
Deferred income tax expense (benefit)	3,465	(6,867
Equity in income of unconsolidated entities, net	(1,015)	(479
Dividends from unconsolidated entities		308
Harsco 2011/2012 Restructuring Program non-cash adjustment	—	21,787
Impairment of long-lived assets	241,323	
Other, net	(1,449)	(29,263
Changes in assets and liabilities:		
Accounts receivable	(21,194)	(1,247
Inventories	(10,671)	(23,298
Accounts payable	28,882	(24,719
Accrued interest payable	6,333	5,786
Accrued compensation	(5,036)	2,286
Harsco Infrastructure Segment 2010 Restructuring Program accrual	(870)	(4,755
Harsco 2011/2012 Restructuring Program accrual	(14,496)	(8,050
Other assets and liabilities	(57,170)	(34,733
Net cash provided by operating activities	164,087	111,369
Cash flows from investing activities:		
Purchases of property, plant and equipment	(181,706)	(172,770
Proceeds from sales of assets	16,947	42,889
Purchases of businesses, net of cash acquired	(2,841)	(517
Other investing activities, net	(4,985)	457
Net cash used by investing activities	(172,585)	(129,941
Cash flows from financing activities:		
Short-term borrowings, net	239	(39,302
Current maturities and long-term debt:		
Additions	284,861	249,034
Reductions	(203,677)	(151,399
Cash dividends paid on common stock	(49,652)	(49,548
Dividends paid to noncontrolling interests	(2,880)	(2,264
Contributions from noncontrolling interests	4,622	8,097
Purchase of noncontrolling interests	(166)	
Common stock issued - options	371	725
Other financing activities, net	(405)	(2,709
Net cash provided by financing activities	33,313	12,634
Effect of exchange rate changes on cash	(4,253)	(1,506
Vet increase (decrease) in cash and cash equivalents		
	20,562	(7,444
Cash and cash equivalents at beginning of period	95,250	121,184
Cash and cash equivalents at end of period	<u>\$ 115,812</u> \$	113,740

See accompanying notes to unaudited condensed consolidated financial statements.

HARSCO CORPORATION

CONDENSED CONSOLIDATED STATEMENTS OF EQUITY (Unaudited)

	Harsco Corporation Stockholders' Equity												
(In thousands, except share and per share amounts)	Common Stock Issued Treasury				Retained Earnings		Accumulated Other Comprehensive Loss		Noncontrolling Interests	Total			
Balances, January 1, 2012	\$	139,914	\$	(744,644)	9	\$ 149,066	\$	1,996,234	9	6 (364,191)	\$	43,539	\$ 1,219,918
Net income								9,746				1,023	10,769
Cash dividends declared:													
Common @ \$0.615 per share								(49,597)					(49,597)
Noncontrolling interests												(2,264)	(2,264)
Translation adjustments, net of deferred income taxes of \$(4,618)										7,479		254	7,733
Cash flow hedging instrument adjustments, net of deferred income taxes of \$(653)										1,941			1,941
Contributions from noncontrolling interests												8,602	8,602
Sale of investment in consolidated subsidiary												(704)	(704)
Pension liability adjustments, net of deferred income taxes of \$189										(1,294)			(1,294)
Marketable securities unrealized gains, net of deferred income taxes of \$(2)										3			3
Stock options exercised, 38,900 shares		49				661							710
Vesting of restricted stock units and other stock grants, net 68,558 shares		117		(561)		959							515
Amortization of unearned portion of stock- based compensation, net of forfeitures						2,257							 2,257
Balances, September 30, 2012	\$	140,080	\$	(745,205)	ŝ	\$ 152,943	\$	1,956,383	9	\$ (356,062)	\$	50,450	\$ 1,198,589

	Harsco Corporation Stockholders' Equity													
(In thousands, except share and per share amounts)		Common Stock Issued Treasury				Additional Paid- in Capital		Retained Earnings		Accumulated Other Comprehensive Loss		Noncontrolling Interests		Total
Balances, January 1, 2013	\$	140,080	\$	(745,205)		\$	152,645	\$	1,675,490		\$ (411,168)	\$	49,782	\$ 861,624
Net income (loss)									(202,613)				7,495	(195,118)
Cash dividends declared:														
Common @ \$0.615 per share									(49,668)					(49,668)
Noncontrolling interests													(2,880)	(2,880)
Translation adjustments, net of deferred income taxes of \$(1,094)											(34,029)		152	(33,877)
Cash flow hedging instrument adjustments, net of deferred income taxes of \$915											(2,076)		(84)	(2,160)
Contributions from noncontrolling interests													4,622	4,622
Purchase of subsidiary shares from noncontrolling interest							(292)						107	(185)
Pension liability adjustments, net of deferred income taxes of \$(2,439)											17,587			17,587
Marketable securities unrealized gains, net of deferred income taxes of \$(13)											21			21
Stock options exercised, net 20,000 shares		25					375							400
Vesting of restricted stock units and other stock grants, net 62,039 shares		119		(841)			2,057							1,335
Amortization of unearned portion of stock- based compensation, net of forfeitures							3,632							3,632
Balances, September 30, 2013	\$	140,224	\$	(746,046)	1	\$	158,417	\$	1,423,209	2	\$ (429,665)	\$	59,194	\$ 605,333

See accompanying notes to unaudited condensed consolidated financial statements.

HARSCO CORPORATION NOTES TO CONDENSED CONSOLIDATED FINANCIAL STATEMENTS (Unaudited)

1. Basis of Presentation

Harsco Corporation (the "Company") has prepared these unaudited condensed consolidated financial statements based on Securities and Exchange Commission rules that permit reduced disclosure for interim periods. In the opinion of management, all adjustments (all of which are of a normal recurring nature) that are necessary for a fair presentation are reflected in the unaudited condensed consolidated financial statements. The December 31, 2012 Condensed Consolidated Balance Sheet information contained in this Quarterly Report on Form 10-Q was derived from the 2012 audited consolidated financial statements, but does not include all disclosures required by accounting principles generally accepted in the United States of America ("U.S. GAAP") for an annual report. The unaudited condensed consolidated financial statements should be read in conjunction with the audited consolidated financial statements, including the notes thereto, included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012.

Operating results and cash flows for the three and nine months ended September 30, 2013 are not indicative of the results that may be expected for the year ending December 31, 2013.

During the second quarter of 2013, the Company recorded non-cash out-of-period adjustments that had the net effect of increasing after-tax income by \$3.0 million, or \$0.04 per diluted share, for the second quarter of 2013, and by \$2.6 million, or \$0.03 per diluted share, for the first half of 2013. The adjustments are primarily the result of correcting claims-paid information for certain international accrued insurance reserves which were identified during the regular, periodic evaluation of these reserves during the quarter. The Company assessed the individual and aggregate impact of these adjustments on the current year and all prior periods and determined that the cumulative effect of the adjustments was not material to the expected full-year 2013 results, and did not result in a material misstatement to any previously issued annual or quarterly financial statements. Consequently, the Company recorded the \$3.0 million net adjustment in the second quarter of 2013 and has not revised any previously issued annual financial statements or interim financial data.

2. Recently Adopted and Recently Issued Accounting Standards

The following accounting standards have been adopted in 2013:

On January 1, 2013, the Company adopted Financial Accounting Standards Board ("FASB") issued changes related to offsetting assets and liabilities. The changes require additional disclosure regarding the offsetting of assets and liabilities to enable users of financial statements to understand the effect on financial position. The adoption of these changes did not have a material impact on the Company's consolidated financial statements. The disclosures required by these changes are included in Note 12, "Derivative Instruments, Hedging Activities and Fair Value."

On January 1, 2013, the Company adopted FASB issued changes related to the presentation of reclassification adjustments out of accumulated other comprehensive income. These changes require additional disclosure related to changes in accumulated other comprehensive income by component and significant items reclassified out of accumulated other comprehensive income. Other than the additional disclosure requirements, the adoption of these changes did not have a material impact on the Company's consolidated financial statements. The disclosures required by these changes are included in Note 15, "Components of Accumulated Other Comprehensive Loss."

On July 17, 2013, the Company adopted FASB issued changes related to the inclusion of the Fed Funds Effective Swap Rate (or Overnight Index Swap Rate) as a benchmark interest rate for hedge accounting purposes. These changes also removed restrictions on using different benchmark rates for similar hedges. The adoption of these changes did not have a material impact on the Company's consolidated financial statements.

The following accounting standards have been issued and become effective for the Company at a future date:

In March 2013, the FASB issued changes related to a parent's accounting for the cumulative translation adjustment upon derecognition of certain subsidiaries or groups of assets within a foreign entity or of an investment in a foreign entity. The changes resolve diversity in practice related to these matters. The changes become effective for the Company on January 1, 2014. Management has determined that these changes will not have a material impact on the Company's consolidated financial statements.

In July 2013, the FASB issued changes related to financial statement presentation of an unrecognized tax benefit when a net operating loss carryforward, a similar tax loss, or a tax credit carryforward exists that could be used to offset the liability for an uncertain tax position. The changes resolve diversity in practice related to these matters. The changes become effective for the Company on January 1, 2014. Management has determined that these changes will not have a material impact on the Company's consolidated financial statements.

3. Acquisitions and Dispositions

On September 15, 2013, the Company entered into an agreement to sell the Company's Harsco Infrastructure Segment into a strategic venture with Clayton, Dubilier & Rice ("CD&R") as part of a transaction that will combine the Harsco Infrastructure Segment with Brand Energy & Infrastructure Services, Inc., which CD&R is simultaneously acquiring (the "Infrastructure transaction"). Under the terms of the purchase agreement, the Company will contribute substantially all of the Company's equity interests in, and the assets of, the Harsco Infrastructure Segment in exchange for \$300 million, subject to working capital and other adjustments, and an approximate 29% equity interest in the strategic venture. The Infrastructure transaction is anticipated to close during the fourth quarter of 2013, subject to certain customary conditions of closing. Upon closing of the Infrastructure transaction, the Company's 29% equity interest in the strategic venture will be accounted for under the equity method of accounting as prescribed by U.S. GAAP.

As a result of the Infrastructure transaction, the Company recorded an estimated non-cash long-lived asset impairment charge of \$241.3 million during the third quarter of 2013. The final non-cash long-lived asset impairment charge may differ, possibly materially, due to a number of factors including, but not limited to (i) the valuation of certain assets and liabilities being transferred to the strategic venture; (ii) final capital expenditure and working capital adjustments required under the purchase agreement; (iii) the recognition of cumulative translation amounts; and (iv) the finalization of current and deferred income tax amounts. The Company determined that the estimated non-cash long-lived asset impairment charge was necessary given the fair value of the consideration received in the Infrastructure transaction. The amount of the estimated non-cash long-lived asset impairment charge represents the Company's estimate of the difference between the current book value of the Harsco Infrastructure Segment, less costs to sell, and the sum of the cash consideration plus the estimated fair value of the Company's approximate 29% equity interest in the strategic venture. The estimated fair value of the equity interest was determined based on the expected future discounted cash flows of the strategic venture. See Note 1, "Summary of Significant Accounting Policies," to the consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012 for additional information on the Company's policy on impairment of long-lived assets (other than goodwill).

The Company has reflected substantially all assets and liabilities of the Harsco Infrastructure Segment as Assets held-for-sale and Liabilities of assets held-for-sale, respectively, on the Condensed Consolidated Balance Sheets at September 30, 2013. The carrying amounts of the assets and liabilities that were aggregated in Assets held-for-sale and Liabilities of assets held-for-sale as of September 30, 2013 are as follows:

(In thousands)	September 30 2013	(In thousands)	S	September 30 2013
Assets held-for-sale:		Liabilities of assets held-for-sale:		
Trade accounts receivable, net	\$ 264,310	Accounts payable	\$	66,759
Other receivables	13,206	Accrued compensation		33,334
Inventories	68,064	Income taxes payable		2,663
Other current assets	20,024	Insurance liabilities - current portion		7,587
Property, plant and equipment, net	334,967	Other current liabilities		42,872
Intangible assets, net	6,461	Insurance liabilities - non-current portion		18,465
Other assets (a)	7,954	Retirement plan liabilities		41,916
		Other liabilities		4,450
	\$ 714,986		\$	218,046

(a) Includes assets held-for-sale of the Harsco Metals & Minerals Segment totaling \$0.2 million at September 30, 2013.

Additionally, the Company incurred \$12.4 million of expenses during the three months ended September 30, 2013 in conjunction with the Infrastructure transaction.

Table of Contents

4. Accounts Receivable and Inventories

Accounts receivable consist of the following:

(In thousands)	Se	ptember 30 2013	Ι	December 31 2012
Trade accounts receivable	\$	358,191	\$	617,517
Less: Allowance for doubtful accounts		(4,070)		(17,253)
Trade accounts receivable, net	\$	354,121	\$	600,264

39,836

Other receivables (a) \$ 29,627 \$

(a) Other receivables include insurance claim receivables, employee receivables, tax claim receivables and other miscellaneous receivables not included in Trade accounts receivable, net.

The provision for doubtful accounts related to trade accounts receivable was as follows:

	Three Mo	nths Er	nded		Nine Mo	nths Ended			
	 Septen	nber 30	0	September 30					
(In thousands)	2013		2012		2013		2012		
Provision for doubtful accounts related to trade accounts receivable	\$ 1,059	\$	1,802	\$	5,897	\$	8,665		

Inventories consist of the following:

(In thousands)	Sep	September 30 2013		December 31 2012		
Finished goods	\$	21,707	\$	69,904		
Work-in-process		35,214		28,944		
Raw materials and purchased parts		83,037		99,058		
Stores and supplies		36,834		38,606		
Inventories	\$	176,792	\$	236,512		

The decreases in Accounts receivable and Inventories since December 31, 2012, primarily relates to the reclassification of substantially all the assets and liabilities of the Harsco Infrastructure Segment to Assets held-for-sale and Liabilities of assets held-for-sale, respectively, on the Condensed Consolidated Balance Sheets at September 30, 2013 in conjunction with the Infrastructure transaction. See Note 3 "Acquisitions and Dispositions," for additional information related to the Infrastructure transaction.

5. Property, Plant and Equipment

Property, plant and equipment consists of the following:

(In thousands)	5	September 30 2013		December 31 2012
Land	\$	17,414	\$	26,336
Land improvements		12,731		14,199
Buildings and improvements		175,008		190,078
Machinery and equipment		1,967,023		2,950,384
Uncompleted construction		95,564		107,633
Gross property, plant and equipment		2,267,740		3,288,630
Less: Accumulated depreciation		(1,561,122)		(2,022,405)
Property, plant and equipment, net	\$	706,618	\$	1,266,225

The decreases in Property, plant and equipment since December 31, 2012, primarily relates to the reclassification of substantially all the assets and liabilities of the Harsco Infrastructure Segment to Assets held-for-sale and Liabilities of assets held-for-sale, respectively, on the Condensed Consolidated Balance Sheets at September 30, 2013 in conjunction with the Infrastructure transaction. See Note 3 "Acquisitions and Dispositions," for additional information related to the Infrastructure transaction.

6. Goodwill and Other Intangible Assets

The following table reflects the changes in carrying amounts of goodwill by segment (there is no goodwill associated with the Harsco Infrastructure or Harsco Industrial Segments) for the nine months ended September 30, 2013:

(In thousands)	Metals & ls Segment	Harsco Rail Segment	Consolidated Totals
Balance at December 31, 2012	\$ 419,888	\$ 9,310	\$ 429,198
Foreign currency translation	(1,171)	—	(1,171)
Balance at September 30, 2013	\$ 418,717	\$ 9,310	\$ 428,027

The Company tests for goodwill impairment annually or more frequently if indicators of impairment exist or if a decision is made to dispose of a business. The Company performs its annual goodwill impairment test as of October 1 and monitors for triggering events on an ongoing basis. The Company determined that as of September 30, 2013, no interim goodwill impairment testing was necessary. There can be no assurance that the Company's annual goodwill impairment testing will not result in a charge to earnings. Any impairment could result in the write-down of the carrying value of goodwill to its implied fair value.

Intangible assets consist of the following:

	September 30, 2013					Decembe	r 31, 2012		
(In thousands)	Gross Carrying Amount		Accumulated Amortization		Gross Carrying Amount			Accumulated Amortization	
Customer related	\$	149,421	\$	109,413	\$	183,862	\$	129,904	
Non-compete agreements		1,123		1,013		1,347		1,310	
Patents		6,196		5,186		6,909		5,503	
Technology related		27,354		18,343		29,588		17,551	
Trade names		4,116		2,838		18,685		11,688	
Other		7,519		4,252		9,947		6,656	
Total	\$	195,729	\$	141,045	\$	250,338	\$	172,612	

Amortization expense for intangible assets was as follows:

	Three Months Ended				Nine Months Ended			
	 Septen			1				
(In thousands)	2013		2012		2013		2012	
Amortization expense for intangible assets	\$ 3,629	\$	4,111	\$	11,481	\$	13,518	

The estimated amortization expense for the next five fiscal years based on current intangible assets is as follows:

(In thousands)		2013		2014		2015		2016		2017	
Estimated amortization expense (a)	\$	14,750	\$	9,000	\$	8,000	\$	7,250	\$	3,500	
(a) These estimated amortization evapores amounts do not	(a) These estimated amortization supports amounts do not velocit the potential effect of future fevoirs surports archange fluctuations										

(a) These estimated amortization expense amounts do not reflect the potential effect of future foreign currency exchange fluctuations.

The decreases in Intangible assets and estimated amortization expense for the next five fiscal years since December 31, 2012, primarily relates to the reclassification of substantially all the assets and liabilities of the Harsco Infrastructure Segment to Assets held-for-sale and Liabilities of assets held-for-sale, respectively, on the Condensed Consolidated Balance Sheets at September 30, 2013 in conjunction with the Infrastructure transaction. See Note 3 "Acquisitions and Dispositions," for additional information related to the Infrastructure transaction.

7. Debt and Credit Agreements

On September 12, 2013, the Company entered into an additional amendment to the Amended and Restated Five-Year Credit Agreement (as amended, the "Credit Agreement"). In addition to certain administrative and conforming modifications, the amendment replaced the debt covenant for total consolidated debt to total consolidated capital ratio. That debt covenant has been replaced by a debt covenant for total consolidated earnings before interest, taxes, depreciation and amortization ratio, which is not to exceed 3.50 to 1.00. During the three months ended September 30, 2013, the Company expensed \$0.4 million of fees associated with the additional amendment to the Credit Agreement.

Additionally, the Company retired the \$150.0 million aggregate principal amount of 5.125% notes, due September 15, 2013, at maturity using borrowings under the Company's Credit Agreement.

8. Employee Benefit Plans

				Three Mo	ths Er	ided		
Defined Benefit Net Periodic Pension Cost		U. S.	Plans		International Plans			
(In thousands)	2013 2012 2013						2012	
Defined benefit plans:								
Service cost	\$	641	\$	472	\$	822	\$	855
Interest cost		2,942		3,209		10,512		10,996
Expected return on plan assets		(3,911)		(3,907)		(11,540)		(10,638)
Recognized prior service costs		36		47		87		95
Recognized loss		1,263		1,158		4,065		3,730
Settlement/curtailment losses		—		1,112		—		6
Defined benefit plans net periodic pension cost	\$	971	\$	2,091	\$	3,946	\$	5,044

			Nine Mont	hs Ei	nded					
	 September 30									
Defined Benefit Net Periodic Pension Cost	 U. S. Plans International Plans									
(In thousands)	2013		2012		2013	2012				
Defined benefit plans:										
Service cost	\$ 1,924	\$	1,415	\$	2,630	\$	2,781			
Interest cost	8,825		9,620		32,058		33,437			
Expected return on plan assets	(11,732)		(11,713)		(35,159)		(32,373)			
Recognized prior service costs	108		142		271		289			
Recognized loss	3,789		3,471		12,364		11,209			
Amortization of transition liability	—						8			
Settlement/curtailment (gains) losses	_		1,112		(289)		(2,055)			
Defined benefit plans net periodic pension cost	\$ 2,914	\$	4,047	\$	11,875	\$	13,296			

	Three Months Ended					Nine Months Ended				
Company Contributions	September 30 September 30						0			
(In thousands)	2013 2012				2013	2012				
Defined benefit pension plans:										
United States	\$	919	\$	3,024	\$	1,967	\$	5,662		
International		4,484		4,127		25,440		26,298		
Multiemployer pension plans		3,527		3,127		12,042		10,901		
Defined contribution pension plans		3,471		5,971		12,292		14,405		

The Company currently anticipates contributing an additional \$0.8 million and \$5.7 million for the U.S. and international defined benefit pension plans, respectively, during the remainder of 2013.

9. Income Taxes

Income tax expense from continuing operations decreased, primarily due to the estimated non-cash long-lived asset impairment charge recorded as part of the Infrastructure transaction for the three and nine months ended September 30, 2013 compared with the three and nine months ended September 30, 2012. This decrease was partially offset by valuation allowances recorded against deferred tax assets within certain foreign jurisdictions in which the Harsco Infrastructure Segment operates, and for the expected tax costs of cash repatriation from the Infrastructure transaction.

The effective income tax rate related to continuing operations for the three and nine months ended September 30, 2013 was (4.9)% and (16.3)%, respectively. The effective income tax rate related to continuing operations for the three and nine months ended September 30, 2012 was 33.7% and 72.5%, respectively. The effective income tax rate changed between 2012 and 2013 primarily due to the jurisdictional mix of the estimated non-cash long-lived asset impairment charge recorded as part of the Infrastructure transaction compared with restructuring charges recorded during 2012; tax expense recorded for valuation allowances on deferred tax assets within certain foreign jurisdictions in which the Harsco Infrastructure Segment operates; and for the expected tax costs of cash repatriation from the Infrastructure transaction.

Due to the estimated non-cash long-lived asset impairment charge, certain foreign jurisdictions' deferred tax assets for the Harsco Infrastructure Segment were no longer realizable on a more-likely-than-not basis after weighing all available positive and negative evidence. One of the significant pieces of objective negative evidence evaluated was the cumulative losses in certain foreign jurisdictions in recent years after considering the effects of the estimated non-cash long-lived asset impairment charge. Such objective evidence outweighed all other available positive evidence. This resulted in tax expense of approximately \$11 million to reduce beginning of the year net deferred tax assets for these foreign jurisdictions.

Due to the expected repatriation of proceeds from the Infrastructure transaction, the Company recorded a deferred tax liability to estimate the cost of distributing prior year earnings that were previously indefinitely reinvested. This resulted in tax expense of approximately \$8 million for the tax-costs of cash repatriation, including a reduction for any available foreign tax credits. Excluding any cash repatriation associated with the Infrastructure transaction, the Company remains indefinitely reinvested for all other undistributed earnings.

An income tax benefit from an uncertain tax position may be recognized when it is more-likely-than-not that the position will be sustained upon examination, based on technical merits, including resolutions of any related appeals or litigation processes. The unrecognized income tax benefit at September 30, 2013 was \$28.0 million, including interest and penalties. Within the next twelve months, it is reasonably possible that up to \$4.3 million of unrecognized income tax benefits will be recognized upon settlement of tax examinations and the expiration of various statutes of limitations. Of this amount, approximately \$1 million relates to uncertain tax positions for entities included in the Infrastructure transaction.

The Company was contacted by the U.S. Internal Revenue Service to audit the Company's 2010 income tax return. The U.S. Internal Revenue Service commenced its audit during the second quarter of 2013.

10. Commitments and Contingencies

Environmental

The Company is involved in a number of environmental remediation investigations and cleanups and, along with other companies, has been identified as a "potentially responsible party" for certain waste disposal sites. While each of these matters is subject to various uncertainties, it is probable that the Company will agree to make payments toward funding certain of these activities and it is possible that some of these matters will be decided unfavorably to the Company. The Company has evaluated its potential liability, and its financial exposure is dependent upon such factors as the continuing evolution of environmental laws and regulatory requirements, the availability and application of technology, the allocation of cost among potentially responsible parties, the years of remedial activity required and the remediation methods selected. The Condensed Consolidated Balance Sheets as of September 30, 2013 and December 31, 2012 include accruals in Other current liabilities of \$1.3 million and \$1.9 million, respectively, for environmental matters. The amounts charged against pre-tax income related to environmental matters totaled \$0.6 million and \$1.2 million for the three and nine months ended September 30, 2013, respectively. The amounts charged against pre-tax income related to environmental matters totaled \$0.2 million and \$0.7 million for the three and nine months ended September 30, 2012, respectively.

The Company evaluates its liability for future environmental remediation costs on a quarterly basis. Although actual costs to be incurred at identified sites in future periods may vary from the estimates (given inherent uncertainties in evaluating environmental exposures), the Company does not expect that any costs that are reasonably possible to be incurred by the Company in connection with environmental matters in excess of the amounts accrued would have a material adverse effect on the Company's financial condition, results of operations or cash flows.

Brazilian Tax Disputes

The Company is involved in a number of tax disputes with federal, state and municipal tax authorities in Brazil. These disputes are at various stages of the legal process, including the administrative review phase and the collection action phase, and include assessments of fixed amounts of principal and penalties, plus interest charges that increase at statutorily determined amounts per month and are assessed on the aggregate amount of the principal and penalties. In addition, the losing party at the collection action or court of appeals phase could be subject to a charge to cover statutorily mandated legal fees, which are generally calculated as a percentage of the total assessed amounts due, inclusive of penalty and interest. A large number of the claims relate to value-added ("ICMS") services and social security ("INSS") tax disputes. The largest proportion of the assessed amounts relate to ICMS claims filed by the State Revenue Authorities from the State of São Paulo, Brazil (the "SPRA"), encompassing the period from January 2002 to May 2005.

In October 2009, the Company received notification of the SPRA's final administrative decision regarding the levying of ICMS in the State of São Paulo in relation to services provided to a customer in the State between January 2004 and May 2005. As of September 30, 2013, the principal amount of the tax assessment from the SPRA with regard to this case was approximately \$3 million, with penalty, interest and fees assessed to date increasing such amount by an additional \$28 million. Any change in the aggregate amount since the Company's Annual Report on Form 10-K for the year ended December 31, 2012 reflects an increase in assessed interest and statutorily mandated legal fees for the period and includes the effect of foreign currency translation.

Another ICMS tax case involving the SPRA refers to the tax period from January 2002 to December 2003, and is still pending at the administrative phase, where the aggregate amount assessed by the tax authorities in August 2005 was approximately \$11 million (the amounts with regard to this claim are valued as of the date of the assessment since it has not yet reached the collection phase), composed of a principal amount of \$2.5 million, with penalty and interest assessed through that date increasing such amount by an additional \$8.5 million. All such amounts include the effect of foreign currency translation.

The Company continues to believe it is not probable that it will incur a loss for these assessments by the SPRA. The Company also continues to believe that sufficient coverage for these claims exists as a result of the Company's customer's indemnification obligations and such customer's pledge of assets in connection with the October 2009 notice, as required by Brazilian procedure.

The Company intends to continue its practice of vigorously defending itself against these tax claims under various alternatives, including judicial appeal. The Company will continue to evaluate its potential liability with regard to these claims on a quarterly basis; however, it is not possible to predict the ultimate outcome of these tax-related disputes in Brazil. No loss provision has been recorded in the Company's consolidated financial statements because the loss contingency is not deemed probable, and the Company does not expect that any costs that are reasonably possible to be incurred by the Company in connection with Brazilian tax disputes would have a material adverse effect on the Company's financial condition, results of operations or cash flows.

Customer Disputes

The Company, through its Harsco Metals & Minerals Segment, provides services to various subsidiaries and affiliates of ArcelorMittal (collectively, "ArcelorMittal") through long-term service contracts on a number of sites worldwide. Currently, ArcelorMittal and the Company are involved in several commercial disputes, a few of which have resulted in legal action. Both the Company and ArcelorMittal are working to resolve these matters. Although results of operations and cash flows for a given period could be adversely affected by a negative outcome in these or other lawsuits, claims and proceedings, management believes that the ultimate outcome of these matters will not have a material adverse effect on the Company's financial condition, results of operations or cash flows.

Other

In the United States, the Company has been named as one of many defendants (approximately 90 or more in most cases) in legal actions alleging personal injury from exposure to airborne asbestos over the past several decades. In their suits, the plaintiffs have named as defendants, among others, many manufacturers, distributors and installers of numerous types of equipment or products that allegedly contained asbestos.

The Company believes that the claims against it are without merit. The Company has never been a producer, manufacturer or processor of asbestos fibers. Any component within a Company product that may have contained asbestos would have been purchased from a supplier. Based on scientific and medical evidence, the Company believes that any asbestos exposure arising from normal use of any Company product never presented any harmful levels of airborne asbestos exposure, and, moreover, the type of asbestos contained in any component that was used in those products was protectively encapsulated in other materials and is not associated with the types of injuries alleged in the pending suits. Finally, in most of the depositions taken of plaintiffs to date in the litigation against the Company, plaintiffs have failed to specifically identify any Company products as the source of their asbestos exposure.

The majority of the asbestos complaints pending against the Company have been filed in New York. Almost all of the New York complaints contain a standard claim for damages of \$20 million or \$25 million against the approximately 90 defendants, regardless of the individual plaintiff's alleged medical condition, and without specifically identifying any Company product as the source of plaintiff's asbestos exposure.

As of September 30, 2013, there are 17,701 pending asbestos personal injury claims filed against the Company. Of these cases, 17,261 are pending in the New York Supreme Court for New York County in New York State. The other claims, totaling 440, are filed in various counties in a number of state courts, and in certain Federal District Courts (including New York), and those complaints generally assert lesser amounts of damages than the New York State court cases or do not state any amount claimed.

As of September 30, 2013, the Company has obtained dismissal by stipulation, or summary judgment prior to trial, in 27,093 cases.

In view of the persistence of asbestos litigation nationwide, the Company expects to continue to receive additional claims. However, there have been developments during the past several years, both by certain state legislatures and by certain state courts, which could favorably affect the Company's ability to defend these asbestos claims in those jurisdictions. These developments include procedural changes, docketing changes, proof of damage requirements and other changes that require plaintiffs to follow specific procedures in bringing their claims and to show proof of damages before they can proceed with their claim. An example is the action taken by the New York Supreme Court (a trial court), which is responsible for managing all asbestos cases pending within New York County in the State of New York. This Court issued an order in December 2002 that created a Deferred or Inactive Docket for all pending and future asbestos claims filed by plaintiffs who cannot demonstrate that they have a malignant condition or discernible physical impairment, and an Active or In Extremis Docket for plaintiffs who are able to show such medical condition. As a result of this order, the majority of the asbestos cases filed against the Company in New York County have been moved to the Inactive Docket until such time as the plaintiffs can show that they have incurred a physical impairment. As of September 30, 2013, the Company has been listed as a defendant in 288 Active or In Extremis asbestos cases in New York County. The Court's Order has been challenged by some plaintiffs.

Except with regard to the legal costs in a few limited, exceptional cases, the Company's insurance carrier has paid all legal and settlement costs and expenses to date related to the Company's U.S. asbestos cases. The Company has liability insurance coverage under various primary and excess policies that the Company believes will be available, if necessary, to substantially cover any liability that might ultimately be incurred on these claims.

The Company intends to continue its practice of vigorously defending these claims and cases. It is not possible to predict the ultimate outcome of asbestosrelated lawsuits, claims and proceedings due to the unpredictable nature of personal injury litigation, and no loss provision has been recorded in the Company's consolidated financial statements because a loss contingency is not deemed probable or estimable. Despite this uncertainty, and although results of operations and cash flows for a given period could be adversely affected by asbestos-related lawsuits, claims and proceedings, the Company does not expect that any costs that are reasonably possible to be incurred by the Company in connection with asbestos litigation would have a material adverse effect on the Company's financial condition, results of operations or cash flows.

The Company is subject to various other claims and legal proceedings covering a wide range of matters that arose in the ordinary course of business. In the opinion of management, all such matters are adequately covered by insurance or by established reserves, and, if not so covered, are without merit or are of such kind, or involve such amounts, as would not have a material adverse effect on the financial position, results of operations or cash flows of the Company.

Insurance liabilities are recorded when it is probable that a liability has been incurred for a particular event and the amount of loss associated with the event can be reasonably estimated. Insurance reserves have been estimated based primarily upon actuarial calculations and reflect the undiscounted estimated liabilities for ultimate losses, including claims incurred but not reported. Inherent in these estimates are assumptions that are based on the Company's history of claims and losses, a detailed analysis of existing claims with respect to potential value, and current legal and legislative trends. If actual claims differ from those projected by management, changes (either increases or decreases) to insurance reserves may be required and would be recorded through income in the period the change was determined. When a recognized liability is covered by third-party

insurance, the Company records an insurance claim receivable to reflect the covered liability. Insurance claim receivables are included in Other receivables on the Company's Condensed Consolidated Balance Sheets. See Note 1, "Summary of Significant Accounting Policies," to the consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012 for additional information on Accrued Insurance and Loss Reserves.

11. Reconciliation of Basic and Diluted Shares

	Three Months Ended					Nine Months Ended			
	September 30					September 30			
(In thousands, except per share amounts)	2013			2012		2013		2012	
Income (loss) from continuing operations attributable to Harsco Corporation common stockholders	\$	(233,255)	\$	26,208	\$	(201,282)	\$	10,278	
Weighted-average shares outstanding - basic		80,775		80,659		80,747		80,623	
Dilutive effect of stock-based compensation		—		251				240	
Weighted-average shares outstanding - diluted		80,775		80,910		80,747		80,863	

Earnings (loss) from continuing operations per common share, attributable to Harsco Corporation common stockholders:

Basic	\$ (2.89)	\$ 0.32	\$ (2.49)	\$ 0.13
Diluted	\$ (2.89)	\$ 0.32	\$ (2.49)	\$ 0.13

The following average outstanding stock-based compensation units were not included in the three and nine months ended computation of diluted earnings per share because the effect was antidilutive:

	Three Months	Ended	Nine Months Ended				
	September	30	September 30				
(In thousands)	2013	2012	2013	2012			
Restricted stock units	324	34	267	12			
Stock options	297	349	306	381			
Stock appreciation rights	1,417	291	1,119	309			
Other	103	—	106	_			

12. Derivative Instruments, Hedging Activities and Fair Value

Derivative Instruments and Hedging Activities

The Company uses derivative instruments, including foreign currency forward exchange contracts, cross currency interest rate swaps and, at times, commodity contracts, to manage certain foreign currency, interest rate and commodity price exposures. Derivative instruments are viewed as risk management tools by the Company and are not used for trading or speculative purposes.

All derivative instruments are recorded on the Condensed Consolidated Balance Sheets at fair value. Changes in the fair value of derivatives used to hedge foreign currency denominated balance sheet items are reported directly in earnings along with offsetting transaction gains and losses on the items being hedged. Derivatives used to hedge forecasted cash flows associated with foreign currency commitments or forecasted commodity purchases may be accounted for as cash flow hedges, as deemed appropriate and if the criteria for hedge accounting are met. Gains and losses on derivatives designated as cash flow hedges are deferred as a separate component of equity and reclassified to earnings in a manner that matches the timing of the earnings impact of the hedged transactions. Generally, at September 30, 2013, these deferred gains and losses are reclassified to earnings over 10 to 15 years from the balance sheet date. The ineffective portion of all hedges, if any, is recognized currently in earnings.

The fair values of outstanding derivative contracts recorded as assets and liabilities on the Condensed Consolidated Balance Sheets at September 30, 2013 and December 31, 2012 were as follows:

	Asset Deriva	tives		Liability Derivatives			
(In thousands)	Balance Sheet Location		Fair Value	Balance Sheet Location	F	air Value	
September 30, 2013							
Derivatives designated as hedging instruments:							
Cross currency interest rate swaps	Other current assets	\$	166	Other current liabilities	\$		
Cross currency interest rate swaps	Other assets		31,801	Other liabilities		12,378	
Foreign currency forward exchange contracts	Other current assets			Other current liabilities		18	
Total derivatives designated as hedging instruments		\$	31,967		\$	12,396	
Derivatives not designated as hedging instruments:							
Foreign currency forward exchange contracts	Other current assets	\$	789	Other current liabilities	\$	3,001	

	Asset Deriva	tives		Liability Deri	vatives	
(In thousands)	Balance Sheet Location Fair Value		air Value	Balance Sheet Location	F	air Value
December 31, 2012						
Derivatives designated as hedging instruments:						
Cross currency interest rate swaps	Other assets	\$	39,058	Other liabilities	\$	14,346
Derivatives not designated as hedging instruments:						
Foreign currency forward exchange contracts	Other current assets	\$	853	Other current liabilities	\$	1,775

All of the Company's derivatives are recorded in the Condensed Consolidated Balance Sheets at gross amounts and not offset. All of the Company's cross currency interest rate swaps and certain foreign currency forward exchange contracts are transacted under International Swaps and Derivatives Association ("ISDA") documentation. Each ISDA master agreement permits the net settlement of amounts owed in the event of default. The Company's derivative assets and liabilities subject to enforceable master netting arrangements resulted in a \$0.1 million net liability at December 31, 2012.

The effect of derivative instruments on the Condensed Consolidated Statements of Operations and the Condensed Consolidated Statements of Comprehensive Income for the three and nine months ended September 30, 2013 and 2012 was as follows:

Amount of Cain

Derivatives Designated as Hedging Instruments

(In thousands)	(Loss) Con Incom Do	unt of Gain Recognized in Other nprehensive e ("OCI") on erivative - tive Portion	Location of Gain (Loss) Reclassified from Accumulated OCI into Income - Effective Portion	Ga Reclas Accun into	nount of in (Loss) ssified from nulated OCI Income - ive Portion	Location of Gain (Loss) Recognized in Income on Derivative - Ineffective Portion and Amount Excluded from Effectiveness Testing	(Loss i on - Ineff an Exc	ount of Gain s) Recognized n Income berivative fective Portion d Amount cluded from fectiveness Testing
Three Months Ended September 30,	2013:							
Foreign currency forward exchange contracts	\$	(18)	Cost of services and products sold	\$	(9)	Cost of services and products sold	\$	(6)
Cross currency interest rate swaps		(2,824)			_	Cost of services and products sold		(19,620) _(a)
	\$	(2,842)		\$	(9)		\$	(19,626)
Three Months Ended September 30,	2012:							
Foreign currency forward exchange contracts	\$	(137)	Cost of services and products sold	\$	2		\$	_
Cross currency interest rate swaps		(865)				Cost of services and products sold		(17,485) _(a)
	\$	(1,002)		\$	2		\$	(17,485)

(a) These gains (losses) offset foreign currency fluctuation effects on the debt principal.

(In thousands)	(Loss) Cor Incorr D	ount of Gain Recognized in Other nprehensive te ("OCI") on erivative - ctive Portion	Location of Gain (Loss) Reclassified from Accumulated OCI into Income - Effective Portion	Amount of Gain (Loss) Reclassified from Accumulated OCI into Income - Effective Portion		Location of Gain (Loss) Recognized in Income on Derivative - Ineffective Portion and Amount Excluded from Effectiveness Testing	(Loss i on - Ineff an Ex	ount of Gain) Recognized n Income Derivative fective Portion of Amount cluded from fectiveness Testing
Nine Months Ended September 30, 20	13:							
Foreign currency forward exchange contracts	\$	(18)	Cost of services and products sold	\$	(9)	Cost of services and products sold	\$	(6)
Cross currency interest rate swaps		(3,057)			_	Cost of services and products sold		(2,749) _(a)
	\$	(3,075)		\$	(9)		\$	(2,755)
Nine Months Ended September 30, 20	12:							
Foreign currency forward exchange contracts	\$	(320)	Cost of services and products sold	\$	258		\$	_
Cross currency interest rate swaps		2,914			_	Cost of services and products sold		(8,740) _(a)
	\$	2,594		\$	258		\$	(8,740)

(a) These gains (losses) offset foreign currency fluctuation effects on the debt principal.

Derivatives Not Designated as Hedging Instruments

	Location of Gain	Income on Derivative for the Three Months Ended September 30 (a)							
(In thousands)	(Loss) Recognized in Income on Derivative		2013		2012				
Foreign currency forward exchange contracts	Cost of services and products sold	\$	(5,076)	\$	(3,186)				

ount of Cain (Loss) Becognized in

(a) These gains (losses) offset amounts recognized in cost of services and products sold principally as a result of intercompany or third party foreign currency exposures.

	Location of Gain	Amount of Gain (Loss) Recognized in Income on Derivative for the Nine Months Ended September 30 (a)						
(In thousands)	(Loss) Recognized in Income on Derivative	2013		2012				
Foreign currency forward exchange contracts	Cost of services and products sold	\$ (7,125)	\$	(681)				

(a) These gains (losses) offset amounts recognized in cost of services and products sold principally as a result of intercompany or third party foreign currency exposures.

Foreign Currency Forward Exchange Contracts

The Company conducts business in multiple currencies and, accordingly, is subject to the inherent risks associated with foreign exchange rate movements. The financial position and results of operations of substantially all of the Company's foreign subsidiaries are measured using the local currency as the functional currency. Foreign currency-denominated assets and liabilities are translated into U.S. dollars at the exchange rates existing at the respective balance sheet dates. Income and expense items are translated at the average exchange rates during the respective periods. The aggregate effects of translating the balance sheets of these subsidiaries are deferred and recorded in Accumulated other comprehensive loss, which is a separate component of equity.

The Company uses derivative instruments to hedge cash flows related to foreign currency fluctuations. Foreign currency forward exchange contracts outstanding are part of a worldwide program to minimize foreign currency exchange operating income and balance sheet exposure by offsetting foreign currency exposures of certain future payments between the Company and its various subsidiaries, suppliers or customers. These unsecured contracts are with major financial institutions. The Company may be exposed to credit loss in the event of non-performance by the contract counterparties. The Company evaluates the creditworthiness of the counterparties and does not expect default by them. Foreign currency forward exchange contracts are used to hedge commitments, such as foreign currency debt, firm purchase commitments and foreign currency cash flows for certain export sales transactions.

The following tables summarize, by major currency, the contractual amounts of the Company's foreign currency forward exchange contracts in U.S. dollars at September 30, 2013 and December 31, 2012. The "Buy" amounts represent the U.S. dollar equivalent of commitments to purchase foreign currencies, and the "Sell" amounts represent the U.S. dollar equivalent of commitments to sell foreign currencies. The recognized gains and losses offset amounts recognized in cost of services and products sold principally as a result of intercompany or third party foreign currency exposures.

Contracted Amounts of Foreign Currency Forward Exchange Contracts Outstanding at September 30, 2013:

(In thousands)	Туре	U.S. Dollar Equivalent	Maturity	Recognized Gain (Loss)	
British pounds sterling	Sell	\$ 25,998	October 2013	\$ (534)	
British pounds sterling	Buy	8,035	October 2013	50	
Euros	Sell	184,091	October 2013 through January 2014	(1,870)	
Euros	Buy	128,023	October 2103 through March 2014	170	
Other currencies	Sell	5,590	October 2013 through July 2014	(16)	
Other currencies	Buy	10,932	October 2013 through February 2014	(30)	
Total		\$ 362,669		\$ (2,230)	

Contracted Amounts of Foreign Currency Forward Exchange Contracts Outstanding at December 31, 2012:

(In thousands)	Туре	U.S. Dollar Equivalent	Maturity	Recognized Gain (Loss)	
British pounds sterling	Sell	\$ —		\$ —	
British pounds sterling	Buy	6,141	January 2013 through February 2013	58	
Euros	Sell	264,234	January 2013 through March 2013	(1,082)	
Euros	Buy	116,618	January 2013 through February 2013	187	
Other currencies	Sell	2,811	January 2013 through March 2013	(15)	
Other currencies	Buy	44,291	January 2013	(71)	
Total		\$ 434,095		\$ (923)	

In addition to foreign currency forward exchange contracts, the Company designates certain loans as hedges of net investments in international subsidiaries. The Company recorded pre-tax net losses of \$4.9 million and \$8.3 million during the three and nine months ended September 30, 2013, respectively, and pretax net losses of \$1.9 million and \$4.4 million during the three and nine months ended September 30, 2012, respectively, into Accumulated other comprehensive loss.

Cross Currency Interest Rate Swaps

The Company uses cross currency interest rate swaps in conjunction with certain debt issuances in order to secure a fixed local currency interest rate. Under these cross currency interest rate swaps, the Company receives interest based on a fixed or floating U.S. dollar rate and pays interest on a fixed local currency rate based on the contractual amounts in dollars and the local currency, respectively. The cross currency interest rate swaps are recorded on the Condensed Consolidated Balance Sheets at fair value, with changes in value attributed to the effect of the swaps' interest spread recorded in Accumulated other comprehensive loss. Changes in value attributed to the effect of foreign currency fluctuations are recorded in the statements of operations and offset currency fluctuation effects on the debt principal. The following table indicates the contractual amounts of the Company's cross currency interest rate swaps at September 30, 2013:

			Interest Rates							
(In millions)	Contra	ctual Amount	Receive	Pay						
Maturing 2018	\$	250.0	Fixed U.S. dollar rate	Fixed euro rate						
Maturing 2020		220.0	Fixed U.S. dollar rate	Fixed British pound sterling rate						
Maturing 2013 through 2017		10.9	Floating U.S. dollar rate	Fixed rupee rate						

Fair Value of Derivative Assets and Liabilities and Other Financial Instruments

Fair value is the price that would be received to sell an asset or paid to transfer a liability in an orderly transaction between market participants at the measurement date (an exit price). The Company utilizes market data or assumptions that the Company believes market participants would use in valuing the asset or liability, including assumptions about risk and the risks inherent in the inputs to the valuation technique.

The fair value hierarchy distinguishes between (1) market participant assumptions developed based on market data obtained from independent sources (observable inputs), and (2) an entity's own assumptions about market participant assumptions based on the best information available in the circumstances (unobservable inputs). The fair value hierarchy consists of three broad levels, which give the highest priority to unadjusted quoted prices in active markets for identical assets or liabilities (Level 1) and the lowest priority to unobservable inputs (Level 3). The three levels of the fair value hierarchy are described below:

- Level 1—Unadjusted quoted prices in active markets that are accessible at the measurement date for identical, unrestricted assets or liabilities.
- Level 2—Inputs other than quoted prices included within Level 1 that are observable for the asset or liability, either directly or indirectly, including quoted prices for similar assets or liabilities in active markets; quoted prices for identical or similar assets or liabilities in markets that are not active; inputs other than quoted prices that are observable for the asset or liability (e.g., interest rates); and inputs that are derived principally from or corroborated by observable market data by correlation or other means.
- Level 3—Inputs that are both significant to the fair value measurement and unobservable.

In instances in which multiple levels of inputs are used to measure fair value, hierarchy classification is based on the lowest level input that is significant to the fair value measurement in its entirety. The Company's assessment of the significance of a particular input to the fair value measurement in its entirety requires judgment, and considers factors specific to the asset or liability.

The following table indicates the fair value hierarchy of the financial instruments of the Company at September 30, 2013 and December 31, 2012:

Level 2 Fair Value Measurements

(In thousands)			December 31 2012	
Assets				
Foreign currency forward exchange contracts	\$ 78	9 \$	853	
Cross currency interest rate swaps	31,96	7	39,058	
Liabilities				
Foreign currency forward exchange contracts	3,01	9	1,775	
Cross currency interest rate swaps	12,37	8	14,346	

The Company primarily applies the market approach for recurring fair value measurements and endeavors to utilize the best available information. Accordingly, the Company utilizes valuation techniques that maximize the use of observable inputs, such as forward rates, interest rates, the Company's credit risk and counterparties' credit risks, and which minimize the use of unobservable inputs. The Company is able to classify fair value balances based on the ability to observe those inputs. Commodity derivatives, foreign currency forward exchange contracts and cross currency interest rate swaps are classified as Level 2 fair value based upon pricing models using market-based inputs. Model inputs can be verified, and valuation techniques do not involve significant management judgment.

The carrying amounts of cash and cash equivalents, accounts receivable, accounts payable, accrued liabilities and short-term borrowings approximate fair value due to the short-term maturities of these assets and liabilities. At September 30, 2013 and December 31, 2012, the total fair value of long-term debt, including current maturities, was \$833.6 million and \$1.0 billion, respectively, compared with a carrying value of \$794.9 million and \$1.0 billion at September 30, 2013 and December 31, 2012, respectively. Fair values for debt are based on quoted market prices (Level 1) for the same or similar issues, or on the current rates offered to the Company for debt of the same remaining maturities.

13. Review of Operations by Segment

	Three Months Ended September 30					Nine Months Ended September 30				
(In thousands)		2013	2012		2013			2012		
Revenues From Continuing Operations										
Harsco Metals & Minerals	\$	335,705	\$	344,867	\$	1,009,175	\$	1,069,741		
Harsco Infrastructure		242,569		229,287		709,800		701,829		
Harsco Rail		66,424		91,423		216,636		239,098		
Harsco Industrial		95,347		91,206		279,565		269,025		
Total revenues from continuing operations	\$	740,045	\$	756,783	\$	2,215,176	\$	2,279,693		
Operating Income (Loss) From Continuing Operations										
Harsco Metals & Minerals	\$	23,036	\$	26,350	\$	66,857	\$	79,662		
Harsco Infrastructure		(242,098)		(6,065)		(256,342)		(83,956)		
Harsco Rail		8,063		13,906		27,554		35,272		
Harsco Industrial		15,901		16,710		48,152		47,663		
Corporate (a)		(13,279)		(506)		(17,966)		(6,294)		
Total operating income (loss) from continuing operations	\$	(208,377)	\$	50,395	\$	(131,745)	\$	72,347		

(a) The Corporate caption includes transactions costs associated with the Infrastructure transaction for the three and nine months ended September 30, 2013.

Reconciliation of Segment Operating Income to Income (Loss) From Continuing Operations Before Income Taxes and Equity Income

	Three Months Ended					Nine Months Ended				
	September 30					September 30				
(In thousands)		2013	2012		2013			2012		
Segment operating income	\$	(195,098)	\$	50,901	\$	(113,779)	\$	78,641		
General Corporate expense		(13,279)		(506)		(17,966)		(6,294)		
Operating income (loss) from continuing operations		(208,377)		50,395		(131,745)		72,347		
Interest income		388		1,449		1,624		3,005		
Interest expense		(12,815)		(11,609)		(37,413)		(36,041)		
Income (loss) from continuing operations before income taxes and equity income	\$	(220,804)	\$	40,235	\$	(167,534)	\$	39,311		

14. Other (Income) Expenses

This Condensed Consolidated Statements of Operations classification includes restructuring costs for employee termination benefit costs, product rationalization, and costs to exit activities; former CEO separation costs; net gains on the disposal of non-core assets; and currency translation adjustments recognized in earnings.

	Three Mo	nths En	Nine Months Ended			
	 Septer	nber 30	 September 30			
(In thousands)	2013		2012	2013		2012
Restructuring Program costs (see Note 16)	\$ _	\$	8,454	\$ 	\$	73,563
Former CEO separation costs						4,125
Net gains	(563)		(2,736)	(5,132)		(3,958)
Currency translation adjustments recognized in earnings			(4,152)			(10,906)
Other (a)	335		817	7,290		2,527
Other (income) expenses	\$ (228)	\$	2,383	\$ 2,158	\$	65,351

(a) Other includes employee termination benefit costs, impaired asset write-downs, and costs to exit activities that are not directly related to the restructuring programs detailed in Note 16, "Restructuring Programs."

Table of Contents

15. Components of Accumulated Other Comprehensive Loss

Accumulated other comprehensive loss is included on the Consolidated Statements of Stockholders' Equity. The components of Accumulated other comprehensive loss, net of the effect of income taxes, and activity for the nine months ended September 30, 2013 are as follows:

		Components of Accumulated Other Comprehensive Income (Loss) - Net of Tax									
(In thousands)	Exchan	ative Foreign ge Translation justments	Deriv	Effective Portion of erivatives Designated as Hedging Instruments		Cumulative Inrecognized Iarial Losses on sion Obligations	Unrealized Loss on Marketable Securities			Total	
Balance at December 31, 2012	\$	62,308	\$	(8,139)	\$	(465,286)	\$	(51)	\$	(411,168)	
Other comprehensive income (loss) before reclassifications		(33,877) _(a)		(2,166) _(b)		2,692 _(a)		21		(33,330)	
Amounts reclassified from accumulated other comprehensive loss		_		6		14,895				14,901	
Total other comprehensive income (loss)		(33,877)		(2,160)	_	17,587		21		(18,429)	
Less: Other comprehensive (income) loss attributable to noncontrolling interests		(152)		84		_				(68)	
Other comprehensive income (loss) attributable to Harsco Corporation		(34,029)		(2,076)		17,587		21		(18,497)	
Balance at September 30, 2013	\$	28,279	\$	(10,215)	\$	(447,699)	\$	(30)	\$	(429,665)	
(-) Drive size allow for uniting any many set flow structions											

(a) Principally foreign currency fluctuation.(b) Net change from periodic revaluations.

Amounts reclassified from accumulated other comprehensive loss for the three and nine months ended September 30, 2013 are as follows:

thousands)	(In]	Three Months Ended September 30 2013		Nine Months Ended September 30 2013		Affected Line Item in the Condensed Consolidated Statements of Operations
Amortization of defined benefit pension items (a):							
Actuarial losses		\$	3,052	\$	9,353		Selling, general and administrative expenses
Actuarial losses			2,275		6,800		Cost of services and products sold
Prior-service costs			62		192		Selling, general and administrative expenses
Prior-service costs			61		187	((b) 5	Cost of services and products sold
Total before tax			5,450		16,532		
Tax benefit			(523)		(1,637)		
Total reclassification of defined benefit pension items, net of tax		\$	4,927	\$	14,895	_	
Amortization of cash flow hedging instruments (a):							
Foreign currency forward exchange contracts		\$	9	\$	9		Cost of services and products sold
Tax benefit			(3)		(3)		
Total reclassification of cash flow hedging instruments (a) Amounts in parentheses indicate credits to profit/loss.		\$	6	\$	6	_	

(a) Amounts in parentheses indicate credits to profit/loss.(b) These accumulated other comprehensive loss components are included in the computation of net periodic pension costs. Please refer to Note 8, "Employee Benefit Plans," for additional details.

16. Restructuring Programs

In recent years, the Company has instituted restructuring programs to balance short-term profitability goals with long-term strategies. A primary objective of these programs has been to establish platforms upon which the affected businesses can grow with reduced fixed investment and generate annual operating expense savings. The restructuring programs have been instituted in response to the continuing impact of global financial and economic uncertainty on the Company's end markets, particularly in the Company's Harsco Infrastructure Segment. Restructuring costs incurred in these programs were recorded in the Other (income) expenses line of the Condensed Consolidated Statements of Operations. The timing of associated cash payments is dependent on the type of restructuring cost and can extend over a multi-year period.

2011/2012 Restructuring Program

Under the 2011/2012 Restructuring Program, the Company optimized rental assets and sale inventories by removing non-core assets under an expanded product rationalization and branch structure reduction program undertaken in its Harsco Infrastructure Segment and optimized office structures and reduced global workforce in the Harsco Infrastructure and Harsco Metals & Minerals Segments. Benefits under this program, in the form of reduced costs when compared with 2011, are expected to be approximately \$63 million for the full year of 2013. This represents an incremental \$8 million in benefits under this program when compared with the benefits realized in 2012. As previously disclosed in the Company's Annual Reports on Form 10-K, the Company incurred approximately \$101 million and \$69 million in pre-tax charges under this program in calendar years 2011 and 2012, respectively. The restructuring accrual for the 2011/2012 Restructuring Program at September 30, 2013 and the activity for the nine months then ended are as follows:

(In thousands)	Accrual December 31 2012	Non-Cash Charges / Other Adjustments	Cash Expenditures	Foreign Currency Translation	Remaining Accrual September 30 2013
Harsco Infrastructure Segment					
Employee termination benefit costs	\$ 6,999	\$ (962)	\$ (4,880)	\$ (250)	\$ 907
Cost to exit activities	9,000	(764)	(2,914)	(63)	5,259
Total Harsco Infrastructure Segment	 15,999	(1,726)	(7,794)	 (313)	 6,166
Harsco Metals & Minerals Segment					
Employee termination benefit costs	6,494	(473)	(4,318)	(59)	1,644
Cost to exit activities	499		(175)	6	330
Total Harsco Metals & Minerals Segment	 6,993	 (473)	 (4,493)	 (53)	 1,974
Harsco Corporate					
Employee termination benefit costs	13	2	(12)		3
Total	\$ 23,005	\$ (2,197)	\$ (12,299)	\$ (366)	\$ 8,143

Cash expenditures related to the remaining employee termination benefit costs accrual at September 30, 2013 are expected to be paid principally throughout the remainder of 2013 with approximately \$6 million of exit activity costs for lease terminations expected to be paid over the remaining life of the leases.

Fourth Quarter 2010 Harsco Infrastructure Program

Under the Fourth Quarter 2010 Harsco Infrastructure Program, the Harsco Infrastructure Segment reduced its branch structure; consolidated and/or closed administrative office locations; reduced its global workforce; and rationalized its product lines. The restructuring accrual for the Fourth Quarter 2010 Harsco Infrastructure Program at September 30, 2013 and the activity for the nine months then ended are as follows:

(In thousands)	D	Accrual December 31 2012	on-cash Charges / her Adjustments	Cash Expenditures	Foreign Currency Translation	Remaining Accrual September 30 2013
Harsco Infrastructure Segment						
Cost to exit activities	\$	6,791	\$ (449)	\$ (421)	\$ (3)	\$ 5,918

Of the remaining accrual at September 30, 2013, \$5.7 million relates to payment of multiemployer pension plan withdrawal liabilities and is expected to be paid through 2023 under contractual payment terms with the related plan administrators. The remaining balance primarily relates to costs for lease terminations that are expected to be paid over the remaining life of the leases.



ITEM 2. MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS

The following discussion should be read in conjunction with the accompanying unaudited condensed consolidated financial statements as well as the audited consolidated financial statements of Harsco Corporation (the "Company"), including the notes thereto, included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012, which includes additional information about the Company's critical accounting policies, contractual obligations, practices and the transactions that support the financial results, and provides a more comprehensive summary of the Company's outlook, trends and strategies for 2013 and beyond.

Certain amounts included in Item 2 of this Quarterly Report on Form 10-Q are rounded in millions and all percentages are calculated based on actual amounts. As a result, minor differences may exist due to rounding.

Forward-Looking Statements

The nature of the Company's business and the many countries in which it operates subject it to changing economic, competitive, regulatory and technological conditions, risks and uncertainties. In accordance with the "safe harbor" provisions of the Private Securities Litigation Reform Act of 1995, the Company provides the following cautionary remarks regarding important factors that, among others, could cause future results to differ materially from the results contemplated by forward-looking statements, including the expectations and assumptions expressed or implied herein. Forward-looking statements contained herein could include, among other things, statements about management's confidence in and strategies for performance; expectations for new and existing products, technologies and opportunities; and expectations regarding growth, sales, cash flows, earnings and Economic Value Added ("EVA®"). Forward-looking statements can be identified by the use of such terms as "may," "could," "expect," "anticipate," "intend," "believe," "likely," "estimate," "plan" or other comparable terms.

Factors that could cause actual results to differ, perhaps materially, from those implied by forward-looking statements include, but are not limited to: (1) changes in the worldwide business environment in which the Company operates, including general economic conditions; (2) changes in currency exchange rates, interest rates, commodity and fuel costs and capital costs; (3) changes in the performance of equity and bond markets that could affect, among other things, the valuation of the assets in the Company's pension plans and the accounting for pension assets, liabilities and expenses; (4) changes in governmental laws and regulations, including environmental, tax and import tariff standards; (5) market and competitive changes, including pricing pressures, market demand and acceptance for new products, services and technologies; (6) unforeseen business disruptions in one or more of the many countries in which the Company operates due to political instability, civil disobedience, armed hostilities, public health issues or other calamities; (7) the seasonal nature of the Company's business; (8) the Company's ability to successfully enter into new contracts and complete new acquisitions or strategic ventures in the timeframe contemplated, or at all; (9) the integration of the Company's strategic acquisitions; (10) the amount and timing of repurchases of the Company's common stock, if any; (11) the prolonged recovery in global financial and credit markets and economic conditions generally, which could result in the Company's customers curtailing development projects, construction, production and capital expenditures, which, in turn, could reduce the demand for the Company's products and services and, accordingly, the Company's revenues, margins and profitability; (12) the outcome of any disputes with customers; (13) the financial condition of the Company's customers, including the ability of customers (especially those that may be highly leveraged and those with inadequate liquidity) to maintain their credit availability; (14) the Company's ability to successfully implement and receive the expected benefits of costreduction and restructuring initiatives, including the achievement of expected cost savings in the expected time frame; (15) the Company's ability to successfully implement its strategic initiatives and portfolio optimization and the impact of such initiatives; (16) the ability and timing to consummate the strategic venture between the Company and Clayton, Dubilier & Rice ("CD&R") that was announced on September 16, 2013, including the ability and timing to obtain required regulatory approvals and satisfy other closing conditions; (17) the strategic venture's ability to promptly and effectively integrate the Company's Infrastructure business and the Brand Energy & Infrastructure Services business and realize the synergies contemplated by the potential transaction; (18) the Company's ability to realize cost savings from the divestiture of the Harsco Infrastructure Segment, as well as the transaction being accretive to earnings and improving operating margins and return on capital; (19) risk and uncertainty associated with intangible assets; and (20) other risk factors listed from time to time in the Company's SEC reports. A further discussion of these, along with other potential risk factors, can be found in Part I, Item 1A, "Risk Factors," of the Company's Annual Report on Form 10-K for the year ended December 31, 2012. The Company cautions that these factors may not be exhaustive and that many of these factors are beyond the Company's ability to control or predict. Accordingly, forward-looking statements should not be relied upon as a prediction of actual results. The Company undertakes no duty to update forward-looking statements except as may be required by law.

Table of Contents

Executive Overview

Three Months Ended								
September 30								
2013 2012 Change %						%		
\$	335.7	\$	344.9	\$	(9.2)	(2.7)%		
	242.6		229.3		13.3	5.8		
	66.4		91.4		(25.0)	(27.3)		
	95.3		91.2		4.1	4.5		
\$	740.0	\$	756.8	\$	(16.7)	(2.2)%		
	\$	\$ 335.7 242.6 66.4 95.3	\$ 335.7 \$ 242.6 66.4 95.3	2013 2012 \$ 335.7 \$ 344.9 242.6 229.3 66.4 91.4 95.3 91.2	2013 2012 \$ 335.7 \$ 344.9 \$ 242.6 229.3 5 66.4 91.4 9 95.3 91.2 5	September 30 2013 2012 Change \$ 335.7 \$ 344.9 \$ (9.2) 242.6 229.3 13.3 66.4 91.4 (25.0) 95.3 91.2 4.1		

	Nine Months Ended							
Revenues by Segment	September 30							
(In millions)	2013 2012 Change %						%	
Harsco Metals & Minerals	\$	1,009.2	\$	1,069.7	\$	(60.6)	(5.7)%	
Harsco Infrastructure		709.8		701.8		8.0	1.1	
Harsco Rail		216.6		239.1		(22.5)	(9.4)	
Harsco Industrial		279.6		269.0		10.5	3.9	
Total revenues	\$	2,215.2	\$	2,279.7	\$	(64.5)	(2.8)%	

	Three Months Ended							
Revenues by Region				Septo	ember	30		
(In millions)		2013		2012		Change	%	
Western Europe	\$	279.4	\$	262.8	\$	16.7	6.3 %	
North America		271.5		300.1		(28.6)	(9.5)	
Latin America (a)		80.4		79.9		0.5	0.6	
Asia-Pacific		49.0		52.0		(3.0)	(5.8)	
Middle East and Africa		36.9		38.0		(1.1)	(2.8)	
Eastern Europe		22.8		24.0		(1.2)	(5.0)	
Total revenues	\$	740.0	\$	756.8	\$	(16.7)	(2.2)%	

(a) Includes Mexico.

	Nine Months Ended								
Revenues by Region	September 30								
(In millions)		2013	2012			Change	%		
Western Europe	\$	816.6	\$	810.2	\$	6.4	0.8 %		
North America		833.1		880.2		(47.1)	(5.4)		
Latin America (a)		241.9		247.7		(5.8)	(2.3)		
Asia-Pacific		140.3		150.0		(9.7)	(6.5)		
Middle East and Africa		121.6		115.6		6.0	5.2		
Eastern Europe		61.7		75.9		(14.2)	(18.7)		
Total revenues	\$	2,215.2	\$	2,279.7	\$	(64.5)	(2.8)%		
(a) Includes Mexico.									

Revenues for the Company during the third quarter and first nine months of 2013 were \$740.0 million and \$2.2 billion, respectively, compared with \$756.8 million and \$2.3 billion, respectively, in the third quarter and first nine months of 2012. Foreign currency translation decreased revenues by \$4.6 million and \$15.4 million, respectively, for the third quarter and first nine months of 2013 in comparison with the third quarter and first nine months of 2013.

		Three Months Ended								
Operating Income (Loss) by Segment	September 30									
(In millions)		2013		2012		Change	%			
Harsco Metals & Minerals	\$	23.0	\$	26.4	\$	(3.3)	(12.6)%			
Harsco Infrastructure		(242.1)		(6.1)		(236.0)	(3,891.7)			
Harsco Rail		8.1		13.9		(5.8)	(42.0)			
Harsco Industrial		15.9		16.7		(0.8)	(4.8)			
Corporate (a)		(13.3)		(0.5)		(12.8)	(2,524.3)			
Total operating income (loss)	\$	(208.4)	\$	50.4	\$	(258.8)	(513.5)%			

(a) The Corporate caption includes transactions costs associated with the Infrastructure transaction for the three and nine months ended September 30, 2013.

Table of Contents

	Nine Months Ended September 30									
Operating Income (Loss) by Segment										
(In millions)	2013			2012		Change	%			
Harsco Metals & Minerals	\$	66.9	\$	79.7	\$	(12.8)	(16.1)%			
Harsco Infrastructure		(256.3)		(84.0)		(172.4)	(205.3)			
Harsco Rail		27.6		35.3		(7.7)	(21.9)			
Harsco Industrial		48.2		47.7		0.5	1.0			
Corporate (a)		(18.0)		(6.3)		(11.7)	(185.4)			
Total operating income (loss)	\$	(131.7)	\$	72.3	\$	(204.1)	(282.1)%			

(a) The Corporate caption includes transactions costs associated with the Infrastructure transaction for the three and nine months ended September 30, 2013.

	Three Months	Ended	Nine Months Ended				
	September	30	September 30				
Operating Margin by Segment	2013	2012	2013	2012			
Harsco Metals & Minerals	6.9 %	7.6 %	6.6 %	7.4 %			
Harsco Infrastructure	(99.8)	(2.6)	(36.1)	(12.0)			
Harsco Rail	12.1	15.2	12.7	14.8			
Harsco Industrial	16.7	18.3	17.2	17.7			
Consolidated operating margin	(28.2)%	6.7 %	(5.9)%	3.2 %			

On September 15, 2013, the Company entered into an agreement to sell the Company's Harsco Infrastructure Segment into a strategic venture with CD&R as part of a transaction that will combine the Harsco Infrastructure Segment with Brand Energy & Infrastructure Services, Inc., which CD&R is simultaneously acquiring (the "Infrastructure transaction"). Under the terms of the purchase agreement, the Company will contribute substantially all of the Company's equity interests in, and the assets of, the Harsco Infrastructure Segment in exchange for \$300 million, subject to working capital and other adjustments, and an approximate 29% equity interest in the strategic venture. The Infrastructure transaction is anticipated to close during the fourth quarter of 2013, subject to certain customary conditions of closing.

As a result of the Infrastructure transaction, the Company recorded an estimated non-cash long-lived asset impairment charge and transactions costs of \$253.7 million during the third quarter of 2013. The final non-cash long-lived asset impairment charge may differ, possibly materially, due to a number of factors including, but not limited to (i) the valuation of certain assets and liabilities being transferred to the strategic venture; (ii) final capital expenditure and working capital adjustments required under the purchase agreement; (iii) the recognition of cumulative translation amounts; and (iv) the finalization of current and deferred income tax amounts. The Company determined that the estimated non-cash long-lived asset impairment charge was necessary given the fair value of the consideration received in the Infrastructure transaction. The amount of the estimated non-cash long-lived asset impairment charge represents the Company's estimate of the difference between the current book value of the Harsco Infrastructure Segment, less costs to sell, and the sum of the cash consideration plus the estimated fair value of the Company's approximate 29% equity interest in the strategic venture. The estimated fair value of the equity interest was determined based on the expected future discounted cash flows of the strategic venture. See Note 1, "Summary of Significant Accounting Policies," to the consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012 for additional information on the Company's policy on impairment of long-lived assets (other than goodwill).

The operating loss from continuing operations for the third quarter and first nine months of 2013 was \$208.4 million and \$131.7 million, respectively, compared with operating income from continuing operations of \$50.4 million and \$72.3 million, respectively, in the third quarter and first nine months of 2012. The changes are primarily related to the \$253.7 million estimated non-cash long-lived asset impairment charge and transaction costs associated with the Harsco Infrastructure Segment which were recorded during the third quarter of 2013. Additionally, the Harsco Metals & Minerals Segment experienced a reduction in customer steel production during the third quarter and first nine months of 2013 and the Harsco Rail Segment experienced decreased machine sales in the third quarter and first nine months of 2013. Partially offsetting these decreases are pre-tax restructuring program costs incurred in 2012 that were not repeated in 2013. These costs were \$8.5 million and

\$73.6 million in the third quarter and first nine months of 2012, respectively, principally in the Harsco Infrastructure Segment.

These changes in operating income (loss) from continuing operations were the primary drivers of the diluted loss per share from continuing operations for the third quarter and first nine months of 2013 of \$2.89 and \$2.49, respectively, compared with diluted earnings per share from continuing operations of \$0.32 and \$0.13 for the third quarter and first nine months of 2012, respectively.

The Company continues to have sufficient available liquidity and has been able to obtain all necessary financing. The Company currently expects operational and business needs to be met by cash from operations supplemented with borrowings from time to time due to historical patterns of seasonal cash flow and for the funding of various projects. See Liquidity and Capital Resources below for further discussion on liquidity, capital resources and cash flows.

Harsco Metals & Minerals Segment:

Significant Impacts on Revenues	Three Months Ended			Nine Months Ended		
(In millions)	S	September 30, 2013	September 30, 2013			
Revenues — 2012	\$	344.9	\$	1,069.7		
Impact of exited underperforming contracts		(8.8)		(30.7)		
Net increased (decreased) volume / price		3.8		(14.9)		
Impact of foreign currency translation		(4.2)		(14.9)		
Revenues — 2013	\$	335.7	\$	1,009.2		

Significant Impacts on Operating Income:

- Continuing lower global steel production was experienced in the steel mills services business. Overall, steel production by customers under services contracts was down 1% in the third quarter and 5% for the first nine months of 2013 compared with the same periods in 2012, including the impact of exited underperforming contracts and an extended production outage at a large customer location in North America (currently expected to resume operation in the fourth quarter of 2013).
- Continuing lower demand and pricing for by-product in North America.
- Lower demand in the roofing granules business.
- Nickel prices have decreased 17% in the third quarter and 14% in the first nine months of 2013 compared with the same periods in 2012.
- Foreign currency translation did not significantly impact operating income in third quarter or first nine months of 2013 compared to the same periods in 2012.

Harsco Infrastructure Segment:

Significant Effects on Revenues

Significant Effects on Revenues	Three Months Ended			Nine Months Ended	
(In millions)	Sep	tember 30, 2013	September 30, 2013		
Revenues — 2012	\$	229.3	\$	701.8	
Impact of exited operations				(12.5)	
Net increased volume / price		12.9		19.9	
Effect of foreign currency translation		0.4		0.6	
Revenues — 2013	\$	242.6	\$	709.8	

Significant Impacts on Operating Income:

- Operating income was negatively affected in third guarter 2013 by the estimated non-cash long-lived asset impairment charge of \$241.3 million, partially offset by the a decrease in depreciation expense of \$3.5 million associated with the reclassification of substantially all the assets of the Harsco Infrastructure Segment to Assets held-for-sale on the Condensed Consolidated Balance Sheet at September 30, 2013. Both adjustments were a result of the Infrastructure transaction, as described in Note 3, "Acquisitions and Dispositions," in Part I, Item I, Financial Statements.
- Pre-tax restructuring program costs totaling \$7.7 million and \$71.7 million incurred in the third quarter and first nine months of 2012, respectively, were not repeated in 2013.
- Improved volume for the third guarter and first nine months of 2013 compared with the same periods of 2012 in North America related to industrial services, partially offset in the United Kingdom and Benelux due to industrial services and overall market conditions in the Asia-Pacific region and Europe.
- Foreign currency translation did not significantly impact the third quarter comparison of 2013 with 2012, but increased operating income by \$1.3 million in the first nine months 2013 compared with the first nine months of 2012.



Harsco Rail Segment:

Significant Impacts on Revenues	Three Months Ended	Nine Months Ended		
(In millions)	September 30, 2013	September 30, 2013		
Revenues — 2012	\$ 91.4	\$	239.1	
Net decreased volume	(24.6)		(21.4)	
Impact of foreign currency translation	(0.4)		(1.1)	
Revenues — 2013	\$ 66.4	\$	216.6	

Significant Impacts on Operating Income:

- Operating income for the third quarter and first nine months of 2013 was impacted by decreases in machine sales primarily due to the completion of the large China Railway Corporation rail grinders order and a decrease in contract services, partially offset by strong parts sales compared with the same periods of 2012.
- Foreign currency translation did not significantly impact operating income in the third quarter or first nine months of 2013 in comparison with 2012.

Harsco Industrial Segment:

Significant Effects on Revenues	Thre	e Months Ended	Nine Months Ended			
(In millions)	Sep	tember 30, 2013	September 30, 2013			
Revenues — 2012	\$	91.2	\$	269.0		
Net increased volume		4.4		10.6		
Effect of foreign currency translation		(0.3)		—		
Revenues — 2013	\$	95.3	\$	279.6		

Significant Effects on Operating Income:

• Operating income for the third quarter and first nine months of 2013 compared with the same period of 2012 was favorably affected by improved demand for industrial grating products and industrial boilers coupled with year to date improvement from the Australian air cooled heat exchanger market.

Outlook, Trends and Strategies

In addition to items noted in the Company's Annual Report on Form 10-K for the year ended December 31, 2012, the following significant items, risks, trends and strategies are expected to affect the Company for the remainder of 2013 and beyond:

- Upon the close of the Infrastructure transaction, Harsco will receive approximately \$300 million in cash, which will significantly strengthen the Company's balance sheet and enable the Company to reallocate capital to higher-return opportunities. On a pro forma basis, Harsco expects the Infrastructure transaction to improve margins, be accretive to earnings in the first year after close and improve the Company's return on capital. The Company believes there are a number of organic growth and bolt-on acquisition opportunities to create differential value and generate improved returns and growth.
- Upon closing of the Infrastructure transaction, the Company's 29% equity interest in the strategic venture will be accounted for under the equity method of accounting as prescribed by accounting principles generally accepted in the United States of America ("U.S. GAAP").
- Management will continue to be very selective and disciplined in allocating capital, choosing projects with the highest EVA potential and return on capital employed. The Company expects capital expenditures in 2013 to be in line with capital expenditures in 2012.
- The Company expects its effective income tax rate to approximate 34% for the full year 2013, excluding any tax effects attributable to the
 Infrastructure transaction. This modest increase from historical levels is due to losses from operations in certain jurisdictions where tax benefits will
 not be able to be recognized, as well as the geographic mix of income. Going forward, there may be some variability in the reported tax rate from
 quarter-to-quarter depending on the actual geographic mix of earnings.



Harsco Metals & Minerals Segment:

- The Metals & Minerals Segment continues to be impacted by the overall weak economic climate in the global steel industry, particularly in developed markets. The Segment has been impacted by lower prices and lower demand for its recycled by-products, as well as lower steel production by its customers. Total customer steel production has declined year-over-year in recent quarters, including an extended production outage at a large customer location in North America (currently expected to resume operation in the fourth quarter of 2013). This decline was 1% in the third quarter of 2013 compared with the third quarter of 2012, and a decline of 5% in the first nine months of 2013 compared with the first nine months of 2012. Market forecasts are weak for the remainder of 2013.
- In addition to renewing its contracts in mature markets, the Company focuses on winning contracts in emerging markets where steel production is
 increasing and where the customers value the Company's environmental solutions. The Company will continue its focus on ensuring that forecasted
 profits for contracts meet certain established requirements. Given this strategy, in some cases when opportunities do not meet established criteria, a
 contract may not be renewed, resulting in exit costs during the period in which such decisions are finalized.
- Another example of the Company's longer-term strategy is the July 2013 announcement of a new 15-year slag management and metal recovery services contract with Essar Steel India Limited, one of India's largest steel producers and exporters. This is the third major contract announcement in India during the current year following new contracts with JSW Steel, a leading private sector steel producer and Jindal Stainless Limited, India's largest stainless steel producer. Combined with other ongoing contracts, this Segment's portfolio of business in the Indian steel sector now totals more than \$500 million in projected future revenues over the next 10 to 15 years.
 - As the Company has disclosed previously, one of the Company's large steel mill customers in Europe has filed for receivership. The Company has approximately \$11 million of receivables with this customer. Based on current facts and circumstances, the Company has determined that there is no probable and estimable loss associated with these receivables, and therefore no reserves have been established for this matter. The situation with this customer is fluid, and the Company continues to monitor this matter closely. Should there be an adverse change in facts and circumstances, it may be necessary to establish a reserve in the future, which would result in a charge against income in future periods.

Harsco Infrastructure Segment:

- The Company expects the Harsco Infrastructure Segment to realize a year-over-year decrease in operating performance due to continued challenges in key end markets, particularly in the European economies and, to a lesser extent, impacts from low commodity prices on activities in locations such as Australia.
- The Harsco Infrastructure Segment continues to focus on driving sales performance and continuing activities to improve efficiencies in yard, asset and
 project management.
- The Infrastructure transaction is anticipated to close during the fourth quarter of 2013, subject to certain customary conditions of closing. Upon closing of the Infrastructure transaction, the Company's 29% equity interest in the strategic venture will be accounted for under the equity method of accounting as prescribed by U.S. GAAP.

Harsco Rail Segment:

- The short-term outlook for this business is unfavorably impacted by the timing of shipments for its large China Railway Corporation rail grinder orders, which were mostly completed during the first quarter of 2013. Compared with 2012, revenues for this Segment are expected to decline approximately \$55 million in 2013 largely due to the completion of the China Railway Corporation order. The success in China has been leveraged to secure several new orders in other geographies, as well as with multiple metro systems in China; however, none of the individual orders are as large as the China Railway Corporation order. These challenges are partially offset by strength in this Segment's part sales.
- Due to dynamics in the North American contract services market served by this Segment, certain rail grinders are idle. Additional opportunities for use of these rail grinders is actively being pursued. While no long-lived asset impairment was necessary as of September 30, 2013, this Segment may be required to record a long-lived asset impairment charge in the future to the extent it cannot generate future undiscounted cash flows at a level sufficient to recover the net book value of the long-lived assets. The net book value of the related long-lived assets was \$13.1 million as of September 30, 2013.
- The longer-term outlook for this Segment continues to be favorable. The global demand for railway maintenance-of-way equipment, parts, and services continues to be strong, giving positive indication of further opportunities for this Segment.



Harsco Industrial Segment:

- The Harsco Industrial Segment is expecting another year of consistent performance with slight growth for revenue and operating income in 2013, and will continue to focus on product innovation and development to drive strategic growth in its businesses.
- Overall, the Segment's markets are stable with steady demand for air-cooled heat exchangers. Improvement is expected in the industrial boilers market for the heat transfer products business, as well as for industrial grating.

Results of Operations

Three Months Ended						Nine Months Ended					
		Septer	otember 30 September 30								
(In millions, except per share amounts)		2013		2012		2013		2012			
Revenues from continuing operations	\$	740.0	\$	756.8	\$	2,215.2	\$	2,279.7			
Cost of services and products sold		567.9		580.0		1,709.3		1,758.4			
Selling, general and administrative expenses		124.0		121.9		374.3		376.7			
Long-lived asset impairment charge and transaction costs		253.7				253.7		_			
Other (income) expenses		(0.2)		2.4		2.2		65.4			
Operating income (loss) from continuing operations		(208.4)		50.4		(131.7)		72.3			
Interest expense		(12.8)		(11.6)		(37.4)		(36.0)			
Income tax expense from continuing operations		(10.8)		(13.5)		(27.3)		(28.5)			
Income (loss) from continuing operations		(231.2)		26.9		(193.8)		11.3			
Diluted earnings (loss) per common share from continuing operations attributable to Harsco Corporation common											
stockholders		(2.89)		0.32		(2.49)		0.13			
Effective income tax rate for continuing operations		(4.9)%		33.7%		(16.3)%		72.5%			

Comparative Analysis of Consolidated Results

Revenues

Revenues for the third quarter of 2013 decreased \$16.7 million or 2.2% from the third quarter of 2012. Revenues for the first nine months of 2013 decreased \$64.5 million or 2.8% from the first nine months of 2012. Changes in revenues for the periods presented were attributable to the following significant items:

Change in Revenues — 2013 vs. 2012	Three Months Ended	Nine Months Ended		
(In millions)	September 30, 2013		September 30, 2013	
Net decreased revenues in the Harsco Metals & Minerals Segment, primarily reflecting lower production by customers and reductions of \$8.8 million and \$30.7 million, for the three and nine months ended September 30, 2013, respectively, due to the exit from underperforming contracts.	\$ (5.0)	\$	(45.7)	
Net decreased revenues in the Harsco Rail Segment due principally to the timing and mix of equipment deliveries and decreased service volume, partially offset by strong parts sales.	(24.6)		(21.4)	
Impact of foreign currency translation.	(4.4)		(15.4)	
Net increased revenues in the Harsco Infrastructure Segment reflecting increasing activity in erection/dismantling services in the three month comparison and increasing rentals and erection/dismantling services activity, partially offset by decreases related to exited operations in certain countries in 2012 in the nine month comparison.	12.9		7.4	
Net increased revenues in the Harsco Industrial Segment, reflecting improving customer demand for industrial grating products, air-cooled heat exchangers, and industrial boilers.	4.4		10.6	
Total change in revenues — 2013 vs. 2012	\$ (16.7)	\$	(64.5)	

Cost of Services and Products Sold

Cost of services and products sold for the third quarter of 2013 decreased \$12.1 million or 2.1% from the third quarter of 2012. Cost of services and products sold for the first nine months of 2013 decreased \$49.1 million or 2.8% from the first nine months of 2012. Changes in cost of services and products sold for the periods presented were attributable to the following significant items:

Change in Cost of Services and Products Sold — 2013 vs. 2012	Three Months Ended		Nine Months Ended		
(In millions)	September 30, 2013		September 30, 2013		
Decreased costs due to changes in revenues (exclusive of the effect of foreign currency translation, and including the effect of restructuring program savings and the impact of fluctuations in commodity costs					
included in selling prices).	\$ (11.4)	\$	(31.0)		
Impact related to exited operations in certain countries in 2012 in the Harsco Infrastructure Segment.	—		(12.1)		
Impact of foreign currency translation.	(3.9)		(13.9)		
Other					
	3.2		7.9		
Total change in cost of services and products sold — 2013 vs. 2012	\$ (12.1)	\$	(49.1)		

Selling, General and Administrative Expenses

Selling, general and administrative expenses for the third quarter of 2013 increased \$2.1 million or 1.8% from the third quarter of 2012. The increase was primarily related to increased professional fees. Selling, general and administrative expenses for the first nine months of 2013 decreased \$2.3 million or 0.6% from the first nine months of 2012. The decrease was primarily related to decreased compensation costs partially offset by increased professional fees.

Long-lived Asset Impairment Charge and Transaction Costs

As a result of the Infrastructure transaction, the Company recorded an estimated non-cash long-lived asset impairment charge of \$241.3 million, which was recognized during the third quarter of 2013. The final non-cash long-lived asset impairment charge may differ due to a number of factors including, but not limited to (i) the valuation of certain assets and liabilities being transferred to the strategic venture; (ii) final capital expenditure and working capital adjustments required under the purchase agreement; (iii) the recognition of cumulative translation amounts; and (iv) the finalization of current and deferred income tax amounts. The Company determined that the estimated non-cash long-lived asset impairment charge was necessary given the fair value of the consideration received in the Infrastructure transaction. The amount of the estimated non-cash long-lived asset impairment charge represents the Company's estimate of the difference between the current book value of the Harsco Infrastructure Segment, less costs to sell, and the sum of the cash consideration plus the estimated fair value of the Company's approximate 29% equity interest in the strategic venture. The estimated fair value of the equity interest was determined based on the expected future discounted cash flows of the strategic venture. See Note 1, "Summary of Significant Accounting Policies," to the consolidated financial statements included in the Company's Annual Report on Form 10-K for the year ended December 31, 2012 for additional information on the Company's policy on impairment of long-lived assets (other than goodwill).

Additionally, the Company incurred \$12.4 million in transaction costs during the three months ended September 30, 2013 in conjunction with the Infrastructure transaction which are included in this caption.

Other (Income) Expenses

This income statement classification includes; restructuring costs for employee termination benefit costs, product rationalization and costs to exit activities; former CEO separation costs; net gains on the disposal of non-core assets; currency translation adjustments recognized in earnings; and business combination accounting adjustments for contingent consideration related to acquisitions by the Company. The most significant change in Other expenses relates to net gains on the disposal of non-core assets. Additional information on Other expenses is included in Note 14, "Other (Income) Expenses," in Part I, Item 1, Financial Statements.

	Three Mo	nths End	led	Nine Mor	iths Ended		
	 Septen	nber 30		Septer	nber 30	ber 30	
(In thousands)	2013		2012	2013		2012	
Restructuring Program costs (see Note 16)	\$ _	\$	8,454	\$ _	\$	73,563	
Former CEO separation costs	—		—	—		4,125	
Net gains	(563)		(2,736)	(5,132)		(3,958)	
Currency translation adjustments recognized in earnings	_		(4,152)	—		(10,906)	
Other (a)	335		817	7,290		2,527	
Other (income) expenses	\$ (228)	\$	2,383	\$ 2,158	\$	65,351	

(a) Other includes employee termination benefit costs, impaired asset write-downs, and costs to exit activities that are not directly related to the restructuring programs detailed in Note 16, "Restructuring Programs."

Interest Expense

Interest expense in 2013 increased \$1.2 million and \$1.4 million from the third quarter and first nine months of 2012, respectively. The increase in the third quarter and first nine months of 2013 compared with the same periods of 2012 reflects interest expense on higher net borrowings outstanding under the Company's Amended and Restated Five-Year Credit Agreement (as amended, the "Credit Agreement").

Income Tax Expense

Income tax expense from continuing operations decreased primarily due to the estimated non-cash long-lived asset impairment charge recorded as part of the Infrastructure transaction for the three and nine months ended September 30, 2013 compared with the three and nine months ended September 30, 2012. This decrease was partially offset by the valuation allowances recorded against deferred tax assets within certain foreign jurisdictions that the Harsco Infrastructure Segment operates and for the expected tax costs of cash repatriation from the Infrastructure transaction.

The effective income tax rate related to continuing operations for the three and nine months ended September 30, 2013 was (4.9)% and (16.3)%, respectively. The effective income tax rate related to continuing operations for the three and nine months ended September 30, 2012 was 33.7% and 72.5%, respectively. The effective income tax rate changed between 2012 and 2013 primarily due to the jurisdictional mix of the estimated non-cash long-lived asset impairment charge on the Infrastructure transaction compared with restructuring charges recorded during 2012, tax expense recorded for valuation allowances on deferred tax assets within certain foreign jurisdictions that the Harsco Infrastructure Segment operates, and for the expected tax costs of cash repatriation from the Infrastructure transaction.

Income (Loss) from Continuing Operations

The loss from continuing operations was \$231.2 million in the third quarter of 2013 compared with the \$26.9 million income from continuing operations in the third quarter of 2012. The loss from continuing operations was \$193.8 million in the first nine months of 2013 compared with the \$11.3 million income from continuing operations in the first nine months of 2012. The changes are primarily related to the \$253.7 million estimated non-cash long-lived asset impairment charge and transaction costs associated with the Infrastructure transaction which was recorded during the third quarter of 2013. Additionally, the Harsco Metals & Minerals Segment experience lower volumes during the third quarter and first nine months of 2013. Partially offsetting these decreases are pre-tax restructuring program costs incurred in 2012 that were not repeated in 2013. These costs were

\$8.5 million and \$73.6 million in the third quarter and first nine months of 2012, respectively, principally in the Harsco Infrastructure Segment.

Liquidity and Capital Resources

Overview

The Company continues to have sufficient available liquidity and has been able to obtain all necessary financing. The Company currently expects operational and business needs to be met with cash from operations supplemented with borrowings from time to time due to historical patterns of seasonal cash flow and for the funding of various projects.

The Company continues to implement and perform capital efficiency initiatives to enhance liquidity. These initiatives have included: focused allocation of capital spending to projects where the highest returns can be achieved while redeploying existing capital investments; optimization of worldwide cash positions; reductions in discretionary spending; and frequent evaluation of customer and business-partner credit risk.

During the first nine months of 2013, the Company's operations provided \$164.1 million in operating cash flow, an increase from the \$111.4 million provided in the first nine months of 2012. Approximately \$13 million of operating cash was disbursed in the first nine months of 2013 for restructuring program costs, which are described in Note 16, "Restructuring Programs," in Part I, Item I, Financial Statements, compared with approximately \$69 million of operating cash disbursed in the first nine months of 2012. In the first nine months of 2013, the Company invested \$181.7 million in capital expenditures, mostly for the Harsco Metals & Minerals Segment (approximately 50% of the total capital expenditures were for revenue-growth projects), compared with \$172.8 million invested in the first nine months of 2012. The Company paid \$49.7 million in common stock dividends in the first nine months of 2013 compared with \$49.5 million in the first nine months of 2012.

The Company's net cash borrowings increased by \$81.4 million in the first nine months of 2013, primarily to fund capital expenditures, principally in the Harsco Metals & Minerals Segment. During mid-September 2013, the Company entered into an additional amendment to the Company's Credit Agreement. In addition to certain administrative and conforming modifications, the amendment replaced the total consolidated debt to total consolidated capital ratio covenant of the Credit Agreement with a total consolidated debt to consolidated earnings before interest, taxes, depreciation and amortization ("EBITDA") ratio covenant, which is not to exceed 3.50 to 1.00. The Company's total consolidated debt to consolidated EBITDA ratio was 2.60 to 1.00 at September 30, 2013.

The Company plans to sustain its balanced portfolio through its strategy of redeploying discretionary cash for disciplined organic growth and international or market-segment diversification; for potential strategic ventures, alliances and partnerships; for growth in long-term, high-return service contracts for the Harsco Metals & Minerals Segment, principally in targeted growth markets or for customer diversification; and for strategic investments or possible acquisitions in the Harsco Rail Segment and Harsco Industrial Segment. The Company also foresees continuing its long and consistent history of paying dividends to stockholders.

The Company continues its focus on improving working capital efficiency. The Company's Continuous Improvement initiatives are being used to further improve effective and efficient use of working capital, particularly in accounts receivable and inventories.

The Company also generated \$16.9 million and \$42.9 million in cash from asset sales in the first nine months of 2013 and 2012, respectively. Asset sales are a normal part of the Company's business model, primarily for the Harsco Infrastructure and Harsco Metals & Minerals Segments.

Sources and Uses of Cash

The Company's principal sources of liquidity are cash from operations and borrowings under its Credit Agreement, augmented by cash proceeds from asset sales. The primary drivers of the Company's cash flow from operations are the Company's revenues and income. The Company's long-term Harsco Metals & Minerals Segment contracts provide predictable cash flows for the near-term years. Cash returns on capital investments made in prior years, for which limited cash is currently required, are a significant source of cash from operations. Depreciation expense related to these investments is a non-cash charge.

Major uses of operating cash flows and borrowed funds include: capital investments, principally in the Harsco Metals & Minerals Segment; payroll costs and related benefits; dividend payments; pension funding payments; inventory purchases for the Harsco Rail and Harsco Industrial Segments; income tax payments; debt principal and interest payments; insurance premiums and payments of self-insured casualty losses; and machinery, equipment, automobile and facility lease payments. Additionally, in 2012, there were significant payments under restructuring programs as noted above. Cash may also be used for targeted strategic acquisitions as appropriate opportunities arise.

Resources available for cash requirements for operations and growth initiatives

In addition to utilizing cash from operations and cash proceeds from asset sales, the Company has bank credit facilities available throughout the world. Public markets are also accessed through discrete-term note issuance to investors. The Company also utilizes capital leases to finance the acquisition of certain equipment when appropriate which allows the Company to minimize capital expenditures. The Company expects to continue to utilize all these sources to meet its future cash requirements for operations and growth initiatives.

The following table illustrates available credit at September 30, 2013:

	September 30, 2013 Exactly Limit Outstanding Balance Available Credit \$ 525.0 \$ 302.0 \$ 223.			
(In millions)	Facility Limit			
Multi-year revolving credit agreement (a U.Sbased program)	\$ 525.0	\$ 302.0	\$ 223.0	

At September 30, 2013 and December 31, 2012, the Company had \$302.0 million and \$89.5 million, respectively, of commercial paper and Credit Agreement borrowings outstanding. At September 30, 2013, \$266.0 million was classified as short-term borrowings and \$36.0 million was classified as long-term borrowings in the Condensed Consolidated Balance Sheets. At December 31, 2012, all such balances were classified as long-term borrowings in the Condensed Consolidated Balance Sheets. Classification of such balances is based on the Company's ability and intent to repay such amounts over the subsequent twelve months, as well as the Company's current intent and ability to borrow for a period longer than a year. To the extent the Company expects to repay any amounts within the subsequent twelve months, the amounts are classified as short-term borrowings. Upon the close of the Infrastructure transaction, Harsco will receive approximately \$300 million, subject to working capital and other adjustments, in cash, which will significantly strengthen the Company's balance sheet and enable the Company to reallocate capital to higher-return opportunities.

During the third quarter of 2013, the Company's U.S. commercial program paper was canceled. Additionally, the Company retired the \$150.0 million aggregate principal amount of 5.125% notes, due September 15, 2013, at maturity using borrowings under the Company's Credit Agreement.

Credit Ratings and Outlook

The following table summarizes the Company's current debt ratings:

Rating Agency	Long-term Notes	Watch / Outlook
Standard & Poor's (S&P)	BBB-	Negative Watch
Moody's	Ba1	Stable Outlook
Fitch	BBB-	Negative Outlook

In September 2013, S&P reaffirmed the Company's BBB- long-term rating while placing the Company on negative watch, Moody's reaffirmed the Company's Ba1 long-term rating and stable outlook while Fitch reaffirmed the Company's BBB- long-term rating and negative outlook. These rating agency actions resulted from reviews initiated upon the announcement of the Infrastructure transaction.

Any future downgrades to the Company's credit ratings may increase borrowing costs to the Company, while an improvement in the Company's credit ratings may decrease such costs. However, any future downgrades in the Company's credit ratings will not reduce availability under the Company's Credit Agreement.



Table of Contents

Working Capital Position

Changes in the Company's working capital are reflected in the following table:

(Dollars in millions)	September 30December 3120132012		Increase (Decrease)	
Current Assets				
Cash and cash equivalents	\$ 115.8	\$ 95.3	\$ 20.6	
Trade accounts receivable, net	354.1	600.3	(246.1)	
Other receivables	29.6	39.8	(10.2)	
Inventories	176.8	236.5	(59.7)	
Assets held-for-sale	715.0	2.4	712.6	
Other current assets	74.2	92.2	(17.9)	
Total current assets	1,465.6	1,066.4	399.1	
Current Liabilities		-		
Short-term borrowings and current maturities	281.6	11.8	269.8	
Accounts payable	196.5	221.5	(24.9)	
Accrued compensation	53.8	94.4	(40.6)	
Income taxes payable	1.5	10.1	(8.6)	
Liabilities of assets held-for-sale	218.0	—	218.0	
Other current liabilities	208.9	299.8	(90.8)	
Total current liabilities	960.5	637.6	322.9	
Working Capital	\$ 505.1	\$ 428.9	\$ 76.3	
Current Ratio (a)	1.5	1.7		

(a) Calculated as Total current assets divided by Total current liabilities

The net \$76.3 million increase in working capital for the first nine months of 2013 is due primarily to the following factors:

- In connection with the Infrastructure transaction, the Company has reflected substantially all assets and liabilities of the Harsco Infrastructure Segment as Assets held-for-sale and Liabilities of assets held-for-sale, respectively, on the Condensed Consolidated Balance Sheets at September 30, 2013. This reclassification resulted in a net increase to working capital of approximately \$284 million; and
- This increase was partially offset by the reclassification from Long-term debt to Short-term borrowings of approximately \$266 million at September 30, 2013, as certain amounts outstanding under the Company's Credit Agreement are expected to be repaid with the cash proceeds from the Infrastructure transaction.

Additional factors affecting working capital for the first nine months of 2013 were as follows:

- Working capital was positively affected by an increase in Trade accounts receivable, net primarily due to the timing of collections (excluding the Harsco Infrastructure Segment which is now classified as Assets held-for-sale) and increased sales in the Harsco Industrial Segment;
- Working capital was positively affected by a decrease in Other current liabilities primarily due to a decrease in customer advances in the Harsco Rail Segment due to the delivery of certain machines; and
- Working capital was negatively impacted by an increase in Accounts payable primarily due to the timing of payments across all segments (excluding the Harsco Infrastructure Segment which is now classified as Liabilities of assets held-for-sale).

Certainty of Cash Flows

The certainty of the Company's future cash flows is underpinned by the long-term nature of the Company's metals services contracts and overall discretionary cash flows (operating cash flows plus cash from asset sales in excess of the amounts necessary for capital expenditures to maintain current revenue levels) generated by the Company. Historically, the Company has utilized these discretionary cash flows for growth-related capital expenditures, strategic acquisitions, debt repayment and dividend payments.

The types of products and services that the Company provides are not subject to rapid technological change, which increases the stability of related cash flows. Additionally, the Company believes each of its businesses in its balanced portfolio is a leader in the industries and major markets the Company serves. Due to these factors, the Company is confident in its future ability to generate positive cash flows from operations.

Cash Flow Summary

The Company's cash flows from operating, investing and financing activities, as reflected in the Condensed Consolidated Statements of Cash Flows, are summarized in the following table:

	Nine Months Ended September 30				
(In millions)		2013		2012	
Net cash provided (used) by:					
Operating activities	\$	164.1	\$	111.4	
Investing activities		(172.6)		(129.9)	
Financing activities		33.3		12.6	
Impact of exchange rate changes on cash		(4.3)		(1.5)	
Net change in cash and cash equivalents	\$	20.6	\$	(7.4)	

Cash Provided by Operating Activities — Net cash provided by operating activities in the first nine months of 2013 was \$164.1 million, an increase of \$52.7 million from the first nine months of 2012. The increase is primarily attributable to higher net income (excluding the estimated non-cash long-lived asset impairment charge), a reduction in restructuring payments and the timing of accounts payable disbursements, partially offset by the timing of accounts receivable collections.

Included in the Cash flows from operating activities section of the Condensed Consolidated Statement of Cash Flows is the caption Other assets and liabilities. For the nine months ended September 30, 2013 and 2012, decreases in this line item were \$57.2 million and \$34.7 million, respectively. A summary of the major components of this caption for the periods presented is as follows:

	Nine Months Ended September 30				
(In millions)		2013	2012		
Net cash provided by (used in):					
Change in net defined benefit pension liabilities	\$	(13.3)	\$	(10.6)	
Change in advance on contracts from customers		(17.5)		(35.2)	
Change in prepaid expenses		(6.4)		20.1	
Change in accrued taxes		(13.1)		(0.4)	
Other		(6.9)		(8.6)	
Total	\$	(57.2)	\$	(34.7)	

Cash Used by Investing Activities — Net cash used in investing activities in the first nine months of 2013 was \$172.6 million, an increase of \$42.6 million from the first nine months of 2012. The increase was primarily due to approximately \$25.9 million in reduced cash proceeds from sales of assets and an increase of approximately \$8.9 million in capital expenditures.

Cash Provided by Financing Activities — Net cash provided by financing activities in the first nine months of 2013 was \$33.3 million, an increase of \$20.7 million from the first nine months of 2012. The change was primarily due to an increase in year-over-year net cash borrowings.

Debt Covenants

During mid-September 2013, the Company entered into an additional amendment to the Company's Credit Agreement. In addition to certain administrative and conforming modifications, the amendment replaced the total consolidated debt to total consolidated capital ratio covenant of the Credit Agreement with a total consolidated debt to consolidated EBITDA ratio covenant, which is not to exceed 3.50 to 1.00.

Additionally, the Credit Agreement contains covenants that limit the proportion of subsidiary consolidated indebtedness to a maximum of 10% of consolidated tangible assets and a minimum total consolidated EBITDA to consolidated interest charges ratio of 3.00 to 1.00. The Company's 5.75% and 2.70% notes include covenants that require the Company to offer to repurchase the notes at 101% of par in the event of a change of control of the Company or disposition of substantially all of the Company's assets in combination with a downgrade in the Company's credit rating to non-investment grade. At September 30, 2013, the Company was in compliance with these covenants as the proportion of subsidiary consolidated EBITDA to consolidated tangible assets was 3.3%, the total consolidated debt to consolidated EBITDA ratio was 2.60 to 1.00 and total consolidated EBITDA to consolidated interest charges of 8.70 to 1.00. Based on balances at September 30, 2013, the Company could increase borrowings by \$383.0 million and still be in compliance with its debt covenants. Alternatively, keeping all other factors constant, the Company's EBITDA could decrease by \$109.4 million and the Company would still be

Table of Contents

within its covenants. The Company expects to continue to be in compliance with these debt covenants for at least the next twelve months.

Cash and Value-Based Management

The Company has various cash management systems throughout the world that centralize cash in various bank accounts where it is economically justifiable and legally permissible to do so. These centralized cash balances are then redeployed to other operations to reduce short-term borrowings and to finance working capital needs or capital expenditures. Due to the transitory nature of cash balances, they are normally invested in bank deposits that can be withdrawn at will or in very liquid short-term bank time deposits and government obligations. The Company's policy is to use the largest banks in the various countries in which the Company operates. The Company monitors the creditworthiness of its banks, and, when appropriate, adjusts its banking operations to reduce or eliminate exposure to less credit worthy banks.

At September 30, 2013, the Company's consolidated cash and cash equivalents included approximately \$109 million held by non-U.S. subsidiaries. At September 30, 2013, less than 10% of the Company's consolidated cash and cash equivalents had regulatory restrictions that would preclude the transfer of funds with and among subsidiaries. The cash and cash equivalents held by non-U.S. subsidiaries also included approximately \$41 million held in consolidated strategic ventures. The strategic venture agreements may require strategic venture partner approval to transfer funds with and among subsidiaries. While the Company's remaining non-U.S. cash and cash equivalents can be transferred with and among subsidiaries, the majority of these non-U.S. cash balances will be used to support the on-going working capital needs and continued growth of the Company's non-U.S. operations.

The Company plans to continue its strategy of targeted, prudent investing for strategic purposes for the foreseeable future and to make more efficient use of existing investments. The long-term goal of this strategy is to create stockholder value by improving the Company's EVA. Under this program, the Company evaluates strategic investments based upon the investment's economic profit. EVA equals after-tax operating profits less a charge for the use of the capital employed to create those profits. Therefore, value is created when a project or initiative produces a return above the risk-adjusted local country cost of capital. In the first nine months of 2013, EVA was lower than the first nine months of 2012.

The Company currently expects to continue paying dividends to stockholders. In October 2013, the Company declared its 255th consecutive quarterly cash dividends, payable in February 2014.

The Company's financial position and debt capacity should enable it to meet current and future requirements. As additional resources are needed, the Company should be able to obtain funds readily and at competitive costs. The Company intends to continue investing in high-return, organic growth projects and prudent, strategic alliances and ventures; reduce debt; and pay cash dividends as a means of enhancing stockholder value.

Recently Adopted and Recently Issued Accounting Standards

Information on recently adopted and recently issued accounting standards is included in Note 2, "Recently Adopted and Recently Issued Accounting Standards," in Part I, Item 1, Financial Statements.

ITEM 3. QUANTITATIVE AND QUALITATIVE DISCLOSURES ABOUT MARKET RISK

Market risks have not changed significantly from those disclosed in the Company's Annual Report on Form 10-K for the fiscal year ended December 31, 2012.

ITEM 4. CONTROLS AND PROCEDURES

Based on the evaluation required by Securities Exchange Act Rules 13a-15(b) and 15d-15(b), the Company's management, including the Chief Executive Officer and Chief Financial Officer, conducted an evaluation of the effectiveness of disclosure controls and procedures, as defined in Securities Exchange Act Rules 15a-15(e) and 15d-15(e), as of September 30, 2013. Based on that evaluation, the Chief Executive Officer and Chief Financial Officer concluded that the disclosure controls and procedures were effective as of September 30, 2013. There have been no changes in internal control over financial reporting that have materially affected, or are reasonably likely to materially affect, internal control over financial reporting during the third quarter of 2013.

PART II - OTHER INFORMATION

ITEM 1. LEGAL PROCEEDINGS

Information on legal proceedings is included in Note 10, "Commitments and Contingencies," in Part I, Item 1, Financial Statements.

ITEM 1A. RISK FACTORS

The Company's risk factors as of September 30, 2013 have not changed materially from those described in Part 1, Item 1A, "Risk Factors," of the Company's Annual Report on Form 10-K for the year ended December 31, 2012.

ITEM 6. EXHIBITS

See the Exhibit Index following the signature page to this Quarterly Report on Form 10-Q for a list of exhibits filed or furnished with this report, which Exhibit Index is incorporated herein by reference.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned thereunto duly authorized.

		HARSCO CORPORATION	
		(Registrant)	
DATE	November 7, 2013	/s/ F. NICHOLAS GRASBERGER, III	
_		F. Nicholas Grasberger, III	
		Senior Vice President and Chief Financial Officer	
		(Principal Financial Officer)	
DATE	November 7, 2013	/s/ BARRY E. MALAMUD	
		Barry E. Malamud	
		Vice President and Corporate Controller	
		(Principal Accounting Officer)	

EXHIBIT INDEX

Exhibit Number	Description
2.1*	Purchase Agreement, dated as of September 15, 2013, by and among Harsco Corporation, on behalf of itself and the other sellers named therein, Bullseye, Inc., on behalf of itself and the other buyers named therein, Bullseye Investors, Inc. and CD&R Bullseye Holdings, L.P.
10.1	Amendment No. 1, dated September 12, 2013, to the Amended and Restated Five-Year Credit Agreement, dated March 2, 2012, among Harsco Corporation, the lenders named therein, Citibank, N.A., as administrative agent, RBS Securities Inc., as syndication agent, and the Bank of Tokyo-Mitsubishi UFJ, Ltd., HSBC Bank USA, National Association, ING Bank N.V., Dublin Branch, JPMorgan Chase Bank, N.A. and Lloyds TSB Bank PLC, as documentation agents.
10.2	Retention and Severance Agreement, made as of October 27, 2013, by and between Harsco Corporation and Mark Kimmel.
31.1	Certification Pursuant to Rule 13a-14(a) or 15d-14(a), as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (Chief Executive Officer).
31.2	Certification Pursuant to Rule 13a-14(a) or 15d-14(a), as Adopted Pursuant to Section 302 of the Sarbanes-Oxley Act of 2002 (Chief Financial Officer).
32	Certifications Pursuant to 18 U.S.C. Section 1350, as Adopted Pursuant to Section 906 of the Sarbanes-Oxley Act of 2002 (Chief Executive Officer and Chief Financial Officer).
101	The following financial statements from Harsco Corporation's Quarterly Report on Form 10-Q for the quarter ended September 30, 2013, filed with the Securities and Exchange Commission on November 7, 2013, formatted in XBRL (Extensible Business Reporting Language): (i) the Condensed Consolidated Balance Sheets; (ii) the Condensed Consolidated Statements of Operations; (iii) the Condensed Consolidated Statements of Comprehensive Income (Loss); (iv) the Condensed Consolidated Statements of Cash Flows; (v) the Condensed Consolidated Statements of Equity; and (vi) the Notes to Condensed Consolidated Financial Statements.

*The registrant has omitted certain immaterial schedules and exhibits to this exhibit pursuant to the provisions of Regulation S-K, Item 601(b)(2). The registrant will furnish a copy of any of the omitted schedules and exhibits to the Securities and Exchange Commission upon request.

PURCHASE AGREEMENT

among

HARSCO CORPORATION

ON BEHALF OF ITSELF And the other Sellers NAMED HereIN,

BULLSEYE, INC.

ON BEHALF OF ITSELF AND THE OTHER BUYERS NAMED HEREIN,

BULLSEYE INVESTORS, INC.

and

CD&R BULLSEYE HOLDINGS, L.P.

dated as of September 15, 2013

TABLE OF CONTENTS

ARTICLE I PURCHASE AND SALE OF TARGET SHARES AND TARGET ASSETS	3
1.1 Purchase and Sale of Target Shares and Target Assets	3
1.2 Consideration	3
1.3 The Closing	5
1.4 UK Entity Closing	5
1.5 Deferred India Entity Closing	6
1.6 Deferred China Entities Closing	7
1.7 Deferred Baroom Closing	8
1.8 Deliveries at Closing	9
1.9 Post Closing Purchase Price Adjustment	11
1.10 Purchase Price Allocation	14
ARTICLE II REPRESENTATIONS AND WARRANTIES OF HARSCO AND SELLERS	15
2.1 Incorporation and Authority	15
2.2 Capitalization and Subsidiaries	16
2.3 No Conflict	17
2.4 Consents and Approvals	17
2.5 Financial Statements	17
2.6 No Undisclosed Material Liabilities	18
2.7 Litigation	19
2.8 Employee Benefits	19
2.9 Taxes	22
2.10 Absence of Charges	23
2.11 Title to Assets; Sufficiency of Assets	23
2.12 Real Property	24
2.13 Intellectual Property	24
2.14 Contracts	26
2.15 Compliance with laws; Permits	27
2.16 Anti-Corruption; Trade Controls	28
2.17 Insurance	29
2.18 Environmental Matters	29
2.19 Financial Advisors	30
2.20 Transactions with Affiliates	30
2.21 Customers and Suppliers	31
2.22 Investment Intent	31
2.23 Harsco Acknowledgment	31
2.24 Disclaimer of Other Representations	32
ARTICLE III REPREESENTATIONS AND WARRANTIES OF CD&R INVESTOR	32
3.1 Incorporation and Authority	
3.2 No Conflict	
3.3 Consents and Approvals	33
3.4 Litigation	34
0	

3.5 Financial Advisors	34
3.6 Financing	34
3.7 Guarantee	35
3.8 Investment Intent	35
3.9 CD&R Investor Acknowledgment	35
3.10 Agreements with CD&R Investor	36
3.11 Disclaimer of Other Representations	36
ARTICLE IV CERTAIN COVENANTS	36
4.1 Conduct of Business Prior to the Closing	36
4.2 Cooperation	39
4.3 Access to Information; Other Contacts	41
4.4 Books and Records; Access; Assistance	42
4.5 Confidentiality	42
4.6 Employees	43
4.7 WARN Act	50
4.8 Intercompany Arrangements; Transactions	51
4.9 Insurance	51
4.10 Financing	53
4.11 Debt Financing Cooperation	54
4.12 Intellectual Property Assignments	58
4.13 Internal IT Systems and Data Separation; Replication of Shared Contract	58
4.14 Certain Payments	59
4.15 Further Assurances	59
4.16 Harsco House Lease	59
4.17 Harsco Letters of Credit; Other Obligations	59
4.18 Indemnification and Insurance	60
4.19 Brand Purchase Agreement	61
4.20 Options to Effect Sale of French Business	61
4.21 Netherlands Business	62
4.22 Real Property	62
4.23 Middle East Arrangements	62
4.24 Repatriation Schedule	63
ARTICLE V TAX MATTERS	63
5.1 Tax Indemnification	63
5.2 Procedures Relating to Certain Tax Indemnification	64
5.3 Tax Returns	65
5.4 Transaction Taxes	66
5.5 Records Retention	66
5.6 Tax Sharing Agreements	66
5.7 Straddle Period	67
5.8 Waiver of Loss Carrybacks	67
5.9 [Intentionally Omitted]	67
5.10 Computational Matters; Indemnification Cap	67
5.11 Certain Tax Elections; Actions	67

5.12 Exclusivity	69
ARTICLE VI CONDITIONS TO THE CLOSING	69
6.1 Conditions to Obligations of the Sellers	69
6.2 Conditions to Obligations of CD&R Investor and Buyer	70
6.3 Frustration of Closing Conditions	71
ARTICLE VII CERTAIN RESTRICTIONS	71
7.1 Non Competition	71
7.2 Non-Solicitation	72
7.3 Specific Performance	72
7.4 Severability	73
7.5 Intellectual Property Matter	73
ARTICLE VIII INDEMNIFICATION	75
8.1 Indemnification by CD&R Investor	75
8.2 indemnification by Harsco	75
8.3 Indemnification by Buyer	73
8.4 Limitations on Indemnification	78
8.5 Sole Remedy	70
8.6 Computation of indemnity Payments	80
8.7 Procedures for Indemnification	81
8.8 Treatment of Indemnification Payments	82
o.o freatment of indemnification rayments	02
ARTICLE IX TERMINATION AND WAIVER	82
9.1 Termination	82
9.2 Effect of Termination	83
9.3 Enforcement	85
9.4 Waiver	86
ARTICLE X GENERAL PROVISIONS	86
10.1 Expenses	86
10.2 Notices	86
10.3 Severability	88
10.4 Entire Agreement	88
10.5 Assignment	88
10.6 No Third Party Beneficiaries	89
10.7 Amendment.	89
10.8 Governing Law; Jurisdiction	89
10.9 Public Announcements	90
10.10 Waiver Jury Trial	90
10.11 Disclosure Generally	90
10.12 No Presumption Against Drafting Party	91
10.13 Time Periods	91
10.14 Execution of Agreement	91
10.15 Harsco and CD&R Investor	91

ARTICLE XI CERTAIN DEFINITIONS	91
11.1 Certain Defined Terms	91
11.2 Interpretation	109

EXHIBIT A - LIST OF STOCK SELLERS AND STOCK BUYERS
EXHIBIT B - LIST OF ASSET SELLERS AND ASSET BUYERS
EXHIBIT C - FORM OF INVESTOR RIGHTS AGREEMENT
EXHIBIT D - FORM OF TRANSITION SERVICES AGREEMENT
EXHIBIT E - BRAND PURCHASE AGREEMENT
EXHIBIT F - PRE-CLOSING TRANSACTIONS
EXHIBIT G - CLOSING TRANSACTIONS
EXHIBIT H - NET WORKING CAPITAL STATEMENT METHODOLOGY
EXHIBIT I - CD&R CONSULTING AGREEMENT
EXHIBIT J - CD&R INDEMNIFICATION AGREEMENT
EXHIBIT K - HARSCO CONSULTING AGREEMENT
EXHIBIT L - HARSCO INDEMNIFICATION AGREEMENT
EXHIBIT M - FORM OF UPPER-TIER PARTNERSHIP AGREEMENT
EXHIBIT N - REPATRIATION SCHEDULE

iv

PURCHASE AGREEMENT

This PURCHASE AGREEMENT (this "<u>Agreement</u>"), dated as of September 15, 2013, is entered into by and among Harsco Corporation, a Delaware corporation ("<u>Harsco</u>"), on behalf of itself, the stock sellers set forth on <u>Exhibit A</u> (collectively, the "<u>Stock Sellers</u>") and the asset sellers set forth on <u>Exhibit B</u> (collectively, the "<u>Asset Sellers</u>" and, together with the Stock Sellers and Harsco, the "<u>Stock Buyers</u>"), Bullseye, Inc., a Delaware corporation ("<u>Buyer</u>"), on behalf of itself, the stock buyers set forth on <u>Exhibit A</u> (collectively, the "<u>Stock Buyers</u>") and the asset buyers set forth on <u>Exhibit B</u> (collectively, the "<u>Asset Buyers</u>" and, together with the Stock Buyers and Buyer, the "<u>Buyers</u>"), Bullseye Investors, Inc., a Delaware corporation (the "<u>Company</u>"), and CD&R Bullseye Holdings, L.P., a Delaware limited partnership ("<u>CD&R Investor</u>" and, together with Buyer, the Company and Harsco, the "<u>Parties</u>").

WHEREAS, the Stock Sellers will own immediately prior to the Closing the issued and outstanding shares of capital stock or other equity securities designated on <u>Schedule 2.2(a)</u> (the "<u>Target Shares</u>") of the entities identified with an asterisk on <u>Schedule 2.2(a)</u> under the column titled "Target Entity";

WHEREAS, the Asset Sellers own or will own immediately prior to the Closing all of the assets, properties and rights primarily used in the conduct of the Business by such Asset Sellers (the "<u>Target Assets</u>");

WHEREAS, Buyer is, and immediately prior to Closing will be, an indirect, wholly-owned subsidiary of the Company, and the Company immediately prior to the Closing will be a wholly-owned subsidiary of Bullseye Holdings, L.P., a Cayman Islands exempted limited partnership ("<u>Holdings Partnership</u>");

WHEREAS, immediately prior to Closing, (i) the sole limited partner of Holdings Partnership will be Bullseye Partnership, L.P., a Cayman Islands exempted limited partnership ("<u>Upper-Tier Partnership</u>"), (ii) the sole limited partner of Upper-Tier Partnership will be CD&R Investor and (iii) the general partner of each of Holdings Partnership and Upper-Tier Partnership will be Bullseye G.P., LLC, a Delaware limited liability company and a wholly-owned subsidiary of CD&R Investor.

WHEREAS, the Sellers and Buyers desire to enter into a series of transactions in connection with the Closing, pursuant to which: (<u>i</u>) the Sellers, the Target Entities and certain of the Sellers' other Subsidiaries shall consummate the Pre-Closing Transactions (as defined below); (<u>ii</u>) the Sellers shall sell the Target Shares (other than the UK Shares) and the Target Assets to the Buyers in exchange for cash in the aggregate amount of three hundred million dollars (\$300,000,000) (as adjusted pursuant to this Agreement) and 2,250,000 shares of common stock of the Company (less the Partnership Company Shares to be issued pursuant to <u>Section 1.4</u>) (the "<u>Harsco Company Shares</u>"); (iii) following the consummation of clause (i) and (ii) of this Recital, Harsco shall contribute the Harsco Company Shares to Upper-Tier Partnership in exchange for 1,914,930 Class B Units of Upper-Tier Partnership, which when combined with the Class B Units of Upper-Tier Partnership to be issued pursuant to <u>Section 1.4</u>, will represent an indirect 29% equity interest in the Company immediately following the Closing (excluding the effect of any equity held by management of the Company and its Subsidiaries) (the "<u>Harsco Partnership Interests</u>"); and (<u>iv</u>) following the consummation of clause (iii) of this Recital, Upper-Tier Partnership shall contribute the Harsco Company Shares to Holdings Partnership in exchange for 1,914,930 units of Holdings Partnership;

WHEREAS, immediately following the Closing, and as a result of the Transactions, (<u>i</u>) CD&R Investor will own 4,650,000 Class A Units of the Upper-Tier Partnership, which after taking into account

the UK Entity Contribution to the Upper-Tier Partnership contemplated by <u>Section 1.4</u>, will represent an indirect 60% equity interest in the Company immediately following the Closing (excluding the effect of any equity held by management of the Company and its Subsidiaries) (the "<u>CD&R Investor Partnership Interests</u>"), and (<u>ii</u>) CD&R Investor and/or other investors (which, for the avoidance of doubt, may include management of the Company's subsidiaries following the Closing who purchase or "roll over" equity) will collectively own 850,000 units of Holdings Partnership which, after taking into account the UK Entity Contribution to the Upper-Tier Partnership contemplated by <u>Section 1.4</u>, will represent an indirect 11% equity interest in the Company immediately following the Closing (excluding the effect of any equity held by management of the Company and its Subsidiaries);

WHEREAS, in connection with the Transactions, CD&R Investor and Harsco desire to enter into an Investor Rights Agreement substantially in the form attached as <u>Exhibit C</u> (the "<u>Investor Rights Agreement</u>") to set forth the governance of the Company following the Closing and to provide for, among other things, transfer restrictions, preemptive rights, consent rights, registration rights and other obligations and rights, in each case, on the terms and conditions contained in the Investor Rights Agreement;

WHEREAS, in connection with the Transactions, Harsco and Buyer desire to enter into a transition services agreement in substantially the form attached as <u>Exhibit D</u> hereto (the "<u>Transition Services Agreement</u>") to provide for certain services and other arrangements among Harsco and/or the Non-Company Affiliates, on the one hand, and Buyer and/or the Target Entities, on the other hand, following the Closing as more fully described herein and therein;

WHEREAS, concurrently with the execution and delivery of this Agreement and as a condition and material inducement to the Sellers' execution of this Agreement, Clayton, Dubilier & Rice Fund IX, L.P., a Cayman Islands exempted limited partnership (the "<u>Guarantor</u>") is executing and delivering to Harsco a limited guarantee, pursuant to which, and subject to the terms and conditions thereof, the Guarantor has guaranteed certain obligations of Buyer hereunder; and

WHEREAS, concurrently with the execution of this Agreement, Buyer and FR Brand Management L.P. ("<u>FRC</u>") are entering into a Sale and Purchase Agreement, dated as of the date of this Agreement (together with all annexes and exhibits thereto, the "<u>Brand Purchase Agreement</u>"), a copy of which is attached as <u>Exhibit E</u> to this Agreement, pursuant to which Buyer will, upon the terms and subject to the conditions set forth in the Brand Purchase Agreement, acquire from FRC, and FRC will cause to be sold to Buyer, all of the issued and outstanding shares of capital stock of Brand Energy, Inc. ("<u>Brand</u>").

NOW, THEREFORE, in consideration of the mutual agreements and covenants hereinafter set forth, and for good and valuable consideration, the receipt, sufficiency and adequacy of which are hereby acknowledged, the Parties hereby agree as follows:

ARTICLE I

PURCHASE AND SALE OF TARGET SHARES AND TARGET ASSETS

1.1 Purchase and Sale of Target Shares and Target Assets

(a) Subject to the terms and conditions set forth in this Agreement, on the Closing Date, the Sellers shall sell, assign and transfer to the Buyers, and the Buyers shall purchase and acquire, all of such Sellers' right, title and interest in and to, the Target Shares (other than the UK Shares) and the Target Assets, free and clear of all Encumbrances other than (<u>A</u>) Encumbrances that may be created by or on behalf of the Buyers, (<u>B</u>) any Encumbrance imposed by federal and state securities Laws, and (<u>C</u>) in the case of the Target Assets, Permitted Encumbrances.

(b) Notwithstanding anything to the contrary herein, the assets and properties of Harsco, the Asset Sellers or Target Entities set forth on <u>Schedule 1.1(b)</u> (the "<u>Excluded Assets</u>") shall (<u>i</u>) be retained by Harsco or the applicable Asset Seller, or (<u>ii</u>) will be transferred to Harsco or a Non-Company Affiliate, as elected by Harsco, prior to Closing.

1.2 Consideration

(a) On the Closing Date and subject to the terms and conditions set forth in this Agreement:

(i) first, in consideration of the sale, assignment and transfer of the Target Shares (other than the UK Shares) and the Target Assets to the Buyers, (\underline{x}) the applicable Asset Buyers shall assume and thereafter pay, discharge and perform in accordance with their terms, all Liabilities of the applicable Asset Sellers to the extent arising from or relating to the Target Assets or the Business, irrespective of whether the same shall arise prior to, on, or following the Closing Date, other than the Excluded Target Liabilities (the "<u>Assumed Target Liabilities</u>"), (y) the applicable Buyer shall assign and transfer, or cause to be assigned and transferred, to Harsco, for the benefit of the applicable Seller, as set forth on <u>Exhibit G</u>, the Harsco Company Shares free and clear of Encumbrances (other than any Encumbrance imposed by federal and state securities Laws or the Ancillary Agreements) and (\underline{z}) Buyer shall pay, or cause to be paid, to Harsco, for the benefit of the Sellers, by wire transfer of immediately available funds in dollars, an amount in cash (the "<u>Initial Cash Purchase Price</u>," and, as adjusted pursuant to <u>Section 1.9</u>, and together with the Harsco Company Shares at the Closing, the "<u>Purchase Price</u>") equal to the sum of:

(1) three hundred million dollars (\$300,000,000) <u>less</u>, (a) if the French Sale Option has not been exercised, \$53,037,000 (which reflects the amount attributable to the purchase of the French Business) and (b) if the Consultations have not been completed pursuant to <u>Section 4.21</u>, \$47,610,000 (which reflects the amount attributable to the purchase of the Netherlands Business); <u>plus</u>

- (2) the Estimated Closing Cash; minus
- (3) the Estimated Closing Indebtedness; plus
 - 3

(4) the excess, if any, of the Estimated Closing Net Working Capital over the Target Net Working Capital; <u>minus</u>

(5) the excess, if any, of the Target Net Working Capital over the Estimated Closing Net Working Capital; <u>minus</u>

(6) the Estimated Closing Cap Ex Shortfall; and

(ii) second, following the consummation of the transactions set forth in clause (i) above, Harsco shall contribute the Harsco Company Shares to Upper-Tier Partnership in exchange for the Harsco Partnership Interests free and clear of Encumbrances (other than Encumbrances imposed by federal or state securities laws or the Ancillary Agreements).

(b) At least three (3) Business Days prior to the Closing Date, Harsco shall deliver to Buyer (<u>i</u>) an estimate of the Closing Cash (the "<u>Estimated Closing Cash</u>"), (<u>ii</u>) an estimate of the Closing Indebtedness (the "<u>Estimated Closing Indebtedness</u>"), (<u>iii</u>) an estimate of the Closing Net Working Capital (the "<u>Estimated Closing Net Working Capital</u>") and (<u>iv</u>) an estimate of the Closing Cap Ex Shortfall (the "<u>Estimated Closing Cap Ex Shortfall</u>"), in each case, calculated in good faith as of immediately prior to the Closing and assuming full completion of the Transactions. The items identified in clauses (i) through (iv), collectively, the "<u>Estimated Closing Statement</u>".

(c) Buyer shall not withhold any Tax from the Purchase Price actually paid to the Sellers pursuant to this <u>Section 1.9</u> except to the extent that Buyer is required by applicable Law (taking into account any applicable Tax treaty) to withhold and remit to the applicable Tax Authority any such Tax (in which case any such amount withheld and remitted to the applicable Tax Authority shall be treated for all purposes of this Agreement as having been paid); <u>provided</u>, <u>however</u>, that if either party becomes aware that any amount is required to be so withheld, it shall promptly (in any event, no later than two (2) Business Days prior to any such withholding) notify the other party of any such required withholding and the Parties shall cooperate with each other in good faith (including by taking all reasonable actions requested by the other party to the extent such actions would not reasonably be expected to have a material adverse impact on the requested party) to minimize or eliminate such withholding Taxes; <u>provided further</u> that Buyer shall provide to Harsco within thirty (30) days of any such Taxes to the extent such a receipt is available, a copy of the return reporting the payment of such Taxes or other evidence of such payment reasonably satisfactory to Harsco. Notwithstanding the foregoing, the Purchase Price is exclusive of any amount arising in respect of any applicable VAT, which shall be payable to the relevant Seller at the same time as the Purchase Price, <u>provided</u> that, for the avoidance of doubt, any such VAT shall be borne by the Parties in accordance with <u>Section 5.4</u>.

(d) For the avoidance of doubt, each of the Target Shares and each Target Asset to be acquired by a Buyer (as contemplated by the Closing Transactions) shall be transferred by the relevant Seller directly to such Buyer and such Buyer shall pay, in U.S. dollars, the portion of the Purchase Price that corresponds to the allocation to such Target Shares or Target Asset determined pursuant to Section 1.10 (as adjusted by Section 1.9) to Harsco for the benefit of the relevant Seller. Notwithstanding anything contained in this Agreement to the contrary, appropriate reconciliation payments shall be made between the relevant Buyer and Harsco, on behalf of the relevant Seller, to reflect the Final Closing Statement (as determined in good faith by the Parties, which reconciliation payments may be further adjusted as mutually agreed by the Parties). To the extent applicable, indemnity or Adjustment Amount

payments required to be made under this Agreement shall be made between the relevant Buyer and Harsco, on behalf of the relevant Seller.

1.3 The Closing

Subject to <u>Sections 1.4</u>, <u>1.5</u> and <u>1.6</u>, unless this Agreement shall have been terminated pursuant to <u>ARTICLE IX</u>, the closing of the transactions contemplated by this Agreement (the "<u>Closing</u>") shall take place at 10:00 a.m. New York time at the offices of Debevoise & Plimpton, 919 Third Avenue, New York, New York, 10022, (i) on the third Business Day after each of the conditions set forth in <u>ARTICLE VI</u> has been satisfied or waived (other than those conditions that by their terms are to be satisfied by actions taken at the Closing, but subject to the satisfaction or written waiver of those conditions); <u>provided</u> that, if the Marketing Period has not ended as of such time, the Closing shall not occur until the earlier to occur of (<u>1</u>) a date during the Marketing Period specified by Buyer on not less than three (<u>3</u>) Business Days' written notice to Harsco and (<u>2</u>) the third Business Day following the final day of the Marketing Period (subject in each case to the satisfaction or waiver of all of the conditions set forth in <u>ARTICLE VI</u> as of the date determined pursuant to this proviso), or (<u>ii</u>) at such other date, time or place as is agreed to in writing by the Parties. The date on which the Closing occurs is the closing date (the "<u>Closing Date</u>"). Unless the Parties agree otherwise, the Closing will be deemed to have occurred at 11:59 p.m. local time in each applicable jurisdiction on the Closing Date.

1.4 UK Entity Closing

(a) Notwithstanding anything to the contrary contained in this Agreement (other than <u>ARTICLE IX</u>), Harsco shall cause Harsco (UK) Group Ltd ("<u>HUG</u>") to contribute all of HUG's right, title and interest in and to, the issued and outstanding shares of capital stock or other equity securities (the "<u>UK Shares</u>") of Harsco Infrastructure Middle East Limited, a company organized under the laws of the United Kingdom (the "<u>UK Entity</u>"), free and clear of all Encumbrances other than (<u>A</u>) Encumbrances that may be created by or on behalf of the Holdings Partnership and (<u>B</u>) any Encumbrance imposed by federal and state securities Law, to Holdings Partnership (the "<u>UK Entity Contribution</u>") on the later of (<u>i</u>) the Closing Date and (<u>ii</u>) January 2, 2014 (the "<u>UK Closing Date</u>") and, in exchange for such UK Entity Contribution, Holdings Partnership shall issue to HUG a number of units in Holdings Partnership (the "<u>Holdings Units</u>") corresponding to the value of the UK Shares (as reflected on <u>Schedule 1.10</u>).

(b) Immediately following the UK Entity Contribution, (i) Harsco shall cause HUG to contribute the Holdings Units to Upper-Tier Partnership (the "<u>Partnership Contribution</u>") in exchange for a number of Class B Units in Upper-Tier Partnership corresponding to the value of the Holdings Units and (<u>ii</u>) Holdings Partnership shall contribute all of its right, title and interest in and to, the UK Shares to the Company (the "<u>Company Contribution</u>") in exchange for shares of common stock of the Company corresponding to the value of the UK Shares (as reflected on <u>Schedule 1.10</u>) (the "<u>Partnership Company Shares</u>"). Immediately following the Company Contribution, the Company shall cause the UK Shares to be contributed to one or more (direct or indirect) wholly-owned subsidiaries of Company (the "<u>Contributions</u>"). If the UK Closing occurs on or before January 1, 2014, each of the Contributions shall be structured in a manner intended to qualify for tax free treatment under Section 351 of the Code.

(c) The transactions contemplated by this <u>Section 1.4</u> shall take place at a closing (the "<u>UK Closing</u>") at the offices of Debevoise & Plimpton, 919 Third Avenue, New York, New York, 10022, on the UK Closing Date.

(d) At the UK Closing, the Parties hereto shall, and shall cause their respective Affiliates to, execute and deliver such documents and instruments as may be reasonably necessary to evidence the transfer of the equity interests in the UK Entity.

(e) Notwithstanding anything contained herein to the contrary, if the Closing shall have occurred in 2013, (<u>i</u>) there shall be no conditions required to be satisfied or waived prior to the UK Closing in order to consummate the transactions contemplated by this <u>Section 1.4</u>, (<u>ii</u>) during the period between the Closing Date and the UK Closing Date, the Parties hereto shall, and shall cause their respective Affiliates to, cooperate fully and use commercially reasonable efforts to take such actions with respect to the UK Entity as may be reasonably requested by the other Party hereto in order to permit the transfer of the UK Entity in accordance with this <u>Section 1.4</u> as promptly as practicable and (<u>iii</u>) from the Closing Date to the UK Closing Date, unless the context clearly requires otherwise, all references in this Agreement to the "Closing" or the "Closing Date" shall, with respect to the UK Entity and its Subsidiaries, be deemed to refer to the UK Closing or the UK Closing Date, as applicable, and the Parties hereto shall continue to comply with all covenants and agreements contained in this Agreement that are required by their terms to be performed prior to the Closing in respect of the UK Entity.

(f) Notwithstanding anything to the contrary contained herein, if the UK Closing occurs after the Closing Date, then, (i) for purposes of applying <u>ARTICLE V</u> of this Agreement with respect to the UK Entity, (<u>A</u>) the "Closing Date" shall be deemed to mean the UK Closing Date, (<u>B</u>) the "Closing" shall be deemed to mean the closing of the acquisition of the UK Entity on the UK Closing Date, (<u>C</u>) "Pre-Closing Tax Period" shall be deemed to mean (<u>1</u>) any Tax period ending on or before the UK Closing Date and (<u>2</u>) in the case of any Straddle Period, the portion of such period ending after the UK Closing Date and (<u>2</u>) in the case of any Straddle Period, the portion of such period ending after the UK Closing Date and (<u>2</u>) in the case of any Straddle Period beginning after, but not including, the UK Closing Date and (<u>E</u>) "Straddle Period" shall be deemed to mean, as the context requires, any Tax period with respect to the UK Entity that begins before but ends after the UK Closing Date, (<u>ii</u>) any action taken by the UK Entity or any of its Subsidiaries in connection with the UK Closing shall be deemed to be a Pre-Closing Transaction and (<u>iii</u>) other appropriate adjustments shall be made to the application of this Agreement to give effect to the intent of this <u>Section 1.4(f)</u>.

1.5 Deferred India Entity Closing

(a) Notwithstanding anything to the contrary contained in this Agreement (other than <u>ARTICLE IX</u>), the assignment, transfer and delivery of the equity interests of the Deferred India Entity, shall occur on the later of (<u>i</u>) the Closing Date and (<u>ii</u>) the third Business Day after the consummation of the transactions and receipt of the consents set forth on <u>Schedule 1.5(a)</u> (the "<u>Deferred India Closing Date</u>").

(b) The assignment, transfer, conveyance and delivery of the Deferred India Entity shall take place at a closing (the "<u>Deferred India Closing</u>") at the offices of Debevoise & Plimpton, 919 Third Avenue, New York, New York, 10022, on the Deferred India Closing Date.

(c) At the Deferred India Closing, the Parties hereto shall, and shall cause their respective Affiliates to, execute and deliver such documents and instruments as may be reasonably necessary to evidence the transfer of the equity interests in the Deferred India Entity.

(d) Notwithstanding anything contained herein to the contrary, but subject to the consummation of the transactions and receipt of the consents set forth on <u>Schedule 1.5(a)</u> and <u>Section 1.5</u>

(<u>h</u>), if the Closing shall have occurred on an earlier date than the Deferred India Closing Date, there shall be no conditions required to be satisfied or waived prior to the Deferred India Closing in order to consummate the transactions contemplated by <u>Section 1.5(b)</u>.

(e) During the period between the Closing Date and the Deferred India Closing Date, the Parties hereto shall, and shall cause their respective Affiliates to, cooperate fully and use commercially reasonable efforts to take such actions with respect to the Deferred India Entity as may be reasonably requested by the other Party hereto in order to permit the transfer of the Deferred India Entity in accordance with <u>Section 1.5(b)</u> as promptly as practicable. The consummation of the transactions and receipt of the consents set forth in Item 2(a) of <u>Schedule 1.5(a)</u> shall be on terms no less favorable to the Target Entities than those contained in the draft documents provided to CD&R Investor as of the date hereof unless CD&R Investor provides written consent to the relevant terms (which consent shall not be unreasonably withheld or delayed) and the consummation of the transactions set forth in Item 2(b) of <u>Schedule 1.5(a)</u> shall be in accordance with the Shareholders Agreement (as defined in <u>Schedule 1.5(a)</u>).

(f) Prior to the occurrence of the Deferred India Closing, the Deferred India Entity and the assets, operations and liabilities of the India Business shall be held for the account of Buyer and its subsidiaries.

(g) From the Closing Date to the Deferred India Closing Date, unless the context clearly requires otherwise, all references in this Agreement to the "Closing" or the "Closing Date" shall, with respect to the Deferred India Entity and its Subsidiaries, be deemed to refer to the Deferred India Closing or the Deferred India Closing Date, as applicable, and the Parties hereto shall continue to comply with all covenants and agreements contained in this Agreement that are required by their terms to be performed prior to the Closing in respect of the Deferred India Entity.

(h) If the Deferred India Closing Date has not occurred by the later of (i) ninety (90) days following the Closing Date and (ii) May 14, 2014, the Parties shall, at the request of Buyer, take all actions necessary to unwind the purchase and sale of the Deferred India Entity and Seller shall, within five (5) Business Days, refund Buyer (or a subsidiary of Buyer selected by Buyer) the purchase price amount that was paid to a Stock Seller in respect of the Deferred India Entity pursuant to the Transactions (and in accordance with the allocations set forth in <u>Schedule 1.10</u>), together with interest thereon at the Prime Rate from and until the date of such payment. Following such unwinding and refund, no Party shall be obligated to consummate the Deferred India Closing.

1.6 Deferred China Entities Closing

(a) Notwithstanding anything to the contrary contained in this Agreement (other than <u>ARTICLE IX</u>), the assignment, transfer and delivery of the equity interests in Harsco Infrastructure Equipment Leasing (Beijing) Co. Ltd., a company organized under the laws of the People's Republic of China, and Harsco Infrastructure Zhejiang Co. Ltd., a company organized under the laws of the People's Republic of China (collectively the "<u>Deferred China Entities</u>"), shall occur on the later of (<u>i</u>) the Closing Date and (<u>ii</u>) the third Business Day after receipt of the approval set forth in <u>Schedule 1.6(d)</u> (the "<u>Deferred China Closing Date</u>").

(b) The assignment, transfer, conveyance and delivery of the Deferred China Entities shall take place at a closing (the "<u>Deferred China Closing</u>") at the offices of Debevoise & Plimpton, 919 Third Avenue, New York, New York, 10022, on the Deferred China Closing Date.

(c) At the Deferred China Closing, the Parties hereto shall, and shall cause their respective Affiliates to, execute and deliver such documents and instruments as may be reasonably necessary to evidence the transfer of the equity interests in the Deferred China Entities.

(d) Notwithstanding anything contained herein to the contrary, but subject to receipt of the approval letters issued by the foreign investment administration authorities supervising the Deferred China Entities approving the aforesaid assignment, transfer and delivery of the equity interests in the Deferred China Entities and <u>Section 1.6(h)</u>, there shall be no conditions required to be satisfied or waived prior to the Deferred China Closing in order to consummate the transactions contemplated by <u>Section 1.6(b)</u>.

(e) During the period between the Closing Date and the Deferred China Closing Date, the Parties hereto shall, and shall cause their respective Affiliates to, cooperate fully and use commercially reasonable efforts to take such actions with respect to the Deferred China Entities as may be reasonably requested by the other Party hereto in order to permit the transfer of the Deferred China Entities in accordance with <u>Section 1.6(b)</u> as promptly as practicable.

(f) Prior to the occurrence of the Deferred China Closing, the Deferred China Entities and their assets, operations and liabilities shall be held for the account of the Buyer and its subsidiaries.

(g) From the Closing Date to the Deferred China Closing Date, unless the context clearly requires otherwise, all references in this Agreement to the "Closing" or the "Closing Date" shall, with respect to the Deferred China Entities, be deemed to refer to the Deferred China Closing or the Deferred China Closing Date, as applicable, for the Deferred China Entity and the Parties hereto shall continue to comply with all covenants and agreements contained in this Agreement that are required by their terms to be performed prior to the Closing in respect of the Deferred China Entities.

(h) If the Deferred China Closing Date has not occurred within two hundred (200) days following the Closing Date, the Parties shall, at the request of Buyer, take all actions necessary to unwind the purchase and sale of the Deferred China Entities and Seller shall, within five (5) Business Days, refund Buyer (or a subsidiary of Buyer selected by Buyer) the purchase price amount that was paid to a Stock Seller in respect of the Deferred China Entities pursuant to the Transactions (and in accordance with the allocations set forth in <u>Schedule 1.10</u>), together with interest thereon at the Prime Rate from and until the date of such payment. Following such unwinding and refund, no Party shall be obligated to consummate the Deferred China Closing.

1.7 Deferred Baroom Closing

(a) Notwithstanding anything to the contrary contained in this Agreement (other than <u>ARTICLE IX</u>), the assignment, transfer and delivery of the equity interests of Harsco (Bermuda) Ltd. (the "<u>Deferred Baroom Entity</u>"), shall occur on the later of (<u>i</u>) the Closing Date and (<u>ii</u>) the third Business Day after the consummation of the transactions and receipt of the consents and approvals set forth on <u>Schedule 1.7(a)</u> (the "<u>Deferred Baroom Closing Date</u>").

(b) The assignment, transfer, conveyance and delivery of the Deferred Baroom Entity shall take place at a closing (the "<u>Deferred Baroom Closing</u>") at the offices of Debevoise & Plimpton, 919 Third Avenue, New York, New York, 10022, on the Deferred Baroom Closing Date.

(c) At the Deferred Baroom Closing, the Parties hereto shall, and shall cause their respective Affiliates to, execute and deliver such documents and instruments as may be reasonably necessary to evidence the transfer of the equity interests in the Deferred Baroom Entity.

(d) Notwithstanding anything contained herein to the contrary, but subject to the consummation of the transactions and receipt of the consents set forth on <u>Schedule 1.7(a)</u> and <u>Section 1.7(h)</u>, if the Closing shall have occurred on an earlier date than the Deferred Baroom Closing Date, there shall be no conditions required to be satisfied or waived prior to the Deferred Baroom Closing in order to consummate the transactions contemplated by <u>Section 1.7(b)</u>.

(e) During the period between the Closing Date and the Deferred Baroom Closing Date, the Parties hereto shall, and shall cause their respective Affiliates to, cooperate fully and use commercially reasonable efforts to take such actions with respect to the Deferred Baroom Entity as may be reasonably requested by the other Party hereto in order to permit the transfer of the Deferred Baroom Entity in accordance with <u>Section 1.7(b)</u> as promptly as practicable.

(f) Prior to the occurrence of the Deferred Baroom Closing, the Deferred Baroom Entity and its assets, operations and liabilities shall be held for the account of Buyer and its subsidiaries.

(g) From the Closing Date to the Deferred Baroom Closing Date, unless the context clearly requires otherwise, all references in this Agreement to the "Closing" or the "Closing Date" shall, with respect to the Deferred Baroom Entity and its Subsidiaries, be deemed to refer to the Deferred Baroom Closing or the Deferred Baroom Closing Date, as applicable, and the Parties hereto shall continue to comply with all covenants and agreements contained in this Agreement that are required by their terms to be performed prior to the Closing in respect of the Deferred Baroom Entity.

(h) If the Deferred Baroom Closing Date has not occurred within 250 days following the Closing Date, the Parties shall, at the request of Buyer, take all actions necessary to unwind the purchase and sale of the Deferred Baroom Entity and Seller shall, within five (5) Business Days, refund Buyer (or a subsidiary of Buyer selected by Buyer) the purchase price amount that was paid to a Stock Seller in respect of the Deferred Baroom Entity pursuant to the Transactions (and in accordance with the allocations set forth in <u>Schedule 1.10</u>), together with interest thereon at the Prime Rate from and until the date of such payment. Following such unwinding and refund, no Party shall be obligated to consummate the Deferred Baroom Closing.

- 1.8 Deliveries at Closing
 - (a) At or prior to the Closing, Harsco shall deliver or cause to be delivered to Buyer the following:

(i) other than in respect of Target Shares in the capital of companies registered in England and Wales, if the Target Shares are certificated, stock certificates (or the local legal equivalent) evidencing the Target Shares to be sold by Harsco and the other Stock Sellers duly endorsed in blank, or accompanied by stock powers duly executed in blank, and if the Target Shares are not certificated, evidence of appropriate changes to the applicable share register (or equivalent corporate record) or other evidence of transfer of the Target Shares to the applicable Buyer consistent with the Laws of the jurisdiction of the applicable Target Entity;

(ii) in respect of Target Shares in the capital of companies registered in England and Wales, (A) unstamped transfers in favor of the applicable Buyers executed by the relevant Stock Sellers, (B) the relevant share certificates (or indemnities in the usual form in respect of any such share certificates that cannot be so delivered) and (C) powers of attorney in the usual form executed by the relevant Stock Sellers in respect of the Target Shares sold by such Stock Sellers appointing the applicable Buyers to be their lawful attorneys in respect of the relevant Target Shares pending registration of the transfer of such Target Shares;

(iii) one or more assignment and assumption agreements (the "<u>Assignment and Assumption Agreements</u>") evidencing the sale and transfer of the Target Assets from the Asset Sellers to the Asset Buyers and the assumption of the Assumed Target Liabilities by the Asset Buyers, in a form reasonably satisfactory to each of CD&R Investor and Harsco;

(iv) one or more quitclaim deeds, bills of sale, transfer tax forms, and any other required filings and/or real estate conveyance documentation evidencing the sale and transfer of any Target Assets which are Owned Real Property or Leased Real Property, each (A) in form and substance customary in the applicable jurisdiction in which such Owned Real Property or Leased Real Property is located and (B) reasonably satisfactory to each of CD&R Investor and Harsco;

(v) the duly executed certificate to be delivered by Harsco pursuant to <u>Section 6.2(c)</u>;

(vi) one or more certificates in form and substance reasonably acceptable to Buyer, duly executed and acknowledged, certifying any facts that would exempt the sale of Target Shares of any Target Entities that are domestic corporations for U.S. federal income tax purposes and the sale of any Target Asset that is a U.S. real property interest from withholding under Section 1445 of the Code;

(vii) the Transition Services Agreement duly executed by Harsco;

(viii) the Upper-Tier Partnership Agreement duly executed by Harsco;

(ix) the Investor Rights Agreement duly executed by Harsco; and

(x) each other Ancillary Agreement to which Harsco or its Affiliates is a party executed by a duly authorized representative of Harsco and/or its Affiliates, as applicable.

(b) At or prior to the Closing, Buyer, on behalf of itself and the other Buyers, and CD&R Investor, on behalf of itself and Upper-Tier Partnership, shall deliver or cause to be delivered to Harsco the following:

(i) the Initial Cash Purchase Price by wire transfer of immediately available funds to one or more accounts designated by Harsco at least two (2) Business Days prior to the Closing;

(ii) stock certificates evidencing the Harsco Company Shares to be transferred to the Sellers by the Buyers duly endorsed in blank, or accompanied by stock powers duly executed in blank;

- (iii) the duly executed certificate to be delivered by CD&R Investor and Buyer pursuant to <u>Section 6.1(c)</u>;
- (iv) the Transition Services Agreement duly executed by Buyer;
- (v) the Upper-Tier Partnership Agreement duly executed by CD&R Investor and General Partner;

(vi) the Investor Rights Agreement duly executed by CD&R Investor, Holdings Partnership, the Company and Upper-Tier Partnership; and

(vii) each other Ancillary Agreement to which CD&R Investor Holdings Partnership, the Company or any of the Buyers is a party executed by a duly authorized representative of CD&R Investor and/or such Buyer, as applicable.

1.9 Post-Closing Purchase Price Adjustment

(a) As soon as reasonably practicable after the Closing Date and, in any event, within sixty (60) days thereof, Buyer shall prepare and deliver to Harsco (i) a calculation of the aggregate amount of Cash of the Target Entities (the "<u>Closing Cash</u>"), (<u>ii</u>) a calculation of the aggregate amount of Indebtedness of the Target Entities (the "<u>Closing Indebtedness</u>"), (<u>iii</u>) a calculation of Net Working Capital (the "<u>Closing Net Working Capital</u>") and (<u>iv</u>) a calculation of the Cap Ex Shortfall (the "<u>Closing Net Working Capital</u>") and (<u>iv</u>) a calculation of the Cap Ex Shortfall (the "<u>Closing Net Working Capital</u>") and (<u>iv</u>), collectively, the "<u>Closing Statement</u>"). The Closing Statement shall be prepared, and each Adjustment Amount shall be calculated, by aggregating each applicable and specified line item in the Closing Statement. Notwithstanding anything herein to the contrary, the Parties agree that the Estimated Closing Statement and the Closing Statement, and the component items thereof, shall take into account the Transactions and be prepared and calculated in a manner consistent with <u>Exhibit H</u> and the definitions of Cash, Indebtedness, Net Working Capital and Closing Cap Ex Shortfall and using the Agreed Calculation Principles. For purposes of calculating Closing Cash, Closing Indebtedness, Closing Net Working Capital and Closing Cap Ex Shortfall, any information available to any of the Parties as of the time of the delivery of the Closing Statement may be taken into account for purposes of such calculation.

(b) From and after the receipt by Harsco of the Closing Statement, Buyer shall, and shall cause the Target Entities to, (<u>i</u>) assist Harsco and representatives of Harsco in their review of, and provide Harsco and representatives of Harsco with reasonable access upon reasonable notice during normal business hours to, the books, records (including work papers, schedules, memoranda and other documents), supporting data, facilities, employees and auditors of the Target Entities for purposes of reviewing and verifying the Closing Statement and the items included therein, and preparing any Objection, and (<u>ii</u>) reasonably cooperate with Harsco and its representatives in connection therewith, including providing other information necessary in connection with such review, verification and preparation as is reasonably requested by Harsco and its representatives.

(c) After receipt of the Closing Statement from Buyer, Harsco shall have forty-five (45) days to review the Closing Statement (the "<u>Review Period</u>"). The Closing Statement shall be binding and conclusive upon, and deemed accepted by, Harsco unless Harsco shall have notified Buyer in writing prior to the expiration of the Review Period of any dispute or objection thereto (any such written dispute or objection, the "<u>Objection</u>"), setting forth the basis for its dispute or objections and the specific adjustments (including dollar amounts) to the Closing Statement which Harsco believes in good faith

should be made. Any items not disputed or objected to in an Objection shall be deemed to have been accepted by Harsco. If no Objection is delivered by Harsco to Buyer prior to the expiration of the Review Period, then the Closing Statement shall be deemed to have been accepted by the Parties and shall become final and binding upon the Parties. Harsco and Buyer shall, within thirty (30) days (or such longer period as Harsco and Buyer may agree in writing) following delivery of an Objection to Buyer (the "<u>Resolution Period</u>"), attempt in good faith to resolve their differences. Any items agreed to by Harsco and Buyer in writing, together with any items not disputed or objected to by Harsco in the Objection, are collectively referred to herein as the "<u>Resolved Matters</u>".

(d) If, at the end of the Resolution Period, Harsco and Buyer have been unable to resolve any differences that they may have with respect to the matters specified in the Objection, Harsco and Buyer shall refer all matters that remain in dispute with respect to the Objection (the "Unresolved Matters") to an internationally recognized independent public accounting firm jointly selected by Harsco and Buyer or, if Harsco and Buyer are unable to agree within five (5) Business Days from the end of the Resolution Period, then such internationally recognized independent public accounting firm, which shall be independent of both Harsco and CD&R Investor, jointly selected by Harsco's and Buyer's independent accountants within five (5) Business Days thereafter (the "<u>CPA Firm</u>"). Harsco, on behalf of the Sellers, and Buyer, each agree to promptly sign an engagement letter, in commercially reasonable form, as may reasonably be required by the CPA Firm. The CPA Firm shall, acting as experts in accounting and not as arbitrators, determine on a basis consistent with the requirements of this Agreement, and only with respect to the Unresolved Matters so submitted, whether and to what extent the Closing Statement requires adjustment. Harsco and Buyer shall request the CPA Firm to use its commercially reasonable efforts to (i) render its final written determination within thirty (30) days after such firm's engagement, and (ii) prepare the Final Closing Statement (which shall be consistent with the Resolved Matters and the final determination of the CPA Firm of the Unresolved Matters) and calculate the applicable Adjustment Amount based on such Final Closing Statement. The final written determination of the CPA Firm shall be based only on the written submissions by Harsco and Buyer, and shall be made in strict accordance with the terms of this Agreement, without regard to principles of equity. With respect to each Unresolved Matter, the CPA Firm's determination, if not in accordance with the position of either Harsco or Buyer, shall not be in excess of the higher, nor less than the lower, of the amounts advocated by Harsco or Buyer with respect thereto. The CPA Firm's final written determination shall be conclusive and binding upon Harsco and Buyer. Harsco and Buyer shall make reasonably available to the CPA Firm, upon the CPA Firm's request, all relevant books and records, any workpapers (including those of the Parties' respective accountants) and supporting documentation relating to the Closing Statement and all other items reasonably requested by the CPA Firm (provided, that they shall contemporaneously provide a copy to the other party of any materials requested by, and provided to, the CPA Firm). None of Harsco, Buyer or any of their respective Affiliates shall have any exparte communications or meetings with the CPA Firm regarding the subject matter hereof without the other party's prior written consent.

(e) The "<u>Final Closing Statement</u>" shall be (<u>i</u>) in the event that no Objection is delivered by Harsco to Buyer prior to the expiration of the Review Period with respect thereto, the Closing Statement delivered by Buyer to Harsco pursuant to <u>Section 1.9(a)</u>, (<u>ii</u>) in the event that an Objection is delivered by Harsco to Buyer prior to the expiration of the Review Period and Harsco and Buyer are able to agree on all matters set forth in the Objection, the Closing Statement delivered by Buyer to Harsco pursuant to <u>Section 1.9(a)</u> as adjusted pursuant to the agreement of Harsco and Buyer in writing or (<u>iii</u>) in the event that an Objection is delivered by Harsco to Buyer prior to the expiration of the Review Period and Harsco and Buyer are unable to agree on all matters set forth in such Objection, the Closing Statement delivered by Buyer to Harsco and Buyer are unable to agree on all matters set forth in such Objection, the Closing Statement delivered by Buyer to Harsco pursuant to <u>Section 1.9(a)</u> as adjusted by the CPA Firm to be consistent with the Resolved Matters and the final determination of the CPA Firm of the Unresolved

Matters in accordance with Section 1.9(d). In the event that the Final Closing Statement is determined (\underline{x}) pursuant to clause (i) or (ii) of the immediately preceding sentence, Buyer shall prepare the Final Closing Statement and calculate the applicable Adjustment Amount, in each case, in strict accordance with the terms of this Agreement (and as otherwise agreed amongst the Parties pursuant to this Section 1.9), and deliver such items to Harsco within three (3) Business Days following the determination thereof or (\underline{y}) pursuant to clause (iii) of the immediately preceding sentence, the CPA Firm shall prepare the Final Closing Statement (which shall be consistent with the Resolved Matters and the final determination of the CPA Firm of the Unresolved Matters) and calculate the applicable Adjustment Amount based on the Final Closing Statement, in each case, in strict accordance with the terms of this Agreement, and deliver such items to Harsco and Buyer within three (3) Business Days following the determination of the CPA Firm of the CPA Firm of the CPA Firm of the Unresolved Matters) of the final written determination of the CPA Firm to Harsco and Buyer.

(f) All fees and disbursements of the CPA Firm shall be borne by Harsco, on the one hand, and Buyer, on the other hand, in the same proportion as the aggregate amount of the Unresolved Matters that is unsuccessfully disputed by each (as determined by the CPA Firm) bears to the total amount of the Unresolved Matters submitted to the CPA Firm.

(g) The Sellers and Buyer agree that, after the Closing, the Initial Cash Purchase Price shall be adjusted as set forth below (each such adjustment, an "<u>Adjustment Amount</u>"):

(i) if the Closing Cash (as set forth in the Final Closing Statement) is less than the Estimated Closing Cash, the Initial Cash Purchase Price shall be reduced by the difference between the Estimated Closing Cash and the Closing Cash;

(ii) if the Closing Cash (as set forth in the Final Closing Statement) is greater than the Estimated Closing Cash, the Initial Cash Purchase Price shall be increased by the difference between the Closing Cash and the Estimated Closing Cash;

(iii) if the Closing Indebtedness (as set forth in the Final Closing Statement) is greater than the Estimated Closing Indebtedness, the Initial Cash Purchase Price shall be reduced by the difference between the Closing Indebtedness and the Estimated Closing Indebtedness;

(iv) if the Closing Indebtedness (as set forth in the Final Closing Statement) is less than the Estimated Closing Indebtedness, the Initial Cash Purchase Price shall be increased by the difference between the Estimated Closing Indebtedness and the Closing Indebtedness;

(v) if the Closing Net Working Capital (as set forth in the Final Closing Statement) is less than the Estimated Closing Net Working Capital, the Initial Cash Purchase Price shall be reduced by the difference between the Estimated Closing Net Working Capital and the Closing Net Working Capital;

(vi) if the Closing Net Working Capital (as set forth in the Final Closing Statement) is greater than the Estimated Closing Net Working Capital, the Initial Cash Purchase Price shall be increased by the difference between the Closing Net Working Capital and the Estimated Closing Net Working Capital;

(vii) if the Closing Cap Ex Shortfall (as set forth in the Final Closing Statement) is less than the Estimated Cap Ex Shortfall, the Initial Cash Purchase Price shall be

increased by the difference between the Estimated Closing Cap Ex Shortfall and the Closing Cap Ex Shortfall; and

(viii) if the Closing Cap Ex Shortfall (as set forth in the Final Closing Statement) is greater than the Estimated Closing Cap Ex Shortfall, the Initial Cash Purchase Price shall be reduced by the difference between the Closing Cap Ex Shortfall and the Estimated Closing Cap Ex Shortfall.

(h) Within five (5) Business Days following the delivery of the Final Closing Statement, Buyer shall deliver to Harsco a schedule setting forth the allocation of each Adjustment Amount among the Target Shares and Target Assets based upon the item to which such adjustment is attributable. Harsco shall have the right to review and object to such allocation in accordance with the procedures set forth in <u>Sections 1.9(c)</u> through <u>1.9(f)</u>, which shall apply *mutatis mutandis*; provided, however, that the "Review Period" shall be ten (10) Business Days. With respect to each Adjustment Amount, (<u>A</u>) if such Adjustment Amount results in an increase to the Initial Cash Purchase Price, Buyer shall cause the applicable Stock Buyers and/or Asset Buyers to pay or cause to be paid to Harsco (on behalf of the applicable Stock Sellers and/or Asset Sellers) such Adjustment Amount and (<u>B</u>) if such Adjustment Amount results in a decrease to the Initial Cash Purchase Price, Harsco (or its designees) shall pay to the applicable Stock Buyers and/or Asset Buyers such Adjustment Amount, in each case, by wire transfer of immediately available funds in Dollars, together with an amount computed as interest thereon at the rate declared by JPMorgan Chase Bank at its "base rate" on the Closing Date (the "<u>Prime Rate</u>") calculated based on the number of days elapsed from the Closing Date to the date of such payment.

1.10 Purchase Price Allocation

The Parties have agreed to allocate the Purchase Price among the Target Shares (other than the Target Shares of any Target Entity that is treated as a flow-through entity for U.S. federal income tax purposes) and the Target Assets (including, for purposes of this <u>Section 1.10</u>, the assets of any Target Entity that is treated as a flow-through-entity for U.S. federal income tax purposes) as set forth in <u>Schedule 1.10</u> to this Agreement. The Parties acknowledge and agree that, to the extent applicable, the provisions of Section 1060 of the Code shall govern the purchase of the Target Assets. In the event an adjustment to the Purchase Price is made pursuant to <u>Section 1.9</u> or otherwise under this Agreement (and any distributions, refunds and/or other payments are made in connection therewith), the allocation of the Purchase Price shall be revised to allocate such adjustment to the Target Shares or Target Assets, as the case may be, based upon the item to which such adjustment is attributable. The Parties acknowledge and agree that, to the extent any such adjustment to the Purchase Price cannot be allocated to a specific asset or jurisdiction, such unallocated Purchase Price shall be allocated across such jurisdictions as the parties, acting in good faith, agree is most appropriate under the circumstances. To the extent permitted by Law, (i) this allocation shall be binding on the Parties for federal, state, local, foreign and other Tax reporting purposes, (<u>ii</u>) no party will assert or maintain a position inconsistent with this allocation and (<u>iii</u>) the applicable Tax Returns to be filed by any of the Parties or their Subsidiaries shall reflect this allocation.

ARTICLE II

REPRESENTATIONS AND WARRANTIES OF HARSCO AND SELLERS

Except as set forth on the corresponding section or subsection of the Schedules attached hereto (collectively, the "<u>Schedules</u>"), Harsco, on behalf of itself, the other Sellers and the Target Entities, hereby represents and warrants to CD&R Investor and Buyer as of the date hereof (except in each case to the extent that such representations and warranties speak only as of another date or dates in which case, only as of such date(s)) as follows, <u>provided</u>, <u>however</u>, that disclosure in any section or subsection of the Schedules shall apply to any other section or subsection of the Schedules to the extent that it is reasonably apparent from the face of such disclosure that such disclosure is relevant to another section or subsection of the Schedules:

2.1 Incorporation and Authority

Each of the Sellers and Target Entities is an entity duly organized, validly existing and (where such concept is applicable) in good standing (or local equivalent) under the Laws of its jurisdiction of organization, formation or incorporation, as applicable, and has all requisite corporate power and authority to conduct its business as currently conducted, except for the entities set forth on Schedule 2.1, which are being liquidated in accordance with applicable Laws. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, each Target Entity is duly gualified as a foreign corporation for the transaction of business and is (where such concept is applicable) in good standing under the laws of each other jurisdiction in which it owns or leases properties, or conducts any business so as to require such qualification. Harsco has made available to CD&R Investor complete copies of the respective articles of incorporation or bylaws (or similar organizational documents) of each of the Target Entities as in effect as of the date hereof. Each of the Sellers and Target Entities has all necessary corporate power and authority to enter into this Agreement and the Ancillary Agreements to which it will be a party, to carry out and perform its obligations hereunder and thereunder and to consummate the transactions contemplated on its part hereby and thereby. The execution and delivery by Harsco of this Agreement and by the Sellers of the Ancillary Agreements to which they will be a party, and the consummation by the Sellers and the Target Entities of the transactions contemplated on their part hereby and thereby, have been duly authorized by all necessary company action on the part of each Seller and each Target Entity. This Agreement has been duly executed and delivered by Harsco, and, to the extent the Sellers are a party thereto, the Ancillary Agreements will at the Closing be, duly executed and delivered by the Sellers, as applicable, and (assuming due authorization, execution and delivery by the other parties thereto) this Agreement constitutes a legal, valid and binding obligation of Harsco, and when executed and delivered, to the extent the Sellers are a party thereto, the Ancillary Agreements will constitute, legal, valid and binding obligations of the Sellers, as applicable, enforceable against the Sellers, as applicable, in accordance with their terms, except as enforceability may be limited by bankruptcy Laws, other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity).

2.2 Capitalization and Subsidiaries

(a) Set forth on Schedule 2.2(a) is (i) the jurisdiction of incorporation or legal organization of each of the Target Entities, (ii) the number of authorized, issued and outstanding shares or other equity securities of each of the Target Entities and the holders of all such shares or other equity securities (exclusive of the effects of the Transactions), and (iii) the number of authorized, issued and outstanding shares or other equity securities of each of the Target Entities and the holders of all such shares or other equity securities (immediately after giving effect to the Transactions) and, in the case of (ii) and (iii), except as set forth on Schedule 2.2(a), there are no other authorized, issued or outstanding shares of capital stock or other equity securities of the Target Entities. Except as set forth on Schedule 2.2(a), all of the issued and outstanding Target Shares are owned of record free and clear of any Encumbrances by the Stock Seller identified on <u>Schedule 2.2(a)</u> as owning such Target Shares. All of the issued and outstanding Target Shares and equity securities of each of the Target Entities have been validly issued, are fully paid and nonassessable and have not been issued in violation of any preemptive or similar rights (to the extent such concepts are applicable). Except as set forth on Schedule 2.2(a), there are (and at the Closing, after giving effect to the Transactions, there will be) no outstanding options, warrants, calls, rights or any other agreements, trusts, proxies or similar arrangements affecting the sale, issuance or voting of any equity securities of any Target Entity, or any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any equity securities of any Target Entity. No Target Entity has any outstanding indebtedness for borrowed money, outstanding bonds, debentures, notes or other securities, the holders of which have the right to vote (or which are convertible into or exchangeable for securities having the right to vote) with the stockholders of the Target Entities.

(b) Immediately prior to the Closing, all of the issued and outstanding equity securities of each Target Entity will be owned of record, directly or indirectly, free and clear of any Encumbrances by one or more Stock Sellers, and all of such issued and outstanding equity securities will have been validly issued, fully paid and nonassessable.

(c) None of the Target Entities has (or at the Closing, after giving effect to the Transactions, will have) any Subsidiaries or owns (or at the Closing, after giving effect to the Transactions, will own) any equity interests in any other Person, except as set forth on <u>Schedule 2.2(a)</u>.

(d) As of the date of this Agreement, <u>Schedule 2.2(d)</u> sets forth each Contract (including all modifications and amendments thereto) currently in effect between a JV Partner, on the one hand, and any Target Entity, Stock Seller or Asset Seller (to the extent relating to the Business), on the other hand. With respect to each such Contract, (<u>i</u>) Harsco has made available to CD&R Investor a true and complete copy of such Contract (including all modifications and amendments thereto and written waivers thereunder), (<u>ii</u>) such Contract is in full force and effect in accordance with its respective terms and is valid, binding and enforceable against the applicable JV Partner thereto in accordance with its terms, except as enforceability may be limited by bankruptcy Laws, other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity) and (<u>iii</u>) the execution, delivery and performance by the Sellers of the transactions contemplated by this Agreement (including, for the avoidance of doubt, the performance of the Transactions) will not result in any breach of, or constitute a default (or event which with the giving of notice or lapse of time, or both, would become a default) under, or give any JV Partner any right of termination, rescission, acceleration or cancellation of, or any right of notice or Consent with respect to such Contract.

2.3 No Conflict

The execution, delivery and performance by the Sellers of the transactions contemplated by this Agreement (including, for the avoidance of doubt, the performance of the Transactions) and the Ancillary Agreements to which they will be a party do not and will not: (a) conflict with or violate the respective articles of incorporation or bylaws (or similar organizational documents) of any of the Sellers or any Target Entity; (b) assuming all Consents referred to in Section 2.4 have been obtained, conflict with or violate any Law applicable to any of the Sellers or any Target Entity, or to any of their respective properties, rights or assets; (c) result in any breach of, or constitute a default (or event which with the giving of notice or lapse of time, or both, would become a default) under, or give to any Person any right of termination, rescission, acceleration or cancellation of, or any right of notice or Consent with respect to, any material Contract to which any of the Target Entities (or any of their controlled Affiliates) is a party or by which any of the Target Assets or any properties or rights of any Target Entity are bound; (d) result in the creation or imposition of any Encumbrance other than Permitted Encumbrances on any Assets; or (e) result in any breach of, or constitute a default (or event which with the giving of notice or lapse of time, or both, would become a default) under, or give to any Person any right of termination, rescission, acceleration or cancellation of, or any right of notice or Consent with respect to, any Permit to which any of the Target Entities (or any of their controlled Affiliates) is a party or by which any of the Target Assets or any properties or rights of any Target Entity are bound except, in the case of clauses (b), (d), and (e), where such conflicts, violations, rescissions, accelerations, breaches, defaults, terminations, cancellations or failures to obtain such Consents would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

2.4 Consents and Approvals

The execution and delivery by Harsco of this Agreement does not, and the execution and delivery by the Sellers of the Ancillary Agreements, solely to the extent they are a party thereto, will not, and the performance by the Sellers of this Agreement (including, for the avoidance of doubt, the Transactions) and the Ancillary Agreements, to the extent they are a party thereto, and the consummation of the transactions contemplated hereby and thereby will not, require any Seller or any Target Entity to seek or obtain any Consent from, or make any filing or notification with, any Governmental Body, except for (<u>a</u>) those Consents set forth on <u>Schedule 2.4(a)</u>, (<u>b</u>) compliance with the applicable requirements of the HSR Act or any other Competition Laws of the jurisdictions set forth on <u>Schedule 2.4(b)</u> and (<u>c</u>) those Consents from, and filings with Governmental Bodies the failure of which to be made or obtained would not, individually or in the aggregate, be reasonably expected to materially impact the ability of any of the Sellers to consummate the transactions contemplated hereby or otherwise adversely affect the Business in any material respect.

2.5 Financial Statements

(a) The audited combined balance sheets of the Business as at December 31, 2011 and December 31, 2012, and the related audited combined statements of operations, comprehensive loss, cash flows and changes in equity of the Business for the years ended December 31, 2011 and December 31, 2012, together with all related notes and schedules thereto, accompanied by the reports thereon of Harsco's independent auditors (collectively, the "<u>Audited Financial Statements</u>") and the unaudited combined balance sheets of the Business as at June 30, 2013 and June 30, 2012, and the related combined statements of operations, comprehensive loss, cash flows and changes in equity of the Business for the three (3) and six (6) month periods ended June 30, 2013 and June 30, 2012, together with all related notes and schedules thereto (together with the Audited Financial Statements, the "<u>Financial Statements</u>") are included in <u>Schedule 2.5(a)</u>. The Financial Statements (<u>i</u>) have been prepared based on the separate books

and records of the Target Entities and Harsco (except as may be indicated in the notes thereto), (<u>ii</u>) have been prepared in accordance with GAAP consistently applied (subject, in the case of unaudited statements, to normal year end adjustments which will not be material in nature or amount to the Business taken as a whole), (<u>iii</u>) present fairly in all material respects the combined financial position, the results of operations and cash flows of the Business as of the respective dates thereof and for the respective periods indicated therein (and, except as set forth on <u>Schedule 2.5(a)</u>, such fair presentation will not be affected by the Pre-Closing Transactions) and (<u>iv</u>) have been prepared in compliance with Regulation S-X of the SEC.

(b) Harsco maintains, and has maintained since January 1, 2011, a system of internal controls over financial reporting (as defined in Rules 13a-15(f) and 15d-15(f) of the Exchange Act) sufficient to provide reasonable assurances regarding the reliability of financial reporting and the preparation of financial statements in accordance with GAAP. Harsco has disclosed, based on the most recent evaluation undertaken in connection with Harsco's financial statements, to its external accountants (i) all significant deficiencies and material weaknesses in the design or operation of internal controls over financial reporting which are reasonably likely to adversely affect Harsco's ability to record, process, summarize and report financial information, and (<u>ii</u>) any fraud, whether or not material, that involves management or other employees who have a significant role in Harsco's internal controls over financial reporting.

(c) Since January 1, 2011, no director, officer, external auditor or external accountant of Harsco or its Affiliates (including the Target Entities) or, to the Knowledge of Harsco, any employee (other than officers) or representative of Harsco or its Affiliates (including the Target Entities), has received or otherwise been made aware of any material complaint, allegation, assertion or claim, whether written or oral, regarding the accounting or auditing practices, procedures, methodologies or methods of the Business or the Target Entities or their respective internal accounting controls, including any material complaint, allegation, assertion or claim that any Target Entity has engaged in questionable accounting or auditing practices.

(d) <u>Schedule 2.5(d)</u> sets forth an accurate and complete copy of the capital expenditure budget for the Business on a combined basis for the fiscal year beginning January 1, 2013 and for January 1, 2014 through February 28, 2014 (the "<u>Capital</u> <u>Expenditure Budget</u>").

2.6 No Undisclosed Material Liabilities

No Target Entity has, after giving effect to completion of the Transactions, any Liabilities, whether known, unknown, absolute, accrued, contingent or otherwise and whether due or to become due, in each case that are required under GAAP to be reflected in the Financial Statements or disclosed in the notes thereto, except (<u>i</u>) as set forth in <u>Schedule 2.6</u>, (<u>ii</u>) Liabilities disclosed or reserved against in the Financial Statements as at the Balance Sheet Date or disclosed in the notes thereto, (<u>iii</u>) Liabilities that were incurred after the Balance Sheet Date in the ordinary course of business, (<u>iv</u>) Liabilities which, individually or in the aggregate, would not be material to the Business, taken as a whole, (<u>v</u>) Liabilities that are Tax liabilities incurred in connection with the Transactions, to the extent Harsco has agreed to provide indemnification pursuant to <u>Section 5.1</u>, and (<u>vi</u>) Liabilities that are specifically contemplated by this Agreement and incurred after the date hereof. No Target Entity is a party to, or has any commitment to become a party to, any joint venture, off balance sheet partnership or any similar Contract (including any Contract relating to any transaction or relationship between or among the Target Entities, on the one hand, and any Non-Company Affiliate, including any structured finance, special purpose or limited purpose Person, on the other hand, or any "off-balance sheet arrangement" (as defined in Item 303(a) of Regulation S-K of the SEC)).

2.7 Litigation

(a) As of the date hereof, there is no suit, action or proceeding (administrative or judicial) by or before any Governmental Body ("<u>Proceeding</u>") pending or, to the Knowledge of Harsco, threatened against Harsco, any other Seller or the Target Entities, or any of their respective assets, rights or properties which bring into question the validity of this Agreement or that would reasonably be expected, individually or in the aggregate, to be adverse in any material respect to the Business, the Target Assets or the Target Entities, taken as a whole. There are no Governmental Orders seeking or purporting to enjoin or restrain the execution, delivery and performance by Harsco or any Seller of this Agreement, or the consummation by Harsco, any other Seller or the Target Entities of the transactions contemplated hereby or that would reasonably be expected to be adverse in any material respect to the Business or the Target Entities, taken as a whole.

(b) <u>Schedule 2.7(b)</u> describes in reasonable detail each accident or incident where the Business operates or provides equipment, in each case occurring since January 1, 2010 and prior to the date hereof, that involves a death or serious bodily injury reasonably likely to result in Losses in excess of \$100,000 where claims, or to the Knowledge of Harsco, potential claims from said accident or incident have not been fully settled.

2.8 Employee Benefits

(a) <u>Schedule 2.8(a)</u> lists, as of the date of this Agreement, each Employee Plan (other than a Harsco Benefit Plan) that any Target Entity sponsors, maintains or contributes to, or is required to maintain or contribute to for the benefit of any current or former Business Employee (the "<u>Target Entity Benefit Plans</u>"). No Target Entity Benefit Plan that is not a Multiemployer Plan provides benefits to any individual other than current or former employees of the Target Entities, Persons who provide or have provided services in respect of the Business and any beneficiaries and dependents of the foregoing.

(b) <u>Schedule 2.8(b)</u> lists, as of the date of this Agreement, each Employee Plan that Harsco or any Non-Company Affiliate sponsors, maintains or contributes to, or is required to maintain or contribute to as of the date hereof for the benefit of any current Business Employee (together with any persons formerly employed by any of the Target Entities) (the "<u>Harsco Benefit</u> <u>Plans</u>," and, together with the Target Entity Benefit Plans, the "<u>Business Benefit Plans</u>").

(c) With respect to each Target Entity Benefit Plan that is not a Multiemployer Plan, Harsco has provided CD&R Investor, to the extent applicable, complete and correct copies of: all written plan documents and descriptions of all unwritten plans (and all amendments thereto); all trust agreements, insurance contracts or other funding arrangements therefor; the two most recent actuarial and trust reports; the two most recent Forms 5500 and all schedules thereto; the most recent IRS determination letters; and all current summary plan descriptions and summaries of material modification (or with respect to the foregoing items, the equivalent thereto required by any foreign Governmental Body).

(d) The Target Entity Benefit Plans that are not Multiemployer Plans have been maintained and operated in material compliance with their terms and, to the extent applicable, provisions of ERISA, the Code and all other applicable Laws, domestic or foreign. None of the Target Entity Benefit Plans are, to the Knowledge of Harsco, presently under audit or examination (nor has notice been received of a potential audit or examination) by the IRS, the Department of Labor or any other Governmental Body, domestic or foreign. Other than routine claims for benefits, there are no material Proceedings pending or, to the Knowledge of Harsco, threatened with respect to any Target Entity Benefit Plan that is not a Multiemployer Plan.

(e) Each Target Entity Benefit Plan, other than a Multiemployer Plan, that is intended to be qualified or otherwise receive favorable Tax treatment under applicable Law has, if necessary, been determined by the applicable Governmental Body or Tax Authority to be so qualified, and to the Knowledge of Harsco, there are no existing circumstances or events that would reasonably be expected to result in any revocation of, or a change to, such qualification or entitlement to favorable Tax treatment under applicable Law except as would not, individually or in the aggregate, reasonably be expected to result in a material liability to the Target Entities taken as a whole.

(f) With respect to Business Employees (or persons formerly employed by the Target Entities), Harsco and its Affiliates (including the Target Entities) are in compliance with all applicable Laws respecting employment and labor, including workplace discrimination, harassment, workers' compensation, classification of workers as exempt or non-exempt or as employees or independent contractors, unlawful retaliation, all titles of the Americans with Disabilities Act, meal and rest breaks, immigration, employee leave issues, wages and hours, occupational safety and health, fair labor standards and the WARN Act and any similar state, local or foreign laws, except such noncompliance that would not, individually or in the aggregate, reasonably be expected to result in a material liability to the Target Entities taken as a whole.

(g) Except as set forth on <u>Schedule 2.8(g)</u>, no Target Entity Benefit Plan that is not a Multiemployer Plan is subject to Section 412 of the Code or Section 302 of ERISA. <u>Schedule 2.8(g)</u> lists each Target Entity Benefit Plan that is an unfunded pension plan or other retirement plan subject to minimum funding standards under applicable Law, and each such Target Entity Benefit Plan satisfies such standards, as applicable, and no waiver of such funding has been sought or obtained. None of the Business Employees is, immediately prior to Closing, a member of any section or sections of the Harsco Pension Scheme, other than Section G or Section M.

(h) Except as set forth in <u>Schedule 2.8(h)</u>, no Target Entity is a party to or bound by, and none of the Business Employees are covered by, any collective bargaining agreement, and there are no labor unions, employee representatives, works councils or other organizations representing or, to the Knowledge of Harsco, purporting to represent or attempting to represent any Business Employees. Since January 1, 2011, there has not occurred or, to the Knowledge of Harsco, been threatened any material strike, sympathy strike, slowdown, picketing, work stoppage, concerted refusal to work overtime or other similar labor activity with respect to any Business Employees. There are no labor disputes currently subject to any grievance procedure, arbitration or litigation which, if adversely decided, may reasonably, individually or in the aggregate, create a liability in excess of \$250,000 and there is no representation petition pending or, to the Knowledge of Harsco, threatened with respect to any Business Employee.

(i) <u>Schedule 2.8(i) sets forth:</u>

(i) Each Business Benefit Plan that constitutes a Multiemployer Plan to which Harsco or one of its ERISA Affiliates (including the Target Entities) currently has or in the past two years had an obligation to contribute with respect to the Business or any current or former Business Employees (each, an "Infrastructure Multiemployer Plan"), including each non-construction Multiemployer Plan (each, a "Non-Construction Multiemployer Plan");

(ii) Each Multiemployer Plan with respect to which Harsco or one of its ERISA Affiliates (including the Target Entities) has or previously had an obligation to contribute with respect to the Business or any current or former Business Employees and currently has any outstanding withdrawal liability obligation (whether resulting from a complete or

partial withdrawal from a Multiemployer Plan), together with the outstanding amount owed for such withdrawal liability and, where available, the relevant payment schedule;

(iii) The aggregate amounts that Harsco and its ERISA Affiliates (including the Target Entities) paid in contributions to all Infrastructure Multiemployer Plans in each of 2011 and 2012, respectively;

(iv) A good-faith summary of the information provided to Harsco and its Subsidiaries by or on behalf of Infrastructure Multiemployer Plan administrators or otherwise provided to Harsco and its Subsidiaries through the date of this Agreement regarding the contributions and premiums paid by Harsco and its ERISA Affiliates with respect to Infrastructure Multiemployer Plans and an estimate of potential withdrawal liability with respect to each such Infrastructure Multiemployer Plan (which summary Harsco shall update, based on information received, using its commercially reasonable efforts, between the date of this Agreement and the Closing Date and provide to Buyer at its reasonable request), substantially in the form previously provided to Buyer; and

(v) Each Business Benefit Plan that constitutes a non-U.S. defined-benefit pension plan for the benefit of employees of multiple unrelated employers to which Harsco or any of its ERISA Affiliates (including the Target Entities) contributes or is or has been required to contribute with respect to the Business or current or former Business Employees.

(j) All contributions and premiums required to have been paid by Harsco and its Affiliates (including the Target Entities) or any of its ERISA Affiliates to any Target Entity Benefit Plan under the terms of any such plan or its related trust, insurance contract or other funding arrangement, or pursuant to any applicable Law (including ERISA and the Code) or collective bargaining agreement have been paid within the time prescribed by any such plan, agreement or applicable Law except as would not, individually or in the aggregate, reasonably be expected to result in a material liability to the Target Entities taken as a whole.

(k) No Target Entity Benefit Plan other than a Multiemployer Plan provides welfare benefits after termination of employment except to the extent required by Section 4980B(f) of the Code or any similar provision of state, local or foreign Law.

(1) Except as set forth on <u>Schedule 2.8(1)</u>, neither this Agreement nor the consummation of the transactions contemplated thereby (either alone or in combination with any other event), including, for the avoidance of doubt, the Transactions, would reasonably be expected to (<u>i</u>) result in any payment becoming due or increase the amount of any compensation due to any current or former employee, officer or director of the Target Entities, (<u>ii</u>) increase any benefits otherwise payable under, or trigger any funding obligation under or to, any Target Entity Benefit Plan, (<u>iii</u>) result in the acceleration of time of payment or vesting of any such benefits, or (<u>iv</u>) result in any one person, or more than one person acting as a group, acquiring during a twelve (12) month period assets from a corporation that have a total gross fair market value equal to or more than one-third of the total gross fair market value of all of the assets of such corporation immediately prior to such acquisition or acquisitions within the meaning of Section 280G of the Code.

(m) Except for any consultation of employees and/or employee representatives required by TUPE (or similar European legislation/regulations (including any national laws or regulations implementing the EU Acquired Rights Directive, Directive 2001/23)), or as otherwise set forth on

<u>Schedule 2.8(m)</u> or <u>Schedule 4.6(b)</u>, the execution and delivery by Harsco of this Agreement and by Harsco of the Ancillary Agreements, as applicable, and the performance by Harsco of this Agreement and the Ancillary Agreements would not reasonably be expected to require Harsco or any of its Affiliates (including the Target Entities) or Buyer or CD&R Investor to seek or obtain any Consent, engage in consultation with, or issue any notice to or make any filing with (as applicable) any Business Employees or any representatives, labor unions, works councils or similar organizations representing Business Employees, or any Governmental Body with respect to Business Employees (except to the extent previously made or obtained prior to the date hereof).

2.9 <u>Taxes</u>

(a) (<u>i</u>) All material Tax Returns required to be filed by or with respect to the Target Entities or the Target Assets have been filed in a timely manner (within any applicable extension periods) and all such Tax Returns are true, correct and complete in all material aspects and (<u>ii</u>) all Taxes shown due on such returns have been timely paid in full. No jurisdiction (whether within or outside the United States) in which any Target Entity has not filed a particular type of Tax Return has asserted in writing that such Target Entity is required to file such Tax Return in such jurisdiction.

(b) There are no outstanding agreements or waivers extending the statutory period for assessment or collection of any material Taxes of the Target Entities or with respect to the Target Assets and no written powers of attorney with respect to any such Taxes. The time for filing any material Tax Return with respect to the Target Entities or the Target Assets has not been extended, other than any extension to which a Target Entity is entitled under applicable Law without the consent of the relevant Tax Authority or Governmental Body.

(c) There are no material audits, Proceedings, investigations or other actions pending or threatened in writing against the Target Entities or the Target Assets in respect of Taxes or Tax Returns. No Tax Authority has asserted or threatened to assert, in each case in writing, any deficiency, claim or issue with respect to Taxes or any adjustment to Taxes against the Target Entities or with respect to the Target Assets for any taxable period for which the period of assessment or collection remains open.

(d) None of the Target Entities will be required to include any item of income in, or exclude any item of deduction from, taxable income for any taxable period (or portion thereof) ending after the Closing Date as a result of (<u>i</u>) any change in method of accounting for a taxable period ending on or prior to the Closing Date, (<u>ii</u>) any installment sale or open transaction, (<u>iii</u>) any prepaid amount received or paid, or (<u>iv</u>) any election pursuant to Section 108(i) of the Code (or any similar provision of state, local or foreign Law) made with respect to any taxable period ending on or prior to the Closing Date.

(e) None of the Target Entities during the five-year period ending on the date hereof, has been a distributing corporation or a controlled corporation in a transaction intended to be governed by Section 355 of the Code.

(f) All material Taxes that the Target Entities were required to withhold or collect have been duly withheld or collected and have been paid, to the extent required by applicable Law, to the proper Tax Authority. The Target Entities have complied in all material respects with the rules and regulations relating to the withholding and remittance of Taxes.

(g) None of the Target Entities (<u>i</u>) has received or applied for a Tax ruling or entered into a closing agreement pursuant to Section 7121 of the Code (or any predecessor provision or any

similar provision of state, local or foreign Law), in either case that would be binding upon any Target Entity after the Closing Date, (<u>ii</u>) is or has been a member of any affiliated, consolidated, combined or unitary group for purposes of filing Tax Returns or paying Taxes or (<u>iii</u>) has any liability for the Taxes of any Person (other than for any Target Entity excluding Asset Sellers) whether under Treasury Regulation Section 1.1502-6 or any similar provision of state, local or foreign Law, as a transferee or successor, pursuant to any Tax sharing or indemnity agreement or other contractual agreements, or otherwise.

(h) None of the Target Entities has engaged in any "listed transactions" within the meaning of Treasury Regulation Section 1.6011-4(b)(2).

(i) <u>Schedule 2.9(i)</u> lists all of the Target Entities for which an election to change the U.S. federal tax classification of such Target Entities has been made within the five-year period ending on the date of this Agreement.

(j) <u>Schedule 2.9(j)</u> lists the Intracompany Obligations as of August 30, 2013.

2.10 Absence of Changes

Since December 31, 2012, there has not been any event, development or state of circumstances that has had or would reasonably be expected to have, individually or in the aggregate, a Material Adverse Effect. Other than as set forth in <u>Section 4.8</u>, between December 31, 2012 and the date hereof (<u>i</u>) the Business has been conducted in the ordinary course consistent with past practices and (<u>ii</u>) except as set forth on <u>Schedule 2.10</u>, neither Harsco nor any of its Affiliates (including the Target Entities) has taken any action that would, if taken after the date hereof, require the consent of CD&R Investor under clauses (5), (7), (8), (11), (14) and (16) of <u>Section 4.1(b</u>), and neither Harsco nor any of its Affiliates (including the Target Entities) has taken any material action outside the ordinary course of business that would, if taken after the date hereof, require the consent of CD&R Investor under clause (6) of <u>Section 4.1(b</u>).

2.11 <u>Title to Assets; Sufficiency of Assets</u>

(a) Except as would not, individually or in the aggregate, reasonably be expected to be adverse in any material respect to the Business or the Target Entities, taken as a whole, the Target Entities have good and valid (and, in the case of Owned Real Property, valid and marketable fee simple) title to, or have a valid leasehold or licensed interest in, all of the tangible and intangible assets, properties and rights (real and personal, including all Intellectual Property) that (i) are primarily used or held for use in connection with the Business, (ii) are reflected on the Financial Statements or (iii) were acquired by the Business after the Balance Sheet Date (collectively, the "Assets"), in each case, free and clear of all Encumbrances, except for Permitted Encumbrances, it being understood that this representation and warranty shall not apply to any Asset disposed of by the Target Entities (\underline{x}) after the Balance Sheet Date and prior to the date hereof in the ordinary course of business or (\underline{y}) after the date hereof and not in violation of this Agreement. This representation and warranty shall not apply to infringement of third party Intellectual Property, which is the subject of <u>Section 2.13(c)</u> hereof.

(b) Except as set forth on <u>Schedule 2.11</u>, the Assets that will be owned, leased or licensed by Buyer and its Subsidiaries immediately after giving effect to the Transactions, together with the Shared Contracts (excluding any Replicated Contracts), Intellectual Property that will be licensed pursuant to <u>Section 7.5</u> of this Agreement and the services and Intellectual Property that will be provided or licensed pursuant to the Transition Services Agreement, constitute all of the assets, properties and rights required for the conduct of the Business in all material respects as of (i) the Balance Sheet Date (other than such assets, properties and rights acquired or disposed of by the Target Entities in the ordinary course

of business, after the Balance Sheet Date and prior to the date hereof) and (ii) the date of this Agreement (other than such assets, properties and rights acquired, or disposed of by the Target Entities in accordance with this Agreement, after the date of this Agreement). After giving effect to the Transactions, Harsco and the Non-Company Affiliates (including, for the avoidance of doubt, the Asset Sellers) will not own or have any interest in (other than to the extent resulting from their ownership of the Harsco Partnership Interests) any material tangible or intangible assets, material properties or rights (real or personal, including Intellectual Property) that are used or held for use in connection with the Business as of date hereof except for Intellectual Property that will be licensed pursuant to <u>Section 7.5</u> of this Agreement, the services and Intellectual Property that will be provided or licensed pursuant to the Transition Services Agreement, the Shared Contracts, the Excluded Assets and as set forth on <u>Schedule 2.11</u>.

2.12 Real Property

(a) <u>Schedule 2.12(a)</u> sets forth an accurate and complete list of each material real property owned by any Target Entity or that otherwise primarily relates to the Business and is owned by Harsco or any Non-Company Affiliate, in each case, as of the date hereof (together with all structures, facilities, improvements and fixtures presently or hereafter located thereon or attached thereto, the "<u>Owned Real Property</u>"), including the address and owner of each parcel of such Owned Real Property.

(b) <u>Schedule 2.12(b)</u> sets forth an accurate and complete list of each material real property leased, subleased or licensed by any Target Entity or that otherwise relates primarily to the Business and is leased, subleased or licensed by Harsco or any Non-Company Affiliate (the "<u>Leased Real Property</u>"), including all leases, subleases or licenses (together with any and all amendments and modifications thereto and any guarantees thereof) in effect as of the date hereof pursuant to which such Leased Real Property is leased, subleased or licensed by any Target Entity or Harsco (or any Affiliate thereof) as lessee, sublessee or licensee (collectively, the "<u>Real Property Leases</u>"), and identifying the landlord, tenant and address for each Real Property Lease. Harsco has delivered to CD&R Investor accurate and complete copies of each Real Property Lease.

(c) Except as set forth on <u>Schedule 2.12(c)</u>, neither Harsco nor any of its Affiliates (including the Target Entities) has leased, subleased or granted to any Person any material right to possess, lease or occupy any portion of the Owned Real Property or the Leased Real Property.

(d) Except as set forth on <u>Schedule 2.12(d)</u>, neither Harsco nor any of its Affiliates (including the Target Entities) holds, has granted or is obligated under any option, right of first offer, right of first refusal or other contractual right to purchase, acquire, sell or dispose of any material Owned Real Property or any material Leased Real Property or any portion thereof or interest therein or any other material real property.

2.13 Intellectual Property

(a) <u>Schedule 2.13(a)</u> sets forth an accurate and complete list of (<u>i</u>) all registered Intellectual Property and all applications for registration for Intellectual Property primarily used or primarily held for use by the Business (the "<u>Registered Intellectual Property</u>") and (<u>ii</u>) all unregistered Marks that are material to the Business, in each case, owned, beneficially or of record, by (<u>x</u>) the Target Entities or (<u>y</u>) Harsco or any Non-Company Affiliates as of the date of this Agreement, and the owner thereof. The Registered Intellectual Property and each of the unregistered Marks, Copyrights and Trade Secrets that are owned by the Target Entities or the Asset Sellers and are material to the Business (collectively, the "<u>Owned Intellectual Property</u>") are owned by such Target Entities or the Asset Sellers

(and upon the Closing will be owned by Buyer and its Subsidiaries) free and clear of all Encumbrances other than Permitted Encumbrances.

(b) Other than as set forth on <u>Schedule 2.13(b)</u>, there is no pending or, to the Knowledge of Harsco, threatened claim, action, opposition, interference, reexamination or proceeding that relates to the rights of any Target Entity, Harsco or any Non-Company Affiliates in respect of any Owned Intellectual Property primarily used in the conduct of the Business.

(c) Other than as set forth on <u>Schedule 2.13(c)</u>, (i) the conduct of the Business does not infringe, misappropriate or violate any Intellectual Property of any Person and (ii) to the Knowledge of Harsco, no Person is infringing, misappropriating or violating the Owned Intellectual Property, except in each case of (i) and (ii) as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(d) Except as would not, individually or in the aggregate, reasonably be expected to have a material and adverse effect on the conduct of the Business, as currently conducted, following the Closing, the Target Entities have used (and Harsco and its Affiliates have otherwise in relation to the Business used) commercially reasonable efforts to ensure protection of the Owned Intellectual Property, including the confidentiality of their Trade Secrets. There is no patentable invention created by an employee or former employee of Harsco or its Affiliates that is material to the conduct of the Business for which Harsco or its Affiliates have sought, or wish to seek, patent protection for which Harsco or its Affiliates (including the Target Entities) of such invention. There is no Intellectual Property created by an independent contractor of Harsco or its Affiliates that is material to the conduct of the Business which was intended to be owned by Harsco or its Affiliates for which Harsco or its Affiliates have not obtained an assignment agreement that effectively vests ownership of such invention is Affiliates for which Harsco or its Affiliates have not obtained an assignment agreement that effectively vests ownership of such intellectual Property in Harsco or its Affiliates (including the Target Entities) of such including the Target Entities or its Affiliates have not obtained an assignment agreement that effectively vests ownership of such Intellectual Property in Harsco or its Affiliates (including the Target Entities) of such including the Target Entities) of such Intellectual Property.

(e) The Target Entities have implemented reasonable backup and disaster recovery arrangements to provide for the continued operation of the Business in the event of a disaster or business interruption. The Internal IT Systems owned by the Target Entities or that constitute Target Assets (i) are in good repair and operating condition and are adequate and suitable in all material respects for the purposes for which they are being used or held for use, (ii) conform in all material respects with their related documentation and (iii) do not contain any virus, Trojan, or other malicious code that would reasonable efforts. The Target Entities have not (and Harsco and its Affiliates have not otherwise in relation to the Business), in the past three (3) years, experienced any data security breach or failure or unauthorized data release incident that has had a material adverse effect on the Business.

(f) Solely relating to the Business, Harsco and each of its Affiliates (including the Target Entities) and each of their respective Assets is in compliance with all contractual and legal requirements pertaining to data protection or information privacy and security, including any privacy policy concerning the collection or use of data or information, except as would not, individually or in the aggregate, reasonably be expected to have an adverse effect in any material respect on the Business or the Target Entities, taken as a whole.

2.14 Contracts

(a) <u>Schedule 2.14(a)</u> sets forth an accurate and complete list of all Contracts in effect as of the date hereof to which any Target Entity, Harsco or any Non-Company Affiliate (but with respect to Harsco and any Non-Company Affiliate, only such Contracts primarily relating to the Business), is a party (other than purchase orders or standard sales orders entered into by any Target Entity, as applicable, in the ordinary course of business) which by its terms:

(i) is not terminable at will within six (6) months and requires future expenditures or other performance with respect to goods, equipment or services having an annual value in excess of, with respect to Contracts listed in <u>Schedule 2.14(ii)</u>, \$2,000,000, and with respect to such other Contracts, \$250,000, in each case, other than any Business Benefit Plan or related Contract;

(ii) relates to the obligation of the Target Entities or the Asset Sellers to purchase products, materials, supplies, goods, equipment, other assets or services pursuant to which payments of \$2,000,000 or more were made during the 12-month period ending on December 31, 2012, other than any Business Benefit Plan or related Contract;

(iii) relates to the obligation of the Target Entities or the Asset Sellers to sell products or services pursuant to which payments of \$250,000 or more were received during the 12-month period ending on December 31, 2012;

(iv) relates to any Indebtedness other than Intercompany Balances or Intracompany Obligations;

(v) (i) limits the ability of any Target Entity to compete in any line of business that is material to the Business or with any Person in any geographic area and that would so limit the freedom of the Target Entities or any of their Affiliates after the Closing or (ii) contains exclusivity obligations binding on any Target Entity or the Asset Sellers;

(vi) requires any capital commitment or capital expenditure (including any series of related expenditures) by the Target Entities or the Asset Sellers of greater than \$250,000;

(vii) relates to the acquisition or disposition of any business, assets, properties or rights under which the Target Entities have any future liability with respect to an "earn-out," contingent purchase price, deferred purchase price or similar contingent payment obligation;

(viii) constitutes a Contract (other than (\underline{x}) purchase orders or standard sales orders entered into by any Asset Seller or Target Entity in the ordinary course of business, (\underline{y}) Business Benefit Plans and related contracts and agreements, or (\underline{z}) intercompany notes that will be terminated pursuant to Section 4.8(a)), to which one of the counterparties is Harsco or any Non-Company Affiliate or any officer, director, stockholder or other Affiliate of Harsco (other than the Target Entities) or any Non-Company Affiliate (each, an "Intercompany Contract");

(ix) constitutes an agreement that contains any indemnification obligations of any Target Entity or any Asset Seller, or credit support relating to such indemnification obligations, other than any of such indemnification obligations or credit support incurred in

the ordinary course of business or that require credit support or indemnification obligations of less than \$250,000 or that are Business Benefit Plans; or

(x) is an Intellectual Property Contract that is material to the Business (excluding Intellectual Property Contracts for commercially available off-the-shelf Software that is not the subject of a negotiated agreement and for which the aggregate annual amounts paid or payable to or by the Target Entities related to such Contract are less than \$250,000), including the Contracts with Oracle Corporation for the benefit of the Target Entities; or

(xi) is a Real Property Lease (the Contracts described in clauses (i) through (xi), the "Material Contracts").

Harsco has made available to CD&R Investor a true and complete copy of (\underline{x}) each Material Contract (including all modifications and amendments thereto and written waivers thereunder) and (y) all form purchase orders or contracts of the Target Entities and the Asset Sellers that are material to the Business. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, each Material Contract to which Harsco or any of its Affiliates (including the Target Entities), as applicable, is a party is in full force and effect in accordance with its respective terms. Each Material Contract is valid, binding and enforceable against Harsco or such Affiliate, as applicable, and, to the Knowledge of Harsco, each other party thereto in accordance with its terms, except as enforceability may be limited by bankruptcy Laws, other similar Laws affecting creditors' rights and general principles of equity affecting the availability of specific performance and other equitable remedies (regardless of whether enforcement is sought in a proceeding at law or in equity) and except as would not reasonably be expected to have, individually or in the aggregate, a material and adverse effect on the Business or the Target Entities, taken as a whole. Neither Harsco nor any of its Affiliates (including the other Sellers and Target Entities) is in default, violation or breach in any material respect under (or, to the Knowledge of Harsco, is alleged to be in default or breach in any material respect under) any such Material Contract to which it is a party, or has within the last 12 months provided or received notice of any intention to terminate any such Material Contract. To the Knowledge of Harsco, no event or circumstance has occurred and is continuing that constitutes or, with notice or the passage of time or both, would constitute, a default, violation or breach in any respect under any such Material Contract by any party thereto, or result in a termination thereof or would cause or permit the acceleration of or other changes of or to any right or obligation or the loss by Harsco or any of its Affiliates (including the Target Entities) of any benefit thereunder, in each case, except for such defaults, breaches, violations, terminations, accelerations or changes as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(b) <u>Schedule 2.14(b)</u> sets forth an accurate and complete list of all Shared Contracts in effect as of the date of this Agreement (excluding any Business Benefit Plan and Contracts for commercially available off-the-shelf Software that is not the subject of a negotiated agreement and for which the aggregate amounts paid or payable to or by Harsco or any of its Affiliates related to such Contract are less than \$250,000).

2.15 Compliance with Laws; Permits

Other than as set forth on <u>Schedule 2.15</u>, each Asset Seller (solely with respect to the Business) and each Target Entity is, and since January 1, 2012, has been, in compliance with all Laws applicable to it, except as would not, individually or in the aggregate, reasonably be expected to have a material and adverse effect on the Business or the Target Entities, taken as a whole. The Target Entities hold all

Permits that are required for the conduct of the Business as now being conducted, except where the failure to have any such Permits would not, individually or in the aggregate, reasonably be expected to be adverse in any material respect to the Business or the Target Entities, taken as a whole. Except as would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect, (i) all such Permits are in full force and effect, (ii) no violations are recorded in respect of any such Permits and (iii) no Proceedings are pending, or to the Knowledge of Harsco, threatened to revoke, suspend, cancel or limit any such Permit. Neither Harsco nor any of its Affiliates (including the Target Entities) makes any representation or warranty in this Section 2.15 with respect to Tax matters, which are governed exclusively by Section 2.9, or environmental matters, which are governed exclusively by Section 2.18.

2.16 Anti-Corruption; Trade Controls

(a) With respect to the Sellers, solely relating to the Business, the Sellers and the Target Entities, and each of their respective officers, directors, employees, agents, distributors and other Persons acting for or on behalf of the Business, have not directly or indirectly violated or taken any act in furtherance of violating any provision of the U.S. Foreign Corrupt Practices Act of 1977 (as amended), the U.K. Bribery Act 2010 or any other anti-corruption or anti-bribery Laws applicable to the Target Entities or the Business (collectively, the "<u>Anti-Corruption Laws</u>").

(b) With respect to the Sellers, solely relating to the Business, the Sellers and the Target Entities, and each of their respective officers, directors, employees, agents, distributors and other Persons acting for or on behalf of the Business, have not directly or indirectly taken any act in furtherance of an offer, payment, promise to pay, gift, bribe, rebate, loan, payoff, kickback or any other transfer of value provided to a Government Official for the purpose of inducing such Government Official to do any act or make any decision in an official capacity, including a decision to fail to perform an official function, or use his or her or its influence with a Governmental Body in order to affect any act or decision of such Governmental Body for the purpose of assisting any Person to obtain or retain any business, or to facilitate efforts of any Person to transact business or for any other improper purpose (*e.g.*, to obtain a tax rate lower than allowed by Law).

(c) To the Knowledge of Harsco, there is no investigation, allegation, request for information, or other inquiry by any Governmental Body regarding the Target Entities' or the Business's actual or possible violation of the Anti-Corruption Laws.

(d) To the Knowledge of Harsco, none of the officers, directors or employees, agents (or their employees) involved in the Business is a Government Official or consultant to any Government Official, and there is no existing family relationship between any officer, director, employee, agents (or their employees) member of the Target Entities or any Subsidiary and any Government Official.

(e) With respect to the Sellers, solely relating to the Business, the Sellers and the Target Entities, and each of their respective officers, directors, employees, agents, distributors and other Persons acting for or on behalf of the Business, have not directly or indirectly (i) circumvented the internal accounting controls of Harsco or any of the Target Entities, (ii) falsified any of the books, records or accounts of Harsco or any of the Target Entities, or (iii) made false or misleading statements to, or attempted to coerce or fraudulently influence, an accountant in connection with any audit, review or examination of the financial statements of Harsco or any of the Target Entities.

(f) With respect to the Sellers, solely relating to the Business, the Sellers and the Target Entities, and each of their respective officers, directors, employees, agents, distributors and other Persons acting for or on behalf of the Business (the "<u>Relevant Persons</u>"), have not in the course of their actions for,

or on behalf of, any Seller or Target Entity or any of their respective Subsidiaries or controlled Affiliates, engaged directly or indirectly in transactions connected with any of North Korea, Cuba, Iran, Syria, Myanmar or Sudan, or otherwise engaged directly or indirectly in transactions connected with any government, country or other entity or Person that is the target of U.S. economic sanctions administered by the U.S. Treasury Department Office of Foreign Assets Control ("<u>OFAC</u>") or any applicable UN, EU or other international sanctions regime, including specially designated nationals and blocked persons designated by the OFAC and no Relevant Person is any such person.

2.17 Insurance

<u>Schedule 2.17</u> contains an accurate and complete list of all material policies of insurance maintained as of the date of this Agreement with respect to any Target Entity (each, a "<u>Company Policy</u>"), together with, with respect to Company Policies identified as "Global Policies" on <u>Schedule 2.17</u>, the limits and any deductible or retention amounts of each such Company Policy. Except as would not, individually or in the aggregate, reasonably be expected to be adverse in any material respect to the Business and the Target Entities, taken as a whole, (<u>a</u>) no Target Entity is in default with respect to its obligations under any Company Policy and (<u>b</u>) no Target Entity has received, since the time any Company Policy was last renewed or issued (whichever is later), written notice of termination, cancellation or non-renewal of any such Company Policy from any of its insurance brokers or carriers. As of the date hereof, there is no material claim by any Target Entity pending under any Company Policy as to which coverage has been denied or disputed by the underwriters of such Company Policy. To the Knowledge of Harsco, Harsco and its Affiliates have properly reported all material claims, acts, omissions, events, circumstances, occurrences and losses relating to the Business to the extent required under each Company Policy except those not reported in the ordinary course and consistent with past practice.

2.18 Environmental Matters

(a) Each Target Entity and each Asset Seller (solely with respect to the Business) complies and, while owned by Harsco, has complied with all applicable Environmental Laws, except where the failure to comply would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(b) Each Target Entity and each Asset Seller (solely with respect to the Business) holds and is in compliance with all Permits required under applicable Environmental Law to operate as it currently operates, except where the failure to hold such Permits would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(c) No Target Entity, nor any Asset Seller (solely with respect to the Business), has received any written notice from any Person regarding any actual or alleged violation of Environmental Laws, or any liabilities or alleged liabilities for investigation costs, cleanup costs, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorney fees under Environmental Laws, the subject of which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(d) There is no Governmental Order, pursuant to any Environmental Laws, under which a Target Entity or the Business has outstanding obligations that would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(e) No Release of Hazardous Substances has occurred on, at, under or from or is affecting any property currently owned or leased by any Target Entity and no Target Entity, Asset Seller

(solely with respect to the Business) nor to the Knowledge of Harsco, has any of their predecessors or Affiliates has caused the Release of Hazardous Substances on, at, under or from any property formerly owned or leased by a Target Entity, in any case, in a manner that would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(f) No Target Entity has, during the period of Harsco's ownership of such Target Entity, and to the Knowledge of Harsco, no Target Entity has, during any period prior to Harsco's ownership of any such Target Entity disposed of or arranged for the disposal of any Hazardous Substance that has resulted, or would reasonably be expected to result, in investigation costs, cleanup costs, response costs, corrective action costs, personal injury, property damage, natural resources damages or attorney fees under or pursuant to any applicable Environmental Laws, which would, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect.

(g) To the Knowledge of Harsco, Harsco has made available to CD&R Investor all environmental assessments, investigations and studies in the possession, custody or control of Harsco or any of its Affiliates (including the Target Entities) prepared since January 1, 2009, and prior to the date hereof (other than those subject to attorney-client privilege), which identify issues at properties owned, leased or operated by any Target Entity or the Business that would, individually or in the aggregate, reasonably be expect to be material to the Target Entities, taken as a whole.

(h) Notwithstanding any other provisions of this <u>ARTICLE II</u>, other than as set forth in <u>Section 2.4</u> (Consents and Approvals) and this <u>Section 2.18</u>, Harsco makes no representations or warranties, either express or implied, with respect to matters arising under any Environmental Law or concerning any Hazardous Substance.

2.19 Financial Advisors

Other than Robert W. Baird & Co. Incorporated and Credit Suisse Securities (USA) LLC, whose expenses will be paid by Harsco or the Non-Company Affiliates, no Person has acted, directly or indirectly, as an agent, broker, investment banker, finder or financial advisor for Harsco or any of its Affiliates in connection with the transactions contemplated by this Agreement, and no Person is entitled to any fee or commission or like payment from any Target Entity in respect thereof. Except as otherwise agreed in writing by CD&R Investor or as expressly contemplated by this Agreement or the Ancillary Agreements, no Target Entity has paid or agreed to pay, reimburse or otherwise indemnify any Person for, any costs, fees or expenses incurred prior to the date of this Agreement by Harsco or any Non-Company Affiliate in connection with this Agreement, the Ancillary Agreements or the transactions contemplated hereby or thereby.

2.20 Transactions with Affiliates.

To the Knowledge of Harsco, no officer, director or employee of any Target Entity, or any family member, relative or Affiliate of any such officer, director or employee, (i) owns, directly or indirectly, any interest in (<u>A</u>) any asset or other property primarily used in or held for use in the Business or (<u>B</u>) any Person that is a supplier, customer or competitor of the Business or any Target Entity, (ii) serves as an officer, director or employee of any Person that is a supplier, customer or competitor of any Target Entity or (<u>iii</u>) is a debtor or creditor of any Target Entity.

2.21 Customers and Suppliers

<u>Schedule 2.21(a)</u> lists the names of the five largest customers in each of the eleven countries that account for the largest shares of the Business's revenue, as determined by annual revenue to the Business for the 2012 fiscal year (the "<u>Major</u> <u>Customers</u>"). <u>Schedule 2.21(b)</u> lists the names of the 20 largest suppliers of the Business, based on payments made by the Business during the 2012 fiscal year (the "<u>Major Suppliers</u>"). As of the date of this Agreement, no Seller or Target Entity (nor Harsco on their behalf) has received written notice that any Major Customer or Major Supplier has terminated or threatened in writing to terminate any material agreement with any Seller or Target Entity.

2.22 Investment Intent

Harsco is acquiring the Harsco Partnership Interests for its own account for investment purposes only and not with a view to any public distribution thereof or with any intention of selling, distributing or otherwise disposing of the Harsco Partnership Interests in a manner that would violate the registration requirements of the Securities Act or any similar provisions of any applicable Law. Harsco agrees that the Harsco Partnership Interests may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act and any applicable state securities Laws, except pursuant to an exemption from such registration under the Securities Act and such Laws. Harsco is able to bear the economic risk of holding the Harsco Partnership Interests for an indefinite period (including total loss of its investment), and (either alone or together with its representatives) has sufficient knowledge and experience in financial and business matters so as to be capable of evaluating the merits and risk of its investment.

2.23 Harsco Acknowledgment

Harsco acknowledges that it has made its own assessment of the present condition and the future prospects of the Company and, following the Closing, its Subsidiaries (including the Target Entities) and the Business and is sufficiently experienced to make an informed judgment with respect thereto. Harsco acknowledges that, except as explicitly set forth herein, neither CD&R Investor nor any of its Affiliates has made any representation or warranty, express or implied, as to the Company or any of its Subsidiaries (including the Target Entities) or the Business following the Closing (including the prospects thereof or their profitability for Harsco), or with respect to any forecasts, projections or business plans or other information prepared by or on behalf of CD&R Investor and delivered to Harsco in connection with Harsco's review of its acquisitions of the Harsco Partnership Interests and the negotiation and the execution of this Agreement and the Ancillary Agreements.

2.24 <u>Disclaimer of Other Representations</u>. Except for the representations and warranties contained in this <u>ARTICLE II</u>, or in any certificate pursuant to this Agreement or in any Ancillary Agreement, neither Harsco nor any of its Affiliates makes any representation or warranty, express or implied, statutory or otherwise, in respect of Harsco or any of its Affiliates (including the other Sellers and the Target Entities) or any of the assets, liabilities or operations of any of the foregoing Persons or the Business, and any such other representation or warranty and liability and responsibility for such representation or warranty are expressly disclaimed. Harsco and its Affiliates make no representations or warranties regarding the probable future success or profitability of the Business, the Target Entities or the Target Assets.

ARTICLE III

REPRESENTATIONS AND WARRANTIES OF CD&R INVESTOR

CD&R Investor hereby represents and warrants to Harsco as of the date hereof (except in each case to the extent that such representations and warranties speak only as of another date or dates in which case, only as of such date(s)) as follows:

3.1 Incorporation and Authority

(a) Each of the Buyers, CD&R Investor, the Company and each other Person owned directly or indirectly by the Company and having a direct or indirect equity interest in Buyer (each an "<u>Intermediate Company</u>") is a corporation or other business organization duly incorporated or organized, validly existing and in good standing under the Laws of its jurisdiction of incorporation or organization. Each of the Buyers, CD&R Investor, the Company and each Intermediate Company has all requisite corporate or other power and authority to own its assets, properties and rights and to carry on its business as now being conducted and is duly qualified or licensed to do business and is in good standing in the jurisdictions in which the ownership of its property or the conduct of its business requires such qualification or license.

(b) Immediately after giving effect to the Closing, the equity interests of the Upper-Tier Partnership will consist of (<u>x</u>) the CD&R Investor Partnership Interests owned by CD&R Investor free and clear of any Encumbrances (other than any Encumbrance imposed by federal and state securities Laws or the Ancillary Agreements), which, after taking into account the Class B Units of Upper-Tier Partnership to be issued pursuant to <u>Section 1.4</u>, will represent an indirect 60% equity interest in the Company immediately following the Closing (excluding the effect of any equity held by management of the Company and its Subsidiaries) and (<u>y</u>) the Harsco Partnership Interests owned by Harsco free and clear of any Encumbrances (other than any Encumbrance imposed by federal and state securities Laws or the Ancillary Agreements), which when combined with the Class B Units of Upper-Tier Partnership to be issued pursuant to <u>Section 1.4</u>, will represent an indirect 29% equity interest in the Company immediately following the Closing (excluding the effect of any equity held by management of the Company and its Subsidiaries). All of the issued and outstanding shares of Buyer are owned, directly or indirectly, beneficially and of record free and clear of any Encumbrances, by the Company, and all of such issued and outstanding shares have been validly issued and fully paid and nonassessable. Except as contemplated by this Agreement and the transactions contemplated hereby (including the Transactions), there currently are no, and, immediately following the Closing, there will be no, outstanding options, warrants, calls, rights or any other agreements, trust, proxies or similar arrangements affecting the sale, issuance or voting of any shares of capital stock or equity interests of each of Buyer, CD&R Investor, the

Company or the Intermediate Companies, or any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any shares of capital stock or equity interests of each of Buyer, CD&R Investor, the Company or the Intermediate Companies. None of Buyer, CD&R Investor, the Company or the Intermediate Companies has any outstanding indebtedness for borrowed money, outstanding bonds, debentures, notes or other securities, the holders of which have the right to vote (or which are convertible into or exchangeable for securities having the right to vote) with the equity holders of Buyer, CD&R Investor, the Company or the Intermediate Companies. Each of Buyer, CD&R Investor, the Company and the Intermediate Companies were formed solely for the purpose of engaging in the transactions contemplated hereby and none of Buyer, CD&R Investor, the Company or the Intermediate Companies has conducted any operations, or has any liabilities, in each case other than in connection with this Agreement and the transactions contemplated hereby.

(c) At the Closing, (i) all of the issued and outstanding shares of each of the Stock Buyers and Asset Buyers will be owned, directly or indirectly, free and clear of any Encumbrances, by Buyer, and all of such issued and outstanding shares have been validly issued and fully paid and nonassessable, (ii) there will be no, outstanding options, warrants, calls, rights or any other agreements, trust, proxies or similar arrangements affecting the sale, issuance or voting of any shares of the capital stock of any Stock Buyer or Asset Buyer, or any securities or other instruments convertible into, exchangeable for or evidencing the right to purchase any shares of capital stock of any Stock Buyer or Asset Buyer, and (iii) none of the Stock Buyers or Asset Buyers will have any outstanding indebtedness for borrowed money, outstanding bonds, debentures, notes or other securities, the holders of which have the right to vote (or which are convertible into or exchangeable for securities having the right to vote) with the stockholders of any Stock Buyer or Asset Buyer; and (iv) each Stock Buyer and Asset Buyer were formed solely for the purpose of engaging in the transactions contemplated hereby and none of them will have conducted any operations, or have any liabilities, in each case other than in connection with this Agreement and the transactions contemplated hereby.

3.2 No Conflict

The execution, delivery and performance by CD&R Investor and Buyer of this Agreement and the Ancillary Agreements do not and will not: (<u>i</u>) conflict with or violate the constituent documents of CD&R Investor or Buyer; (<u>ii</u>) assuming all Consents referred to in <u>Section 3.3</u> have been obtained, conflict with or violate any Law applicable to CD&R Investor, Buyer or their respective Affiliates, or to any of their respective properties, rights or assets; or (<u>iii</u>) result in any breach of, or constitute a default (or event which, with the giving of notice or lapse of time, or both, would become a default) under, or give to any Person any right of termination, rescission, acceleration or cancellation of, or any right of notice or Consent with respect to, any note, bond, mortgage, indenture, contract, agreement, lease, license, Permit, franchise or other instrument to which CD&R Investor or Buyer is a party or by which any of the properties, rights or assets of CD&R Investor or Buyer is bound or affected except, in the case of clauses (ii) and (iii), where such conflicts, violations, rescissions, accelerations, breaches, defaults, terminations, cancellations or failures to obtain such Consents would not, individually or in the aggregate, be reasonably expected to materially impair the ability of CD&R Investor or Buyer to consummate the transactions contemplated hereby and by the Ancillary Agreements.

3.3 Consents and Approvals

The execution and delivery by CD&R Investor and Buyer of this Agreement does not, and the execution and delivery by CD&R Investor or Buyer of the Ancillary Agreements to which any of them is a party will not, and the performance by CD&R Investor and Buyer of this Agreement and the Ancillary Agreements to which it is or they are a party or parties and the consummation of the transactions

contemplated hereby and thereby will not, require CD&R Investor or Buyer to seek or obtain any Consent from, or make any filing or notification with, any Governmental Body, except for (i) those Consents set forth on <u>Schedule 3.3(i)</u>, and (ii) compliance with the applicable requirements of the HSR Act or any other Competition Laws of the jurisdictions set forth on <u>Schedule 3.3(ii)</u>.

3.4 Litigation

As of the date hereof, there is no Proceeding pending or, to the knowledge of CD&R Investor, threatened against CD&R Investor, Buyer or any of their respective Affiliates or any of their respective properties, rights and assets, which bring into question the validity of this Agreement or that, individually or in the aggregate, would reasonably be expected to prevent or materially delay the performance by CD&R Investor or Buyer of their obligations hereunder or under any Ancillary Agreement. Neither CD&R Investor nor Buyer, nor any of their respective Affiliates, is subject to any order, writ, judgment, injunction, decree or award that, individually or in the aggregate, would prevent, materially delay or materially impede the consummation of the transactions contemplated hereby or by any Ancillary Agreement.

3.5 Financial Advisors

Except as set forth in <u>Schedule 3.5</u>, no Person has acted, directly or indirectly, as an agent, broker, investment banker, finder or financial advisor for CD&R Investor in connection with the transactions contemplated by this Agreement, and no Person is entitled to any fee or commission or like payment from CD&R Investor in respect thereof.

3.6 Financing

CD&R Investor has delivered to Harsco true, complete and correct copies of: (i) the executed commitment letter, dated as of the date hereof by and among Morgan Stanley Senior Funding, Inc. ("MSSF"), Citigroup Global Markets Inc. ("CGMI"), Goldman Sachs Bank USA ("GS"), UBS Securities LLC ("UBSS") and UBS Loan Finance LLC ("UBSLF", collectively and together with MSSF, CGMI, GS and UBSS and with any Additional Commitment Lender (as defined in the Debt Commitment Letter), the "Debt Commitment Parties") and Bullseye MergerSub (the "Debt Financing Commitment"), pursuant to which, upon the terms and subject to the conditions set forth therein, MSSF, CGMI, GS and UBSLF have agreed to lend the amounts set forth therein to Bullseye MergerSub (the "Debt Financing"); (ii) the executed equity commitment letter, dated as of the date hereof among Buyer, Guarantor and the other parties thereto (the "Equity Financing Commitment" and together with the Debt Financing Commitment, the "Financing Commitments"), pursuant to which, upon the terms and subject to the conditions set forth therein, Guarantor has committed to invest the cash amount in Buyer set forth in its Equity Financing Commitment (the "Equity Financing" and together with the Debt Financing, the "Financing"); (iii) the fee letter executed in connection with the Debt Financing Commitment (the "<u>Fee Letter</u>"); (iv) the engagement letter in connection with the potential offering of high yield bonds in lieu of a portion of the Debt Financing (the "Engagement Letter") and (v) the fee credit letter executed in connection with the Engagement Letter (the "Fee <u>Credit Letter</u>"). None of the Financing Commitments, the Fee Letter , the Engagement Letter or the Fee Credit Letter has been amended or modified prior to the date of this Agreement (provided that the existence or exercise of the "flex" provisions contained in the Fee Letter shall not constitute an amendment or modification of the Financing Commitments), and, as of the date hereof, the respective commitments contained in the Financing Commitments have not been reduced, withdrawn, terminated or rescinded in any respect. There are no other agreements, side letters, contracts or arrangements, whether written or oral, pursuant to which CD&R Investor or any of its Affiliates is a party relating to any the Financing Commitments. As of the date hereof, the Financing Commitments are

in full force and effect and constitute the legal, valid, binding and enforceable obligations of CD&R Investor or any of its Affiliates, as applicable, and, to the knowledge of CD&R Investor, the other parties thereto (subject in each case to the effect of bankruptcy, insolvency, fraudulent conveyance, reorganization, moratorium, receivership or similar Laws relating to or affecting creditors' rights generally and by general equity principles, whether considered in Proceedings in equity or at law) and are not subject to any conditions that are not set forth in the copies of the Financing Commitments. Other than as expressly set forth in the Financing Commitments or as set forth in any such documents amended after the date hereof and not in violation of the provisions hereof, there are no conditions precedent related to the funding of the full net proceeds of the Financing (including any "flex" provisions) under any agreement relating to the Financing to which CD&R Investor or any of its Affiliates is a party. As of the date hereof, no event has occurred which with or without notice, lapse of time or both would constitute a breach of any covenants or other obligations by each of CD&R Investor or any of its Affiliates under the Financing Commitments, to the knowledge of CD&R Investor, or the other parties to the Financing Commitments. Bullseye MergerSub has fully paid or caused to be fully paid all commitment fees or other fees required to be paid on or prior to the date hereof pursuant to the Financing Commitments. As of the date hereof, CD&R Investor has no reason to believe that any of the conditions to the funding of the Financing Commitments will not be satisfied or that the Financing Commitments will not be made available to Bullseve MergerSub and Buyer on or prior to the Closing Date. Assuming the accuracy of the representations and warranties set forth in ARTICLE II and performance by the Sellers of their respective obligations hereunder, upon receipt of the proceeds contemplated by the Financing Commitments, CD&R Investor and Buyer will have access as of the Closing to sufficient cash funds to pay all amounts contemplated by this Agreement to be paid by them and to perform their respective obligations hereunder and to have sufficient cash funds to cover the immediate liquidity needs of the Company.

In no event shall the receipt or availability of the Financing Commitments or any other funds or financing by CD&R Investor or any Affiliate be a condition to any of Buyers' and CD&R Investor's obligations hereunder.

3.7 Guarantee

Concurrently with the execution of this Agreement, the Guarantor has delivered to Harsco the Guarantee, dated as of the date hereof, in favor of Harsco. The Guarantee is in full force and effect and constitutes a valid and binding obligation of the Guarantor, enforceable against the Guarantor.

3.8 Investment Intent

Buyers are purchasing the Target Shares for their own account and solely for investment, with no intention to sell, transfer or distribute any Target Shares to any other Person. At Closing, each Buyer will be an "accredited investor" as defined in Regulation D promulgated under the Securities Act. Each Buyer acknowledges that it is informed as to the risks of the transactions contemplated hereby and of ownership of the Target Shares. Each Buyer acknowledges that none of the Target Shares are registered under the Securities Act or under any state or foreign securities laws, and Buyers will not sell, transfer or distribute any Target Shares except in compliance with the registration requirements or exemption provisions under the Securities Act and the rules and regulations promulgated thereunder, or any other applicable securities Law.

3.9 CD&R Investor Acknowledgement

CD&R Investor acknowledges that it has made its own assessment of the present condition and the future prospects of Target Entities, the Target Assets and the Business and is sufficiently experienced to

make an informed judgment with respect thereto. CD&R Investor acknowledges that, except as explicitly set forth herein, neither Harsco nor any of its Affiliates has made any representation or warranty, express or implied, as to Target Entities, the Target Assets or the Business (including the prospects thereof or their profitability for CD&R Investor), or with respect to any forecasts, projections or business plans or other information prepared by or on behalf of Harsco and delivered to CD&R Investor in connection with CD&R Investor's review of Target Entities, the Target Assets and the Business and the negotiation and the execution of this Agreement and the Ancillary Agreements.

3.10 Agreements with CD&R Investor

As of the date hereof, there are no agreements, arrangements or understandings (other than as provided in this Agreement, the Brand Purchase Agreement and a confidentiality agreement) between the CD&R Investor or its Affiliates, on the one hand, and FRC or its Affiliates, on the other hand. True and complete copies of the Brand Purchase Agreement and such confidentiality agreement have been delivered to Harsco.

3.11 Disclaimer of Other Representations.

Except as expressly set forth in this <u>ARTICLE III</u>, in any certificate pursuant to this Agreement or in any Ancillary Agreement, neither CD&R Investor nor any of its Affiliates makes any representation or warranty, express or implied, statutory or otherwise, in respect of CD&R Investors or any of its Affiliates (including the Buyers) or any of the assets, liabilities or operations of any of the foregoing Persons, and any such other representations or warranties are expressly disclaimed. Without limiting the generality of the foregoing, CD&R Investor and Buyer acknowledge that neither Harsco nor any other Person makes any representation or warranty with respect to any projections, estimates, statements, or budgets of future revenues, future results of operations (or any component thereof), future cash flows or future financial condition (or any component thereof) of the Business or the future business and operations of the Business.

ARTICLE IV

CERTAIN COVENANTS

4.1 Conduct of Business Prior to the Closing

(a) Except as expressly permitted or contemplated by this Agreement, including pursuant to <u>Section 4.8</u>, as set forth on <u>Schedule 4.1(b)</u> or with respect to such further matters as may be consented to by CD&R Investor in writing, such consent not to be unreasonably withheld, conditioned or delayed, between the date hereof and the Closing Date, Harsco shall, and shall cause the Target Entities and the Non-Company Affiliates to the extent relating to the Business (which, for the avoidance of doubt, does not include any matter relating to the funding of the Harsco Pension Scheme, which shall be the sole responsibility of Harsco and the Non-Company Affiliates) to, conduct the Business in the ordinary course of business, consistent with past practices.

(b) Without limiting the generality of <u>Section 4.1(a)</u>, except (i) as expressly permitted or contemplated by this Agreement, including pursuant to <u>Section 4.8</u>, (ii) with respect to Harsco and the Non-Company Affiliates to the extent not relating to the Business, (iii) as required by applicable Law or (iv) as set forth on <u>Schedule 4.1(b)</u>, between the date hereof and the Closing Date, Harsco shall not, and shall not permit its Affiliates (including the Target Entities) to, do any of the following unless the prior

written consent of CD&R Investor (which shall not be unreasonably withheld, delayed or conditioned) has been obtained:

(1) amend or otherwise modify the respective articles of incorporation or bylaws (or similar organizational documents) of any Target Entity or enter into, amend or modify any Contract with a JV Partner;

(2) incur, guaranty or assume any Indebtedness or debt obligation other than Intercompany Obligations and obligations under Contracts, in each case, incurred in the ordinary course of business consistent with past practices;

(3) cancel, forgive, pay, discharge, satisfy, compromise or settle any material debt, claim or obligation of any Target Entity other than (<u>A</u>) in the ordinary course of business or as required pursuant to any Contract in effect as of the date hereof or (<u>B</u>) any Intercompany Contract, Intercompany Balance or Intracompany Obligations, in each case incurred in the ordinary course of business consistent with past practice;

(4) create or otherwise incur any Encumbrance on any material Asset of any Target Entity other than Permitted Encumbrances and other than such Encumbrances that will be released in full prior to the Closing;

(5) acquire (<u>A</u>) by merging or consolidating with, or by purchasing all or a substantial portion of the assets, properties or rights of, or by any other manner, any business of any Person or any division thereof, or (<u>B</u>) any capital stock, assets, properties or rights of any other Person, in each case, other than those acquired in the ordinary course of business consistent with past practices;

(6) except (A) as required pursuant to the terms of any Business Benefit Plan or by Law, (B) for increases in annual base salary or hourly base wages in the ordinary course of business and consistent with past practice for employees (other than the President of the Business and his direct reports) or (C) as may be required by any applicable collective bargaining, works council or similar agreement as in effect on the date hereof, (i) grant to any Business Employee any material increase in compensation or benefits, including any retention or change in control bonus, (ii) grant to any Business Employee any increase in severance or termination pay, (iii) enter into or amend in any material respect any employment, consulting, indemnification, severance or termination agreement with any Business Employee, (iv) enter into or amend any collective bargaining agreement with any Business Employee, except to transfer collective bargaining agreements to Buyer or a Target Entity, (v) establish, adopt, enter into or amend or terminate any Target Entity Benefit Plan or, with respect to Business Employees, any Harsco Benefit Plan, in a manner that would increase the liability of any Target Entity or (vi) accelerate the vesting, material payment or funding of any compensation or benefits under any Target Entity Benefit Plan or, if it would result in liability to Buyer or a Target Entity, under any Harsco Benefit Plan; provided, however, that the foregoing restrictions shall not restrict or prohibit any Target Entity from entering into with, or making available to, newly hired employees or to employees in the context of promotions based on job performance and workplace requirements (but, in each case, excluding the President of the Business and his direct reports and any employees hired for or promoted to such a position), any plan, agreement, benefit or compensation arrangement that does not contain change in control provisions and is made in the ordinary course of business consistent with past practices;

(7) make any (<u>i</u>) change in any material accounting principles, methods or practices followed by any Target Entity, or any material change in amortization policies or rates, or (<u>ii</u>) revaluation of any material Asset, in any such case in clause (i) or (ii), other than as required by applicable Law or GAAP, as then in effect;

(8) (<u>A</u>) sell, lease, encumber, exclusively license or otherwise dispose of any material properties, rights or Assets, except operational assets (i.e., excluding real property) in the ordinary course of business and non-exclusive licensing of Owned Intellectual Property, or (<u>B</u>) acquire any real property;

(9) make any loans, advances or capital contributions to, or investments in, any Person (other than pursuant to any existing obligations under any Contract) other than the advancement of trade credit to customers and expenses to employees or other than between or among Sellers, Non-Company Affiliates and any Target Entity in the ordinary course of business consistent with past practices;

(10) authorize for issuance, issue, sell, deliver or agree or commit to issue, sell or deliver (whether through the issuance or granting of options, warrants, convertible or exchangeable securities, commitments, subscriptions, rights to purchase or otherwise) any shares of capital stock or any other securities of any Target Entity, or amend any of the terms of any such securities or make any changes (by combination, reorganization or otherwise) in the equity capital structure of any Target Entity;

(11) incur any capital expenditures or commitments or additions to property, plant or equipment, other than (i) those capital expenditures that are paid in full prior to the Closing with cash or pursuant to financing lease commitments in the aggregate amount set forth on <u>Schedule 2.5(d)</u> and (ii) any individual expenditure or commitment or addition in an amount less than \$1,000,000;

(12) make any material payments or grant any material discounts to customers or suppliers of any Target Entity, other than those in the ordinary course of business consistent with past practices;

(13) fail to pay or satisfy when due any material account payable or other material liability of any Target Entity incurred in the ordinary course of business consistent with past practices (including, in each case, the timing of any such payments), other than any such liability that is being contested in good faith by any Target Entity;

(14) amend, cancel, compromise or waive any material claim or right of any Target Entity;

(15) (i) enter into (other than extensions at the end of a term in the ordinary course of business), transfer or terminate any Contract pursuant to which payments of \$3,000,000 or more are or will be made or received in a 12-month period (other than any Business Benefit Plan or related Contract) or materially modify or amend any such Contract, (ii) waive any right of material value under any Material Contract or other Contract that, if in effect on the date hereof, would have been a Material Contract or (iii) enter into any fixed price contracts in excess of \$3,000,000 relating to the provision of equipment, goods or services by the Business;

(16) except as otherwise required by Law, and other than in respect of a Consolidated Tax Return relating to income Taxes, (i) make or change any material Tax election, (ii) change any annual Tax accounting period, (iii) adopt or change any method of Tax accounting, (iv) amend any material Tax Returns or file any claims for material Tax refunds, (v) enter into any material closing agreement, or (vi) settle any material Tax claim, audit or assessment; provided that Harsco or any of its Affiliates (including the Target Entities) may take any such action described in clauses (i) through (vi) hereof without the prior written consent of CD&R Investor so long as Sellers indemnify and hold harmless the Buyer Indemnified Parties from and against any Taxes incurred as a result of any such action;

(17) amend or modify the Capital Expenditure Budget;

(18) (<u>i</u>) terminate, cancel, materially amend or modify any insurance policy maintained for the benefit of the Target Entities or the Business or any of their respective properties, in each case, in a manner that would have the effect of reducing coverage available to the Business with respect to periods prior to the Closing Date or (<u>ii</u>) allow any of the Target Entities to enter into any new insurance policy (other than any renewals of any insurance policy in effect on the date hereof) outside of the ordinary course of Business to the extent such policy would be in effect following the Closing; or

(19) authorize or announce an intention to do any of the foregoing, or enter into any Contract or arrangement to do any of the foregoing or any other action which would reasonably be expected to prevent or materially delay or materially impede the consummation of the Closing or the other transactions contemplated by this Agreement.

4.2 Cooperation

(a) On the terms and subject to the conditions of this Agreement, each Party shall use its reasonable best efforts to cause the Closing to occur (subject to the limitation in the proviso to Section 4.2(c)), including taking all actions reasonably necessary to comply promptly with all legal requirements that may be imposed on it or any of its Affiliates with respect to the Closing, including proffering and consenting to a Governmental Order providing for the sale or other disposition, or the holding separate, of particular assets, categories of assets or lines of business of any Party or its Affiliates (including any assets, categories of assets or lines of business held by Buyer after Closing), in order to remedy any material competition concerns that any Governmental Body may have; provided, however, that any such action is conditioned upon the consummation of the transactions contemplated in this Agreement; and, provided, further, that notwithstanding anything in this Agreement to the contrary, (x) no Seller shall agree to any such action with respect to the Target Entities, the Target Assets or the Business without the consent of CD&R Investor and (y) no Party shall be obligated to take or agree to take any such action that would, individually or in the aggregate with any other such action, reasonably be expected to have a material adverse effect on (1) the Business taken as a whole or (2) the Company and its Subsidiaries (immediately after giving effect to the Transactions), taken as a whole (a "Regulatory" Material Adverse Effect"). The entry of any Governmental Order permitting the consummation of the transactions contemplated hereby but requiring any of the assets or lines of business of any Party or its Affiliates (including any assets, categories of assets or lines of business held by Buyer after Closing), to be held separate thereafter shall not be deemed a failure to satisfy the conditions specified in <u>Section 6.1(d)</u> and <u>Section 6.2(d)</u>, <u>provided</u> that satisfying such requirement would not result in a Regulatory Material Adverse Effect.

(b) Without limiting the terms of this Section 4.2, Harsco and CD&R Investor shall (i) as soon as practicable after the date of this Agreement (and in any event within five (5) Business Days following the date hereof) make any filings required by any Governmental Body pursuant to the HSR Act or other Competition Laws in connection with the transactions contemplated hereunder, (ii) respond promptly to inquiries from the applicable Governmental Bodies in connection with such filings, including providing any supplemental information that may be requested by such Governmental Bodies and (iii) provide to the other copies of any filings made under the HSR Act or other Competition Laws at the time they are filed with the applicable Governmental Bodies, excluding any information included in any such filings that the party reasonably determines it needs to keep confidential. Harsco and CD&R Investor shall cooperate and use their respective best efforts to oppose any motion or action for temporary, preliminary or permanent injunction against the transactions contemplated by this Agreement and the Ancillary Agreements. For the avoidance of doubt, except as otherwise required by applicable Law, CD&R Investor shall have the primary responsibility for making any such filings and seeking necessary approvals, and Harsco shall cooperate with CD&R Investor to complete such filings or applications at the reasonable request of CD&R Investor. The Parties hereto shall furnish to each other such necessary information and reasonable assistance as the others may request in connection with its preparation of any filing or submission that is necessary under the HSR Act or other Competition Laws. Each party hereto shall keep the other apprised of the status of any communications with, and any inquiries or request for additional information from, Governmental Bodies pursuant to the HSR Act or any other Competition Laws, and each party hereto shall use all reasonable efforts to defend against any lawsuit, action or proceeding, judicial or administrative, challenging this Agreement or the transactions contemplated hereby. Except as otherwise provided herein, CD&R Investor shall be responsible for all fees and all costs and submissions of all regulatory filings related to any required governmental or regulatory approvals, including the HSR Act, with respect to the Applicable Non-U.S. Antitrust Approvals and with respect to any other applicable Laws.

(c) In connection with the matters contemplated by this Agreement, Harsco shall, and shall cause its Affiliates to, use commercially reasonable efforts prior to the Closing Date to give all notices to third parties and to obtain all third party Consents and assignments in connection with the transactions contemplated hereby and by the Ancillary Agreements and the Debt Financing Commitments, provided, however, that Harsco and its Affiliates shall not be required to pay or commit to pay any material amount to (or incur any extraordinary or unreasonable obligation in favor of) any Person from whom any such Consent or assignment may be required (excluding, for the avoidance of doubt, nominal filing or application fees) or agree to any material amendments to any Contract. To the extent that Harsco is unable to obtain any required third party Consents prior to the Closing despite such efforts, each of Harsco, Buyer and each of their respective Affiliates shall use its respective commercially reasonable efforts to obtain (or cause to be obtained), as promptly as practicable following the Closing, all such Consents and to complete any relevant transfers, leases or assignments following obtaining those Consents; provided, however, that no Person shall be required to pay or commit to pay any amount to (or incur any extraordinary or unreasonable obligation in favor of) any Person from whom any such Consent or assignment may be required (other than nominal filing or application fees) or agree to any amendments to any Contract except that, in respect of the Leased Real Property in the UK, the Buyer must provide a guaranty, rent deposit or other form of security to the relevant landlord if legally required by any of the leases as a valid condition or circumstance of obtaining consent. With respect to any Contract requiring a third party Consent after the Closing, Harsco and the Non-Company Affiliates shall use their commercially reasonable efforts to cooperate with Buyer and the Target Entities in any lawful and economically feasible arrangement (including by license) to provide Buyer and the Target Entities with the benefits under such Contract; provided that Buyer and the Target Entities shall undertake to pay or satisfy the corresponding liabilities for the enjoyment of such benefit to the extent Buyer and the Target

Entities would have been responsible therefor if such Consent had been obtained prior to the Closing. Notwithstanding any other provision of this Agreement to the contrary, this Agreement shall not constitute an agreement to contribute, convey, assign, transfer or deliver to the Buyers any Target Shares or Target Assets or any claim or right or any benefit arising thereunder or resulting thereform if an attempted contribution, conveyance, assignment, transfer or delivery thereof, or an agreement to do any of the foregoing, without the consent of a third party (including any Government Body), would constitute a breach or other contravention thereof or a violation of Law or would in any way adversely affect the rights of Buyer (as assignee or transferee of the applicable transferor, or otherwise), any other Target Entity thereto or thereunder.

(d) Each party hereto shall use its reasonable best efforts to finalize the terms of all Ancillary Agreements to be entered into at or prior to the Closing.

(e) Each party hereto shall keep the other reasonably apprised of the status of matters relating to completion of the transactions contemplated by this Agreement, including promptly furnishing the other with copies of written notices or other written communications received by CD&R Investor, on the one hand, or Harsco, on the other hand, as the case may be, or any of its Affiliates or representatives, from any Governmental Body with respect to the transactions contemplated by this Agreement or any Ancillary Agreement or from any Person alleging that the Consent of such Person is or may be required in connection with the transactions contemplated by this Agreement or any Ancillary Agreement.

4.3 Access to Information; Other Contacts

(a) From and after the date of this Agreement until the earlier of the Closing Date or termination of this Agreement, Harsco shall allow CD&R Investor and its accountants, counsel and other representatives, at CD&R Investor's expense, to make such reasonable investigation of the business, operations and properties of the Business and the Target Entities as reasonably necessary in connection with the transactions contemplated by this Agreement. Such investigation shall include reasonable access to the respective directors, officers, employees, agents and representatives (including legal counsel and independent accountants) of the Target Entities, and Harsco and the Non-Company Affiliates to the extent related to the Business, and the properties and books and records of the Target Entities and Harsco and the Non-Company Affiliates to the extent related to the Business. The Target Entities, Harsco and the Non-Company Affiliates, shall furnish CD&R Investor and its representatives with such financial, operating and other data and information and copies of documents with respect to the Target Entities, or Harsco or the Non-Company Affiliates to the extent related to the Business, or any of the transactions contemplated by this Agreement as CD&R Investor shall from time to time reasonably request. All access and investigation pursuant to this <u>Section 4.3</u> shall occur only upon reasonable notice and during normal business hours. Notwithstanding anything to the contrary in this Agreement, Harsco and tis Affiliates, including the Target Entities, shall not be required to disclose any information to CD&R Investor if such disclosure would reasonably be expected to (i) jeopardize any attorney-client or other legal privilege or (<u>ii</u>) contravene any applicable Laws, fiduciary duty or Contract entered into prior to the date hereof.

(b) During the period between the date hereof and the Closing, Buyer and its representatives may contact or communicate with the customers and suppliers of the Business in connection with the transactions contemplated hereby only with the prior written consent of Harsco (which shall not be unreasonably withheld but may be conditioned upon a designee of Harsco being present at any meeting or conference or included in any communication). For the avoidance of doubt, nothing in this <u>Section 4.3</u> shall prohibit CD&R Investor's Affiliates from contacting the customers, suppliers and licensors of the Business in the ordinary course of the businesses of CD&R Investor's

Affiliates for the purpose of selling such Affiliates' products or for any other purpose unrelated to the Business and the transactions contemplated by this Agreement and the Ancillary Agreements.

4.4 Books and Records; Access; Assistance

(a) Subject to <u>Section 5.5</u>, for a period of seven years after the Closing Date, subject to their document retention policies and procedures, Harsco and Buyer shall preserve and retain (or cause to be preserved and retained) all accounting, legal, auditing and other books and records (including any documents relating to any Proceedings arising out of or with respect to the operation or conduct of the Business) over which it has control to the extent relating to the conduct of the Business prior to the Closing Date. Notwithstanding the foregoing, during such seven-year period, any party (and its Affiliates) may dispose of any such books and records that are offered to, but not accepted by, the other party.

(b) From and after the Closing, if, in order to properly prepare documents required to be filed with Governmental Bodies (including Tax Authorities) or its financial statements, or for any other reasonable business purpose (including in connection with any securities offering), it is necessary that Harsco (or its Affiliates or its or their successors), on the one hand, or CD&R Investor (or its Affiliates or its or their successors), on the other hand, be furnished with additional information of the type described in <u>Section 4.4(a)</u> above, and such information is in the possession of the other party, except as would, in such party's reasonable discretion (<u>i</u>) jeopardize any attorney-client or other legal privilege or (<u>ii</u>) contravene any applicable Laws, fiduciary duty or Contract, such other party agrees to use its commercially reasonable efforts to furnish such information to the party that requires such information and provide reasonable access to any employees or other relevant personnel that may be reasonably required to be consulted with in connection with any such purpose, in each case, at the cost and expense of the party being furnished such information or access to such employees or personnel, as applicable.

(c) Nothing in this <u>Section 4.4</u> shall be deemed to modify or diminish any information rights to which a party is entitled pursuant to the Investor Rights Agreement.

4.5 Confidentiality

(a) Each of CD&R Investor and Buyer acknowledges that the information being provided to it in connection with the transactions contemplated hereby is subject to the terms of the Confidentiality Agreement, the terms of which are incorporated herein by reference. Effective upon the Closing, the obligations under the Confidentiality Agreement shall terminate except with respect to provisions regarding disclosure and use of confidential information not related to either the Business or the Target Entities, which shall continue in accordance with the terms of the Confidentiality Agreement. If for any reason this Agreement is terminated prior to the Closing Date, the Confidentiality Agreement shall nonetheless continue in full force and effect in accordance with its terms.

(b) Except as required by applicable Law, CD&R Investor and Buyer shall, and shall cause the Target Entities after the Closing, and shall direct their other representatives, to, to the extent any of them has any nonpublic information relating to Harsco and its Subsidiaries, but excluding information relating to either the Business or the Target Entities (any such information, "<u>Harsco Information</u>"), (i) treat the Harsco Information strictly confidentially, (ii) not use the Harsco Information for any purpose other than to consummate the transactions contemplated hereby and (<u>iii</u>) return to Harsco or destroy (at CD&R Investor's option) any Harsco Information that is documented or exists in written, photographic or other physical form (and any copies made thereof) upon Harsco's request; <u>provided</u>, <u>however</u> that in the event any disclosure of Harsco Information is required pursuant to applicable Law, to the extent permissible under applicable Law, CD&R Investor shall notify Harsco prior to the disclosure of such Harsco

Information pursuant to the terms of the Confidentiality Agreement and otherwise comply with Section 2.3 of the Confidentiality Agreement.

(c) Except as required by applicable Law, Harsco shall, and shall cause its Subsidiaries after the Closing, to the extent any of them has any nonpublic information relating to the Business, to treat such information strictly confidentially; <u>provided</u>, <u>however</u>, that in the event any disclosure of such confidentiality information is required pursuant to applicable Law, to the extent permissible under applicable Law, Harsco shall notify Buyer prior to the disclosure thereof.

4.6 Employees

(a) Subject in each case to all applicable Laws and the provisions of any applicable collective bargaining, works council or similar agreement and <u>Section 4.20</u>, effective no later than immediately prior to the Closing (or, in the case of any individuals employed in the UK, on Closing), Harsco shall cause the employment of all Business Employees employed in the U.S., the UK or the Netherlands and all shared employees mutually agreed by Buyer and Harsco prior to the Closing Date to transfer from the applicable Seller or Non-Company Affiliate to a Buyer or a Target Entity (and such individuals shall be Business Employees). Buyer shall use commercially reasonable efforts to cooperate with Harsco to effectuate such employment transfers; provided that Buyer shall have no obligation to agree to any terms or conditions with respect to such employment that are materially different from the terms and conditions applicable to such Business Employees as of the date hereof except to the extent required by applicable local Law. Harsco shall bear and be responsible for, and indemnify CD&R Investor, Buyer and the Target Entities and each of their respective Affiliates from and against any and all Liabilities:

(i) relating to any severance or termination payment or benefits payable or due to Business Employees employed in the U.S., the UK or the Netherlands or shared employees mutually agreed by CD&R Investor and Harsco prior to the Closing Date (who will transfer prior to or at Closing from the applicable Seller or Non-Company Affiliate to a Buyer or a Target Entity) required under the Business Benefit Plans solely as a result of the transfer of such employees, if any;

(ii) solely with respect to Liabilities arising on or prior to the Closing Date, such Liabilities arising from or relating to the Harsco Infrastructure Executive Retention Plan, any other Business Benefit Plans listed on <u>Schedule</u> <u>2.8(1)</u> or any other retention, severance or change in control agreement entered into with Business Employees prior to the Closing Date, including, without limitation, all Liabilities in respect of retention bonuses, guaranteed annual bonuses, enhanced severance pay and equity awards and other long-term incentives granted by Harsco and its Affiliates prior to the Closing Date and all associated employment taxes, except, in each case, for (\underline{x}) any such Liabilities accrued as a liability in Closing Net Working Capital and (\underline{y}) severance pay to the extent otherwise due under an applicable Business Benefit Plan generally applicable to Business Employees as in effect on the date hereof; and

(iii) relating to any claims for failure to inform and/or consult with such employees or their representatives as required by applicable Laws, whether pursuant to the Transfer of Undertakings (Protection of Employment) Regulations 2006 ("<u>TUPE</u>") or any other provisions under national laws or regulations implementing the EU Acquired Rights Directive, Directive 2001/23) ("<u>TUPE-Equivalent Regulations</u>"), except that Harsco shall not bear or be responsible for or indemnify CD&R Investor and/or any Buyer and/or Target

Entity and/or each of their respective Affiliates against any Liabilities arising out of any failure by any Buyer or Buyer Affiliate to provide information to any Seller, Target Entity or Non-Company Affiliate that is required under regulation 13(4) of TUPE or any other similar provisions under TUPE-Equivalent Regulations), and Buyer shall indemnify all Sellers and/or all Non-Company Affiliates from any Liabilities arising from any failure by any Buyer or Buyer Affiliate to provide such information required under regulation 13(4) of TUPE or any other similar provisions under TUPE-Equivalent Regulations and provisions under TUPE-Equivalent Regulation 13(4) of TUPE or any other similar provisions under TUPE-Equivalent Regulations.

The Parties agree that it would not be reasonable for any Buyer or Buyer Affiliate or Target Entity to enforce any of its or their rights against any Seller or Non-Company Affiliate in relation to "employee liability information" under regulations 11 and/or 12 of TUPE or any other similar provision under TUPE-Equivalent Regulations relating to any Business Employees employed in the UK or the Netherlands or any shared employees mutually agreed by CD&R Investor and Harsco prior to the Closing Date (who will transfer prior to or at Closing from any Seller or a Non-Company Affiliate to a Buyer or any Target Entity) and, consequently, Buyer shall hold harmless all Sellers and/or all Non-Company Affiliates against any Liabilities solely to the extent arising out of any claim by any Buyer or Buyer Affiliate or Target Entity against any Seller or Non-Company Affiliate in respect of the foregoing.

In the event that the employment of any individual other than a Business Employee is found or alleged to have transferred to Buyer or a Target Entity by operation of Law pursuant to the transactions contemplated hereby, then Buyer or the relevant Target Entity may terminate the employment of such individual and Harsco shall indemnify the relevant Buyer or Target Entity against all Liabilities arising from the employment of such individual, or its termination, <u>provided</u> that Harsco shall not bear or be responsible for or indemnify any Buyer or Target Entity if: (i) the relevant Buyer or Target Entity does not first allow Harsco or (at the option of Harsco) a Non-Company Affiliate a period of twenty-one (21) days to offer employment to any such individual before terminating the employment of such individual; and/or (<u>ii</u>) the relevant Buyer or Target Entity does not terminate the employment of any such individual within thirty (30) days following the expiration of the twenty-one (21) day period described in clause (i) above.

(b) Except as otherwise set forth on <u>Schedule 4.6(b)</u> or where any failure to do so arises as a result of any failure by any Buyer or Buyer Affiliate to provide any information it is required by applicable Laws to provide, prior to the Closing, the relevant Seller or Target Entity or Non-Company Affiliate will notify, or, where applicable, will consult with (and Harsco will cause them to do so), all employees, employee representatives, work councils, unions, labor boards and relevant government agencies where such notification or consultation is required by applicable Law or any relevant agreement, regarding the transactions contemplated hereby, and will use its reasonable best efforts to complete such consultation as soon as reasonably possible. The relevant Seller or Target Entity or Non-Company Affiliate will use reasonable endeavors to keep CD&R Investor reasonably informed about all material steps in the notification and/or consultation processes in the jurisdictions concerned. Reasonably in advance of any meeting between Harsco or its representatives and the Works Council at the level of the Netherlands (the "<u>Works Council</u>"), Harsco and CD&R Investor shall consult in good faith as to whether it is appropriate for a representative of CD&R Investor to attend such meeting, which Harsco shall decide in its reasonable discretion. All notifications and consultations pursuant to this <u>Section 4.6(b)</u> shall be made in accordance with the requirements of applicable Laws. To the extent required by, and in accordance with, applicable Law, and as otherwise set forth on <u>Schedule 4.6(b)</u>, Harsco will observe, and

will cause the relevant Seller or Target Entity or Non-Company Affiliate to observe, the rules that apply to the information and consultation of any works council (including, without limitation in France and the Netherlands) in relation to the transactions contemplated by this Agreement. In connection with these consultations and notifications, Harsco shall not and shall cause the relevant Sellers, Target Entities and Non-Company Affiliates not to, make or accept any commitments, obligations or undertakings which could materially affect or prejudice the financial, legal or other position of Buyer, CD&R Investor or the Target Entities, without the prior written consent of CD&R Investor.

(c) (1) Effective immediately after the Closing and subject to and without prejudice to the requirements of applicable Laws including the provisions of TUPE and/or TUPE Equivalent Regulations, the relevant Buyer or Target Entity shall continue the employment of all Persons who are Business Employees as of the Closing; provided that Harsco shall bear and be responsible for, and indemnify CD&R Investor, Buyer and the Target Entities and each of their respective Affiliates from and against, all Liabilities to any Business Employee who, as of the Closing, is on long-term disability leave or other long-term leave and who is not on the payroll of Harsco or one of its Affiliates (an "Inactive Employee"), to the extent such Liability arises on or after the Closing Date, but prior to the date on which such Inactive Employee presents himself or herself to Buyer or any Target Entity for reemployment and has been cleared for a return to active employment (including with any reasonable accommodation) with Buyer or Target Entities (the "Return Date"), including without limitation Liabilities (a) arising out of or related to (i) the loss of eligibility for or entitlement to disability payments or continued coverage under the long-term disability plans of Harsco or its Affiliates (other than Buyer and Target Entities) for such Inactive Employee or related medical or life insurance coverage and (ii) the loss of long-term disability benefits or coverage or related medical or life insurance coverage of any Inactive Employee currently on long-term disability, in each case, to the extent caused by or resulting from the consummation of the transactions contemplated in this Agreement or any action taken by Harsco or its Affiliates following the Closing Date (any event described in either clause (i) or (ii), a "Loss of Coverage"), (b) for severance or termination payment or benefits for such Inactive Employee required under the Business Benefit Plans or applicable Laws borne by Buyer or the Target Entities following the Closing Date, or (c) any compensation payable to such Inactive Employees by Buyer or the Target Entities. For the avoidance of doubt, (A) in the event of a Loss of Coverage, Harsco shall pay to Buyer or the Target Entities the payments that the affected employee (or his dependents) would have been entitled to receive under the arrangement with respect to which the Loss of Coverage occurred and (B) Harsco shall bear no responsibility for any costs or Liabilities with respect to an Inactive Employee arising on or after the Return Date. Notwithstanding any provision of this <u>Section 4.6(c)(1)</u>, to the extent that any applicable non-U.S. Law mandates a different treatment than as provided in this Section 4.6(c)(1), the requirements of such Law shall supersede this Section 4.6(c)(1).

(2) As of the Closing Date and until December 31, 2014, Buyer or its Affiliates (including any Target Entities) shall provide the Transferred Employees (other than any Transferred Employees whose employment is demoted or terminated for any reason following the Closing) with compensation (including short-term target incentive bonus opportunities) and other employee benefits (but excluding equity-based incentives, retention bonuses and other change in control entitlements and retirement compensation and benefits, other than under a defined contribution plan) that are substantially comparable in the aggregate to, in Buyer's discretion, (i) the compensation and benefits (subject to the same exclusions) provided to each Transferred Employee immediately prior to the Closing Date, (ii) the compensation and benefits (subject to the same exclusions) that Buyer or its Affiliate is then providing to similarly-situated non-union-represented employees of Brand Energy, Inc. and its Subsidiaries or (iii) some combination of the foregoing; provided that for any Transferred Employee who is demoted at or following the Closing, the level of compensation and benefits (subject to the exclusions described above

to be provided to such Transferred Employee pursuant to this <u>Section 4.6(c)(2)</u> shall be determined by the position of such Transferred Employee after giving effect to such demotion, and <u>provided</u>, <u>further</u>, that changes to the annual cash bonus plan(s) of Buyer and the Target Entities that are approved by the board of directors of the Company, including the approval of at least one "Harsco Director" (as defined in the Investor Rights Agreement), shall be disregarded for purposes of the application of this <u>Section 4.6(c)(2)</u>.

(3) In the event that prior to the Closing Date, the Parties agree in writing to an employee leasing arrangement from Harsco to Buyer and its Subsidiaries with respect to some or all of the Business Employees in the U.S. ("<u>Leased Employees</u>") for a specified period following the Closing Date (the "<u>Leasing Period</u>"), then the employment of the Leased Employees will not transfer to Buyer until the conclusion of the Leasing Period (or as otherwise agreed by the Parties), subject, in all respects, to terms and conditions to be mutually agreed by the Parties.

(d) Buyer agrees that it will cause, and Harsco shall cooperate with Buyer to cause, the Target Entities or the Buyer's Affiliates to assume, effective as of the Closing Date, the collective bargaining, works council and similar agreements that, as of immediately prior to the Closing, are applicable to, pertain to or cover the Business Employees and are listed on <u>Schedule 4.6(d)</u> (each, a "Labor Agreement," and collectively, the "Labor Agreements"), and Buyer further agrees that it will comply and will cause the Target Entities to comply, in all material respects, with all of the terms and conditions set forth in each such Labor Agreement through the duration of each such agreement. For the avoidance of doubt, the compensation and benefits of Business Employees covered by each Labor Agreement following the Closing will be those set forth in the relevant Labor Agreement.

(e) To the extent not paid prior to the Closing Date, Transferred Employees who are employed by Harsco and its Affiliates on the Closing Date shall be vested (i.e., no further employment requirement) in respect of any portion of the annual bonuses attributable to the portion of the 2013 calendar year prior to closing, determined on a pro rata basis based on actual results. Any such bonuses (together with the bonus earned for the remainder of the calendar year) shall be payable to Transferred Employees by the Buyer or a Target Entity in the ordinary course of business (but in no event later than the later of (x) two and a half months after the end of the applicable fiscal year, and (y) subject to compliance with Section 409A of the Code, thirty (30) days after the Closing). All such pro rata portions of the 2013 bonuses (and associated employment taxes) will be accrued as a liability in Closing Net Working Capital to the extent not paid prior to the Closing. Notwithstanding any provision of this <u>Section 4.6(e)</u>, to the extent that any applicable non-U.S. Law mandates a different treatment than as provided herein, the requirements of such Law shall supersede this <u>Section 4.6(e)</u>.

(f) Buyer and the Target Entities shall (i) give each Transferred Employee credit under each employee benefit plan and personnel policy that covers such Transferred Employee after the Closing Date (including any vacation, sick leave and severance policies) for purposes of eligibility, vesting and entitlement to vacation, sick leave and severance benefits for such Transferred Employee's service with Harsco and its Affiliates prior to the Closing Date, but in each case not for purposes of benefit accrual under any defined benefit plan (except to the extent required by applicable Law or a Target Entity Benefit Plan as in effect on the date hereof), (ii) allow such Transferred Employee to participate in each plan providing welfare benefits (including, but not limited to, medical, dental, vision, life and disability insurance) without regard to preexisting-condition limitations, waiting periods, evidence of insurability or other exclusions or limitations not imposed on such Transferred Employee by the corresponding Business Benefit Plans immediately prior to the Closing for purposes of determining deductibles, co-pays and other applicable limits under any similar replacement plans, except in each case of clauses (i) - (iii) above, (<u>A</u>) where such crediting would result in duplicate benefits with

respect to the same period of service and only to the same extent such service was credited under the applicable Business Benefit Plan immediately prior to the Closing and (\underline{B}) to the extent such credit or participation is not permitted by the terms and conditions of the employee benefit or welfare plan that covers such Transferred Employee following the Closing.

(g) Buyer and the Target Entities shall credit each Transferred Employee all vacation and personal holiday pay that such Transferred Employee is entitled to use but has not used as of the Closing Date (including any earned vacation or personal holiday pay to be used in future years), and Buyer and each Target Entity shall assume all Liability for the payment of such amounts. Prior to or on the Closing Date, Harsco will pay, or cause to be paid, all accrued vacation and personal holiday pay of Business Employees (including employees covered by a collective bargaining agreement) employed in the State of California or any other jurisdiction where applicable Law requires such payment in connection with the Closing, other than any such Business Employees who have requested in writing prior to the Closing that their accrued vacation pay balances transfer to Buyer or a Target Entity.

(h) Buyer and the Target Entities shall (<u>i</u>) on and following the Closing, for the avoidance of doubt, be solely responsible for any severance-type benefits owed to any individual who was a Business Employee employed by the Target Entities prior to the Closing Date and whose employment terminated prior to the Closing Date and (<u>ii</u>) from the Closing Date until December 31, 2014, provide each Transferred Employee with a severance policy comparable in the aggregate to either, in Buyer's discretion, (<u>x</u>) the severance policy provided by Harsco and its Affiliates with respect to such Transferred Employee immediately prior to the Closing Date and set forth in <u>Schedule 4.6(h)</u> or (<u>y</u>) the severance policy provided by Brand Energy, Inc. and its Subsidiaries to their similarly-situated non-union-represented employees. All such severance-type benefits described in clause (<u>i</u>) of this <u>Section 4.6(h)</u> and associated employment taxes will be accrued as a liability in Closing Net Working Capital to the extent not paid prior to the Closing.

(i) After the Closing, Buyer and the Target Entities shall permit Harsco (and its Affiliates) to have access to Business Employees that Harsco or its Affiliates may reasonably need in order to defend or prosecute any legal or administrative action to which Harsco or any of its Affiliates is a party and which relates to the conduct of the Target Entities prior to the Closing (except to the extent CD&R Investor, Buyer or the Target Entities or their respective Affiliates are an opposing party in such action), provided that such access will not interfere with such employee's work obligations and provided, further, that Harsco shall pay for any travel or other expenses incurred by such employee in connection with such access and reimburse CD&R Investor for the loss of such employee's services for the period of such access at a rate to be mutually agreed. After the Closing, Buyer will, and will cause the Target Entities to, cooperate with Harsco (and its Affiliates) in providing access to relevant data and employment records of Business Employees reasonably necessary to administer the benefits of Business Employees under any Business Benefit Plan, subject to any limitations imposed under applicable Law.

(j) After the Closing, Harsco and its Affiliates will permit CD&R Investor, Buyer and the Target Entities (and their respective Affiliates) to have access to employees of Harsco and its Affiliates that CD&R Investor, Buyer and the Target Entities (or their respective Affiliates) may reasonably need in order to defend or prosecute any legal or administrative action to which CD&R Investor, Buyer and the Target Entities (or their respective Affiliates) is a party and which relates to the conduct of the Target Entities prior to the Closing (except to the extent Harsco or its Affiliates (other than the Target Entities) are an opposing party in such action), <u>provided</u> that such access will not interfere with such employee's work obligations and <u>provided</u>, <u>further</u>, that CD&R Investor, Buyer and the Target Entities, as the case may be, shall pay for any travel or other expenses incurred by such employee in connection with such access and reimburse Harsco for the loss of such employee's services for the period of such access at a

rate to be mutually agreed. After the Closing, Harsco (and its Affiliates) will cooperate with CD&R Investor, Buyer and the Target Entities (and their respective Affiliates) in providing access to relevant data and employment records of Transferred Employees reasonably necessary to administer the benefits of Transferred Employees under any Business Benefit Plan or other employee benefit plan, agreement or arrangement, subject to any limitations imposed under applicable Law.

(k) As of the Closing Date, or other such date as agreed upon by CD&R Investor and Harsco in writing, the active participation of each Transferred Employee in the Harsco Benefit Plans set forth on <u>Schedule 4.6(k)</u> (the "<u>Harsco Defined</u> <u>Contribution Plans</u>") shall cease. Prior to the Closing Date, Harsco will fully fund, or cause to be fully funded, through the Closing Date, all employee contributions and employer matching contributions required under the Harsco Defined Contribution Plans with respect to Transferred Employees. As soon as reasonably practical following the Closing Date, or other such date as agreed upon by CD&R Investor and Harsco in writing, Buyer or one of its Affiliates shall designate or establish a tax-qualified defined contribution retirement plan(s) (the "<u>Buyer 401(k) Plan</u>") and, to the extent permitted under applicable Law, Buyer shall take reasonable steps to cause the trustee of such plan(s), if requested by the Transferred Employee, to accept a direct "rollover" in cash of all or a portion of such Transferred Employee's distribution in accordance with the terms and conditions of such applicable Harsco Defined Contribution Plan, Buyer shall take all reasonable actions, to the extent permitted by applicable Law and the terms of the Buyer 401(k) Plan, to permit the Transferred Employee to rollover such outstanding loan balance to the Buyer 401(k) Plan; provided, however, that any such outstanding loans shall be transferred only if the relevant Transferred Employee elects to rollover all of his or her entire account balance under the Harsco Defined Contribution Plan to the Buyer 401(k) Plan.

(l) [Intentionally Omitted.]

(m) Unless the Parties mutually agree in writing on or before the date that is thirty (30) days before the end of the current plan year of any Multiemployer Plan, the parties intend that Section 4204 of ERISA should apply to this transaction with respect to each such Multiemployer Plan, and the following provisions (i) through (vi) shall be applicable with respect to each such Multiemployer Plan; <u>provided</u>, <u>however</u>, the Parties agree that no Section 4204 of ERISA election shall be made with respect to the Central States, Southeast and Southwest Areas Pension Fund ("<u>Central States</u>") and Harsco or its Affiliates shall be solely responsible for withdrawal liability imposed in connection with such withdrawal. Notwithstanding any other provisions of this Agreement, Buyer shall not be obligated to participate in, or to continue participation in, Central States with respect to Business Employees.

(i) Buyer agrees to make contributions after the Closing Date to such Multiemployer Plan with respect to the operations of the Business, for substantially the same number of contribution base units for which Harsco has an obligation immediately prior to the Closing to contribute to such Multiemployer Plan, as described in Section 4204(a)(1)(A) of ERISA, and Buyer agrees to assume the applicable collective bargaining agreements related to such Multiemployer Plan;

(ii) Harsco and Buyer shall take all necessary steps, including complying with the provisions of Section 4204 of ERISA, to prevent the transactions contemplated by this Agreement from being treated as a full or partial withdrawal from such Multiemployer Plan;

(iii) Prior to the first plan year of such Multiemployer Plan beginning after the Closing Date, Harsco and Buyer jointly shall seek a variance from the requirement of

Section 4204(a)(1)(B) of ERISA that a bond be obtained or an amount be held in escrow as provided in said Section. In the event such Multiemployer Plan determines that the request does not qualify for a variance, Buyer shall obtain any required bond or establish any required escrow as of the first plan year beginning after the Closing Date and shall maintain such bond or escrow until the earliest of (<u>A</u>) the date a variance is obtained from such Multiemployer Plan; (<u>B</u>) the date a variance or exemption is obtained from the Pension Benefit Guaranty Corporation; or (<u>C</u>) the last day of the fifth plan year commencing after the Closing Date. Harsco shall reimburse and indemnify Buyer for 50% of the cost of such application for variance and 50% of the cost of any bond or escrow that may be required hereunder with respect to any Multiemployer Plan for which Harsco has requested application of Section 4204 pursuant to the first sentence of this <u>Section 4.6(m</u>);

(iv) If, at any time during the first five plan years of such Multiemployer Plan beginning after the Closing Date, Buyer withdraws from such Multiemployer Plan or fails to make a contribution when due to such Multiemployer Plan, the amount owed to such Multiemployer Plan (or, if less, the total amount payable through the bond, escrow or letter of credit) shall be paid to that Multiemployer Plan through the bond, escrow or letter of credit described in subsection (iii) above;

(v) If, during the first five plan years of such Multiemployer Plan beginning after the Closing Date, Buyer withdraws from such Multiemployer Plan in a complete or partial withdrawal with respect to operations of the Business, Harsco shall be secondarily liable for any withdrawal liability it would have had to such Multiemployer Plan with respect to the Business but for this Agreement, if and to the extent that the liability of Buyer with respect to such Multiemployer Plan is not paid either by Buyer or pursuant to the bond, escrow or letter of credit described in subsection (iii) above; and

(vi) In the event Buyer withdraws from a Multiemployer Plan for which Harsco has requested application of Section 4204 pursuant to the first sentence of this <u>Section 4.6(m)</u> in a complete or partial withdrawal during the plan year in which the Closing occurs or during the next four plan years (which period shall be extended if and to the extent that Harsco's contribution history continues to affect Buyer's withdrawal liability under the allocation method adopted by such Multiemployer Plan), Harsco shall reimburse Buyer for the portion of Buyer's withdrawal liability attributable to Harsco's contribution history assumed by Buyer as a result of the application of Section 4204 of ERISA to the transaction contemplated by this Agreement; <u>provided</u>, <u>however</u>, that such reimbursement shall not apply to any withdrawal resulting from an action or determination by Buyer or one of its Affiliates that has not been approved by the board of directors of the Company, including the approval of at least one "Harsco Director" (as defined in the Investor Rights Agreement). For the avoidance of doubt, the Parties agree that the refusal to approve such action or determination by a "Harsco Director" shall be deemed not to be a breach of fiduciary duties by such director, regardless of the circumstances.

As soon as practicable after the date of this Agreement, Buyer and Harsco shall cooperate to determine whether the transactions contemplated by this Agreement would otherwise result in the imposition of full or partial withdrawal liability with respect to any Multiemployer Plan, and to fulfill any requirements such Multiemployer Plan may set in order to ensure that the transactions contemplated by this Agreement comply with Section 4204 of ERISA if the transactions otherwise would result in the imposition of withdrawal liability.

(n) Harsco and the Non-Company Affiliates shall remain responsible for all Liabilities under each Harsco Benefit Plan whether arising prior to, on or after the Closing Date, except to the extent such Liabilities are expressly assumed by Buyer or Target Entities under this Agreement (including, for avoidance of doubt, Liabilities relating to current and former employees of the Business), and Harsco shall indemnify CD&R Investor, Buver and the Target Entities and each of their respective Affiliates (other than Harsco and the Non-Company Affiliates) against all and any liabilities, obligations, costs, claims or payments which CD&R Investor, Buyer and the Target Entities or any of their respective Affiliates may incur or make in relation to any such Harsco Benefit Plan, and any liabilities arising solely out of the status of any of the Target Entities as an ERISA Affiliate of Harsco or any of its Affiliates (other than Buyer and the Target Entities) ("ERISA Affiliate Liabilities"). Without limiting the foregoing, if it is established that Rule 4.3 of Section G of the rules relating to the Harsco Pension Scheme applies with respect to any Business Employees following the Closing, Harsco and the Seller of the UK Business shall procure that the "principal employer" of the Harsco Pension Scheme shall not give its consent to the granting of any enhanced early retirement benefits (payable on redundancy or otherwise) under Rule 4.3 thereof, and neither Buyer nor the Target Entities and neither Harsco nor any Non-Company Affiliate shall communicate with or make any statement to the Business Employees in respect of early retirement pension rights under the Harsco Pension Scheme without the prior written consent of Harsco and the Seller of the UK Business, and neither Buyer nor the Target Entities shall encourage any Business Employee to request that any party consent or otherwise agree to an enhanced early retirement pension under or in respect of benefits accrued pre-Closing in the Harsco Pension Scheme. For the avoidance of doubt, the Target Entities shall retain all responsibility and Liabilities under each Target Entity Benefit Plan whether arising prior to, on or after the Closing Date, except to the extent retained by Harsco or its Affiliates.

(o) This <u>Section 4.6</u> shall survive the Closing and shall be binding on all successors and assigns of Harsco, CD&R Investor, Buyer and the Target Entities. Nothing set forth in this <u>Section 4.6</u> shall confer any rights or remedies upon any employee or former employee of the Target Entities, any Business Employee or upon any other Person other than the Parties hereto and their respective successors and assigns or shall constitute an amendment to any Business Benefit Plan or any other plan or arrangement covering the Business Employees or employees of Harsco. Nothing in this <u>Section 4.6</u> shall obligate CD&R Investor, Buyer and the Target Entities to continue the employment of any Transferred Employee following the Closing for any specific period or in any specific position, title or level of employment.

4.7 WARN Act

(a) Buyer shall not, and shall cause the Target Entities not to, at any time prior to ninety (90) days after the Closing Date, effectuate a "Plant Closing" or "Mass Layoff", as those terms are defined in the WARN Act, affecting in whole or in part any site of employment, facility, operating unit or employee with respect to the Business, and regardless of whether the employment losses occur before or after the Closing Date, without complying with the notice requirements and all other provisions of the WARN Act. Harsco agrees that between the date hereof and the Closing Date, it will cause the Target Entities and the Business not to effect or permit a "Plant Closing" or "Mass Layoff" as these terms are defined in the WARN Act without notifying CD&R Investor in advance and without complying with the notice requirements and all other provisions of the WARN Act.

(b) A breach by Buyer, the Target Entities or Harsco of their respective obligations under this <u>Section 4.7</u> shall give rise to an obligation by the breaching party to indemnify, defend and hold harmless the non-breaching Parties from and against any and all damages incurred thereby or caused thereto under or pursuant to the WARN Act based on, arising out of, resulting from or relating to any act

or omission to act by or of the breaching party with regard to any single site of employment, facility, operating unit or employee of the breaching party.

4.8 Intercompany Arrangements; Transactions

(a) Harsco shall cause (<u>i</u>) each Intercompany Balance, other than intercompany trade payables and receivables arising in the ordinary course of the Business, to be eliminated or transferred to a Non-Company Affiliate prior to the Closing, and (<u>ii</u>) each Intercompany Contract, other than this Agreement, the Ancillary Agreements and those agreements and arrangements set forth in <u>Exhibit F</u>, to be terminated prior to or at the Closing without any obligation under any Intercompany Contract surviving such termination. Harsco shall use commercially reasonable efforts to distribute any Cash out of the Target Entities prior to Closing, <u>provided</u> that Harsco may cause the Target Entities to engage in one or more transactions designed to reduce repatriation tax costs so long as such transactions (including the repayment or creation of any debt) will not (directly or indirectly) increase the repatriation tax costs out of the Target Entities to Buyer or its Affiliates for any post-Closing period (as compared to what such costs would have been if such transactions had not been undertaken and such cash had been distributed prior to the Closing).

(b) Prior to or substantially concurrently with the Closing, Harsco shall (and shall cause its Affiliates to) complete the Pre-Closing Transactions and the Parties shall (and shall cause their Affiliates to) complete the Closing Transactions, in each case in the sequence set forth in <u>Exhibit F</u> or <u>Exhibit G</u>, as applicable; <u>provided</u> that (<u>i</u>) Harsco may modify or supplement the Pre-Closing Transactions in a manner that does not materially and adversely affect Buyer and its Subsidiaries at or following the Closing (but Harsco shall consult with CD&R Investor about any such modification or supplement) and (<u>ii</u>) the Parties may modify the Closing Transactions with, and only with, the prior written consent of the other Parties, which consent shall not be unreasonably withheld, conditioned or delayed.

(c) Unless otherwise expressly set forth in <u>ARTICLES II</u>, <u>IV</u> or <u>V</u>, (i) the satisfaction or termination of any Intercompany Balance, (ii) the termination of any Intercompany Contract or (iii) the implementation or completion of the Transactions pursuant to this <u>Section 4.8</u> shall not be deemed a breach or violation of any provision of this Agreement.

4.9 Insurance

(a) The Target Entities will use commercially reasonable efforts to continue to carry their existing insurance up to the Closing, including renewing any policies expiring during such period, and shall not cause any breach, default or cancellation (other than expiration and replacement of policies in the ordinary course of business consistent with past practices) of such insurance policies or agreements that have had or would reasonably be expected to have a material and adverse effect on the Business or the Target Entities, taken as a whole.

(b) From and after the Closing Date, the Target Entities, Target Assets, Assumed Liabilities, Business Employees and former employees of the Target Entities shall cease to be insured by, have access or availability to, be entitled to make claims on, be entitled to claim benefits from or seek coverage under any of the Sellers' or their respective Subsidiaries' and/or Affiliates' insurance policies or any of their self-insured programs (for purposes of this <u>Section 4.9</u>, "<u>Sellers</u>" shall include, where appropriate to the context, their respective Subsidiaries and/or Affiliates), other than with respect to any claim, act, omission, event, circumstance, occurrence or loss that both (i) occurred or existed on or prior to the Closing Date (and then only to the extent that such claim, act, omission, event, circumstance, occurrence or loss occurred or existed on or prior to the Closing Date) and (<u>ii</u>) was properly reported in

accordance with the then-applicable reporting requirements of the applicable insurance policy, as the same may be modified from time to time in accordance with <u>Section 4.9(d)</u> below (a "<u>Triggering Event</u>").

(c) Notwithstanding subparagraph (b) above, with respect to any Triggering Event relating to the Target Entities, the Target Assets, Assumed Liabilities, Business Employees or former employees of the Target Entities that would be covered by any of the Sellers' occurrence-based (but not claims-made or occurrence-reported) insurance policies (the "<u>Available Insurance</u> <u>Policies</u>"), the Buyers, on behalf of the Target Entities, may access, make and settle claims on, claim benefits from or seek coverage under such policies following the Closing Date subject to the terms and conditions of such policies and programs and this Agreement and in each case <u>provided</u> that:

(i) the Buyers shall, or shall cause the Target Entities to, as promptly as reasonably practicable notify the Sellers' Corporate Insurance department of all such claims and/or efforts to seek benefits or coverage and shall cooperate with the Sellers in pursuing all such claims, <u>provided</u> that the Buyers and the Target Entities shall be solely responsible for complying with all policy conditions for such claims. The Buyers shall, or shall cause the Target Entities to, keep Harsco reasonably informed of any settlements or proposed settlements of \$1,000,000 or more under or in connection with any Available Insurance Policies. In addition, the Buyers shall cause the Target Entities to pursue or cause to be pursued any rights of recovery against third parties with respect to Triggering Events and shall cooperate with the Sellers with respect to pursuit of such rights. To the extent permitted by the applicable policies, the order of priority of any such recoveries shall inure first to the Sellers to reimburse any and all costs incurred by the Sellers directly or indirectly as a result of such claims or losses.

(ii) the Sellers shall have the right but not the duty to monitor and/or associate with coverage claims or requests for benefits asserted by the Buyers or the Target Entities under any Available Insurance Policies (collectively, "<u>Coverage Claims</u>"), including the coverage positions and arguments asserted therein;

(iii) the Buyers and the Target Entities (A) shall notify the Sellers as far in advance as reasonably practicable of asserting any Coverage Claims, (B) shall be liable for any out-of-pocket fees, costs and expenses incurred by or on behalf of any Seller and any Affiliates thereof (as evidenced by reasonable documentation thereof) relating to any Coverage Claim, and (C) shall not assign in whole or in part any Available Insurance Policies or any rights, interests or claims thereunder; and

(iv) the Buyers and the Target Entities shall exclusively bear and be liable (and the Sellers shall have no obligation to repay or reimburse the Target Entities or the Buyers) for all deductibles and retentions and uninsured, uncovered, unavailable or uncollectible amounts relating to or associated with any Coverage Claims, whether made by or on behalf of the Buyers, the Target Entities or their employees or third parties.

(d) Notwithstanding anything contained herein, the Sellers shall retain exclusive right to control all of their insurance policies and programs, including the Available Insurance Policies, and the benefits and amounts payable thereunder, including the right to exhaust, settle, release, commute, buy-back or otherwise resolve disputes with respect to any of its insurance policies and programs and to amend, modify, terminate or waive any such insurance policies and programs or any rights thereunder, notwithstanding whether any such policies or programs apply to any Liabilities and/or claims the Buyers or the Target Entities has made or could make in the future, including coverage claims with respect to

Triggering Event(s); <u>provided</u> that this paragraph shall not limit the rights of Buyer set forth in <u>Section 4.9(c)</u> above, and the Sellers shall not amend, modify, terminate or waive any coverage under any Available Insurance Policies with respect to periods prior to the Closing Date in a manner that would impair coverage thereunder available for Coverage Claims; <u>provided</u>, <u>further</u>, that the Buyers and the Target Entities shall cooperate with the Sellers and share such information as is reasonably necessary in order to permit the Sellers to manage and conduct their insurance matters as the Sellers deem appropriate and that the Buyers and the Target Entities hereby give consent for the Sellers to inform any affected insurer of this Agreement and to provide such insurer with a copy hereof.

(e) Nothing in this Agreement is intended to waive or abrogate in any way the Sellers' own rights to insurance coverage for any liability, whether relating to the Target Entities or otherwise.

4.10 Financing

(a) Subject to the terms and conditions of this Agreement (including Harsco's compliance with <u>Section 4.11</u>), CD&R Investor shall use its reasonable best efforts to take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable to arrange for Bullseye MergerSub to obtain the Debt Financing at or prior to the Closing subject to the terms and conditions of, and on terms no less favorable than those described in, the Debt Financing Commitments (including any "flex" provisions) and the Fee Letter, including using its reasonable best efforts to (i) except as otherwise permitted by this Section 4.10, maintain in effect the Debt Financing Commitments in accordance with the terms and subject to the conditions thereof, (ii) assist in the satisfaction on a timely basis (giving effect to the anticipated timing of the Marketing Period) of all conditions applicable to Bullseye MergerSub in obtaining the Debt Financing at the Closing set forth therein (including consummating the Equity Financing on the terms and subject to the conditions set forth in the Equity Financing Commitment at or prior to Closing), (iii) negotiate definitive agreements (the "Definitive Agreements") with respect to the Debt Financing on the terms and conditions (including any "flex" provisions) contemplated by the Debt Financing Commitments and the Fee Letter (provided that, upon request by Harsco, CD&R Investor shall provide copies thereof to Harsco on a current basis and otherwise keep Harsco reasonably informed on a current basis of the status of its efforts to arrange the Debt Financing) and (iv) consummate the Debt Financing Commitments at or prior to the Closing Date. CD&R Investor shall give Harsco prompt written notice of (A) any material breach by any party to any Financing Commitment of which it or its Affiliates become aware or (B) the receipt, on or prior to the Closing Date, of any written notice or other written communication from any financing source with respect to any breach, default, event of default, termination or repudiation by any party to any Financing Commitment or pursuant to which the investors or lenders thereunder have indicated that they will not perform their obligations to make the investments or loans and other extensions of credit thereunder at or prior to the Closing. Neither CD&R Investor nor any of its Affiliates shall agree with Guarantor to, enter into any amendment, supplement or other modification of, or waive any of its rights under, the Equity Financing Commitment. Neither CD&R Investor nor any of its Affiliates shall be permitted to amend, amend and restate, assign, modify, supplement, substitute or replace any of the Debt Financing Commitments without the written consent of Harsco; provided that CD&R Investor or its Affiliates may, without the consent of Harsco, (x) amend, amend and restate, modify supplement or replace the Debt Financing Commitments to add or replace lenders, lead arrangers, bookrunners, syndication agents or similar entities or (y) otherwise amend, supplement, replace, or consent to any waiver of any provision or remedy under, the Debt Financing Commitments, so long as such amendment, amendment and restatement, restatement, assignment, modification, supplement, replacement or waiver does not (A) reduce the aggregate principal amount of the Financing, or (B) impose new or additional conditions, or otherwise expand, amend or modify any of the conditions, to the receipt of Debt Financing in a manner

that would reasonably be expected to, (I) hinder, delay or prevent the Closing, (II) make the funding of the Debt Financing (or satisfaction of the conditions to obtaining the Debt Financing) less likely to occur or (III) extend or permit the extension of the Marketing Period. For the avoidance of doubt, nothing contained herein shall prevent CD&R Investor from reallocating the Debt Financing among the Term Loan Facility (as defined in the Debt Financing Commitment) and the Bridge Facility (as defined in the Debt Financing Commitment), in each case in accordance with the terms of the Debt Financing Commitment or reducing the total amount of funds available under the Debt Financing, provided that, in either case, the representation and warranties set forth in the last sentence of Section 3.6 remain true and correct. CD&R Investor shall obtain the Equity Financing contemplated by the Equity Financing Commitment upon satisfaction or waiver of the conditions to closing in Section 6.2 (other than those conditions that by their nature are to be satisfied by actions taken at the Closing and subject to and in accordance with the terms of the Equity Financing Commitment). Subject to the terms and conditions of this Agreement (including Section 4.11), in the event any portion of the Debt Financing becomes unavailable on the terms and conditions contemplated in the Debt Financing Commitments, CD&R Investor shall promptly notify Harsco and shall use its reasonable best efforts to arrange for alternative financing (an "<u>Alternative</u> Financing") from alternative sources in an amount such that the aggregate funds that would be available to Buyer at the Closing will be sufficient to perform its obligations hereunder; provided that CD&R Investor shall not be required to accept any Alternative Financing that is materially less favorable to it than the Debt Financing. CD&R Investor shall promptly (upon reasonable request by Harsco) deliver to Harsco true and complete copies of all drafts of any Alternative Financing commitments and all final agreements pursuant to which any such alternative source shall have committed to provide CD&R Investor with any portion of the Debt Financing. In furtherance and not in limitation of the foregoing, in the event that (i) all or any portion of the Debt Financing Commitment structured as high yield financing has not been consummated, (ii) all closing conditions contained in Section 6.2 shall have been satisfied or waived (other than those conditions that by their nature are satisfied at the Closing) and (iii) the bridge facilities described in or contemplated by the Debt Financing Commitments (or Alternative Financing obtained in accordance with this Section 4.10) are available on terms and conditions described in or contemplated by the Debt Financing Commitments (or Alternative Financing obtained in accordance with this Section 4.10), then CD&R Investor shall consummate, or cause to be consummated, and shall use, or cause to be used, the proceeds of such bridge financing (or such Alternative Financing) to replace such high yield financing no later than the last day of the Marketing Period. For purposes of this Section 4.10, Section 3.6 and Section 4.11, references to "Debt Financing" shall include the financing contemplated by the Debt Financing Commitments as permitted by this Section 4.10 to be amended, modified or replaced by Alternative Financing and any financing in lieu of the Debt Financing Commitments, and references to "Debt Financing Commitments" and "Fee Letter" shall include such documents as permitted by this <u>Section 4.10</u> to be amended, modified or replaced, in each case from and after such amendment, modification or replacement and any analogous documents contemplated by the immediately preceding sentence.

4.11 Debt Financing Cooperation

(a) Prior to the Closing, Harsco shall, and shall cause its Affiliates (including the Target Entities) to, and shall use its reasonable best efforts to cause its and their representatives to, provide to CD&R Investor such cooperation reasonably requested by CD&R Investor to assist CD&R Investor in causing the conditions in the Debt Financing Commitment to be satisfied and such cooperation as is otherwise necessary or reasonably requested by CD&R Investor or the Financing Sources in connection with the Buyers obtaining the Debt Financing in accordance with its terms, including cooperation that consists of:

(1) participating in a customary and reasonable number of meetings, presentations, road shows, due diligence sessions, drafting sessions and sessions with rating agencies;

(2) providing authorization letters to Financing Sources authorizing the distribution of information to prospective lenders or investors (including customary 10b-5 and material non-public information representations);

(3) causing the Target Entities to execute and deliver any securities purchase agreement, credit agreement, indentures, notes, guarantees, pledge and security documents, supplemental indentures, currency or interest hedging arrangements, other definitive financing documents, a certificate of the chief financial officer of Harsco or another officer of Harsco or its subsidiaries with respect to the solvency of the Target Entities and Target Assets on a consolidated basis, and other customary certificates or documents and back-up therefor in connection with the legal opinions of Buyer's counsel as may be reasonably requested by CD&R Investor or Financing Sources and otherwise reasonably facilitating the pledging of collateral; <u>provided</u> that none of the Target Entities or any of their respective officers or employees shall be required to or execute any document in connection with this <u>Section 4.11(a)(3)</u>, which document would be effective at any time before the time that will be immediately prior to the Closing;

(4) (A) furnishing CD&R Investor and its Financing Sources as promptly as practicable with (x) audited combined balance sheets of the Business for the two most recent completed fiscal years ended at least ninety (90) days prior to the Closing Date, statements of operations, comprehensive loss, cash flows and changes of the Business for the two most recently completed fiscal years ended at least ninety (90) days prior to the Closing Date, together with all related notes and schedules thereto, accompanied by the reports thereon of the independent auditors and (y) unaudited combined balance sheets and related combined statements of operations, comprehensive loss, cash flows and changes in equity of the Business prepared in accordance with GAAP for the six months ended June 30, 2013, and for any subsequent interim Financing period (*e.g.*, the nine months ended September 30, 2013) ended at least forty five (45) days prior to the Closing Date and for the comparable quarter of the prior fiscal year, together with all related notes and schedules thereto, in the case of each of clauses (x) and (y), prepared in accordance with GAAP and in compliance with Regulation S-X (other than Rule 3-10 and 3-16 of Regulation S-X and, with respect to any subsequent interim Financing period, other than the fact that such statements will not show quarter-to-quarter comparability (*e.g.*, for the three months ended September 30, 2013)) promulgated under the Securities Act^(B) furnishing CD&R Investor and its Financing Sources as promptly as practicable with all financial statements and pro forma financial information (including for the most recent four fiscal quarter period ended at least forty five (45) days prior to the Closing Date) prepared in accordance with GAAP, financial data, audit reports and other information regarding the Business of the type that would be required by Regulation S-X (including Item 3-05 thereof, but excluding Rule 3-10 and 3-16 of Regulation S-X and, with respect to any subsequent interim Financing period, other than the fact that such statements will not show quarter-to-quarter comparability (e.g., for the three months ended September 30, 2013)) and Regulation S-K (other than Item 402 of Regulation S-K) promulgated under the Securities Act for a registered public offering of non-convertible debt securities of Buyer (including for Buyer's preparation of pro forma financial statements), in each case to the extent the same

is of the type and form customarily included in an offering memorandum for private placements of non-convertible high-yield bonds under Rule 144A promulgated under the Securities Act, or otherwise necessary to receive from the independent accountants of the Business (and any other accountant to the extent financial statements audited or reviewed by such accountants are or would be included in such offering memorandum) customary "comfort" (including "negative assurance" comfort) with respect to the financial information to be included in such offering memorandum and which, with respect to any interim financial statements, shall have been reviewed by the independent accountants of the Business as provided in AU 722, (the authorization letter referred to in clause (2) and all such information in this clause (4), together with any replacements or restatements thereof, and supplements thereto, if any such information would go stale or otherwise be unusable for such purposes the "Harsco Required Information");

(5) cooperating with CD&R Investor and CD&R Investor's efforts to obtain customary and reasonable corporate and facilities ratings, legal opinions, surveys and title insurance (including providing reasonable access to CD&R Investor and its representatives to all owned Real Property and Leased Real Property) as reasonably requested by CD&R Investor;

(6) facilitating the granting and/or pledging of collateral (it being understood that no pledge shall be required to be effective until the Closing occurs);

(7) taking all corporate actions, subject to the occurrence of the Closing, reasonably requested by a Financing Source to permit the consummation of the Debt Financing;

(8) obtaining customary payoff letters, Encumbrance terminations and instruments of discharge to be delivered at Closing to allow for the payoff, discharge and termination in full on the Closing Date of all Indebtedness (other than that comprising the Debt Financing);

(9) furnishing CD&R Investor and its Financing Sources promptly with all documentation and other information which any lender providing or arranging Debt Financing has reasonably requested and that such lender has determined is required by regulatory authorities in connection with such Debt Financing under applicable "know your customer" and anti-money laundering rules and regulations, including the PATRIOT Act;

(10) furnishing CD&R Investor and its Financing Sources promptly with drafts of customary comfort letters that the independent accountants of the Business (and any other accountant to the extent financial statements audited or reviewed by such accountants are or would be included in such offering memorandum) are prepared to deliver upon "pricing" of any high-yield bonds being issued in lieu of all or a portion of the Debt Financing; and

(11) assisting CD&R Investor in the preparation of customary rating agency presentations, lender presentations and high yield road show presentations or memoranda, customary bank offering memoranda, syndication memoranda, private offering memoranda, and other marketing materials or memoranda, including business and financial projections reasonably requested by CD&R Investor, in each case, required in connection

with the Debt Financing (the "<u>Offering Materials</u>"), and consents of accountants for use of their unqualified audit reports in any materials relating to the Debt Financing or any high-yield bonds being issued in lieu of all or a portion of the Debt Financing.

provided that (u) nothing in this Section 4.11(a) shall require such cooperation to the extent it would require Harsco to waive or amend any terms of this Agreement or agree to pay any fees or reimburse any expenses, in each case that is not contingent upon the Closing or for which it has not received prior reimbursement by or on behalf of CD&R Investor (except to the extent CD&R Investor has provided the indemnities set forth in <u>Section 4.11(c)</u>), (v) nothing herein shall require such cooperation from Harsco or its Affiliates to the extent it would unreasonably interfere with the ongoing operations of Harsco or its Affiliates, (w) except to the extent CD&R Investor has provided the indemnities set forth in Section 4.11(c), none of Harsco or any of its Affiliates or any of their respective representatives shall have any liability under any certificate, agreement, arrangement, document or instrument relating to the Debt Financing except upon the Closing, (x) except for the authorization letter set forth in clause (2) above and representation letters required by the Business' auditors in connection with the delivery of "comfort" letters, neither Harsco nor its Affiliates shall be required to commit to take any action that (i) is not contingent upon the Closing, (ii) would be effective prior to the Closing, (iii) would encumber any assets of Harsco or its Affiliates prior to the Closing, or (iv) will conflict with or violate Harsco's or its Affiliates' organizational documents or any laws or result in a violation or breach, default or event of default under, any contract to which Harsco or its Affiliates is a party, (y) results in any officer or director of Harsco or its Affiliates incurring any personal liability with respect to any matters relating to the Financing (except to the extent CD&R Investor has provided the indemnities set forth in <u>Section 4.11(c)</u>) or (<u>z</u>) requires Harsco or its Affiliates to enter into any financing or purchase agreement for the Financing that would be effective prior to the Closing Date.

(b) CD&R Investor shall promptly, upon request by Harsco, reimburse Harsco for all of its and its Affiliates' documented and reasonable out-of-pocket fees, costs and expenses (including reasonable attorneys' fees) incurred by Harsco and its Affiliates in connection with the cooperation of Harsco and its Affiliates contemplated by this <u>Section 4.11</u>.

(c) Harsco, its Affiliates and their respective officers, advisors and representatives shall be indemnified and held harmless by CD&R Investor for and against any and all liabilities, losses, damages, claims, costs, expenses, interest, awards, judgments and penalties suffered or incurred by them in connection with the arrangement of the Debt Financing and/or the provision of information utilized in connection therewith (other than information provided in writing specifically for such use by or on behalf of Harsco or any of its Affiliates) to the fullest extent permitted by applicable Law, other than to the extent any of the foregoing arises as a result of (i) the willful misconduct, gross negligence or material breach of the obligation of any of Harsco, its Affiliates (including Target Entities) or its or their respective directors, officers, employees, attorneys, accountants or other advisors or representatives or (ii) any information provided by or on behalf of Harsco or any of its Affiliates (including the Target Entities) in connection with the Debt Financing to the extent such information is the subject of any of the representations or warranties set forth in <u>ARTICLE II</u> and where such information would constitute a breach of any such representation or warranty.

(d) Harsco shall or shall cause the Target Entities and Non-Company Affiliates' (if applicable) to supplement the Required Information on a reasonably current basis to the extent that any such Required Information, to the Knowledge of Harsco or the Target Entities, when taken as a whole and in light of the circumstances under which such statements were made, contains any material misstatement of fact or omits to state any material fact necessary to make such information not materially misleading.

(e) The Target Entities' logos may be used in connection with the Financing; <u>provided</u> that such logos are used solely in a manner that is not intended to or reasonably likely to harm or disparage the Target Entities or the reputation or goodwill of any Target Entity.

4.12 Intellectual Property Assignments

As soon as practicable after the Closing Date, Harsco shall (and shall cause the Non-Company Affiliates and the Target Entities to), at Harsco's expense, prepare or cause to be prepared and shall file or cause to be filed, and thereafter use commercially reasonable efforts to pursue in the United States Patent and Trademark Office and those other foreign trademark and patent registries or domestic or foreign registries set forth in <u>Schedule 4.12</u>, assignment documents in substance and form reasonably satisfactory to CD&R Investor and Harsco with respect to the registered and applied for Marks, Patents and Domain Names, as applicable, listed on <u>Schedule 4.12</u>, and to record a Target Entity or other Subsidiary of Buyer, as determined by Buyer, as the record and beneficial owner of such registered or applied for Marks, Patents and Domain Names. If any Marks, Patents or Domain Names that are used primarily by the Business as of the Closing but that are not owned by a Target Entity as of the date hereof or as of the Closing (after giving effect to the Transactions) are not included on <u>Schedule 4.12</u>, and are not Marks or Domain Names which incorporate a Harsco Designation (defined in <u>Section 7.5(c)</u> below), <u>Schedule 4.12</u> shall be deemed to include such Marks, Patents or Domain Names. If any other Intellectual Property is owned by Harsco or any Non-Company Affiliate and is used primarily in the Business as of the date hereof or as of the Closing, and does not incorporate a Harsco Designation, Harsco shall, or shall cause such Non-Company Affiliate, as applicable, to assign all such party's right, title and interest in and to such Intellectual Property to a Target Entity or other Subsidiary of Buyer, as determined by Buyer, free and clear of all Encumbrances except Permitted Encumbrances.

4.13 Internal IT Systems and Data Separation; Replication of Shared Contracts

(a) As soon as practicable after the date hereof, Harsco shall (and shall cause its Affiliates to), in consultation with CD&R Investor, separate logically and physically the Internal IT Systems and data of Harsco and the Non-Company Affiliates from the Internal IT Systems and data of the Target Entities at the expense of Buyer with respect to costs paid to third parties by Harsco and its Affiliates in connection with this <u>Section 4.13(a)</u>, in such a manner that the Internal IT Systems and data of the Target Entities are not accessible to Harsco and the Non-Company Affiliates and the Internal IT Systems and data of Harsco and the Non-Company Affiliates are not accessible to the Target Entities, in each case, after such separation date, except as and to the extent otherwise set forth in or required by the Transition Services Agreement and subject to any restrictions or requirements in any agreements concerning the Internal IT Systems, including licenses to Software which may comprise Internal IT Systems.

(b) Upon the request of CD&R Investor, Harsco shall, and shall cause its Affiliates to, at Harsco's expense, except with respect to any amounts paid to counterparties to the Replicated Contracts (which shall be at the expense of Buyer), (i) use commercially reasonable efforts, with the reasonable cooperation of Buyer, to either (\underline{A}) cause the counterparty to any Replicated Contract to enter into a new agreement with Buyer or its designee with respect to matters addressed by such Replicated Contract, on terms substantially similar to such Replicated Contract, including with respect to pricing, or (\underline{B}) if possible, assign to Buyer or its designee the benefits and obligations under such Replicated Contract as they relate to the Business, or (\underline{ii}) until such time as a new agreement or assignment can be arranged pursuant to this <u>Section 4.13(b)(i)</u>, with the reasonable cooperation of Buyer and at Buyer's expense, take such other commercially reasonable action as may be necessary and feasible in order for Buyer or its designee to be placed in a substantially similar position as if Buyer or its designee were a party to such

Replicated Contract, it being understood and agreed that any transition services which may be performed under any Replicated Contract shall be governed by the Transition Services Agreement.

4.14 Certain Payments

(a) At least three (3) Business Days prior to the Closing, CD&R Investor shall deliver to Harsco a schedule setting forth a reasonable, good faith estimate of the CD&R Investor Transaction Expenses, and, after Closing and upon reasonable request, supporting invoices relating to such estimate.

(b) At the Closing, Buyer shall, without duplication, (<u>a</u>) pay the CD&R Deal Fee to Clayton, Dubilier & Rice, LLC, and (<u>b</u>) pay or, as appropriate, reimburse CD&R Investor for the CD&R Investor Transaction Expenses, (including all fees and other amounts required to be paid to the Financing Sources pursuant to the Debt Financing Commitments or otherwise in connection with the Debt Financing (the "<u>Debt Financing Fees and Expenses</u>")), in each case by wire transfer of immediately available funds in Dollars to the account (or accounts) designated in writing by CD&R Investor or the Financing Sources, as the case may be, at least two (2) Business Days prior to the Closing.

(c) At least three (3) Business Days prior to the Closing, Harsco shall deliver to CD&R Investor a schedule setting forth a reasonable, good faith estimate of the reasonable, out-of-pocket incremental transaction expenses of Harsco and its Affiliates described on <u>Schedule 4.14(c)</u> (the "<u>Harsco Incremental Transaction Expenses</u>"), and, after Closing and upon reasonable request, supporting invoices relating to such estimate.

(d) At the Closing, Buyer shall (<u>a</u>) pay the Harsco Deal Fee to Harsco and (<u>b</u>) pay or, as appropriate, reimburse Harsco for the Harsco Incremental Transaction Expenses up to an aggregate amount of \$2,500,000, by wire transfer of immediately available funds in Dollars to the account (or accounts) designated in writing by Harsco at least two (2) Business Days prior to the Closing.

4.15 Further Assurances

Following the Closing, each of CD&R Investor, on the one hand, and Harsco, on the other hand, shall (and shall cause their Affiliates and their representatives to) from time to time, at the other's reasonable request, execute and deliver, or cause to be executed and delivered, such further instruments, documents, conveyances or assurances and perform such further acts, as such other party may reasonably require in order to fully effect the Transactions and to otherwise consummate the transactions contemplated by this Agreement.

4.16 Harsco House Lease

Prior to the Closing, Harsco and CD&R Investor shall use reasonable best efforts to cause their Affiliates to enter into a sublease, subleasing one floor of Harsco House to Harsco or its designee. Subject to <u>Section 4.2</u>, Harsco shall use its reasonable best efforts to cause the applicable landlord under the Harsco House Sublease to grant any and all consents required in connection therewith.

4.17 Harsco Letters of Credit; Other Obligations

Within three months following the Closing, Buyer shall, (<u>a</u>) arrange for substitute letters of credit, Buyer guarantees and other obligations to replace any letters of credit, guarantees, surety bonds, performance bonds and other contractual obligations entered into by or on behalf of Harsco or any of the Non-Company Affiliates solely for the benefit of the Business that are either (<u>i</u>) entered into prior to the date hereof and set forth on <u>Schedule 4.17</u> or (<u>ii</u>) entered into between the date hereof and the Closing

Date in the ordinary course of business consistent with past practices (collectively, the "<u>Harsco Letters of Credit</u>") that are outstanding as of immediately prior to the Closing or (<u>b</u>) assume all obligations under each Harsco Letter of Credit and obtain from the creditor or other counterparty a full release (in a form reasonably satisfactory to Harsco) of all Sellers or Non-Company Affiliate liable, directly or indirectly, for reimbursement to the creditor or fulfillment of other obligations to a counterparty in connection with amounts drawn under such Harsco Letters of Credit. Buyer further agrees that to the extent the beneficiary or counterparty under any such Harsco Letter of Credit does not accept any such substitute letter of credit, Buyer guarantee or other obligation proffered by Buyer, Buyer shall indemnify, defend and hold harmless Harsco against, and reimburse Harsco for, any and all amounts paid, including costs or expenses in connection with such Harsco Letters of Credit, whether or not any such Harsco Letter of Credit is drawn upon or required to be performed, and shall in any event promptly reimburse Harsco to the extent any such Harsco Letter of Credit is called upon and Harsco or the Non-Company Affiliates make any payment or are obligated to reimburse the party issuing such Harsco Letter of Credit.

4.18 Indemnification and Insurance

(a) Following the Closing, Buyer agrees that all rights to exculpation, advancement of expenses, and indemnification with respect to acts or omissions occurring at or prior to Closing (including any matters arising in connection with the transactions contemplated hereby), whether asserted or claimed prior to, at or after the Closing, existing in favor of the current or former directors and officers of the Target Entities (collectively, the "<u>D&O Indemnitees</u>") by virtue of any provision of the articles of incorporation, by-laws or comparable organizational documents of the Target Entities or any indemnification or other agreement shall survive the consummation of the transactions contemplated hereby and shall continue in full force and effect following the Closing. Following the Closing, Buyer shall maintain in effect (<u>A</u>) the current provisions regarding exculpation, advancement of expenses, and indemnification of officers and directors contained in the articles of incorporation, by-laws or other comparable organizational documents of the Target Entities with any of their respective directors, officers and employees existing as of the date hereof.

(b) Buyer shall either (\underline{A}) for a period of six (6) years from and after the Closing Date, maintain, or cause the Target Entities to maintain, directors' and officers' liability insurance, on terms and amounts no less favorable than those liability insurance policies of Harsco and the Target Entities in effect on the date of this Agreement, for current or former directors and officers of the Target Entities covering acts or omissions occurring prior to the Closing Date to the extent relating to the Business; provided, however, that if such insurance policies are not available at an annual cost not greater than three hundred percent (300%) of the last annual premium paid prior to the date of this Agreement under such policy (the "Insurance Cap"), then Buyer shall maintain, or shall cause the Target Entities to maintain, as much comparable insurance as can reasonably be obtained in Buyer's good faith judgment at a cost up to but not exceeding the Insurance Cap or (\underline{B}) obtain a six (6) year tail policy for the Persons currently covered by the Harsco's directors' and officers' liability insurance policy providing for terms that are consistent with those set forth in clause (A) above and with policy limits commercially reasonable for the Target Entities.

(c) The D&O Indemnitees to whom this <u>Section 4.18</u> applies shall be third party beneficiaries of this <u>Section 4.18</u>. The provisions of this <u>Section 4.18</u> are intended to be for the benefit of each D&O Indemnitee and his or her heirs. Following the Closing, the obligations of Buyer under this <u>Section 4.18</u> shall not be terminated or modified in such a manner as to adversely affect any D&O Indemnitee to whom this <u>Section 4.18</u> applies without the consent of each such affected D&O Indemnitee.

(d) If Buyer or any of its successors or assigns (i) consolidates with or merges into any other Person and shall not be the continuing or surviving corporation or entity of such consolidation or merger or (ii) transfers or conveys all or substantially all of its properties and assets to any Person, then, and in each such case, proper provision shall be made so that the successors and assigns of Buyer, as the case may be, shall assume the obligations set forth in this <u>Section 4.18</u>.

4.19 Brand Purchase Agreement

(a) Without the prior written consent of Harsco, Buyer shall not amend, modify or waive any provision of the Brand Purchase Agreement in a manner that would (i) increase the purchase price to be paid by Buyer to FRC or its designees under the Brand Purchase Agreement or any ancillary agreement related thereto or (ii) following the Closing, disproportionally affect Harsco relative to CD&R Investor. Subject to the foregoing, the CD&R Investor shall deliver promptly to Harsco copies of all amendments, modifications and waivers to the Brand Purchase Agreement.

(b) Except as otherwise prohibited by Law, the CD&R Investor shall (<u>i</u>) keep Harsco informed on a current basis in reasonable detail of the status of the satisfaction of the conditions to closing set forth in the Brand Purchase Agreement, (<u>ii</u>) promptly notify Harsco of the commencement of the BEI Marketing Period and of any suspension thereof and (<u>iii</u>) promptly inform Harsco if it becomes aware that the closing of the transactions contemplated by the Brand Purchase Agreement is not reasonably expected to occur prior to January 24, 2014.

(c) Buyer shall comply in all material respects with its material obligations under the Brand Purchase Agreement (including, for the avoidance of doubt, its obligations to use efforts to consummate the Brand Closing thereunder), <u>provided</u>, for the avoidance of doubt, that nothing herein shall require Buyer to waive any of its rights, or conditions to its obligations to close, under the Brand Purchase Agreement.

4.20 Options to Effect Sale of French Business.

Buyer hereby irrevocably offers to Harsco that, at any time after the date the consultation process with the works councils applicable to the Business Employees in France is completed in accordance with applicable Law and <u>Section 4.6(b)</u>, Harsco may, at its sole option (the "French Sale Option" with respect to Harsco Infrastructure France SAS and the Business in Luxembourg and Algeria (collectively, the "French Business")), by delivery of a written notice to CD&R Investor, include step 4.1on Exhibit G as part of the Closing Transactions and effect such step as part of the Closing Transactions pursuant to Section 4.8(a). Upon the exercise of the French Sale Option, step 4.1 set forth on Exhibit G shall be deemed a part of the Closing Transactions for all purposes of this Agreement, and the exercise of the French Sale Option may not be revoked without CD&R Investor's written consent. If the Closing occurs prior to Harsco's exercise of the French Sale Option, then Buyer shall, at the Closing, extend an irrevocable offer to purchase the French Business at any time within one hundred and seventy (170) days following the Closing Date, at a price equal to \$53,037,000, and on substantially the same terms as would have been provided if Harsco had exercised the French Sale Option, as the case may be, prior to the Closing or on such other terms as may be agreed upon by Buyer and Harsco, in their sole and absolute discretion, after careful consideration of the notification and consultation process with the works councils applicable to the Business Employees in France (the "Irrevocable Offer"). Harsco agrees that it shall not directly or indirectly sell or commit to sell Harsco Infrastructure France SAS or the related business to any other Person so long as the French Sale Option or the Irrevocable Offer remain outstanding. Notwithstanding anything to the contrary herein, Harsco shall not be required to exercise the French Sale Option or accept the Irrevocable Offer.

4.21 Netherlands Business.

The Parties agree that the inclusion of step 13.1 set forth on Exhibit G regarding the Business in the Netherlands (together with the terms and conditions that apply thereto in accordance with this Agreement, the "<u>Netherlands Transaction</u>"), which, for the avoidance of doubt, shall include the shares held by Harsco Infrastructure Industrial Services B.V. in Harsco Infrastructure Industrial Services SA (the "<u>Netherlands Business</u>"), will be subject to the completion by the management of the Netherlands Business of its obligations under and in accordance with the Dutch Works Council Act (*Wet op de Ondernemingsraden*) in connection with the Netherlands Transaction (the "<u>Consultations</u>"), pursuant to the terms and conditions set forth in <u>Schedule 4.21</u>, to the extent not already fulfilled or complied with by the date of this Agreement. Except as set forth on <u>Schedule 4.21</u>, upon completion of the Consultations, step 13.1 set forth on <u>Exhibit G</u> shall automatically be deemed to be part of the Closing Transactions. Buyer and Seller shall, and shall cause their Subsidiaries to, use reasonable best efforts to complete the Consultations prior to the Closing (or as soon as reasonably practicable following the Closing) and, subject to completion of the Consultations, to consummate step 13.1 set forth on <u>Exhibit G</u> at the Closing (or as soon as reasonably practicable following the Closing).

4.22 Real Property.

(a) With respect to each of the properties listed on <u>Schedule 4.22</u>, at the Closing, (i) if such property is owned or leased by Harsco or a Non-Company Affiliate, such party will retain ownership and/or its leasehold interest in such property and (ii) if such property is owned or leased by a Target Entity, the property will be transferred to a Seller or Non-Company Affiliate (as designated by Harsco). An amount equal to the aggregate of the Fair Market Value of each of the properties listed on <u>Schedule 4.22</u> shall be included in the definition of Indebtedness for purposes of <u>Sections 1.2</u> and <u>Section 1.9</u>. At the Closing, Harsco shall, or shall cause its Subsidiary which owns or leases the relevant property, subject to <u>Section 4.2(c)</u>, to lease or sublease each of the properties listed on <u>Schedule 4.22</u> to one of the Target Entities for Market Rent. Each such lease shall (<u>A</u>) have an initial term of ten (10) years, (<u>B</u>) be renewable at tenant's option for up to two additional five-year terms and (<u>C</u>) contain other customary terms and conditions for a lease reasonably acceptable to both Harsco and CD&R Investor.

(b) As soon as practicable following the date hereof, each of Harsco and CD&R Investor shall select a third-party appraiser to determine (<u>i</u>) the fair market value and (<u>ii</u>) a market rent, in each case, as of the Closing, based on current use of each of the properties listed on <u>Schedule 4.22</u>. The "<u>Fair Market Value</u>" or "<u>Market Rent</u>" of each property shall be the average of the two (2) valuations (the "<u>Original Valuations</u>") unless the valuations differ by more than 5%, in which case, the two appraisers must mutually agree and select a third independent and reputable appraiser to conduct an appraisal (the "<u>Third Valuation</u>"), and the Fair Market Value or Market Rent for that property will be the average of the Third Valuation and the Original Valuation which is closest to the Third Valuation.

4.23 <u>Middle East Arrangements</u>. From the date hereof through the Closing Date, Seller shall, and from and after the Closing Date, CD&R Investor and Buyer shall, and Buyer shall cause its Subsidiaries to, use reasonable best efforts to resolve, as promptly as practical, the matters set forth on <u>Schedule 8.2(e)(ii)</u>.

4.24 <u>Repatriation Schedule</u>. Harsco and Buyer shall, prior to the Closing, cooperate in good faith and acting reasonably to revise the Repatriation Schedule so that it reflects the applicable tax rate for each non-U.S. jurisdiction in which the Target Entities have Cash taken into account in computing the Repatriation Costs.

ARTICLE V

TAX MATTERS

5.1 <u>Tax Indemnification</u>

(a) <u>Indemnification by Harsco</u>. From and after the Closing, the Sellers, on a joint and several basis, agree to indemnify the Buyer Indemnified Parties against, and to pay and hold the Buyer Indemnified Parties harmless from, all Losses arising out of (i) any Taxes of or payable by any Target Entity or with respect to the Business or the Target Assets, in each case, with respect to any Pre-Closing Tax Period, together with any interest, penalty or additions to Tax accruing after the Closing on Taxes described in this clause (i); (ii) any Liability of any Target Entity for the Taxes of any Person by reason of (A) being a transferee or successor to such Person prior to the Closing Date, (B) pursuant to Section 1.1502-6 of the Treasury Regulations (or comparable provision under any other applicable Law) by reason of being affiliated with such Person prior to the Closing or (<u>C</u>) any Contract entered into by such Target Entity prior to the Closing (other than by reason of any customary provisions contained in (1) any Contract for the provision of goods or services entered into in the ordinary course of business or (2) commercial lending arrangements entered into with third parties), including any secondary liability for Taxes imposed in connection with any such Contract; (iii) any Taxes attributable to any inaccuracy of or breach by Harsco of any representation or warranty made by Harsco in Section 2.9(d), (g),(i) or (j); (iv) any Taxes attributable to any failure by Harsco to perform or comply with any covenant or agreement in this Agreement relating to Taxes; (v) any Taxes that arise as the result of any inclusion under Section 951 of the Code at the end of the taxable year of any Target Entity that is a controlled foreign corporation (as defined under Section 957 of the Code) that includes the Closing Date to the extent such inclusion results from any transactions or activities occurring between the beginning of the taxable year of such controlled foreign corporation that includes the Closing Date and through the Closing; (vi) subject to Section 5.4, any Taxes that arise from the Transactions and any other transactions or activities contemplated by Section 4.8; and (vii) any Transaction Taxes that are the responsibility of Sellers pursuant to Section 5.4; provided, however, that Harsco in all events shall have no obligation to indemnify the Buyer Indemnified Parties under this Section 5.1(a) from and against any Taxes to the extent such Taxes are taken into account on the Final Closing Statement (and assuming full completion of the Transactions) as finally determined pursuant to <u>Section 1.9</u>. The amount of any payments required to be made pursuant to this Section 5.1(a) shall be computed without regard to any net operating loss, net capital loss or other Tax deduction, credit or benefit that is attributable to, arises from or relates to any Post-Closing Tax Period. Notwithstanding any other provision of this Agreement and for the avoidance of doubt, the limitations in <u>Section 8.4</u> shall not apply to this <u>Section 5.1(a)</u>.

(b) <u>Indemnification by Buyer</u>. From and after the Closing Date, Buyer agrees to indemnify CD&R Investor, Harsco and the Non-Company Affiliates against, and to pay and hold CD&R Investor, Harsco and the Non-Company Affiliates harmless from, all Losses arising out of (<u>i</u>) any Taxes attributable to any failure by any Target Entity to perform or comply with any covenant or agreement in this Agreement after the Closing related to Taxes, (<u>ii</u>) any Transaction Taxes that are the responsibility of the Buyer pursuant to <u>Section 5.4</u>, (<u>iii</u>) any Taxes to the extent such Taxes are taken into account on the

Final Closing Statement (and assuming full completion of the Transactions) as finally determined pursuant to <u>Section 1.9</u> and (<u>iv</u>) notwithstanding <u>Section 5.1(a)</u>, any Taxes attributable to any transaction undertaken by Buyer or any Target Entity outside of the ordinary course of business occurring after the Closing on the Closing Date and not contemplated by this Agreement or directed by Harsco or any Non-Company Affiliate prior to the Closing.

(c) <u>Indemnification by CD&R Investor</u>. From and after the Closing Date, CD&R Investor agrees to indemnify Harsco and the Non-Company Affiliates against, and to pay and hold Harsco and the Non-Company Affiliates harmless from, any Transaction Taxes that are allocated to the CD&R Investor under <u>Section 5.4</u>.

5.2 Procedures Relating to Certain Tax Indemnification

(a) If a claim for Taxes, including notice of a pending audit, shall be made by any Tax Authority in writing, which, if successful, might result in an indemnity payment pursuant to <u>Section 5.1</u> (any such claim, a "<u>Tax Claim</u>"), the party which receives such notice shall notify the other party in writing of the Tax Claim within fifteen (15) Business Days after receipt of such Tax Claim. If the party seeking indemnification (the "<u>Tax Indemnified Party</u>") does not give notice of a Tax Claim to the other party (the "<u>Tax Indemnifying Party</u>") within such period or in detail sufficient to apprise the Tax Indemnifying Party of the nature of the Tax Claim, the Tax Indemnifying Party shall not be liable to the Tax Indemnified Party to the extent that the Tax Indemnifying Party's position is prejudiced as a result thereof.

(b) Harsco shall control all proceedings in connection with any Tax Claim (including selection of counsel) related to a Consolidated Tax Return and, without limiting the foregoing, may pursue or forgo any and all administrative appeals, proceedings, hearings and conferences with any Tax Authority with respect thereto, and may either pay the Tax Claim and sue for a refund where applicable Law permits such refund suits or contest such Tax Claim in any permissible manner; provided, however, that (i) Harsco shall keep the relevant Tax Indemnified Party regularly informed as to the status of such proceedings (including by providing copies of all notices received from the relevant Tax Authority) to the extent such proceedings relates to, or could have an impact on, the Business, a Target Asset or a Target Entity and (ii) Harsco shall not settle or otherwise compromise such Tax Claim without the relevant Tax Indemnified Party's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed) unless Harsco indemnifies and holds harmless the Tax Indemnified Party from and against such Taxes actually incurred as a result of such settlement or compromise. With respect to any Tax Claim (or proceeding arising therefrom) relating exclusively to a Pre-Closing Tax Period (other than any such claim or proceeding related to a Consolidated Tax Return), Harsco shall control all proceedings in connection with any such Tax Claim (including selection of counsel) and, without limiting the foregoing, may pursue or forgo any and all administrative appeals, proceedings, hearings and conferences with any Tax Authority with respect thereto, and may either pay the Tax Claim and sue for a refund where applicable Law permits such refund suits or contest such Tax Claim in any permissible manner; provided, however, that (i) Harsco shall keep the relevant Tax Indemnified Party regularly informed as to the status of such proceedings (including by providing copies of all notices received from the relevant Tax Authority) and such Tax Indemnified Party shall have the right to review and comment on any correspondence from Harsco to the relevant Tax Authority prior to submission of such correspondence to the Tax Authority and (ii) Harsco shall not settle or otherwise compromise such Tax Claim without the relevant Tax Indemnified Party's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed) unless Harsco indemnifies and holds harmless the Tax Indemnified Party from and against such Taxes actually incurred as a result of such settlement or compromise. With respect to any Tax Claim (or proceedings arising therefrom) relating to a Straddle Period, Buyer shall control such proceedings; provided, however,

that (\underline{x}) Buyer shall keep Harsco reasonably informed as to the status of such proceedings (including by providing copies of all notices received from the relevant Tax Authority) and Harsco shall have the right to review and comment on any correspondence from Buyer to the relevant Tax Authority prior to submission of such correspondence to the Tax Authority and (\underline{y}) Buyer shall not settle or compromise such Tax Claim without Harsco's prior written consent (which consent shall not be unreasonably withheld, conditioned or delayed) unless Buyer indemnifies and holds harmless Harsco from and against such Taxes actually incurred as a result of such settlement or compromise.

5.3 Tax Returns

(a) Harsco shall timely prepare and file or cause to be timely prepared and filed with the appropriate Tax Authorities (i) all Tax Returns required under applicable Law to be filed (taking into account any applicable extensions) on or prior to the Closing Date by or with respect to any Target Entity and (ii) all Tax Returns required under applicable Law to be filed (taking into account any applicable extensions) after the Closing Date that are Consolidated Tax Returns. All such Tax Returns shall be prepared in a manner consistent with most recent past practice, except as otherwise required by applicable Law. After the Closing Date, Buyer shall timely prepare and file or cause to be timely prepared and filed with the appropriate Tax Authorities all other Tax Returns required to be filed with respect to Buyer or any Target Entity, and to the extent such Tax Returns are for Pre-Closing Tax Periods, such Tax Returns shall be prepared on a basis consistent with past practices and prior Tax reporting positions (except as otherwise required by applicable Law). Buyer shall provide Harsco, at least twenty (20) Business Days prior to the applicable deadline for filing any such Tax Return with respect to a Pre-Closing Tax Period, a copy of the Tax Return for Harsco's review and comment. Harsco shall have ten (10) Business Days to provide Buyer with a statement of any disputed items with respect to such Tax Return. If the disputed items are not resolved by Harsco and Buyer within five (5) calendar days following Harsco's submission of its statement of disputed items, the matter shall be submitted to one or more tax experts at the CPA Firm who shall be directed to, within ten (10) calendar days after such submission, render a decision with respect to all matters in dispute, and such decision shall be final, binding and conclusive on the Parties hereto. The fees and disbursements of the CPA Firm shall be shared equally by Harsco and Buyer. Harsco shall pay to the applicable Target Entity, no later than two (2) Business Days prior to the due date therefor, all Taxes of such Target Entity shown as due on any Tax Return the extent such Taxes are attributable to the Pre-Closing Tax Period and not accrued as a current tax payable in the Closing Net Working Capital as finally determined.

(b) Except as required by applicable Law, Buyer shall not file or cause or permit any Target Entity to file any amended Tax Return after the Closing Date with respect to any Pre-Closing Tax Period without the prior written consent of Harsco (which consent shall be at the sole discretion of Harsco).

(c) Harsco and Buyer shall reasonably cooperate, and shall cause their respective Affiliates, officers, employees, agents, auditors and representatives to reasonably cooperate, in preparing and filing all Tax Returns relating to Buyer or any Target Entity, including by provision of any required power-of-attorney (or other form of authorization), and in maintaining and making available to each other all records necessary in connection with Taxes and in resolving all disputes and audits, and in connection with any other legitimate matters (including, for the avoidance of doubt, reasonable requests for information by Harsco or by CD&R Investor, its Affiliates or their employees or representatives, relating to the tax planning of Buyer and the Target Entities) with respect to all taxable periods relating to Taxes.

(d) Any overpayments, refunds or credits of, Taxes attributable to Pre-Closing Tax Periods of the Target Entities (including in respect of the Straddle Period) for which Harsco is responsible

pursuant to <u>Section 5.1(a)</u> to the extent not taken into account in the Final Closing Statement (and assuming full completion of the Transactions), plus any interest actually received with respect thereto from an applicable Tax Authority (and including refunds or credits in respect of such Taxes arising by reason of amended Tax Returns filed after the Closing Date), shall be for the account of Harsco unless such refunds or credits result from a carryback of losses or other Tax attributes from a Post-Closing Tax Period. Buyer shall pay or cause to be paid such amount to Harsco less reasonable out-of-pocket expenses incurred in connection with obtaining such refunds less any Taxes incurred by Buyer or any Target Entity as a result of such refunds or credits (including interest thereon). Buyer shall, if reasonably requested by Harsco and solely at Harsco's cost, use its commercially reasonable efforts to file for, or cause to be filed for, and to obtain the receipt of, any refund to which Harsco is entitled under this <u>Section 5.3(d)</u>.

5.4 Transaction Taxes

Notwithstanding any other provision of this Agreement to the contrary, all transfer, documentary, sales, use, excise, stamp, registration and other such Taxes (including amounts in respect of VAT) imposed by or payable to any jurisdiction or any Governmental Body arising from the Transactions (other than the Pre-Closing Transactions) or the transfer of the Harsco Company Shares to Harsco (collectively, "<u>Transaction Taxes</u>") shall be borne fifty percent (50%) by Buyer and fifty percent (50%) by Harsco. The Person primarily responsible for such filing under applicable Law will, at its own expense, file all necessary Tax Returns and other documentation with respect to all such Transaction Taxes, and, if required by applicable Law, Buyer and Harsco will, and will cause their respective Affiliates to, join in the execution of any such Tax Returns and other documentation. If Buyer is not the Person primarily responsible for such filing under applicable Law, Buyer and Harsco shall, upon the request of the Person responsible for such filing under applicable Law, Buyer and Harsco shall, upon the request of the Person responsible for such filing under applicable Law, Buyer and Harsco agree, upon the request of the other Party, to use commercially reasonable efforts to obtain any certificate or other document from each other, any Governmental Body or any other Person as may be necessary to mitigate, reduce or eliminate any Transaction Taxes that could be imposed in connection with the transaction contemplated by this Agreement. For the avoidance of doubt, all Transaction Taxes, if any, arising as a result of the Pre-Closing Transactions shall be borne exclusively by the Sellers.

5.5 Records Retention

The Sellers and Buyer shall (and shall cause their respective Affiliates to) (i) provide the Parties and their respective Affiliates with such assistance as may be reasonably requested in connection with the preparation of any Tax Return or any audit or other examination by any Tax Authority or any judicial or administrative proceeding relating to Taxes, <u>provided</u> that the foregoing shall be done in a manner so as not to interfere unreasonably with the conduct of the business of the Parties, and (<u>ii</u>) retain (and provide the Parties and their respective Affiliates with reasonable access to), until the expiration of the later of (<u>A</u>) the seventh anniversary of the Closing Date and (<u>B</u>) the date on which Taxes may no longer be assessed under the applicable statute of limitation, including any waivers or extensions thereof, all records or information which may be relevant to such Tax Return, audit, examination or proceeding in a manner consistent with most recent past practice.

5.6 Tax Sharing Agreements

Harsco shall release or cause the release of each Target Entity from any obligation under any agreement relating to the allocation, indemnification or sharing of Taxes other than this Agreement ("<u>Tax Sharing Agreements</u>") with any Person (other than any Target Entity) prior to the Closing Date. For the

avoidance of doubt, Harsco shall cause each Target Entity to have no obligation under any Tax Sharing Agreement to any Person (other than any Target Entity) for the payment of Taxes.

5.7 Straddle Period

For purposes of this <u>ARTICLE V</u>, in the case of any Straddle Period, the amount of any (i) Taxes based on or measured by income or receipts of the Target Entities for the Pre-Closing Tax Period and (<u>ii</u>) all other Taxes that otherwise can be reasonably allocated to the Pre-Closing Tax Period shall be determined based on an interim closing of the books as of the close of business on the Closing Date (and for such purpose, the taxable period of any partnership or other pass-through entity in which a Target Entity (or Non-Company Affiliate with respect to the Business) holds a beneficial interest shall be determined to terminate at such time). The amount of any Taxes of the Target Entities for a Straddle Period to be attributed to the Pre-Closing Tax Period that is not susceptible to allocation based on the methodology described in the preceding sentence shall be determined by multiplying the amount of such Tax for the entire taxable period by a fraction the numerator of which is the number of days in the taxable period ending on the Closing Date and the denominator of which is the number of days in such Straddle Period.

5.8 Waiver of Loss Carrybacks

To the extent permitted under applicable Law, no Target Entity shall carry any losses or other Tax attributes arising in any Post-Closing Tax Period in respect of a Consolidated Tax Return to any Pre-Closing Tax Period, <u>provided</u> that in cases where any Target Entity is required by law to carry such losses or Tax attributes to any Pre-Closing Tax Period before it can carry forward such losses or Tax attributes, (<u>i</u>) any net Tax benefit actually realized by Harsco or any of its Affiliates shall be remitted by Harsco to Buyer at the time such Tax benefit is actually realized and (<u>ii</u>) Taxes of any Target Entity shall be determined without regard to any such carry backs for purposes of <u>Section 5.1(a)</u>.

5.9 [Intentionally Omitted]

5.10 Computational Matters; Indemnification Cap

(a) The indemnification obligations of Harsco, CD&R Investor, and Buyer pursuant to this <u>ARTICLE V</u> shall, in respect of computational matters, be subject to <u>Section 8.6</u>.

(b) Under no circumstance shall Harsco's aggregate obligation to provide indemnification for matters covered by <u>Section 5.1(a)</u> (but excluding Liabilities described in <u>Section 5.1(a)(ii)</u>, which shall not be subject to any cap) exceed an amount equal to the Purchase Price.

5.11 Certain Tax Elections; Actions

(a) Buyers shall not, without the prior written consent of Harsco, (<u>i</u>) make (or cause to be made) any election under Treasury Regulation Section 301.7701-3 to change the U.S. federal tax classification of Buyer or any Target Entity where such election would be effective on or before the day after the Closing Date or (<u>ii</u>) unless otherwise required by applicable Law, make (or cause to be made) any election to change any accounting method that would have a material adverse effect on any Target Entity or Seller with respect to a Pre-Closing Tax Period.

(b) With respect to each Target Entity identified on <u>Schedule 5.11</u>, Buyers shall not, without the prior written consent of Harsco, pay (or cause to be paid) any dividends or other distributions that would be treated as a dividend for U.S. federal income tax purposes out of such Target Entity prior to the close of the taxable year (determined for U.S. federal income tax purposes) of such entity in which the

Closing occurs, <u>provided</u> that, for the avoidance of doubt, nothing in this <u>Section 5.11(b)</u> shall restrict the payment of any dividend or other distribution if (i) the Buyer and its wholly-owned Subsidiaries do not control the payment of such dividend, (ii) the payment of such dividend or other distribution is legally required by contract or otherwise, or (iii) a Person (other than Buyer and its wholly owned Subsidiaries) own an interest in such Target Entity and such Person objects to the absence of the payment of such dividend or other distribution.

(c) An election under each of Section 22 and Section 85 of the Income Tax Act (Canada) shall be made with respect to the purchase under this Agreement of New Harsco Canada ULC.

(d) If the Closing occurs on or before January 1, 2014 and if any of the Contributions is by a domestic corporation to a foreign corporation (each as determined for U.S. federal income tax purposes), the Company shall, to the extent permitted by Law, cause (i) a "gain recognition agreement" within the meaning of Treasury Regulation Section 1.367(a)-8 (a "<u>GRA</u>") to be timely filed with respect to any gain realized in such Contribution and (ii) a protective GRA to be timely filed with respect to any such gain but as if such Contribution occurred in 2013 (the "<u>Protective GRA</u>" and, collectively with the GRA, the "<u>GRAs</u>"). The Company shall provide Harsco, at least twenty (20) Business Days prior to the applicable deadline for filing the Protective GRA, a copy of such Protective GRA for Harsco's review and comment. Notwithstanding any other provision of this Agreement and for the avoidance of doubt, none of the GRAs shall create any obligation or liability in respect of any taxable income or gain realized prior to the UK Closing. On or before January 1, 2014, the Company shall not (directly or indirectly) agree to sell the UK Shares (or any stock or other equity interests received in exchange therefor in connection with the Contributions) to any other third party. Assuming compliance with the immediately preceding sentence, it is understood and agreed that neither the Company nor any of its Affiliates shall be responsible for any Tax on any gain realized prior to the Closing that relates to the UK Entity or the UK Shares and any such Tax shall be deemed to be described in <u>Section 5.1(a)(i)</u>.

(e) The Parties agree (and agree to use reasonable efforts to secure) that by virtue of the provisions of Article 5 of the VAT (Special Provisions) Order 1995 and section 49(1) of the UK Value Added Tax Act 1994, the sale and purchase of the Harsco UK infrastructure business pursuant to this Agreement shall not be treated as a supply of goods or services for VAT purposes. If for any reason H.M. Revenue & Customs decide that VAT is chargeable in respect of any part of the sale and purchase of the Harsco UK infrastructure business, such VAT shall be payable by the relevant Buyer to the relevant Seller in addition to the Purchase Price upon delivery by the relevant Seller of a valid VAT invoice. Notwithstanding the foregoing, and for the avoidance of doubt, any VAT paid in accordance with this <u>Section 5.11(e)</u> shall be borne by the Parties in accordance with <u>Section 5.4</u>.

(f) With respect to any Target Entity identified on <u>Schedule 5.11</u>, if the Closing occurs on or before January 1, 2014, Buyers shall not, without the prior written consent of Harsco, take any action on or before January 1, 2014 that would have the effect of closing the taxable year of such entity for U.S. federal income tax purposes prior to the close of the taxable year of such entity in which the Closing occurs for purposes of the tax law of the jurisdiction in which such Target Entity is resident for tax purposes.

(g) Between the date hereof and the Closing Date, the Parties shall cooperate with each other and negotiate in good faith to revise the Upper-Tier Partnership Agreement such that the allocations of income, gain, loss or deduction (or items thereof) made pursuant to the Upper-Tier Partnership Agreement comply with the substantial economic effect safe harbor under Treasury Regulations Section 1.704-1(b)(2) (which, for the avoidance of doubt, will require that liquidating distributions be made in accordance with the positive capital account balances of the partners); provided that, for the avoidance of

doubt, (i) no such revised Form of Upper-Tier Partnership Agreement shall be adopted unless Harsco and the CD&R Investor each determine in its discretion that (A) the revisions will not have any economic impact to any partner in any situation (as compared to the economic results under the existing Form of Upper-Tier Partnership Agreement) and (B) the revised form is understandable and (ii) Harsco or the CD&R Investor may discontinue such cooperation and negotiations if it determines that the conditions set forth in clause (i) cannot be satisfied. For the avoidance of doubt, the obligations of the Parties to effect the Closing under this Agreement shall not be subject to any such revision to the Upper-Tier Partnership Agreement.

5.12 Exclusivity

To the extent of any inconsistency between this <u>ARTICLE V</u> and <u>ARTICLE VIII</u>, this <u>ARTICLE V</u> shall control as to Tax matters.

ARTICLE VI

CONDITIONS TO THE CLOSING

6.1 Conditions to Obligations of the Sellers

The obligations of the Sellers to effect the Closing under this Agreement will be subject to the fulfillment (or written waiver by Harsco), on or prior to the Closing Date, of each of the following conditions:

(a) <u>Accuracy of Representations and Warranties</u>. The representations and warranties of CD&R Investor contained in this Agreement shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or "material adverse effect" included therein) on and as of the date hereof and the Closing Date as if made on and as of such date (other than representations and warranties which address matters only as of a certain date which shall be accurate as of such certain date), except where the failure of such representations and warranties to be true and correct would not, individually or in the aggregate, reasonably be expected to have a material adverse effect on CD&R Investor's ability to consummate the transactions contemplated by this Agreement and the Ancillary Agreements.

(b) <u>Compliance with Agreements and Covenants</u>. All the covenants and agreements contained in this Agreement to be complied with by the Buyers or CD&R Investor on or before the Closing shall have been complied with in all material respects.

(c) <u>Certificate of Compliance</u>. Each of CD&R Investor and Buyer shall have delivered to Harsco a certificate dated as of the Closing Date, signed by a duly authorized officer of CD&R Investor, certifying as to satisfaction of the conditions set forth in <u>Section 6.1(a)</u> and <u>Section 6.1(b)</u>.

(d) <u>Antitrust Approvals</u>. All applicable waiting periods and clearances pursuant to the Applicable Non-U.S. Antitrust Approvals shall have unconditionally expired, been terminated or been obtained, as applicable.

(e) <u>No Adverse Order</u>. No Governmental Body shall have enacted, issued, promulgated, enforced or entered any Law or Governmental Order which is in effect and has the effect of

making the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise restraining or prohibiting consummation of such transactions.

(f) <u>Other Agreements</u>. CD&R Investor and Buyer shall have delivered to Harsco copies of each of the Ancillary Agreements to which it is a party, duly executed by it and approved in accordance with applicable Law and the organizational documents of Buyer and CD&R Investor, as applicable.

(g) <u>Brand Closing</u>. The Brand Closing shall have occurred or shall occur simultaneously with the Closing.

6.2 Conditions to Obligations of CD&R Investor and Buyer

The obligations of CD&R Investor and Buyer to effect the Closing under this Agreement will be subject to the fulfillment (or written waiver by CD&R Investor or Buyer), on or prior to the Closing Date, of each of the following conditions:

(a) <u>Accuracy of Representations and Warranties</u>. (i) The representations and warranties of Harsco and the other Sellers contained in this Agreement (other than the representations and warranties identified in clause (ii) hereof) shall be true and correct in all respects (without giving effect to any qualifications or limitations as to "materiality" or "Material Adverse Effect" included therein other than the materiality qualifier contained in <u>Section 2.3(c)</u>), except where the failure of such representations and warranties to be true and correct would not, individually or in the aggregate, reasonably be expected to have a Material Adverse Effect and (<u>ii</u>) the representations and warranties in <u>Sections 2.1</u>, <u>2.2</u> and <u>2.19</u> shall be true and correct in all respects, in the case of the representations and warranties referenced in each of the foregoing clauses (i) and (ii), on and as of the date hereof and the Closing Date as if made on and as of such date (other than representations and warranties which address matters only as of a certain date which shall be accurate as of such certain date).

(b) <u>Compliance with Agreements and Covenants</u>. The covenants and agreements contained in <u>Section 4.8</u> to be complied with by the Sellers shall have been complied with in all respects (or, with respect to those actions to occur at the Closing, are expected to occur at the Closing) and all the other covenants and agreements contained in this Agreement to be complied with by the Sellers on or before the Closing shall have been complied with in all material respects.

(c) <u>Certificate of Compliance</u>. Harsco shall have delivered to Buyer a certificate dated as of the Closing Date, signed by a duly authorized officer of Harsco certifying as to the satisfaction of the conditions set forth in <u>Section 6.2(a)</u> and <u>Section 6.2(b)</u>.

(d) <u>Antitrust Approvals</u>. All applicable waiting periods and clearances pursuant to the Applicable Non-U.S. Antitrust Approvals shall have unconditionally expired, been terminated or been obtained, as applicable.

(e) <u>No Adverse Order</u>. No Governmental Body shall have enacted, issued, promulgated, enforced or entered any Law or Governmental Order which is in effect and has the effect of making the transactions contemplated by this Agreement or the Ancillary Agreements illegal or otherwise restraining or prohibiting consummation of such transactions.

(f) <u>Other Agreements</u>. The Sellers shall have delivered to Buyer copies of each of the Ancillary Agreements, to the extent a party thereto, duly executed by and approved in accordance with

applicable Law and the organizational documents of Harsco and/or their Affiliates (to the extent a party thereto).

(g) <u>No Material Adverse Effect</u>. No change, effect, event, development or occurrence shall have occurred since the date hereof that, individually or in the aggregate, has had or would reasonably be expected to have a Material Adverse Effect.

- (h) <u>Brand Closing</u>. The Brand Closing shall have occurred or shall occur simultaneously with the Closing.
- 6.3 Frustration of Closing Conditions

None of the Sellers, Buyer or CD&R Investor may rely, either as a basis for not consummating the transactions contemplated under this Agreement or terminating this Agreement and abandoning such purchase and sale, on the failure of any condition set forth in <u>Section 6.1</u> or <u>6.2</u>, as the case may be, to be satisfied if such failure was caused by such party's breach in any material respect of any provision of this Agreement or failure to use all the requisite efforts required to consummate the transactions contemplated hereby, as required by and subject to <u>Section 4.2</u> and any other applicable provisions of this Agreement.

ARTICLE VII

CERTAIN RESTRICTIONS

7.1 Non-Competition

(a) Harsco agrees that for the period from the Closing Date until the one year anniversary of the date on which Harsco (or its Affiliates) no longer has a right to designate any directors to the board of directors (or similar governing body) of the Company (the "<u>Non-Competition Period</u>") it shall not and shall cause the Non-Company Affiliates not to, engage in the Business anywhere in the world (each, a "<u>Competitive Activity</u>"); provided that the foregoing shall not prohibit Harsco or any of its Affiliates from collectively owning (\underline{x}) the Harsco Partnership Interests or other equity interests in the Company (or any successor entity) or participating in the management of the Company, Buyer and the Target Entities pursuant to this Agreement and the Ancillary Agreements or (y) up to an aggregate of five percent of the outstanding shares of any class of capital stock of any Person that engages in any Competitive Activity (a "<u>Competing Person</u>") so long as neither Harsco nor any of its Affiliates has any participation in the management of such Competing Person.

(b) Notwithstanding anything to the contrary in the foregoing, nothing in <u>Section 7.1(a)</u> shall:

(i) prohibit Harsco or any of its Affiliates from acquiring the whole or any part of a Person or business which engages in any Competitive Activity or the whole or any part of a business which includes any Competitive Activity; <u>provided</u> that where such Competitive Activities of such Person or business represent greater than 15% of the revenues of such Person or business acquired as set out in the latest available annual financial statements of that Person or business or portion thereof to the extent engaging in such Competitive Activity within 18 months after the consummation of such acquisition;

(ii) prohibit Harsco or any of its Affiliates from acquiring a Minority Investment in a Person or business which engages in, or includes, any Competitive Activity. As used in this Agreement, the term "<u>Minority Investment</u>" means any minority equity investment by Harsco or any of its Affiliates in any Person in which Harsco and any of Harsco's Affiliates, as applicable, collectively hold less than 15% of the outstanding voting securities or similar equity interests of such Person entitled to elect the board of directors (or similar governing body) of such Person; or

(iii) apply to or restrict the ownership or operations of each of the UK Entity, the Deferred India Entity, the Deferred China Entities or the Deferred Baroom Entity until such time as the relevant entity is held directly or indirectly by the Buyer.

7.2 Non-Solicitation

(a) Harsco agrees that from and after the date of this Agreement until the eighteen month anniversary of the Closing Date (the "<u>Non-Solicitation Period</u>"), it shall not, and shall cause its Affiliates not to, request or induce any Person who is at any time from the date of this Agreement to the Closing Date employed in the Business, other than those employees whose responsibilities are solely ministerial in nature, to terminate his or her employment with Buyer and the Target Entities, except in the ordinary course of business, and except for employees that are not Transferred Employees; <u>provided</u>, <u>however</u>, that the foregoing shall not apply (i) to solicitations made by job opportunity advertisements and headhunter searches directed to the general public rather than targeting any employees of Buyer or any of its Subsidiaries (including the Target Entities) (but shall apply to any hiring pursuant thereto of any employees of Buyer or any of its Subsidiaries (including the Target Entities) with a base salary of \$100,000 or higher) or (<u>ii</u>) with respect to any employee who has been terminated by Buyer or the Target Entities (or has voluntarily left his or her employment) more than six months prior to such solicitation.

(b) CD&R Investor agrees that during the Non-Solicitation Period, it shall not, and it shall cause its Subsidiaries (including, after the Closing, the Target Entities) not to, directly or indirectly, request or induce any Person who is at any time during the Non-Solicitation Period employed by Harsco or any of the Non-Company Affiliates (excluding, for the avoidance of doubt, Transferred Employees), other than those employees whose responsibilities are solely ministerial in nature, to terminate his or her employment with Harsco or any of its Non-Company Affiliates; provided, however, that the foregoing shall not apply (i) to solicitations made by job opportunity advertisements and headhunter searches directed to the general public rather than targeting any employees of Harsco or the Non-Company Affiliates (but shall apply to any hiring pursuant thereto of any employees of Harsco or any of its Affiliates, as applicable, (or has voluntarily left his or her employment) more than six months prior to such solicitation.

7.3 Specific Performance

Harsco, Buyer and CD&R Investor recognize and affirm that in the event of breach by any such party or its Affiliates of any of the provisions of this <u>ARTICLE VII</u>, money damages would be inadequate and the other Parties may have no adequate remedy at law. Accordingly, Harsco, on the one hand, and CD&R Investor and Buyer, on the other hand, agree that the other party shall have the right, in addition to any other rights and remedies existing in its favor, to enforce its rights and the other party's obligations under this <u>ARTICLE VII</u> not only by an action or actions for damages, but also by an action or actions for specific performance, injunction and/or other equitable relief in order to enforce or prevent any violations (whether anticipatory, continuing or future) of the provisions of this <u>ARTICLE VII</u>. Harsco, Buyer and

CD&R Investor agree that the other party is not required to post a bond in order for the other party to secure an injunction.

7.4 <u>Severability</u>

If at any time any of the provisions of this <u>ARTICLE VII</u> shall be determined to be invalid or unenforceable by reason of being vague or unreasonable as to duration, area or scope of activity, or otherwise, then this <u>ARTICLE VII</u> shall be considered divisible (with the other provisions to remain in full force and effect) and the invalid or unenforceable provisions shall become and be deemed to be immediately amended to include only such time, area, scope of activity and other restrictions, as shall be determined to be reasonable and enforceable by the court or other body having jurisdiction over the matter, and the Parties hereto expressly agree that this Agreement, as so amended, shall be valid and binding as though any invalid or unenforceable provision had not been included herein. Without limiting the foregoing, if the length of the Non-Competition Period is determined to be unacceptable under applicable Law, the Non-Competition Period shall be modified as determined by a court or other agency of competent jurisdiction or statute, as applicable, to be of the maximum length permitted under applicable Law.

7.5 Intellectual Property Matters

(a) Subject to the terms and conditions of this Agreement, Buyer hereby grants, effective as of the Closing, to Harsco and its Affiliates a non-exclusive, perpetual, irrevocable, non-sublicensable and non-assignable (except as provided in <u>Section 7.5(b)</u>), royalty-free, fully paid-up, worldwide license, in connection with the current and future operation of their businesses, to use and exercise all rights under all Intellectual Property (other than Marks and Domain Names) that is owned by the Target Entities and is or was used by Harsco and its Affiliates (other than the Target Entities) as of or prior to the Closing Date, but only to the same extent and manner that Harsco and its Affiliates used such Intellectual Property in their businesses as of the Closing Date.

(b) Harsco or its Affiliates may assign or sub-license the license set forth in <u>Section 7.5(a)</u> to any Affiliate, or in connection with a merger, reorganization, or sale of all, or substantially all, of any of the businesses to which this license relates, so long as: (<u>i</u>) the assigning or sublicensing party provides CD&R Investor and Buyer or its successor or assignee with prompt written notice of such transaction and (<u>ii</u>) the assignee or sublicensee cannot extend the benefits of such license to its other businesses, <u>provided</u> that no such assignment or sublicense pursuant to this <u>Section 7.5(b)</u> shall relieve the assigning or sublicensing party of its obligations hereunder.

(c) Subject to the terms and conditions of this Agreement, each of Harsco and the Non-Company Affiliates hereby grants, effective as of the Closing, to Buyer and its Subsidiaries (including the Target Entities) a non-exclusive, irrevocable (except as set forth below), non-sublicensable and non-assignable (except as provided in <u>Section 7.5(e)</u>), royalty-free, fully paid-up, worldwide license, in connection with the current and future operation of the Business, to use and exercise all rights under all Intellectual Property that is owned by Harsco or any Non-Company Affiliate and is or was used by the Target Entities or the Asset Sellers in connection with the Business as of or prior to the Closing Date, but only to the same extent and manner that the Target Entities or the Asset Sellers used such Intellectual Property in the Business as of the Closing Date. Such license shall continue (<u>i</u>) for one year with respect to the designation "Harsco" or any variation, derivation, or translation thereof or any related mark, logo or domain name, or any name likely to cause confusion with the mark "Harsco" (the "<u>Harsco Designations</u>"), and (<u>ii</u>) perpetually with respect to all other Intellectual Property described in this <u>Section 7.5(c</u>). With respect to the license of any and all Intellectual Property incorporating a Harsco

Designation, Buyer and its Subsidiaries (including the Target Entities) agree that all goods and services offered under or in connection with such Intellectual Property shall be materially consistent with the standard and quality of such goods and services prior to the Closing. If, in the sole reasonable discretion of Harsco, such standard or quality is not being met, Harsco shall have the right to immediately terminate the license to such Intellectual Property.

(d) Upon the expiration or termination of the license granted pursuant to <u>Section 7.5(c)(i)</u>, Buyer shall, and shall cause the Target Entities to, and shall instruct its third-party sublicensees to, refrain from using the Harsco Designations. Without limiting the generality of the preceding sentence, Buyer shall, and shall cause the Target Entities to, remove or render illegible all references to the Harsco Designations appearing on any sales, marketing, advertising, shipping and related materials (including websites and stationary) used in connection with goods and services owned or offered by Buyer or Target Entities. To the extent any references to the Harsco Designations on such materials cannot be removed or rendered illegible, Buyer shall, and shall cause the Target Entities to, destroy such materials as soon as practicable after the expiration or termination of such license. Buyer covenants that it and its Subsidiaries (including the Target Entities) shall, within one (1)-year after the Closing, change any applicable business names (whether registered or otherwise), including making any applicable filings with the appropriate authorities, which incorporate a Harsco Designation to a name which does not incorporate a Harsco Designation.

(e) Buyer and its Subsidiaries (including the Target Entities) may (\underline{x}) sub-license the license set forth in <u>Section 7.5(c)(i)</u> to any Affiliate, sales agent, distributor or other Person in the ordinary course of the Business, consistent with past practice, and (\underline{y}) assign or sub-license the license set forth in <u>Section 7.5(c)(ii)</u> to any Affiliate, or in connection with a merger, reorganization, or sale of all, or substantially all, of any of the businesses to which such license relates, so long as: (\underline{i}) the assigning or sublicensing party provides Harsco with prompt written notice of such transaction and (\underline{ii}) the assignee or sublicensee cannot extend the benefits of such license to its other businesses, <u>provided</u> that no such assignment or sublicense pursuant to this <u>Section 7.5(e)</u> shall relieve the assigning or sublicensing party of its obligations hereunder.

(f) All rights not expressly granted by a party hereunder or under the Ancillary Agreements are reserved to such

party.

(g) Regarding any Domain Names containing any of the Harsco Designations that are used as of the Closing Date by the Business (the "<u>Redirected Domains</u>"), Harsco shall, and shall cause its Affiliates to, for the one (1)-year period from the Closing Date, use such Redirected Domains solely to direct or redirect Internet traffic (i) at Buyer's or any of its Subsidiaries' request, to one or more Internet websites designated and controlled by Buyer or its Subsidiaries or (ii) if Buyer or its Subsidiaries do not request such redirection, to an inactive web page. After such one (1) -year period, if Harsco or its Affiliates continue to maintain or renew any of the Redirected Domains, Harsco and its Affiliates will use such Domain Name(s) solely to direct or redirect Internet traffic to an inactive page.

ARTICLE VIII

INDEMNIFICATION

8.1 Indemnification by CD&R Investor

Subject to the other terms, conditions and limitations of this Agreement (including the provisions of <u>ARTICLE V</u> and <u>Sections 8.4</u>, <u>8.6</u> and <u>8.7</u>), from and after the Closing, CD&R Investor agrees to indemnify Harsco and its Affiliates and its and their officers, directors, employees, agents, successors and assigns (the "<u>Harsco Indemnified Parties</u>") against, and to pay and to hold Harsco Indemnified Parties harmless from, all Losses suffered or incurred by any of Harsco Indemnified Parties to the extent arising from or as a result of:

(a) the failure of any representation or warranty made by CD&R Investor in <u>ARTICLE III</u> of this Agreement (other than <u>Section 3.10</u>) to be true and correct on and as of the date of this Agreement or on and as of the Closing Date (or with respect to representations and warranties that are made as of a specific date, the failure of such representations and warranties to be true and correct as of such date); and

(b) any failure by CD&R Investor to perform or comply with any of its covenants (other than <u>Section 4.19(c)</u>) or agreements in this Agreement.

8.2 Indemnification by Harsco

Subject to the other terms, conditions and limitations of this Agreement (including the provisions of <u>ARTICLE V</u> and <u>Sections 8.4, 8.6</u> and <u>8.7</u>), from and after the Closing, the Sellers, on a joint and several basis, agree to indemnify (\underline{x}) in the case of the indemnification under <u>Sections 8.2(a), 8.2(b), 8.2(d), 8.2(e), 8.2(f), 8.2(g), 8.2(h)</u> or <u>8.2(i)</u>, Buyer and its Subsidiaries (including the Target Entities) and its and their officers, directors, agents, successors and assigns (the "<u>Buyer Indemnified Parties</u>") and (\underline{y}) in the case of the indemnification under <u>Section 8.2(c)</u>, CD&R Investor and its officers, directors, employees, agents, successors and assigns (the "<u>CD&R Indemnified Parties</u>") against, and to pay and to hold the Buyer Indemnified Parties or the CD&R Indemnified Parties, as applicable, harmless from, all Losses suffered or incurred by any of the Buyer Indemnified Parties or the CD&R Indemnified Parties, as applicable, to the extent arising from or as a result of:

(a) the failure of any representation or warranty made by Harsco in <u>ARTICLE II</u> of this Agreement (other than <u>Sections 2.1</u> and <u>2.2</u>, and <u>2.9</u> (such <u>Section 2.9</u> shall be governed by <u>ARTICLE V</u>)) to be true and correct on and as of the date of this Agreement or on and as of the Closing Date (or with respect to representations and warranties that are made as of a specific date, the failure of such representations and warranties to be true and correct as of such date);

(b) any failure by any Seller or, with respect to any pre-Closing period, any Target Entity, to perform or comply with any covenant or agreement in this Agreement (other than any covenant or agreement in this Agreement relating to Taxes, which shall be governed by <u>ARTICLE V</u>);

(c) the failure of any representation or warranty made by Harsco in <u>Sections 2.1</u> or <u>2.2</u> of this Agreement to be true and correct on and as of the date of this Agreement or on and as of the Closing Date (or with respect to representations and warranties that are made as of a specific date, the failure of such representations and warranties to be true and correct as of such date) or any other breach of

this Agreement by Harsco or any other Seller, or, with respect to any pre-Closing period, any Target Entity under this Agreement that causes a loss to be suffered by a CD&R Indemnified Party, but not Buyer and its Subsidiaries (such as a diminution in value); <u>provided</u>, that for the avoidance of doubt the limitations set forth in <u>Section 8.5(c)</u> shall apply hereto;

(d) the consummation of the Pre-Closing Transactions (<u>provided</u> that such indemnity shall exclude Losses arising out of events occurring after the Closing Date, including changes in Law or accounting rules which Harsco had no reasonable expectation of at the time of such consummation) or the failure by Harsco or its Affiliates to take or complete any of the steps set forth in the Pre-Closing Transactions, in each case, measured as of the Closing Date (but excluding any matter to the extent indemnification with respect thereto is provided pursuant to <u>ARTICLE V</u>);

(e) the matters set forth on <u>Schedule 8.2(e)(i)</u> and, with respect only to Losses incurred and claims made prior to the third anniversary of the Closing Date, except to the extent such Losses are caused by Buyer's breach of its obligations under <u>Section 4.23</u>, the matters set forth on <u>Schedule 8.2(e)(ii)</u>;

(f) (i) any Excluded Target Liabilities (including, for the avoidance of doubt, the operations and Liabilities of Harsco and the Non-Company Affiliates before or after the Closing (except to the extent such Liabilities arise out of the Business)), other than to the extent indemnification with respect thereto is provided pursuant to <u>ARTICLE V</u>) and (ii) any restructuring liabilities in excess of the liabilities used to determine the Restructuring Expenses;

(g) any businesses discontinued, or disposed of, by any Target Entity or any Asset Seller prior to the Closing (but only to the extent assets, liabilities or employees of such operations were not transferred or assigned to, and utilized in connection with, continuing operations of the Business as of the date hereof), including, for the avoidance of doubt, asbestos, mesothelioma and any other latent disease claims as discussed in the Towers Watson report dated January 22, 2013;

(h) (i) any Release of Hazardous Substances at, on, to, under or from the Owned Real Property or Leased Real Property prior to the Closing Date; (ii) any Third Party Claim alleging liability for a Release of Hazardous Substances prior to the Closing Date at, on, to, under or from any property owned, leased or operated by the Business or formerly owned, leased or operated by the Business, any Target Entity or any of their predecessors or Affiliates; (iii) any Third Party Claim alleging liability for actions taken by any Target Entity or any of their predecessors or Affiliates on or prior to the Closing Date in connection with the transportation, recycling, storage, treatment, handling, use or disposal of Hazardous Substances at a property not owned or leased by any Target Entity or any of their predecessors or Affiliates; and (iv) any violations of, or non-compliance with, any applicable Environmental Laws (or Permits required thereunder) by the Business, any Target Entity or any of their predecessors or Affiliates on or prior to the Closing Date; and

(i) any Liabilities or claims relating to or arising from:

(i) the Harsco Employees Pension Plan and any other single-employer defined-benefit pension plan maintained or sponsored in the United States by Harsco or any of its ERISA Affiliates;

(ii) the Harsco Pension Scheme or any current or former Business Employee's membership in any other defined benefit pension plan maintained in the UK at any time prior to the Closing Date;

(iii) withdrawal liability under any Infrastructure Multiemployer Plan contributed to by Harsco or any of its ERISA Affiliates that is assessed and outstanding as of the Closing Date, other than withdrawal liability imposed in connection with the transactions contemplated under this Agreement, including <u>Section 4.6(m)</u> hereof; and

(iv) with respect to the Business Benefit Plans listed on Schedule 8.2(i)(iv) in which current or former Business Employees participated prior to the Closing Date and for which Buyer or any of its Affiliates assumes liability, an amount determined, as of the Closing Date, with respect to all such plans on a net cumulative basis, equal to the net aggregate present value, if any, of the accumulated benefit obligation ("ABO") for all such plans, less the net aggregate fair value of plan assets, each as determined under GAAP consistently applied, with such obligation calculated (A) utilizing the actuarial assumptions and methods applicable to each such plan for the most recent Financial Statements, subject to (B) applicable interest and discount rate assumptions and any currency exchange rate updated to the Closing Date on a basis consistent with GAAP, as reasonably agreed by the Parties (for such plans, the "Pension Obligation"). The Pension Obligation shall be satisfied by payments from Harsco to the relevant Buyer (or its successor) that, after giving effect to the Closing, owns the Target Entity that maintains such Business Benefit Plan (the "Non-U.S. Pension Indemnitees"), in five substantially equal installments paid on the first, second, third, fourth and fifth anniversaries of the Closing Date. The Parties agree that such Pension Obligation shall be allocated in respect of the Non-U.S. Pension Indemnitees in proportion of the respective unfunded ABO of such Business Benefit Plans and that such payments shall be treated by the Parties as an adjustment to the purchase price in respect of the sale transactions described on Exhibit G (and shall be allocated among such transactions in accordance with Section 1.10).

Notwithstanding anything in this Agreement to the contrary, solely for determining whether there is a failure of any representation or warranty to be true and correct for purposes of <u>Section 8.2(a)</u>, except with respect to the representations and warranties set forth in <u>Section 2.3(c)</u>, <u>Section 2.6(a)</u> and the first sentence of <u>Section 2.10</u>, no effect shall be given to any qualification as to "materiality" or "Material Adverse Effect" and phrases of similar import.

8.3 Indemnification by Buyer

Subject to the other terms, conditions and limitations of this Agreement (including the provisions of <u>ARTICLE V</u> and <u>Sections 8.4</u>, <u>8.6</u> and <u>8.7</u>), from and after the Closing, the Buyers agree to indemnify the Harsco Indemnified Parties against, and to pay and to hold the Harsco Indemnified Parties harmless from, all Losses suffered or incurred by any of the Harsco Indemnified Parties to the extent arising from or as a result of:

(a) the operations and Liabilities of the Business, Buyer and the Target Entities, including the Assumed Target Liabilities, before or after the Closing (but excluding Losses in respect of Taxes, which shall be governed by <u>ARTICLE V</u>); <u>provided</u>, <u>however</u>, that this <u>Section 8.3(a)</u> shall not obligate Buyer to indemnify any Harsco Indemnified Party for any Loss (<u>i</u>) for which a Buyer Indemnified Party or CD&R Indemnified Party would be entitled to indemnification under <u>Sections 8.2(a)</u>, <u>8.2(b)</u>, <u>8.2(c)</u>, <u>8.2(f)</u>, <u>8.2(g)</u>, <u>8.2(g)</u>, <u>8.2(i)</u> or (<u>ii</u>) to the extent suffered as a result of any Harsco Indemnified Parties' direct or indirect ownership of Buyer or any Target Entity following the Closing; and

(b) any failure by the Buyers to perform or comply with any covenant or agreement in this Agreement.

8.4 Limitations on Indemnification

(a) The representations and warranties set forth in this Agreement (other than those in Section 2.9 (Taxes)), and the right to assert a claim for indemnification with respect thereto pursuant to Section 8.1(a) or 8.2(a) shall survive the Closing until the 18-month anniversary of the Closing Date and shall thereafter be of no further force or effect; provided that (i) the representations and warranties in Sections 2.1, 2.2, 2.19, 3.1 and 3.5 (the "Fundamental Representations") and the right to assert a claim for indemnification with respect thereto pursuant to Section 8.1(a) or 8.2(a) shall survive the Closing indefinitely, (ii) the representations and warranties in <u>Section 2.9(d)</u>, (g), (i) and (j) and the right to assert a claim for indemnification with respect thereto pursuant to <u>Section 5.1(a)(iii)</u> shall survive the Closing until sixty (60) days following the expiration of the applicable statute of limitations for assessments, after giving effect to any waiver, mitigation or extension of such period, and (iii) the representations and warranties in <u>Section 2.16</u> shall survive until the third anniversary of the Closing. The covenants and agreements set forth in this Agreement, and the right to assert a claim for indemnification with respect thereto shall survive the Closing for a period of 24 months, provided that those covenants and agreements that by their terms contemplate performance in whole or in part after the Closing shall remain in full force and effect until thirty (30) days after the date by which such covenant or agreement is required to be performed. If a party delivers an indemnification notice to the other party before the expiration of a representation or warranty or covenant or agreement or indemnity, then the right to assert a claim for indemnification with respect to the applicable representation or warranty or covenant or agreement or indemnity shall survive until, but only for purposes of, the resolution of the matter covered by such notice. No claim for indemnification under <u>Sections 8.1(a)</u>, <u>8.1(b)</u>, <u>8.2(a)</u>, <u>8.2(b)</u> or <u>8.3(b)</u> may be made after the expiration of the applicable survival period.

(b) Notwithstanding anything to the contrary contained herein, except with respect to ERISA Affiliate Liabilities and the Fundamental Representations, an Indemnified Party shall not be entitled to indemnification under <u>Section 8.1(a)</u> or <u>8.2(a)</u> until such party (together with its Affiliates) shall have incurred (A) with respect to any particular claim or series of related claims based on the same or related facts, an indemnifiable Loss in excess of \$75,000 (such amount, the "<u>Minimum Per Claim Amount</u>"), and (B) as to all claims, indemnifiable Losses in excess of an amount equal to \$4,875,000 in the aggregate (the "<u>Deductible</u>").

(c) Except with respect to ERISA Affiliate Liabilities and the Fundamental Representations, an Indemnifying Party shall be obligated to indemnify, defend or hold harmless any Indemnitee with respect to the matters covered by <u>Section 8.1(a)</u> or <u>8.2(a)</u> only to the extent the aggregate amount of claims to be payable by such Indemnitee exceeds the Deductible and then only to the extent of such amount that is in excess of the Deductible; <u>provided</u> that no individual claim or series of related claims based on the same or related facts shall be so asserted unless and until the aggregate amount that would be payable pursuant to each such claim or series of related claims based on the same or related facts exceeds an amount equal to the Minimum Per Claim Amount (it being understood that any such individual claims or series of related claims for amounts less than the Minimum Per Claim Amount shall be ignored in determining whether the Deductible has been exceeded). Under no circumstance shall Harsco's aggregate obligation to provide indemnification for matters covered by <u>Section 8.2(a)</u> exceed an amount equal to \$60,000,000.

(d) The Buyer Indemnified Parties shall have no right to indemnification hereunder with respect to any Losses to the extent that such Losses (<u>A</u>) arise out of any invasive environmental sampling conducted by or on behalf of a Buyer Indemnified Party, or with Buyer's express consent, unless such sampling is affirmatively required of such Buyer Indemnified Party by applicable Environmental Laws, a written, unsolicited Governmental Order or unsolicited demand by a Governmental Body or with

respect to the Leased Real Property, specifically required under an applicable lease agreement; (B) arise out of any environmental investigation, assessment or remediation or other action taken by or on behalf of a Buyer Indemnified Party to address any contamination to the extent that such activity is not reasonably appropriate to comply with the minimum standards required under Environmental Law applicable to the use of such property as of the Closing Date, taking into account the availability of any engineering controls, deed restrictions and other risk-based standards; (C) arise out of any change in use of the Owned Real Property or Leased Real Property after the Closing Date to a non-industrial use; (D) arise at any Owned Real Property or Leased Property after the sale, transfer, lease or closure of such Owned Real Property or after any transfer, sublease, closure or termination of lease of such Leased Real Property; (<u>E</u>) are indemnifiable under <u>Section 8.2(h)</u> but arise at any time after the second anniversary of the first date on which the CD&R Investor and its Affiliates own, in the aggregate, less than twenty percent (20%) of the equity of the Company, directly or indirectly, or (\underline{F}) arise out of any action taken by, or on behalf of, a Buyer Indemnified Party that exacerbates any environmental condition, including the presence of Hazardous Substances at, on or under the Owned Real Property or Leased Real Property. In the event that a Buyer Indemnified Party believes that it is required to conduct any investigation, remediation or other action that would give rise to a claim for indemnification pursuant to Section 8.2(a) or Section 8.2(h) (the "Required Remediation Activity"), the Buyer Indemnified Party shall first provide Harsco with reasonable notice and the opportunity to conduct such Required Remediation Activity at Harsco's sole cost and expense. Harsco shall notify the Buyer Indemnified Party in writing within ten (10) business days of receipt of such notice whether it shall conduct such Required Remediation Activity. In the event that Harsco declines to undertake such Required Remediation Activity, then the Buyer Indemnified Party shall be entitled to conduct such Required Remediation Activity and entitled to such indemnification as otherwise provided herein. The Buyer Indemnified Parties shall not be entitled to indemnification for any Required Remediation Activity to the extent that Harsco is not first provided with the opportunity to conduct such Required Remediation Activity, except to the extent such Required Remediation Activity is necessary to comply with Environmental Law and notice to Harsco is impractical.

(e) No party shall be entitled to indemnification pursuant to this <u>ARTICLE VIII</u>, to sue for damages or to assert any other right or remedy with respect to any Loss, cause of action or other claim to the extent such Loss resulted from the fact that such party or any of its Affiliates has taken action (or caused action to be taken) the primary purpose of which is to accelerate the time period in which such matter is asserted or payable in order to ensure the applicability of indemnification pursuant to this <u>ARTICLE VIII</u>.

8.5 Sole Remedy.

(a) Each of Harsco and CD&R Investor acknowledges and agrees that its sole and exclusive post-Closing remedy with respect to any and all claims (arising at law, in equity, under contract, in tort or otherwise) relating to this Agreement and the transactions contemplated hereby (other than (<u>x</u>) claims arising from fraud, (<u>y</u>) any disputes relating to Closing Net Working Capital, Closing Cash, Closing Indebtedness and/or Cap Ex Shortfall, which shall be governed by <u>Section 1.9</u> and (<u>z</u>) claims arising under any Ancillary Agreement) shall be pursuant to the indemnification provisions set forth in <u>ARTICLE V</u> and this <u>ARTICLE VIII</u>. In furtherance of the foregoing, each of Harsco and CD&R Investor hereby waives on its own behalf and on behalf of its Affiliates (including in the case of CD&R Investor, Buyer and the Target Entities following the Closing), from and after the Closing, to the fullest extent permitted under Law, any and all claims (other than claims arising from fraud and other than any disputes relating to Closing Net Working Capital, Closing Cash, Closing Indebtedness and/or Cap Ex Shortfall, which shall be governed by <u>Section 1.9</u>) it may have against the other Parties hereto or any of their Affiliates arising under or based upon this Agreement or any document or certificate delivered in

connection herewith (other than any Ancillary Agreement or any document executed in connection with the Transactions), except pursuant to the indemnification provisions set forth in <u>ARTICLE V</u> and this <u>ARTICLE VIII</u>. The foregoing notwithstanding, nothing in this <u>Section 8.5</u> shall limit or restrict the ability or right of CD&R Investor, Buyer or Harsco to seek injunctive relief or other equitable relief for any breach or alleged breach of any provision of this Agreement (subject to any applicable limitations set forth in <u>Sections 9.2</u> and <u>9.3</u>); <u>provided</u> that any procedures in respect of, and limitations on, Losses or Liabilities in <u>ARTICLE V</u> and this <u>Article VIII</u> shall in no event be diminished or circumvented by such relief.

(b) Notwithstanding any other provision in this Agreement to the contrary, there shall be no right to indemnification hereunder for any Losses if and to the extent expressly taken into account in the final determination of Closing Net Working Capital (except with respect to reserves, which are governed by Section 8.6(a)), Closing Cash, Closing Indebtedness or Closing Cap Ex Shortfall pursuant to <u>Section 1.9</u>.

(c) Each party shall take, and shall cause its Affiliates to take, actions to mitigate Losses, to the extent required by applicable Law, upon and after becoming aware of any event which could reasonably be expected to give rise to a claim for indemnification hereunder.

(d) Notwithstanding anything herein to the contrary, in no event shall any Indemnifying Party be liable for any indirect, special, consequential, exemplary or punitive damages except (<u>i</u>) in the event of fraud or willful misconduct or (<u>ii</u>) to the extent awarded against an Indemnitee pursuant to a claim by a third party.

8.6 Computation of Indemnity Payments

(a) The amount payable under this <u>ARTICLE VIII</u> in respect of any Loss shall be calculated net of (<u>A</u>) with respect to any Loss under <u>Section 8.2(a)</u> or <u>Section 8.2(e)</u>, any reserves set forth in the Final Closing Statement (including the balance sheet delivered in connection therewith) that are taken into account in the determination of the Adjustment Amount and (<u>B</u>) any insurance proceeds or other amounts under indemnification agreements with third parties actually received by the Indemnitee on account thereof; <u>provided</u> in each case that the Indemnitee shall (and shall cause its Affiliates to) use commercially reasonable efforts to seek full recovery under all insurance policies and/or indemnification agreements covering any indemnifiable Loss to the same extent as the Indemnitee would if such Loss were not subject to indemnification under this Agreement. In the event that an insurance or other recovery is made at any time after an indemnification payment by the Indemnitor has been made with respect to any indemnified Loss, then, to the extent of the amount of such indemnification payment, a refund equal to the aggregate net amount of the recovery (less the costs of recovery to the Indemnitee) shall be made promptly to the Person or Persons that provided such indemnity payment to such Indemnitee. The party making any indemnification payment hereunder shall be subrogated to all rights of the Indemnitee in respect of any Losses indemnified by such party.

(b) To the extent the Loss (including any Tax) that gives rise to an indemnity payment pursuant to <u>ARTICLE V</u> or this <u>ARTICLE VIII</u> is included in income, the indemnitor shall increase such indemnity payment to the applicable Indemnitee to compensate for any Tax detriment to the Indemnitee that is actually realized by it prior to the end of the close of the taxable year in which the fourth anniversary of such indemnity payment occurs (the aggregate of such indemnity payment and such increase, an "<u>Indemnity Payment</u>"). To the extent the Loss that gave rise to an Indemnity Payment results in a Tax benefit to the Indemnitee that is actually realized by it prior to the end of the close of the taxable year in which the fourth anniversary of such Indemnity Payment occurs, the Indemnity Payment to the Indemnity Payment occurs, the Indemnity Payment to the

applicable indemnitor such Tax benefit (determined on a with and without basis) (a "<u>Tax Benefit Payment</u>"); provided that in no event shall the cumulative Tax benefit remitted by the Indemnitee exceed the amount of the applicable Indemnity Payment. If any such Tax benefit is subsequently disallowed or unnecessary (*e.g.*, a carryback of losses into the taxable year in which such Tax Benefit was taken into account), the applicable indemnitor shall make an appropriate reconciliation payment to the Indemnitee. Notwithstanding the foregoing provisions of this <u>Section 8.6</u>, no Tax Benefit Payment shall be required to be made in respect of any portion of an indemnity payment that is subject to the cap described in <u>Section 5.10</u> or <u>Section 8.4</u>.

8.7 Procedures for Indemnification

(a) <u>Procedures</u>.

(i) Any Person making a claim for indemnification under <u>Section 8.1</u>, <u>Section 8.2</u> or <u>Section 8.3</u> (an "<u>Indemnified Party</u>") shall notify the party against whom indemnification is sought (an "<u>Indemnifying Party</u>") of the claim in writing promptly after receiving notice of any action, lawsuit, proceeding, investigation, demand or other claim against the Indemnified Party by a third party (a "<u>Third Party Claim</u>"), describing the Third Party Claim, the amount thereof (if known and quantifiable) and the basis thereof in reasonable detail; <u>provided</u> that the failure to so notify an Indemnifying Party shall not relieve the Indemnifying Party of its obligations hereunder except to the extent that (and only to the extent that) such failure shall have caused the indemnifiable Losses to be greater than such Losses would have been had the Indemnified Party given the Indemnifying Party prompt notice hereunder.

(ii) Any Indemnifying Party shall be entitled to participate in the defense of such Third Party Claim at such Indemnifying Party's expense, and at its option shall be entitled to assume the defense thereof; provided that (i) the Indemnifying Party acknowledges in writing its obligation to indemnify the Indemnified Party for all Losses related to such Third Party Claim and (ii) the Indemnified Party shall be entitled to participate in the defense of such Third Party Claim and to employ counsel of its choice for such purpose (provided that the fees and expenses of such separate counsel shall be borne by the Indemnified Party and shall not be recoverable from such Indemnifying Party under this <u>ARTICLE VIII</u>). Notwithstanding the foregoing, if the Indemnified Party shall have determined in good faith and upon advice of counsel that (a) an actual or likely conflict of interest makes representation of the Indemnifying Party and the Indemnified Party by the same counsel inappropriate or (b) the defendants in, or targets of, any such action or proceeding include both the Indemnified Party and an Indemnifying Party, and the Indemnified Party's counsel shall have reasonably concluded that there may be legal defenses available to it or to other Indemnified Parties which are different from or additional to those available to the Indemnifying Party (in which case the Indemnifying Party shall not have the right to direct the defense of such action or proceeding on behalf of the Indemnified Party), then, in each case, the Indemnified Party may, upon notice to the Indemnifying Party, engage one (1) separate counsel, and the reasonable fees and expenses of such separate counsel shall be borne by the Indemnifying Party to the extent the Third Party Claim is indemnifiable hereunder.

(iii) Upon assumption of the defense of any such Third Party Claim by the Indemnifying Party, the Indemnified Party will not pay, or permit to be paid, any part of the Third Party Claim, unless the Indemnifying Party consents in writing (such consent not to

be unreasonably withheld or delayed) to such payment or unless a final judgment from which no appeal may be taken by or on behalf of the Indemnified Party is entered against the Indemnified Party for such Liability. Notwithstanding anything to the contrary herein, the Indemnifying Party shall not compromise or settle, or admit any Liability with respect to, any Third Party Claim without the prior written consent of the Indemnified Party (which consent shall not be unreasonably withheld or delayed), unless (\underline{x}) the relief consists solely of money damages (all of which the Indemnifying Party shall pay), (\underline{y}) such settlement or compromise includes a provision whereby the plaintiff or claimant in the matter releases the Indemnified Party from all Liability with respect thereto and (\underline{z}) such settlement or compromise does not include any admission of fault or wrongdoing on the part of the Indemnified Party.

(iv) In all cases with respect to Third Party Claims, the Parties shall provide reasonable cooperation to each other in defense of such Third Party Claims, including by making employees, information and documentation reasonably available (including for purposes of fact finding, consultation, interviews, depositions and, if required, as witnesses) and providing such information, testimony and access to their books and records, during normal business hours and upon reasonable notice, in each case as shall be reasonably necessary in connection with the contest or defense.

(v) If the Indemnifying Party shall not reasonably assume the defense of any such Third Party Claim, or fails to prosecute or withdraws from the defense of any such Third Party Claim, the Indemnified Party may defend against such matter, at the Indemnifying Party's expense, in a manner consistent with the above provisions regarding conduct of the defense by the Indemnified Party.

(b) In the event that any Harsco Indemnified Party, CD&R Indemnified Party or Buyer Indemnified Party alleges that it is entitled to indemnification hereunder, and that its claim is covered under more than one provision of this <u>ARTICLE VIII</u>, such party or Affiliates shall be entitled to elect the provision or provisions under which it may bring a claim for indemnification.

(c) A claim for indemnification for any matter not involving a Third Party Claim may be asserted by notice to the party from whom indemnification is sought.

8.8 <u>Treatment of Indemnification Payments</u>. To the fullest extent permitted by applicable Law, for all purposes (including Tax purposes), the Parties shall treat any adjustments made pursuant to <u>Section 1.9</u> and any payment made under <u>Sections 5.1, 8.1, 8.2</u> or <u>8.3</u> as an adjustment to the Purchase Price.

ARTICLE IX

TERMINATION AND WAIVER

9.1 <u>Termination</u>

This Agreement may be terminated:

(a) at any time prior to the Closing by the mutual written consent of Harsco and CD&R Investor;

(b) by CD&R Investor, if any Seller breaches or fails to perform in any respect any of its representations, warranties or covenants contained in this Agreement and such breach or failure to perform (<u>A</u>) would give rise to the failure of a condition set forth in <u>Section 6.2</u>, (<u>B</u>) cannot be or has not been cured within thirty (30) days following delivery of written notice of such breach or failure to perform and (<u>C</u>) has not been waived in writing by CD&R Investor;

(c) by Harsco, if (<u>x</u>) Buyer or CD&R Investor breaches or fails to perform in any respect any of its representations, warranties or covenants contained in this Agreement (including, for the avoidance of doubt, Buyer's or CD&R Investor's failure to perform its obligations under <u>Section 4.19(c)</u>) and such breach or failure to perform (<u>i</u>) would give rise to the failure of a condition set forth in <u>Section 6.1</u>, (<u>ii</u>) cannot be or has not been cured within thirty (30) days following delivery of written notice of such breach or failure to perform and (<u>iii</u>) has not been waived in writing by Harsco, or (<u>y</u>) (<u>i</u>) all of the conditions set forth in <u>Section 6.2</u> have been satisfied or waived (other than those conditions that by their terms are to be satisfied by actions taken at the Closing) and (<u>ii</u>) Harsco has given notice to CD&R Investor in writing that it is prepared and able to consummate the Closing (assuming the Debt Financing would be available at the Closing and the conditions set forth in <u>Section 6.1</u> that by their terms are to be satisfied at the Closing would be satisfied by actions taken at the Closing) and (<u>iii</u>) CD&R Investor fails to consummate the transactions contemplated by this Agreement on the date the Closing should have occurred pursuant to <u>Section 1.3</u>; and

(d) by CD&R Investor or by Harsco, upon written notice to the other party, if the Closing has not occurred on or prior to (<u>i</u>) January 24, 2014, if the Marketing Period has not commenced prior to, and would be continuing on or any day after, such date or (<u>ii</u>) otherwise, February 14, 2014 (the date in either (i) or (ii), as applicable, the "<u>Outside Date</u>").

Notwithstanding anything else contained in this Agreement, the right to terminate this Agreement under this <u>ARTICLE IX</u> (other than pursuant to <u>Section 9.1(d)</u>) shall not be available to any party (<u>a</u>) that is in material breach of its representations, warranties, covenants or other agreements set forth herein or (<u>b</u>) whose failure to fulfill its obligations or to comply with its covenants under this Agreement has been the primary cause of, or primarily resulted in, the failure to satisfy any condition to the obligations of the other party hereunder.

9.2 Effect of Termination

(a) In the event of termination in accordance with <u>Section 9.1</u>, this Agreement will forthwith become void and there will be no Liability on the part of any party hereto, except that <u>Sections 2.19</u> and <u>3.5</u> relating to brokers' fees, <u>Section 4.5</u> relating to confidentiality, <u>Sections 4.11(b)</u> and <u>4.11(c)</u>, this <u>Section 9.2</u>, <u>ARTICLE X</u> and any corresponding definitions set forth in <u>ARTICLE XI</u>, shall survive termination; <u>provided</u>, <u>however</u>, that except as set forth below, nothing herein shall relieve any party hereto from Liability for any willful breach hereof prior to such termination.

(b) In the event that this Agreement is terminated (<u>I</u>) pursuant to <u>Section 9.1(c)(x)</u> and, at such time, (<u>x</u>) Harsco has confirmed in a written notice to CD&R Investor that it was prepared to consummate the Closing at such time (assuming the Debt Financing would be available at the Closing and the conditions set forth in <u>Section 6.1</u> that by their terms are to be satisfied by actions taken at Closing would be satisfied or waived at the Closing), (<u>y</u>) all the conditions set forth in <u>Section 6.2</u> were satisfied or waived (other than those conditions that by their terms are to be satisfied by actions taken at Closing and those conditions which fail to be satisfied as a result of CD&R Investor's or Buyer's breach) and (<u>z</u>) but for its breach or failure to perform, CD&R Investor was required to complete the Closing on the Closing Date pursuant to <u>Section 1.3</u>, or (<u>ii</u>) <u>Section 9.1(c)(y)</u>, then CD&R Investor shall pay Harsco an amount

equal to \$20,000,000, <u>provided</u> that notwithstanding anything to the contrary set forth herein, CD&R Investor shall not be required to pay the Termination Fee with respect to the foregoing clause (ii) if the Financing was not available as a result of the failure of the representations and warranties of Harsco set forth in clauses (ii), (iii) and (iv) of the last sentence of <u>Section 2.5(a)</u> to be true and correct in all respects (subject to the disclosures set forth in <u>Schedule 2.5</u>) or (<u>II</u>) as a result of the Debt Financing being unavailable other than as a result of a breach by Harsco and a termination fee is payable pursuant to <u>Section 8.3(b)</u> of the Brand Purchase Agreement, and no Termination Fee is payable pursuant to clause (I), then CD&R Investor shall reimburse Harsco for all of Harsco's and its Subsidiaries' reasonable and documented out-of-pocket fees and expenses (including reasonable attorney's fees) incurred in connection with this Agreement and the transactions contemplated hereby up to an amount not to exceed \$10 million (the amount set forth in clause (I) or (II), as applicable, the "<u>Termination Fee</u>").

(c) In the event the Termination Fee is payable, such fee will be paid to Harsco by CD&R Investor in immediately available funds within three (3) Business Days after the date of termination of this Agreement. Solely for purposes of establishing the basis for the amount thereof, and without in any way increasing the amount of the Termination Fee or expanding the circumstances in which the Termination Fee is to be paid, it is agreed that the Termination Fee is liquidated damages, and not a penalty, and the payment of the Termination Fee in the circumstances specified herein is supported by due and sufficient consideration. While Harsco may pursue both a grant of specific performance under <u>Section 9.3(b)</u> and the payment of the Termination Fee under this <u>Section 9.2(c)</u>, under no circumstances shall Harsco be permitted or entitled to receive both a grant of specific performance which would result in consummation of the Closing and monetary damages in connection with this Agreement or the termination of this Agreement, including all or any portion of the Termination Fee.

(d) Harsco and CD&R Investor acknowledge and agree that the agreements contained in this <u>Section 9.2</u> are an integral part of the transactions contemplated hereby, and that without these agreements, the Parties would not have entered into this Agreement.

(e) Notwithstanding anything to the contrary in this Agreement, if CD&R Investor fails to effect the Closing when required by Section 1.3 for any or no reason or otherwise prior to the Closing breaches this Agreement or fails to perform hereunder (in any case, whether willfully, intentionally, unintentionally or otherwise), then (i) except for the right of Harsco to seek an injunction, specific performance or other equitable relief as and only to the extent expressly permitted by Section 9.3(b) and except for the right of Harsco to bring a Pre-Closing Damages Proceeding pursuant to the third sentence of this Section 9.2(e), Harsco's and its Affiliates' sole and exclusive remedy (whether at law, in equity, in contract, in tort or otherwise) against any of the CD&R Investor Parties for any pre-Closing breach, loss or damage shall be to terminate this Agreement as provided in <u>Section 9.1(c)(x)</u> or 9.1(c)(y), and receive payment of the Termination Fee, in each case to the extent provided by Section 9.2(b), or with respect to the obligation to pay the Termination Fee, the Guarantee, as applicable, and (ii) upon payment of the Termination Fee, as provided in the immediately foregoing clause (i), none of the CD&R Investor Parties will have any liability to Harsco or any of its Affiliates or any other Person, whether at Law or equity, in contract in tort or otherwise arising from or in connection with pre-Closing breaches by CD&R Investor of its representations, warranties, covenants and agreements contained in this Agreement or arising from any claim or cause of action that Harsco or any of its Affiliates may have relating to pre-Closing matters under this Agreement, including for a breach of <u>Section 1.3</u> hereof, and no Person will have any rights or claims against any of the CD&R Investor Parties relating to any such matters. Harsco agrees to cause any Proceeding pending in connection with this Agreement or any of the transactions contemplated hereby (including any Proceeding related to the Financing, the Equity Financing Commitments, the Debt Financing Commitment or the Guarantee) by Harsco or any of its Affiliates and any of their respective

former, current or future directors, officers, employees, agents, general or limited partners, managers, members, stockholders or Affiliates (collectively, the "Harsco Parties") against any of the CD&R Investor Parties to be dismissed with prejudice promptly after payment of the Termination Fee. Notwithstanding clause (i) of the second preceding sentence, the Parties agree that in the event that (x) Harsco has terminated this Agreement pursuant to Section 9.1(c)(x) because of the willful and intentional failure by CD&R Investor to perform any of its covenants contained in this Agreement, (y) the Termination Fee is not payable and (z) Harsco has sought an injunction, specific performance and/or other equitable relief with respect to the acts or omissions that gave rise to Harsco's ability to terminate this Agreement pursuant to Section 9.1(c)(x) as contemplated by clause (x) of this sentence and such injunction, specific performance and/or other equitable relief was determined by a court of competent jurisdiction to not be available, then, Harsco shall be entitled to seek monetary damages against CD&R Investor and/or under the Guarantee with respect to the acts or omissions that gave rise to Harsco's ability to terminate this Agreement pursuant to Section 9.1(c) as contemplated by clause (x) of this sentence (a "<u>Pre-Closing Damages Proceeding</u>"); provided that the monetary damages payable by CD&R Investor and Guarantor under all Pre-Closing Damages Proceedings shall not exceed, in the aggregate, the amount of the Termination Fee. If the Closing has not occurred, in no event shall any of the Harsco Parties seek or permit to be sought on behalf of any Harsco Party any damages from, or otherwise bring any Proceeding against, any of the CD&R Investor Parties in connection with pre-Closing breaches of this Agreement or any of the pre-Closing transactions contemplated hereby (including any Proceeding related to the Financing, the Equity Financing Commitments, the Debt Financing Commitment or the Guarantee), other than a Pre-Closing Damages Proceeding, when permitted and to the extent set forth in this <u>Section 9.2(e)</u>, or a Proceeding to recover payment of the Termination Fee when permitted and to the extent set forth in <u>Section 9.2(b)</u> or for specific performance solely under the circumstances and as specifically set forth in Section 9.3(b). In no event shall Harsco be entitled to seek the remedy of specific performance of this Agreement other than solely under the circumstances and as specifically set forth in Section 9.3(b). Nothing in this <u>Section 9.2(e)</u> shall in any way expand or be deemed or construed to expand the circumstances in which CD&R Investor or any other CD&R Investor Party may be liable under this Agreement or any of the transactions contemplated hereby (including the Financing).

(f) The Parties acknowledge and agree that in no event will CD&R Investor be required to pay the Termination Fee on more than one occasion or pay the Termination Fee if it has paid any monetary damages under a Pre-Closing Damages Proceeding.

9.3 Enforcement

(a) The Parties agree that irreparable damage would occur in the event that either party does not perform or otherwise breaches provisions of this Agreement in accordance with their specific terms and that, subject to the limitations provided in <u>Section 9.2</u> and this <u>Section 9.3</u>, any such breach of this Agreement could not be adequately compensated in all cases by monetary damages alone. The Parties acknowledge and agree that CD&R Investor shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement by Harsco or any Seller, and to enforce specifically the terms and provisions hereof, in addition to any other remedy to which it is entitled at law or in equity.

(b) Subject to the last sentence of <u>Section 9.2(c)</u>, it is acknowledged and agreed that Harsco shall be entitled to an injunction, specific performance and other equitable relief to prevent breaches of this Agreement and to enforce specifically the terms and provisions hereof (including any expense reimbursement and indemnification obligations). Notwithstanding the foregoing, Harsco shall be entitled to specific performance of CD&R Investor's obligations pursuant to the terms of this Agreement to cause the Equity Financing to be funded to fund the Transactions and to consummate the Closing only

in the event that each of the following conditions has been satisfied: (i) CD&R Investor is required to complete the Closing on the Closing Date pursuant to <u>Section 1.3</u>, (ii) the Debt Financing has been obtained or will be obtained at the Closing if the Equity Financing is funded at the Closing, (iii) CD&R Investor fails to complete the Closing in accordance with <u>Section 1.3</u> and (iv) Harsco has irrevocably confirmed that if specific performance is granted and the Equity Financing funded and the Debt Financing is obtained, and CD&R Investor otherwise complies with its obligations hereunder, then the Closing will occur.

(c) Each party hereby agrees not to raise any objections to the availability of the equitable remedy of specific performance (other than on the basis that such remedy is not available pursuant to the terms of this Agreement in respect of the particular instance in question) to prevent or restrain breaches of this Agreement by such party, and to specifically enforce the terms and provisions of this Agreement, to prevent breaches or threatened breaches of, or to enforce compliance with, the covenants and obligations of such party under this Agreement all in accordance with the terms of this <u>Section 9.3</u>. Any party seeking an injunction or injunctions to prevent breaches of this Agreement and to enforce specifically the terms and provisions of this Agreement, in each case when expressly available pursuant to the terms of this Agreement, shall not be required to provide any bond or other security in connection with such order or injunction.

(d) To the extent any party hereto brings any action prior to the Closing to enforce specifically the performance of the terms and provisions of this Agreement when available to such party pursuant to the terms of this Agreement, the Outside Date shall automatically be extended by (i) the amount of time during which such action is pending, plus five (5) Business Days, or (ii) such other time period established by the court presiding over such action.

9.4 Waiver

Any party hereto may: (a) extend the time for the performance of any of the obligations or other acts of the other party hereto; (b) waive any inaccuracies in or breaches of the representations and warranties of the other party contained herein or in any document delivered pursuant hereto or (c) waive compliance by the other party with any of the agreements or conditions contained herein. Any such extension or waiver will only be valid if set forth in an instrument in writing signed by the party to be bound thereby.

ARTICLE X

GENERAL PROVISIONS

10.1 Expenses

Except as otherwise set forth herein (including <u>Section 4.14</u>), all costs and expenses, including fees and disbursements of counsel, financial advisors and accountants, incurred in connection with this Agreement and the transactions contemplated hereby will be paid by the party incurring such costs and expenses, whether or not the Closing shall have occurred.

10.2 Notices

All notices, requests, claims, demands and other communications hereunder will be in writing and will be given or made (and will be deemed to have been duly given or made upon receipt) by delivery in

person, by courier service, by confirmed telecopy, or by registered or certified mail (postage prepaid, return receipt requested) to the Parties at the following addresses (or at such other address for a party as will be specified by like notice):

(a) if to Harsco:

Harsco Corporation 350 Poplar Church Road Camp Hill, PA 17011 Attn: General Counsel Fax: (717) 265-8153

with a copy to:

Weil, Gotshal & Manges LLP

767 Fifth Avenue

New York, NY 10153

Attn: Howard Chatzinoff

Attn: Jaclyn L. Cohen

Fax: (212) 310-8007

(b) if to CD&R Investor:

c/o Clayton, Dubilier & Rice, LLC

375 Park Avenue, 18th Floor

New York, NY 10152

Attn: Nathan Sleeper

Fax: (212) 407-5252

with a copy to:

Debevoise & Plimpton LLP

919 Third Avenue

New York, NY 10022

Attn: Kevin A. Rinker

Fax: (212) 521-7569

(c) if to Buyer:

c/o Clayton, Dubilier & Rice, LLC
375 Park Avenue, 18th Floor
New York, NY 10152
Attn: Nathan Sleeper
Fax: (212) 407-5252

and

Debevoise & Plimpton LLP 919 Third Avenue New York, NY 10022 Attn: Kevin A. Rinker Fax: (212) 521-7569

10.3 <u>Severability</u>

If any term or other provision of this Agreement is held invalid, illegal or incapable of being enforced by any rule of law or public policy, all other terms and provisions of this Agreement will nevertheless remain in full force and effect and there shall be deemed substituted for the provision at issue a valid, legal and enforceable provision that effects the original intent of the Parties as closely as possible in order that the transactions contemplated hereby be consummated as originally contemplated to the greatest extent possible.

10.4 Entire Agreement

This Agreement (including the Exhibits and Schedules hereto), the Ancillary Agreements, together with the Confidentiality Agreement, constitute the entire agreement of the Parties hereto with respect to the subject matter hereof and thereof and supersede all prior agreements and undertakings with respect to the subject matter hereof and thereof, both written and oral. Neither this Agreement nor any Ancillary Agreement shall be deemed to contain or imply any restriction, covenant, representation, warranty, agreement or undertaking of any party with respect to the transactions contemplated hereby or thereby other than those expressly set forth herein or therein, and none shall be deemed to exist or be inferred with respect to the subject matter hereof.

10.5 Assignment

This Agreement shall not be assigned by any party without the prior written consent of the non-assigning Parties; <u>provided</u>, <u>however</u>, that CD&R Investor or Buyer may transfer or assign (including by way of a pledge), in whole or in part, to one or more of its Affiliates, without the consent of any other party, its rights hereunder, but no such transfer or assignment will limit CD&R Investor's or Buyer's obligations hereunder. Upon any such permitted assignment, the references in this Agreement to CD&R

Investor or Buyer, as the case may be, shall also apply to any such assignee unless the context requires otherwise.

10.6 No Third Party Beneficiaries

Except as otherwise expressly provided in <u>ARTICLE VIII</u>, except for Clayton, Dubilier & Rice, LLC in respect of the CD&R Deal Fee and except for the D&O Indemnitees with respect to the rights granted to them under <u>Section 4.18</u>, this Agreement is for the sole benefit of the Parties hereto and their permitted assigns and nothing herein, express or implied, is intended to or will confer upon any other Person any legal or equitable right, benefit or remedy of any nature whatsoever under or by reason of this Agreement; <u>provided</u> that the Financing Sources shall be express third party beneficiaries of and have the right to enforce <u>Sections 9.2</u>, <u>10.6</u>, <u>10.7</u>, <u>10.8</u> and <u>10.10</u>.

10.7 Amendment

This Agreement may not be amended or modified except by an instrument in writing signed by each of the Parties hereto. To the extent that any amendment or modification to <u>Sections 9.2</u>, <u>10.6</u>, <u>10.7</u>, <u>10.8</u> or <u>10.10</u> is sought which is adverse to the Financing Sources, the prior written consent of the Debt Commitment Parties shall be required before such amendment or modification is rendered effective. For purposes of this <u>Section 10.7</u>, references to "Debt Commitment Parties" shall include any parties that provide Alternative Financing as contemplated by <u>Section 4.10</u>.

10.8 Governing Law; Jurisdiction

(a) This Agreement will be governed by, and construed in accordance with, the Laws of the State of New York, without regard to any principles of conflicts of law thereof that are not mandatorily applicable by Law and would permit or require the application of the Laws of another jurisdiction.

(b) Each of the Parties irrevocably agrees that any legal action or proceeding arising out of or relating to this Agreement brought by the other party or its successors or assigns shall be brought and determined in any New York State or Federal court sitting in the Borough of Manhattan in The City of New York (or, if such court lacks subject matter jurisdiction, in any appropriate New York State or Federal court), and each of the Parties hereby irrevocably submits to the exclusive jurisdiction of the aforesaid courts for itself and with respect to its property, generally and unconditionally, with regard to any such action or proceeding arising out of or relating to this Agreement and the transactions contemplated hereby. Each of the Parties agrees not to commence any action, suit or proceeding relating thereto except in the courts described above in New York, other than actions in any court of competent jurisdiction to enforce any judgment, decree or award rendered by any such court in New York as described herein. Each of the Parties hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any legal action or proceeding arising out of or relating to this Agreement or the transactions contemplated hereby, (i) any claim that it is not personally subject to the jurisdiction of the courts in New York as described herein for any reason, (ii) that it or its property is exempt or immune from jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (iii) that (A) the action or proceeding in any such court is brought in an inconvenient forum, (B) the venue of such suit, action or proceeding is improper or (C) this Agreement, or the subject matter hereof, may not be enforced in or by such courts. Nothing in this Section 10.8 shall be deemed to prevent any party from seeking to remove any action to a Federal court in the State of New York. In addition, each of the Parties hereto hereby agrees that it will not bring

or support any action, cause of action, claim, cross-claim or third party claim of any kind or description, whether in law or in equity, whether in contract or in tort or otherwise, against the Financing Sources in any way relating to this Agreement or any of the transactions contemplated by this Agreement, including any dispute arising out of or relating in any way to the Debt Financing Commitments or the performance thereof, in any forum other than a court of competent jurisdiction sitting in the borough of Manhattan in the City of New York, New York, whether a state or Federal court, and that the provisions of <u>Section 10.10</u> relating to the waiver of jury trial shall apply to any such action, cause of action, claim, cross-claim or third party claim.

10.9 Public Announcements

From and after the date hereof through (and including) the Closing Date, the timing and content of all press releases or other written public announcements regarding any aspect of this Agreement to the financial community, employees or the general public by CD&R Investor or Harsco (or Buyer) shall be subject to prior consultation with and the consent of the other (which shall not be unreasonably withheld or delayed); provided, however, that a party may, without the prior consent of any other party, issue such a press release or other similar public statement as may be required by applicable Law or any listing agreement with a securities exchange to which the disclosing party is a party, if the disclosing party has used all commercially reasonable efforts to consult with the other and to obtain the other's consent but has been unable to do so in a timely manner.

10.10 Waiver of Jury Trial

EACH PARTY ACKNOWLEDGES AND AGREES THAT ANY CONTROVERSY WHICH MAY ARISE UNDER THIS AGREEMENT OR THE DEBT FINANCING COMMITMENTS IS LIKELY TO INVOLVE COMPLICATED AND DIFFICULT ISSUES, AND THEREFORE EACH SUCH PARTY HEREBY IRREVOCABLY AND UNCONDITIONALLY WAIVES ANY AND ALL RIGHT TO TRIAL BY JURY IN ANY LEGAL PROCEEDING ARISING OUT OF OR RELATED TO THIS AGREEMENT OR THE DEBT FINANCING COMMITMENTS, THE GUARANTEE, THE ANCILLARY AGREEMENTS OR THE CONFIDENTIALITY AGREEMENT OR BY THE TRANSACTIONS CONTEMPLATED HEREBY OR THEREBY. EACH OF THE PARTIES CERTIFIES AND ACKNOWLEDGES THAT (i) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (ii) EACH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (ii) EACH PARTY MAKES THIS WAIVER VOLUNTARILY AND (iv) EACH PARTY HAS BEEN INDUCED TO ENTER INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS <u>SECTION 10.10</u>.

10.11 Disclosure Generally

Information reflected in any Schedule is not necessarily limited to matters required by this Agreement to be reflected in such Schedule. Such additional information is set forth for informational purposes and does not necessarily include other matters of a similar nature. Disclosure of such additional information shall not be deemed to constitute an acknowledgment that such information is required to be disclosed and disclosure of such information shall not be deemed to enlarge or enhance any of the representations or warranties in this Agreement or otherwise alter in any way the terms of this Agreement. Inclusion of information in any Schedule shall not be construed as an admission that such information is material to the business, assets, liabilities, financial position, operations or results of operations of the Target Entities or of the Business.

10.12 No Presumption Against Drafting Party

CD&R Investor and Harsco each acknowledge that they have been represented by counsel in connection with this Agreement and the transactions contemplated by this Agreement. Accordingly, any rule of law or any legal decision that would require interpretation of any claimed ambiguities in this Agreement against the drafting party has no application and is expressly waived.

10.13 <u>Time Periods</u>

Unless specified otherwise, any action required hereunder to be taken within a certain number of days shall be taken within that number of calendar days (and not Business Days); <u>provided</u>, <u>however</u>, that if the last day for taking such action falls on a weekend or a holiday in the United States, the period during which such action may be taken shall be automatically extended to the next Business Day.

10.14 Execution of Agreement

This Agreement may be executed in one or more counterparts, and by the different Parties hereto in separate counterparts, each of which will be deemed to be an original copy of this Agreement and all of which, when taken together, will be deemed to constitute one and the same agreement. The exchange of copies of this Agreement and of signature pages by facsimile or electronic mail transmission shall constitute effective execution and delivery of this Agreement as to the Parties and may be used in lieu of the original Agreement for all purposes.

10.15 Harsco and CD&R Investor

Harsco acknowledges and agrees with CD&R Investor that Harsco shall cause the other Sellers, and CD&R Investor acknowledges and agrees with Harsco that CD&R Investor shall cause the Buyers to perform their respective obligations under this Agreement.

ARTICLE XI

CERTAIN DEFINITIONS

11.1 <u>Certain Defined Terms</u>

As used in this Agreement, the following terms shall have the following meanings:

"ABO" has the meaning given in Section 8.2(i)(iv).

"<u>Adjustment Amount</u>" has the meaning given in <u>Section 1.9(g)</u>.

"<u>Affiliate</u>" means, with respect to a specified Person, any other Person that directly or indirectly controls, is controlled by or is under common control with such specified Person.

"<u>Agreed Calculation Principles</u>" means such accounting principles, policies, methods, practices, categories, estimates, judgments and assumptions as are in accordance with <u>Exhibit H</u>, and, to the extent not inconsistent therewith, the accounting principles, policies, methods, practices, categories, estimates, judgments and assumptions as were used in preparing the Financial Statements as of and for the fiscal year ended December 31, 2012.

"<u>Agreement</u>" has the meaning given in the Preamble.

"Alternative Financing" has the meaning given in Section 4.10(a).

"<u>Ancillary Agreements</u>" means the Upper-Tier Partnership Agreement, the Holdings Partnership Agreement, the Investor Rights Agreement, the Guarantee, the Assignment and Assumption Agreements, the Transition Services Agreement, the CD&R Consulting Agreement, the Harsco Consulting Agreement, the CD&R Indemnification Agreement and the Harsco Indemnification Agreement.

"<u>Anti-Corruption Laws</u>" has the meaning given in <u>Section 2.16(a)</u>.

"<u>Applicable Non-U.S. Antitrust Approvals</u>" means the necessary approvals, orders, permits or other Consents under any Competition Law set forth on <u>Schedule 4.2</u>.

"<u>Asset Buyers</u>" has the meaning given in the Preamble.

"<u>Asset Sellers</u>" means has the meaning given in the Preamble.

"<u>Assets</u>" has the meaning given in <u>Section 2.11(a)</u>.

"Assignment and Assumption Agreements" has the meaning given in Section 1.8(a)(iii).

"Assumed Target Liabilities" has the meaning given in Section 1.2(a).

"Audited Financial Statements" has the meaning given in Section 2.5(a).

"<u>Available Insurance Policies</u>" has the meaning given in <u>Section 4.9(c)</u>.

"Balance Sheet Date" means June 30, 2013.

"Brand" has the meaning given in the Recitals.

"Brand Closing" means the Closing as defined in the Brand Purchase Agreement.

"Brand Marketing Period" has the meaning given to such term in the Brand Purchase Agreement.

"Brand Purchase Agreement" has the meaning given in the Recitals.

"Brand Required Information" has the meaning given to such term in the Brand Purchase Agreement.

"Bullseye MergerSub, Inc., a Delaware corporation.

"<u>Business</u>" means (i) the business of renting and selling scaffolding, shoring, concrete forming and other similar access related systems and equipment and related services, including solely to the extent conducted in connection with the foregoing, related engineering and design, erection and dismantling, insulation and painting, equipment sales and plant maintenance, conducted by Harsco and its Subsidiaries and (<u>ii</u>) any other operations or activities conducted by the Target Entities, as of the date hereof.

"Business Benefit Plans" has the meaning given in Section 2.8(b).

"<u>Business Day</u>" means any day other than (<u>i</u>) any Saturday or Sunday or (<u>ii</u>) any other day on which banks located in New York, New York are required or authorized by Law to be closed for business.

"<u>Business Employee</u>" means (<u>i</u>) any employee of a Target Entity and (<u>ii</u>) any employee of the Sellers or the Non-Company Affiliates whose employment is transferred to Buyer or a Target Entity pursuant to <u>Section 4.6(a)</u>, in each case, including active and inactive and full-time and temporary employees.

"<u>Buyer</u>" has the meaning given in the Preamble.

"Buyer 401(k) Plan" has the meaning given in Section 4.6(k).

"Buyer Indemnified Parties" has the meaning given in Section 8.2.

"<u>Buyers</u>" has the meaning given in the Preamble.

"<u>Cap Ex Shortfall</u>" means any shortfall in the amount of capital expenditures by the Business (net of disposals) during the period beginning on January 1, 2013 and ending on the earlier of the Closing Date or February 28, 2014, relative to the budgeted capital expenditures for the same period as set forth on Schedule 2.5(d). For purposes of this definition, "capital expenditures" shall include (i) property, plant and equipment purchased with cash, (ii) property, plant and equipment acquired pursuant to financing leases (to the extent such leases are treated as Indebtedness), and (iii) property, plant and equipment which has been acquired for which payment has not yet been made (i.e., accounts payable). In the event that the Closing does not occur on a month-end date, the budgeted capital expenditures for the final month will be calculated pro rata based on the timing of Closing.

"Capital Expenditure Budget" has the meaning given in Section 2.5(d).

"<u>Cash</u>" shall mean the sum of cash, cash equivalents and liquid investments (plus all bank deposits in transit and less outstanding checks and overdrafts) of the Target Entities that are not subject to any restrictions on use; <u>provided</u> that in the case of Cash held by any entity for which the Business owns less than 100% of the economic interest, such amount shall be reduced proportionally to reflect the percentage of the economic interest not owned by the Business.

"<u>Central States</u>" has the meaning given in <u>Section 4.6(m)</u>.

"<u>CD&R Consulting Agreement</u>" means the letter agreement to be entered into by Buyer and the other parties identified therein as of the Closing substantially in the form of <u>Exhibit I</u> hereto.

"<u>CD&R Deal Fee</u>" means the amount of the deal fee specified by CD&R Investor at least two Business Days prior to the Closing, up to a maximum of \$20,750,000.00.

"<u>CD&R Indemnification Agreement</u>" means the indemnification agreement to be entered into by Buyer and the other parties identified therein as of the Closing substantially in the form of <u>Exhibit J</u> hereto.

"<u>CD&R Investor</u>" has the meaning given in the Preamble.

"CD&R Indemnified Parties" has the meaning given in Section 8.2.

"<u>CD&R Investor Parties</u>" means (<u>i</u>) CD&R Investor and the Guarantor, (<u>ii</u>) the Financing Sources, (<u>iii</u>) the former, current and future holders of any equity, partnership or limited liability company interest, controlling persons, directors, officers, employees, agents, attorneys, Affiliates, members, managers, general and limited partners, stockholders and assignees of the persons named in clause (i) above and (<u>iv</u>) any future holders of any equity, partnership or limited liability company interest, controlling persons, directors, officers, employees, agents, attorneys, Affiliates, members, managers, general and limited partners, stockholders and assignees of the foregoing.

"CD&R Investor Partnership Interests" has the meaning given in the Recitals.

"<u>CD&R Investor Transaction Expenses</u>" means the costs, fees and expenses incurred by CD&R Investor and its Affiliates (including fees and expenses of legal, accounting and financial advisors and including the Debt Financing Fees and Expenses, but excluding, for the avoidance of doubt, original issue discount which shall be treated as ordinary interest payable under the Debt Financing Documents post-Closing) in connection with the transactions contemplated hereby (including the transaction contemplated by the Brand Purchase Agreement); <u>provided</u> that the CD&R Investor Transaction Expenses shall be subject to an overall cap of \$130,000,000.00 less the amount of the CD&R Deal Fee (or, if the bridge loan under the Debt Financing is utilized, \$137,500,000.00 less the amount of the CD&R Deal Fee).

"CGMI" has the meaning given in Section 3.6.

"<u>Closing</u>" has the meaning given in <u>Section 1.3(a)</u>.

"<u>Closing Cap Ex Shortfall</u>" has the meaning given in <u>Section 1.9(a)</u>.

"Closing Cash" has the meaning given in Section 1.9(a)(i).

"<u>Closing Date</u>" has the meaning given in <u>Section 1.3(a)</u>.

"Closing Indebtedness" has the meaning given in Section 1.9(a).

"Closing Net Working Capital" has the meaning given in Section 1.9(a).

"<u>Closing Statement</u>" has the meaning given in the <u>Section 1.9(a)</u>.

"<u>Closing Transactions</u>" means (<u>i</u>) any step set forth on <u>Exhibit G</u> other than the steps step 4.1 and step 13.1 set forth on <u>Exhibit G</u> (i.e., related to the transactions contemplated by <u>Section 4.20</u> and <u>Section 4.21</u>), (<u>ii</u>) if the French Sale Option is exercised pursuant to the express terms of <u>Section 4.20</u>, step 4.1 set forth on <u>Exhibit G</u>, (<u>iii</u>) subject to the completion of the Consultations pursuant to the express terms of <u>Section 4.21</u>, step 13.1 set forth on <u>Exhibit G</u> and (iv) the transactions described in clauses (<u>iii</u>) and (<u>iv</u>) of the fifth Recital to this Agreement; <u>provided</u> that any step on <u>Exhibit F</u> which occurs between Harsco or any of its Affiliates, on the one hand, and Buyer or any of its Affiliates, on the other hand, on the Closing shall also be a "Closing Transaction."

"Code" means the Internal Revenue Code of 1986 and any successor statute, as amended from time to time.

"<u>Company</u>" has the meaning given in the Recitals.

"Company Contribution" has the meaning given in Section 1.4(b).

"<u>Company Policy</u>" has the meaning given in <u>Section 2.17</u>.

"<u>Competing Person</u>" has the meaning given in <u>Section 7.1(a)</u>.

"<u>Competition Laws</u>" mean any Laws, including the HSR Act, relating to the regulation of monopolies or competition in any jurisdiction.

"Competitive Activity" has the meaning given in Section 7.1(a).

"<u>Confidentiality Agreement</u>" means the Confidentiality Agreement, dated as of December 20, 2012, between Clayton, Dubilier & Rice, LLC and Harsco.

"Consent" means a consent, authorization or approval of a Person or a waiver by a Person.

"<u>Consolidated Tax Return</u>" means any Tax Return filed or required to be filed with respect to which any of the Target Entities joins with Harsco or any of its Affiliates (other than any Target Entity) to file such Tax Return on a consolidated, combined, unitary or similar basis.

"Consultations" has the meaning given in Section 4.21.

"<u>Contract</u>" means any written contract, agreement, license, lease, sales order, purchase order, indenture, mortgage, note, bond or warrant (including all amendments, supplements and modifications thereto), except any common law contracts with employees of the Target Entities.

"Contributions" has the meaning given in Section 1.4(b).

"Copyrights" has the meaning given in the definition of Intellectual Property.

"<u>Coverage Claims</u>" has the meaning given in <u>Section 4.9(c)(ii)</u>.

"<u>CPA Firm</u>" has the meaning given in <u>Section 1.9(d)</u>.

"Debt Commitment Parties" has the meaning given in Section 3.6.

"Debt Financing" has the meaning given in Section 3.6.

"Debt Financing Commitment" has the meaning given in Section 3.6.

"Debt Financing Fees and Expenses" has the meaning given in Section 4.14(b).

"Deductible" has the meaning given in Section 8.4(b).

"Deferred Baroom Closing" has the meaning set forth in Section 1.7(b).

"Deferred Baroom Closing Date" has the meaning set forth in Section 1.7(a).

"Deferred Baroom Entity" has the meaning set forth in Section 1.7(a).

"Deferred China Closing" has the meaning given in Section 1.6(b).

"<u>Deferred China Closing Date</u>" has the meaning given in <u>Section 1.6(a)</u>.

"Deferred China Entities" has the meaning given in Section 1.6(a).

"Deferred India Closing" has the meaning given in Section 1.5(b).

"Deferred India Closing Date" has the meaning given in Section 1.5(a).

"<u>Deferred India Entity</u>" means a private limited company to be incorporated in India and wholly owned by Harsco Investments Europe BV and MultiServ International BV prior to Closing, which, pursuant to a business transfer agreement by and among Harsco India Private Ltd., the Deferred India Entity and the other parties thereto, the Deferred India Entity will own all of the India Business.

"Definitive Agreements" has the meaning given in Section 4.10(a).

"Dollars" or "<u>\$</u>" means U.S. dollars.

"Domain Names" has the meaning given in the definition of Intellectual Property.

"D&O Indemnitees" has the meaning given in Section 4.18(a).

"Employee Plan" means any material (i) employee benefit plan (within the meaning of Section 3(3) of ERISA, whether or not subject to ERISA) including any pension, profit-sharing, retirement, death, disability, supplemental retirement, welfare benefit, retiree health, and life insurance plans, agreements or arrangements, and any material (<u>ii</u>) bonus, stock option, stock purchase, restricted stock, phantom stock or other equity-based arrangement, incentive, deferred compensation, termination, severance, retention, change of control, health, group insurance, vacation or other employee benefit plan, program, policy or arrangement, whether written or unwritten, other than, in each case, an employee benefit plan, program, policy or arrangement maintained pursuant to a requirement of applicable Law and that does not provide retirement benefits.

"<u>Encumbrance</u>" means any pledge, lien (including a Tax lien), collateral assignment, security interest, mortgage, option, easement, deed of trust, title retention, conditional sale or other security arrangement, or any license, order or charge, or any material adverse claim of title, ownership or use, or any agreement of any kind restricting transfer.

"Engagement Letter" has the meaning given in Section 3.6.

"Environment" means the ambient air, surface water, ground water, land, and surface or subsurface strata.

"<u>Environmental Law</u>" means any Law governing or relating to pollution, protection of the Environment or natural resources or exposure to Hazardous Substances.

"Equity Financing" has the meaning given in Section 3.6.

"Equity Financing Commitment" has the meaning given in Section 3.6.

"ERISA" means the Employee Retirement Income Security Act of 1974, as amended.

"<u>ERISA Affiliate</u>" means, with respect to any entity, trade or business, any other entity, trade or business that is, or was at the relevant time, a member of a group described in Section 414(b), (c), (m) or (o) of the Code or Section 4001(b)(1) of ERISA that includes or included the first entity, trade or business, or that is, or was at the relevant time, a member of the same "controlled group" as the first entity, trade or business pursuant to Section 4001(a)(14) of ERISA.

"ERISA Affiliate Liabilities" has the meaning given in Section 4.6(n).

"Estimated Closing Cap Ex Shortfall" has the meaning given in Section 1.2(b)(vi).

"Estimated Closing Cash" has the meaning given in Section 1.2(b)(i).

"Estimated Closing Indebtedness" has the meaning given in Section 1.2(b)(ii).

"Estimated Closing Net Working Capital" has the meaning given in Section 1.2(b)(iii).

"Estimated Closing Statement" has the meaning given in <u>Section 1.2(b)</u>.

"<u>Exchange Act</u>" means the Securities Exchange Act of 1934, as amended, and the rules and regulations promulgated thereunder.

"Excluded Assets" has the meaning set forth in <u>Section 1.1(b)</u>.

"Excluded Target Liabilities" means the Liabilities of Harsco or the Asset Sellers set forth on Schedule 1.2(a).

"Exhibit" or "Exhibits" means respectively the document or documents attached to this Agreement as exhibits.

"Fair Market Value" has the meaning given in Section 4.22(b).

"Fee Credit Letter" has the meaning given in Section 3.6.

"<u>Fee Letter</u>" has the meaning given in <u>Section 3.6</u>.

"<u>Final Closing Statement</u>" has the meaning given in <u>Section 1.9(e)</u>.

"Financial Statements" has the meaning given in Section 2.5(a).

"<u>Financing</u>" has the meaning given in <u>Section 3.6</u>.

"Financing Commitments" has the meaning given in Section 3.6.

"<u>Financing Sources</u>" means the Persons that have committed to provide or have otherwise entered into agreements in connection with the Debt Financing Commitment or alternative debt financings in connection with the transactions contemplated hereby, including the parties named in <u>Section 3.6</u> and any joinder agreements, indentures or credit agreements entered into pursuant thereto or relating thereto, together with their Affiliates, and the officers, directors, employees, agents, representatives, successors and assigns of the foregoing.

"French Business" has the meaning given in Section 4.20.

"French Sale Option" has the meaning given in Section 4.20.

"FRC" has the meaning given in the Recitals.

"Fundamental Representations" has the meaning given in Section 8.4(a).

"GAAP" means United States generally accepted accounting principles.

"<u>Governmental Body</u>" means any domestic or foreign government, including any foreign, federal, state, provincial, local, territorial or municipal government or any governmental division, agency or authority thereof, court or judicial authority, tribunal or commission.

"<u>Government Official</u>" means any (<u>i</u>) officer or employee of a Governmental Body or instrumentality thereof (including any state-owned or state-controlled enterprise), or of a public international organization, (<u>ii</u>) holder of public office, candidate for public office, political party, official of a political party or member of a royal family or (<u>iii</u>) any Person acting for or on behalf of any such Governmental Body or instrumentality thereof (including any state-owned or state-controlled enterprise).

"<u>Governmental Order</u>" means any order, writ, judgment, award, ruling, injunction, decree or consent decree entered by or with any Governmental Body.

"GS" has the meaning given in Section 3.6.

"<u>Guarantee</u>" means the limited guarantee by Guarantor, dated as of the date hereof, in favor of Harsco with respect to certain obligations of CD&R Investor arising under, or in connection with, this Agreement.

"<u>GRA</u>" has the meaning given in <u>Section 5.11(d)</u>.

"Guarantor" has the meaning given in the Recitals.

"Harsco" has the meaning given in the Preamble.

"Harsco Benefit Plans" has the meaning given in Section 2.8(b).

"Harsco Company Shares" has the meaning given in the Recitals.

"<u>Harsco Consulting Agreement</u>" means the letter agreement to be entered into by Buyer and the other parties identified therein as of the Closing substantially in the form of <u>Exhibit K</u> hereto.

"Harsco Deal Fee" means \$3,800,000.

"Harsco Defined Contribution Plans" has the meaning given in Section 4.6(k).

"Harsco Designations" has the meaning given in Section 7.5(c).

"<u>Harsco House</u>" shall mean the premises located at 299 Regent Park, Kingston Road, Leatherhead, Surrey, United Kingdom.

"Harsco Incremental Transaction Expenses" has the meaning given in Section 4.14(c).

"<u>Harsco Indemnification Agreement</u>" means the indemnification agreement to be entered into by Buyer and the other parties identified therein as of the Closing substantially in the form of <u>Exhibit L</u> hereto.

"Harsco Indemnified Parties" has the meaning given in Section 8.1.

"Harsco Information" has the meaning given in <u>Section 4.5(b)</u>.

"Harsco Letters of Credit" has the meaning given in Section 4.17.

"Harsco Parties" has the meaning given in <u>Section 9.2(e)</u>.

"Harsco Partnership Interests" has the meaning given in the Recitals.

"<u>Harsco Pension Scheme</u>" means the occupational pension scheme currently governed by Rules dated April 28, 2004, as amended.

"Harsco Required Information" has the meaning given in Section 4.11(a)(4).

"<u>Hazardous Substance</u>" means any hazardous substance, hazardous waste, toxic substance, pollutant, and any other similarly described substance as defined as such in any applicable Environmental Law, including petroleum, petroleum products, asbestos and asbestos containing materials.

"Holdings Partnership" has the meaning given in Recitals.

"<u>Holdings Partnership Agreement</u>" means the limited partnership agreement of Holdings Partnership to be entered into by its partners substantially concurrently with the Closing.

"HSR Act" means the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended.

"<u>HUG</u>" has the meaning given in <u>Section 1.4(a)</u>.

"Inactive Employees" has the meaning given in Section 4.6(c).

"Indebtedness" means, with respect to any Person, without duplication, (i) all obligations of such Person for borrowed money, including accrued and unpaid interest, (ii) all obligations of such Person evidenced by bonds, debentures, notes or similar instruments, (iii) all obligations of such Person under conditional sale or other title retention agreements relating to any property purchased by such Person, (iv) all obligations of such Person issued or assumed as the deferred purchase price of property or services (excluding obligations of such Person for raw materials, inventory, services and supplies incurred in the ordinary course of business consistent with past practices), (v) any obligations of such Person under any lease of property, real or personal, which obligations are required to be classified as capital leases in accordance with GAAP, (vi) all obligations of such Person under interest rate, currency or commodity derivatives or hedging transactions (valued at the termination value thereof), (vii) all guarantees and keepwell arrangements of such Person of any Indebtedness of any other Person other than a Subsidiary of such Person, (viii) Restructuring Expenses (to be included for purposes of <u>Sections 1.2</u> and <u>1.9</u> only) and (ix) Repatriation Costs (to be included for purposes of <u>Sections 1.2</u> and <u>1.9</u> only), other than, in each case (other than clause (viii), in relation to any pension arrangement in the UK or any operating lease for fixed assets.

"Indemnified Party" has the meaning given in Section 8.7(a)(i).

"Indemnifying Party" has the meaning given in Section 8.7(a)(i).

"Indemnitee" means any Indemnified Party or Tax Indemnified Party.

"Indemnity Payment" has the meaning given in Section 8.6(b).

"India Business" the assets, operations and liabilities relating to the Business of Harsco India Private Ltd.

"Infrastructure MEPP Summary" has the meaning given in Section 2.8(i).

"Infrastructure Multiemployer Plan" has the meaning given in Section 2.8(h)(i).

"Initial Cash Purchase Price" has the meaning given in Section 1.2(a)(i).

"Insurance Cap" has the meaning given in Section 4.18(b).

"Intellectual Property" means any of the following: United States or foreign (i) patents and patent applications (collectively, "Patents"); (ii) registered and unregistered trademarks, service marks, trade dress and other indicia of origin, pending trademark and service mark registration applications and intent-to-use registrations or similar reservations of marks (collectively, "Marks"); (iii) registered and unregistered copyrights (including in Software) and applications for registration (collectively, "Copyrights"); (iv) internet domain names and URLs ("Domain Names"); (v) trade secrets and proprietary information, including unpatented inventions, invention disclosures, know-how, methods, processes, customer lists, data and databases, in each case to the extent any of the foregoing derives economic value (actual or potential) from not being generally known to other Persons who can obtain economic value from its disclosure or use, excluding any Copyrights or Patents that may cover or protect any of the foregoing (collectively, "Trade Secrets"); (vi) rights or paternity, integrity, disclosure and withdrawal and any other rights that may be known as "moral rights"; and (vii) any other intellectual property rights.

"Intellectual Property Contract" means any Contract to which (i) any Target Entity is a party or (ii) any Asset Seller is a party and which primarily relates to the Business, in each case, pursuant to which (\underline{A}) any Target Entity or any such Asset Seller permits any Person to use any Owned Intellectual Property, (\underline{B}) any Person permits any Target Entity or any such Asset Seller to use Intellectual Property not owned by any Target Entity, or (\underline{C}) any Target Entity or any such Asset Seller's right to use or register Intellectual Property is restricted.

"<u>Intercompany Balances</u>" means, as of any date, all balances as of such date between Harsco and the Non-Company Affiliates, on the one hand, and the Target Entities, on the other hand, including intercompany accounts receivable and intercompany accounts payable.

"Intercompany Contract" has the meaning given in Section 2.14(a)(viii).

"<u>Intercompany Obligation</u>" means any obligation between any Target Entity and any other Affiliate of Harris (that is not a Target Entity).

"Intermediate Company" has the meaning given in Section 3.1.

"<u>Internal IT Systems</u>" means the hardware, Software, network and telecommunications equipment and Internet-related information technology infrastructure owned or leased by Harsco or any of its Affiliates (including the Target Entities).

"<u>Intracompany Obligation</u>" means any obligation between any two or more of the Target Entities, including intercompany accounts receivable and intercompany accounts payable between such entities.

"Investor Rights Agreement" has the meaning set forth in the Recitals.

"Irrevocable Offer" has the meaning given in Section 4.2.

"IRS" means the United States Internal Revenue Service.

"<u>JV Partner</u>" means any Person that will own, immediately after giving effect to the Closing Transactions and the Closing, any capital stock or other equity interests in any Target Entity (other than Buyer and its subsidiaries as of such time).

"<u>Knowledge</u>" means, when used with respect to Harsco, the actual knowledge, after reasonable inquiry, of any person set forth on <u>Schedule 11.1(b)</u>.

"Labor Agreement" has the meaning given in Section 4.6(d).

"<u>Law</u>" means any applicable foreign, federal, state or local law, statute, regulation, ordinance, rule, order, decree, judgment, consent decree or other binding directive issued, enacted, promulgated, entered into, agreed or imposed by any Governmental Body.

"Leased Employees" has the meaning given in Section 4.6(c)(3).

"Leased Real Property" has the meaning given in Section 2.12(b).

"Leasing Period" has the meaning given in Section 4.6(c)(3).

"<u>Liability</u>" or "<u>Liabilities</u>" means debts, commissions, duties, fees, salaries, performance or delivery penalties, warranty liabilities and other liabilities and obligations (whether pecuniary or not, including obligations to perform or forebear from performing acts or services), fines or penalties.

"<u>Loss</u>" means any and all Liabilities, losses, damages, expenses (including reasonable expenses of investigation, enforcement and collection and reasonable attorneys' and accountants' fees and expenses, in each case, in connection with any Proceeding), costs, fines, fees, penalties and obligations, whether or not involving a Third Party Claim.

"Loss of Coverage" has the meaning given in Section 4.6(c).

"Major Customers" has the meaning given in Section 2.21.

"Major Suppliers" has the meaning given in Section 2.21.

"Market Rent" has the meaning given in Section 4.22(b).

"<u>Marketing Period</u>" means the first period of twenty-five (25) consecutive calendar days after the date hereof throughout and at the end of which (a) CD&R Investor shall have the Harsco Required Information and the Brand Required Information, (b) the Offering materials shall be in customary complete form (except for portions thereof and information that would customarily be provided by lead arrangers) and (c) (x) the conditions set forth in <u>Section 6.2</u> (other than <u>Section 6.2(h)</u>) shall be satisfied (excluding conditions that, by their nature, are satisfied at Closing) and nothing has occurred and no condition exists that would be reasonably likely to cause any of the conditions set forth in <u>Section 6.2</u> (other than <u>Section 6.2(h)</u>) to fail to be satisfied assuming the Closing were to occur any time during such 25 consecutive calendar day period and (y) the conditions set forth in <u>Sections 7.1</u> and <u>7.3</u> of the Brand Purchase Agreement shall be satisfied and nothing has occurred and no condition exists that would be reasonably likely to cause any of the conditions set forth in <u>Sections 7.1</u> and <u>7.3</u> of the Brand Purchase Agreement to fail to be satisfied, assuming the closing were to occur at any time during such 25 consecutive calendar day period; <u>provided</u> that (<u>i</u>) (<u>A</u>) if such consecutive calendar day period commences between (and including) November 8, 2013 and November 13, 2013, it shall end no earlier than December 14, 2013, (<u>B</u>) the period from (and including) November 27, 2013 through (and including) December 1,

2013 shall be disregarded for purposes of calculating such 25 consecutive calendar day period (and for purposes of determining whether such period is consecutive) and (C) if such period shall not have ended on or prior to December 20, 2013, such period shall not commence before January 6, 2014, and (ii) the Marketing Period shall not be deemed to have commenced if, after the date hereof and prior to the completion of the Marketing Period, (A) financial statements included in the Harsco Required Information becomes stale under such that it would not permit a registration statement including such financial statements to be declared effective by the SEC on the last day of such 25 consecutive calendar day period under Rule 3-12 of Regulation S-X promulgated under the Securities Act, in which case the Marketing Period shall not be deemed to commence unless and until the earliest date on which Harsco has furnished CD&R Investor with updated Harsco Required Information, (B) PricewaterhouseCoopers LLP shall have withdrawn its audit opinion with respect to any of the annual financial statements included in the Required Information, in which case the Marketing Period shall not be deemed to commence unless and until, at the earliest, a new ungualified audit opinion is issued with respect to such financial statements by PricewaterhouseCoopers LLP or another nationally-recognized independent public accounting firm, or (C) Harsco or any of its Affiliates restates or Harsco's board of directors has determined to restate any historical financial statements of the Business included in the Required Information, in which case the Marketing Period shall not be deemed to commence unless and until, at the earliest, such restatement has been completed or Harsco's board of directors subsequently concludes that no restatement shall be required in accordance with GAAP, and (iii) the Marketing Period shall not be deemed to have commenced if, the Marketing Period, the Brand Marketing Period shall have been deemed not to have commenced pursuant to the terms of the Brand Purchase Agreement, in which case the Marketing Period shall not be deemed to commence unless and until the Brand Marketing Period has commenced, provided, further, that if Harsco in good faith reasonably believes that Marketing Period has commenced, it may deliver to the CD&R Investor a written notice to that effect (stating when it believes it had commenced), in which case the Marketing Period shall be deemed to have commenced on the date specified in that notice unless the CD&R Investor in good faith reasonably believes the Marketing Period has not commenced and, three (3) calendar days following receipt of such notice delivers a written notice to Harsco to that effect (stating with reasonable specificity why the CD&R Investor reasonably believes the Marketing Period had not commenced).

"Marks" has the meaning given in the definition of Intellectual Property.

"<u>Material Adverse Effect</u>" means any change, effect, occurrence or state of facts that, individually or in the aggregate, has a materially adverse effect, (<u>a</u>) on the financial condition, business, properties, rights, assets or results of operations of the Business, taken as a whole, other than any change, effect, occurrence or state of facts to the extent relating to (<u>i</u>) changes in business, economic or regulatory conditions as a whole or in the industries in which the Target Entities operate, (<u>ii</u>) changes in political conditions in any country or jurisdiction in which the Business or any Target Entity operates, (<u>iii</u>) an outbreak or escalation in hostilities involving the United States or any country or jurisdiction in which the Business or any Target Entity operates, whether or not pursuant to the declaration of a national emergency or war, or the occurrence of any military or terrorist attack upon the United States or any such country or jurisdiction, or any of their respective territories, possessions, or military installations, (<u>iv</u>) changes in GAAP or Law, (<u>vi</u>) changes in commodity prices, (<u>vii</u>) the announcement of, or the taking of any action contemplated by, this Agreement and the other agreements contemplated hereby, including (<u>A</u>) the loss of any customers, suppliers, employees or joint venture partners (if, and only if, such loss primarily results from such announcement) and (<u>B</u>) compliance with the covenants set forth herein (other than for purposes of the representations and warranties contained in <u>Sections 2.3</u> and <u>2.15</u> and the conditions in <u>Section 6.2(a)</u> to the extent they relate to the representations and warranties

contained in <u>Sections 2.3</u> and <u>2.15</u>), and (<u>viii</u>) any failure by the Target Entities or the Business to meet any projections, forecasts or estimates of revenue or earnings (<u>provided</u> that the underlying cause of such failure may be considered in determining whether there is a Material Adverse Effect), except, in the cases of clauses (i), (ii), (iii), (iv) and (v), to the extent that such adverse effects materially and disproportionately have a greater adverse impact on the Target Entities, taken as a whole, as compared to the adverse impact such changes have on companies in the industry in which the Target Entities operate or (<u>b</u>) would, or would reasonably be expected to, prevent, materially delay, or materially impede the performance by Harsco of its obligations under this Agreement or the consummation of the transactions contemplated hereby.

"Material Contracts" has the meaning given in Section 2.14(a)(xi).

"Minimum Per Claim Amount" has the meaning set forth in Section 8.4(c).

"Minority Investment" has the meaning given in Section 7.1(b)(ii).

"MSSF" has the meaning given in Section 3.6.

"<u>Multiemployer Plan</u>" means a "multiemployer plan" within the meaning of Section 3(37) of ERISA that is maintained in the United States.

"<u>Net Working Capital</u>" means, as of the applicable time for determination thereof, (i) the aggregate of the Current Assets minus (ii) the aggregate of the Current Liabilities, calculated in accordance with the Agreed Calculation Principles. For purposes of this definition, (A) "Current Assets" means the sum of all current assets of the Target Entities, including in each case, accounts receivable, VAT receivables, prepaid expenses and inventory, but excluding cash and cash equivalents, deferred Tax assets and any accounts receivable due or owing to any Target Entity from Harsco or any of its Affiliates (other than intercompany trade receivables arising in the ordinary course of the Business to the extent that they are not eliminated prior to Closing), and (\underline{B}) "Current Liabilities" means the sum of all current liabilities of the Target Entities, including in each case all accounts payable and accrued expenses and deferred revenue, but excluding any current portion of Closing Indebtedness, Restructuring Expenses (irrespective of when such restructuring liabilities are payable), deferred Tax liabilities and any accounts payable owed by any Target Entity to Harsco or any of its Affiliates (other than intercompany trade payables arising in the ordinary course of the Business to the extent that they are not eliminated prior to Closing). For the avoidance of doubt, (\underline{x}) current assets shall not include any receivable of any Target Entity under any Tax sharing, grouping or similar arrangement with Harsco or any Non-Company Affiliate, (y) current assets shall include any intercompany trade payables and receivables due or owing to any Target Entity from Harsco or any of its Affiliates arising in the ordinary course of the Business to the extent that they are not eliminated prior to Closing and (z) current liabilities shall include any intercompany trade payables and receivables arising in the ordinary course of the Business due or owing by any Target Entity from Harsco or any of its Affiliates, to the extent that they are not eliminated prior to Closing, in each case, calculated in accordance with the principles set forth on Exhibit H. For the avoidance of doubt, Net Working Capital shall exclude all cash, cash equivalents and liquid investments (plus all bank deposits in transit and less all outstanding checks and overdrafts) of the Target Entities, including amounts in both U.S. and non-U.S. entities.

"Netherlands Business" has the meaning given in Section 4.21.

"Netherlands Transaction" has the meaning given in Section 4.21.

"<u>Non-Company Affiliate</u>" means any Affiliate of Harsco other than the Target Entities (after giving effect to the Transactions).

"<u>Non-Competition Period</u>" has the meaning given in <u>Section 7.1(a)</u>.

"<u>Non-Solicitation Period</u>" has the meaning given in <u>Section 7.2(a)</u>.

"Non-U.S. Pension Indemnitees" has the meaning given in Section 8.2(i)(iv)(B).

"<u>Objection</u>" has the meaning given in <u>Section 1.9(c)</u>.

"<u>OFAC</u>" has the meaning given in <u>Section 2.16(f)</u>.

"Offering Materials" has the meaning given in <u>Section 4.11(a)(4)</u>.

"Original Valuations" has the meaning given in <u>Section 4.22(b)</u>.

"Outside Date" has the meaning given in Section 9.1(d).

"Owned Intellectual Property" has the meaning given in Section 2.13(a).

"Owned Real Property" has the meaning given in Section 2.12(a).

"<u>Parties</u>" has the meaning given in the Preamble.

"Partnership" has the meaning given in the Recitals.

"Partnership Company Shares" has the meaning given in Section 1.4(b).

"Partnership Contribution" has the meaning given in Section 1.4(b).

"Patents" has the meaning given in the definition of Intellectual Property.

"Pension Obligation" has the meaning given in Section 8.2(i)(iv).

"<u>Permit</u>" means any permit, license, approval, consent, registration, variance, certification, endorsement or qualification granted by or obtained from any Governmental Body pursuant to Law.

"<u>Permitted Encumbrances</u>" means (<u>i</u>) any Encumbrance imposed by federal and state securities Laws; (<u>ii</u>) liens for current Taxes or other governmental charges not yet due and payable or the amount or validity of which is being contested in good faith by appropriate proceedings by Harsco (or any of its Affiliates) or the Target Entities and for which adequate reserves are maintained on the financial statements of Harsco (or any of its Affiliates) or the Target Entities in accordance with GAAP consistently applied; (<u>iii</u>) mechanics', construction, carriers', workers', repairers' and similar statutory liens imposed by law arising or incurred in the ordinary course of business for amounts which are not delinquent and for which adequate reserves are maintained on the financial statements of Harsco (or any of its Affiliates), the Target Entities in accordance with GAAP consistently applied; (<u>iv</u>) zoning, entitlement, building and other land use regulations imposed by any Governmental Body having jurisdiction over the Leased Real Property or Owned Real Property which if violated do not, individually or in the aggregate, (<u>A</u>) materially impair the occupancy or use of the Leased Real Property for the purposes for which it is currently used or proposed to be used or (<u>B</u>) materially affect the value of the Leased Real Property or the Owned Real Property; (<u>v</u>) defects or imperfections in title, Encumbrances, covenants,

conditions, restrictions, reservations, limitations, exemptions, rights-of-way, easements and other similar matters of record affecting title to the Leased Real Property or Owned Real Property which do not, individually or in the aggregate, (<u>A</u>) materially impair the occupancy or use of the Leased Real Property or Owned Real Property for the purposes for which it is currently used or proposed to be used or (<u>B</u>) materially affect the value of the Leased Real Property or the Owned Real Property; (<u>vi</u>) defects or imperfections in title, Encumbrances, covenants, conditions, restrictions, reservations, limitations, and other similar matters affecting the Owned Intellectual Property which do not, individually or in the aggregate, (A) materially impair the use of the Owned Intellectual Property for the purposes for which it is currently used or proposed to be used or (B) materially affect the value of the Owned Intellectual Property; (<u>vii</u>) liens arising under worker's compensation, unemployment insurance, social security, retirement and similar Laws; (<u>viii</u>) liens securing rental payments under capital lease arrangements; (<u>ix</u>) liens of any lessor under any lease agreement; (<u>x</u>) rights of expropriation, access or user or any similar rights conferred on or reserved by or in any statute of a Governmental Body; and (<u>xi</u>) liens set forth on <u>Schedule 11.1(c)</u>.

"<u>Person</u>" means any natural person, partnership, limited liability company, corporation, joint stock company, trust, estate, joint venture, group, association or unincorporated organization or any other form of business or professional entity, but does not include a Governmental Body.

"<u>Post-Closing Tax Period</u>" means (<u>i</u>) any Tax period ending after the Closing Date and (<u>ii</u>) in the case of any Straddle Period, the portion of such period beginning after, but not including, the Closing Date.

"Pre-Closing Damages Proceeding" has the meaning given in Section 9.2(e).

"<u>Pre-Closing Tax Period</u>" means (<u>i</u>) any Tax period ending on or before the Closing Date and (<u>ii</u>) in the case of any Straddle Period, the portion of such period ending on and including the Closing Date.

"<u>Pre-Closing Transactions</u>" means the restructuring transactions set forth on <u>Exhibit F</u>; <u>provided</u> that any step on <u>Exhibit G</u> which occurs among Harsco or any of its Affiliates prior to the Closing shall also be a "Pre-Closing Transaction."

"<u>Prime Rate</u>" has the meaning given in <u>Section 1.9(h)</u>.

"Proceeding" has the meaning given in Section 2.7.

"Protective GRA" has the meaning given in <u>Section 5.11(d)</u>.

"Purchase Price" has the meaning given in Section 1.2(a).

"Real Property Leases" has the meaning given in Section 2.12(b).

"<u>Redirected Domains</u>" has the meaning given in <u>Section 7.5(g)</u>.

"<u>Registered Intellectual Property</u>" has the meaning given in <u>Section 2.13(a)</u>.

"<u>Release</u>" means any actual or threatened release, spill, emission, leaking, dumping, injection, pouring, deposit, disposal, discharge, dispersal, leaching or migration into or through the Environment.

"Repatriation Schedule" is the schedule of repatriation costs on Exhibit N.

"<u>Repatriation Costs</u>" shall mean the costs that would be incurred as a result of repatriating to the United States, after the Closing, an amount of Cash equal to the foreign cash balance at Closing less the net Cash receipts in the five business days prior to Closing, calculated on a jurisdiction by jurisdiction basis, including any costs that would arise as a result of withholding, income or other Taxes, computed without regard to net operating losses, using the tax rates set forth on the Repatriation Schedule.

"<u>Replicated Contracts</u>" means the contracts set forth on <u>Schedule 4.13(b)</u> as well as any Shared Contract which is not set forth on <u>Schedule 4.13(b)</u> but which, in the good faith judgment and mutual agreement of CD&R Investor, Harsco and Buyer, should be deemed a Replicated Contract.

"Required Information" means the Harsco Required Information and the Brand Required Information, collectively.

"Required Remediation Activity" has the meaning given in Section 8.4(d).

"Relevant Persons" has the meaning given in Section 2.16(f).

"Resolution Period" has the meaning given in Section 1.9(c).

"Resolved Matters" has the meaning given in Section 1.9(c).

"<u>Restructuring Expenses</u>" means restructuring liabilities of the Business, calculated in a manner consistent with past practice (including with respect to discounting), <u>provided</u> that to the extent that any such liabilities are not already discounted to reflect their net present value, such undiscounted liabilities shall be discounted to reflect their net present value at a rate of 10%, but not including any reserves relating to the leases in Items 2 and 3 on <u>Schedule 1.1(b)</u>.

"Return Date" has the meaning given in Section 4.6(c).

"<u>Review Period</u>" has the meaning given in <u>Section 1.9(c)</u>.

"<u>Schedule</u>" or "<u>Schedules</u>" means the Schedules referenced in the introductory paragraph to <u>ARTICLE II</u> (for the avoidance of doubt, whether or not relating to the provisions of <u>ARTICLE II</u>).

"SEC" means the U.S. Securities and Exchange Commission.

"Securities Act" means the Securities Act of 1933, as amended, and the rules and regulations promulgated thereunder.

"<u>Sellers</u>" has the meaning given in the Preamble.

"<u>Shared Contract</u>" means any Contract (i) to which Harsco or any Non-Company Affiliate is a party and (<u>ii</u>) which relates to, or provides a benefit to, the Business; <u>provided</u> that Shared Contracts shall not include any Contract which is a Target Asset unless such Contract relates to, or provides a benefit to, any business of Harsco (other than the Business).

"<u>Software</u>" means all computer software, including application software, system software firmware, source code and object code versions thereof, in any and all forms and media, and all related documentation.

"Stock Buyers" has the meaning given in the Preamble.

"<u>Stock Sellers</u>" has the meaning given in the Preamble.

"<u>Straddle Period</u>" means, as the context requires, any Tax period with respect to any Target Entity that begins before but ends after the Closing Date.

"Subsidiary" or "Subsidiaries" means, with respect to any Person, any other Person of which (i) if a corporation, a majority of the total voting power of shares of capital stock entitled (without regard to the occurrence of any contingency) to vote in the election of directors, managers or trustees thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more of the other Subsidiaries of that Person or a combination thereof or (ii) if a limited liability company, partnership, association or other business entity (other than a corporation), a majority of partnership or other similar ownership interest thereof is at the time owned or controlled, directly or indirectly, by that Person or one or more other Subsidiaries of that Person or Persons owns a majority of partnership interest in such a business entity (other than a corporation) if such Person or Persons shall be allocated a majority of such business entity's gains or losses or shall be or control any managing director or general partner of such business entity (other than a corporation); <u>provided</u> that each of the entities identified on <u>Schedule 2.2(a)</u> without an asterisk beside such entities name shall be deemed a Subsidiary of Harsco and/or a Target Entity, as applicable. For the purposes hereof, the term Subsidiary shall include all Subsidiaries of such Subsidiary.

"Target Assets" has the meaning given in the Recitals.

"<u>Target Entities</u>" means, collectively, the entities identified on <u>Schedule 2.2(a)</u> under the columns titled "Target Entity"; <u>provided</u> that for purposes of calculating the Estimated Closing Statement and the Closing Statement, and the applicable provisions and definitions relating thereto, "Target Entities" shall be deemed to include the Asset Sellers to the extent of any Cash, Indebtedness or Net Working Capital that is transferred to an Asset Buyer as part of the Target Assets or Assumed Liabilities (but excluding, for the avoidance of doubt, any such items that are not transferred to an Asset Buyer).

"Target Entity Benefit Plans" has the meaning given in Section 2.8(a).

"Target Net Working Capital" shall mean \$193,505,000.

"Target Shares" has the meaning given in the Recitals.

"<u>Tax</u>" or "<u>Taxes</u>" means all taxes of any kind whatsoever (whether payable directly or by withholding), including franchise, income, gross receipts, personal property, real property, ad valorem, value added, VAT, sales, use, documentary, stamp, intangible personal property, social security, wages, pension, withholding or other taxes, together with any interest and penalties, additions to tax or additional amounts with respect thereto imposed by any Tax Authority.

"<u>Tax Authority</u>" means any Governmental Body, including social security administrators, or any agent thereof (third party or otherwise), legally authorized to assess, lien, levy or otherwise collect, litigate or administer Taxes.

"Tax Benefit Payment" has the meaning given in Section 8.6(b).

"<u>Tax Claim</u>" has the meaning given in <u>Section 5.2(a)</u>.

"Tax Indemnified Party" has the meaning given in Section 5.2(a).

"Tax Indemnifying Party" has the meaning given in Section 5.2(a).

"<u>Tax Return</u>" means any report, return, document, declaration, payee statement or other information or filing required to be supplied to any Tax Authority with respect to Taxes.

"<u>Tax Sharing Agreements</u>" has the meaning given in <u>Section 5.6</u>.

"Termination Fee" has the meaning given in Section 9.2(b).

"Third Party Claim" has the meaning given in <u>Section 8.7(a)(i)</u>.

"Third Valuation" has the meaning given in <u>Section 4.22(b)</u>.

"Trade Secrets" has the meaning given in the definition of Intellectual Property.

"Transactions" means, collectively, the Pre-Closing Transactions and the Closing Transactions.

"Transaction Taxes" has the meaning given in Section 5.4.

"<u>Transferred Employees</u>" means all individuals who are Business Employees as of the Closing, excluding (\underline{x}) any Business Employees covered by a collective bargaining agreement, works council or similar agreement and (\underline{y}) any Inactive Employees who do not return to active employment with the Target Entities on or prior to the first anniversary of the Closing Date.

"<u>Transition Services Agreement</u>" has the meaning given in the Recitals.

"<u>Treasury Regulations</u>" means the Federal income tax regulations, including any temporary or proposed regulations, promulgated under the Code, in effect as of the date hereof.

"Triggering Event" has the meaning given in Section 4.9(b).

"TUPE" has the meaning given in Section 4.6(a)(i).

"TUPE-Equivalent Regulations" has the meaning given in Section 4.6(a)(i).

"<u>UBSLF</u>" has the meaning given in <u>Section 3.6</u>.

"<u>UBSS</u>" has the meaning given in <u>Section 3.6</u>.

"<u>UK Closing</u>" has the meaning given in <u>Section 1.4(b)</u>.

"<u>UK Closing Date</u>" has the meaning given in <u>Section 1.4(a)</u>.

"<u>UK Entity</u>" has the meaning given in <u>Section 1.4(a)</u>.

"<u>UK Entity Contribution</u>" has the meaning given in <u>Section 1.4(a)</u>.

"<u>UK Shares</u>" has the meaning given in <u>Section 1.4(a)</u>.

"Unresolved Matters" has the meaning given in Section 1.9(d).

"<u>Upper-Tier</u> Partnership" has the meaning given in the Recitals.

"<u>Upper-Tier Partnership Agreement</u>" means the limited partnership agreement of the Upper-Tier Partnership, by and amongst Harsco, CD&R Investor and the other parties thereto, substantially in the form attached as <u>Exhibit M</u>.

"<u>VAT</u>" means any tax imposed in compliance with the council directive of 28 November 2006 on the common system of value added tax (EC Directive 2006/112) and any other tax of a similar nature, irrespective of the jurisdiction in which it is imposed.

"<u>WARN Act</u>" means the Worker Adjustment and Retraining Notification Act of 1988, as amended, and any other similar Law.

"Works Council" has the meaning given in Section 4.6(b).

11.2 Interpretation

The headings contained in this Agreement are for reference purposes only and will not affect in any way the meaning or interpretation of this Agreement. In the event of a conflict between language or amounts contained in the body of the Agreement and language or amounts contained in the Exhibits or Schedules attached hereto, the language or amounts in the body of the Agreement shall control. The use of the masculine, feminine or neuter gender or the singular or plural form of words herein shall not limit any provision of this Agreement. The use of the terms "including" or "include" shall in all cases herein mean "including, without limitation" or "include, without limitation," respectively. Reference to any Person in, or with respect to, an agreement includes such Person's successors and assigns to the extent such successors and assigns are permitted by the terms of such agreement (including this Agreement), document or instrument means such agreement, document or instrument as amended or modified and in effect from time to time in accordance with the terms thereof and, if applicable, the terms hereof. Reference to any Law means such Law as amended, modified, codified, replaced or re-enacted, in whole or in part, including rules, regulations, enforcement procedures and any interpretations promulgated thereunder, all as in effect on the date hereof. Underscored references to Articles, Sections or Schedules shall refer to those portions of this Agreement. The use of the terms "hereunder," "hereof,"

[Remainder of page intentionally left blank]

IN WITNESS WHEREOF, Harsco, CD&R Investor, the Company and Buyer have caused this Agreement to be executed as of the date first written above by their respective duly authorized representatives.

HARSCO CORPORATION

By: <u>/s/ Patrick K. Decker</u> Name: Patrick K. Decker Title: President and Chief Executive Officer

CD&R BULLSEYE HOLDINGS, L.P.

By: CD&R Investment Associates IX, Ltd., its general partner

By: <u>/s/ Theresa A. Gore</u> Name: Theresa A. Gore Title: Vice President, Treasurer and Secretary

BULLSEYE INVESTORS, INC.

By: <u>/s/ Theresa A. Gore</u> Name: Theresa A. Gore Title: Vice President and Secretary

BULLSEYE, INC.

By: <u>/s/ Theresa A. Gore</u> Name: Theresa A. Gore Title: Vice President and Secretary

EXHIBIT A - LIST OF STOCK SELLERS AND STOCK BUYERS EXHIBIT B - LIST OF ASSET SELLERS AND ASSET BUYERS EXHIBIT C - FORM OF INVESTOR RIGHTS AGREEMENT EXHIBIT D - FORM OF TRANSITION SERVICES AGREEMENT EXHIBIT E - BRAND PURCHASE AGREEMENT EXHIBIT F - PRE-CLOSING TRANSACTIONS EXHIBIT G - CLOSING TRANSACTIONS EXHIBIT H - NET WORKING CAPITAL STATEMENT METHODOLOGY EXHIBIT I - CD&R CONSULTING AGREEMENT

EXHIBIT J - CD&R INDEMNIFICATION AGREEMENT EXHIBIT K - HARSCO CONSULTING AGREEMENT EXHIBIT L - HARSCO INDEMNIFICATION AGREEMENT EXHIBIT M - FORM OF UPPER-TIER PARTNERSHIP AGREEMENT EXHIBIT N - REPATRIATION SCHEDULE

<u>AMENDMENT NO. 1</u> <u>TO AMENDED AND RESTATED FIVE-YEAR CREDIT AGREEMENT</u>

AMENDMENT NO. 1 TO AMENDED AND RESTATED FIVE-YEAR CREDIT AGREEMENT (this "<u>Amendment</u>"), dated as of September 12, 2013, among HARSCO CORPORATION, a Delaware corporation, as Borrower (the "<u>Company</u>"), the Lenders party hereto and CITIBANK, N.A., as Administrative Agent.

WITNESSETH

WHEREAS, the Company, the Lenders and the Administrative Agent have entered into that certain Amended and Restated Five-Year Credit Agreement dated as of March 2, 2012 (as amended, supplemented or otherwise modified through the date hereof, the "Credit Agreement");

WHEREAS, the Company, the Required Lenders and the Administrative Agent have agreed to amend the Credit Agreement as hereinafter set forth on the terms and conditions set forth herein;

NOW THEREFORE, in consideration of the premises and the mutual agreements, representations and warranties set forth herein, and for other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto hereby agree as follows:

SECTION 1. <u>Definitions and References</u>. Capitalized terms not otherwise defined herein shall have the meanings attributed thereto in the Credit Agreement.

SECTION 2. <u>Amendments</u>. Effective as of the First Amendment Effective Date (as defined below), the Credit Agreement is hereby amended as follows:

(a) Section 1.01 of the Credit Agreement is amended by deleting the table in the definition of "Applicable Margin" and substituting in lieu thereof the following table:

	Applicable Margin (basis points per annum)
<u>Category 1</u> BBB+ or higher by S&P Baal or higher by Moody's	137.5
<u>Category 2</u> BBB by S&P Baa2 by Moody's	150.0
<u>Category 3</u> BBB- by S&P Baa3 by Moody's	175.0
<u>Category 4</u> BB+ by S&P Ba1 by Moody's	200.0
<u>Category 5</u> BB or lower by S&P Ba2 or lower by Moody's	250.0

(b) Section 1.01 of the Credit Agreement is amended by deleting the table in the definition of "Commitment Fee Percentage" and substituting in lieu thereof the following table:

	Commitment Fee <u>Percentage</u>
<u>Category 1</u> BBB+ or higher by S&P Baal or higher by Moody's	0.175%
<u>Category 2</u> BBB by S&P Baa2 by Moody's	0.200%
<u>Category 3</u> BBB- by S&P Baa3 by Moody's	0.250%
<u>Category 4</u> BB+ by S&P Ba1 by Moody's	0.350%
<u>Category 5</u> BB or lower by S&P Ba2 or lower by Moody's	0.500%

(c) Section 1.01 of the Credit Agreement is amended by amending and restating in its entirety the definition of "Consolidated EBITDA" as follows:

""<u>Consolidated EBITDA</u>" shall mean, at any date of determination, for the Company and its Subsidiaries on a consolidated basis, an amount equal to Consolidated Net Income for the most recently completed consecutive four fiscal quarters plus (a) the following to the extent deducted in calculating Consolidated Net Income: (i) Consolidated Interest Charges for such period, (ii) the provision for Federal, state, local and foreign income or excise taxes payable by the Company and its Subsidiaries for such period, (iii) depreciation and amortization expense, (iv) losses on sales of assets outside the ordinary course of business and losses from discontinued operations and (v) any other nonrecurring or noncash items for such period minus (b) the following to the extent included in calculating such Consolidated Net Income: (i) any extraordinary income or gains, (ii) gains on sales of assets outside the ordinary course of business and gains from discontinued operations and (iii) any other nonrecurring or non-cash income."

(d) Section 1.01 of the Credit Agreement is amended by amending and restating in its entirety the definition of "Consolidated Interest Charges" as follows:

"<u>Consolidated Interest Charges</u>" shall mean, for the most recently completed consecutive four fiscal quarters, for the Company and its Subsidiaries on a consolidated basis, the sum of (a) all interest, premium payments, debt discount, fees, charges and related expenses of the Company and its Subsidiaries in connection with borrowed money (including capitalized interest) or in connection with the deferred purchase price of assets, in each case to the extent treated as interest in accordance with GAAP, and (b) the portion of rent expense of the Borrower and its Subsidiaries with respect to such period under capital leases that is treated as interest in accordance with GAAP.

(e) Section 1.01 of the Credit Agreement is amended by adding the following new defined terms in proper alphabetical order:

""<u>Anti-Terrorism Law</u>" shall mean any applicable law related to money laundering or financing terrorism including the USA Patriot Act (Title III of Pub. L. 107-56 (signed into law October 26, 2001)), The Currency and Foreign Transactions Reporting Act (also known as the "<u>Bank Secrecy Act</u>", 31 U.S.C. §§ 5311-5330 and 12 U.S.C. §§ 1818(s), 1820(b) and 1951-1959), the Trading With the Enemy Act (50 U.S.C. § 1 et seq., as amended) and Executive Order 13224 (effective September 24, 2001)."

""<u>Blocked Person</u>" shall mean any person that (a) is publicly identified on the most current list of "Specially Designated Nationals and Blocked Persons" published by the Office of Foreign Assets Control of the US Department of the Treasury ("<u>OFAC</u>") or resides, is organized or chartered, or has a place of business in a country or territory subject to OFAC sanctions or embargo programs or (b) is publicly identified as prohibited from doing business with the United States under the International Emergency Economic Powers Act, the Trading With the Enemy Act, or any other applicable law."

""<u>First Amendment</u>" shall mean the Amendment No. 1 to Amended and Restated Five-Year Credit Agreement dated as of September 12, 2013, among the Borrower, the Lenders party thereto, and the Administrative Agent."

""<u>First Amendment Effective Date</u>" shall have the meaning given to such term in the First Amendment."

(f) Section 1.01 of the Credit Agreement is amended by deleting the definitions of "Total Capital" and "Net Worth".

(g) Article III of the Credit Agreement is amended by adding the following new Section 3.20:

SECTION 3.20. <u>Anti-Terrorism Regulations</u>. Neither the Company nor any of its Subsidiaries:

"(a) (i) has violated any Anti-Terrorism Laws or (ii) has engaged in any transaction, investment, undertaking or activity that conceals the identity, source or destination of the proceeds from any category of prohibited offenses designated by the Organization for Economic Co-operation and Development's Financial Action Task Force on Money Laundering;

(b) is a Blocked Person; or

(c) (i) conducts any business or engages in making or receiving any contribution of goods, services or money to or for the benefit of any Blocked Person, (ii) deals in, or otherwise engages in any transaction related to, any property or interests in property blocked pursuant to any Anti-Terrorism Law or (iii) engages in or conspires to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempts to violate, any of the prohibitions set forth in any Anti-Terrorism Law."

(h) Section 6.06 of the Credit Agreement is amended by deleting such Section in its entirety and substituting in lieu thereof the following new Section 6.06:

"SECTION 6.06 <u>Total Debt to Consolidated EBITDA</u>. The Company will not permit the ratio of Total Debt to Consolidated EBITDA at any time on or after the First Amendment Effective Date to exceed the ratio of 3.50 to 1.00."

(i) Article VI of the Credit Agreement is amended by adding the following new Section 6.09:

"SECTION 6.09. <u>Compliance with Anti-Terrorism Regulations</u>. (a) (i) Violate any Anti-Terrorism Laws or (ii) engage in any transaction, investment, undertaking or activity that conceals the identity, source or destination of the proceeds from any category of prohibited offenses designated by the Organization for Economic Co-operation and Development's Financial Action Task Force on Money Laundering;

(b) Become a Blocked Person; or

(c) (i) Conduct any business or engage in making or receiving any contribution of goods, services or money to or for the benefit of any Blocked Person, (ii) deal in, or otherwise engage in any transaction related to, any property or interests in property blocked pursuant to any Anti-Terrorism Law, or (iii) engage in or conspire to engage in any transaction that evades or avoids, or has the purpose of evading or avoiding, or attempt to violate, any of the prohibitions set forth in any Anti-Terrorism Law."

SECTION 3. <u>Representations and Warranties</u>. The Company represents and warrants as of the date hereof and as of the First Amendment Effective Date that:

(a) This Amendment has been duly authorized, executed and delivered by the Company;

(b) Each of this Amendment and the Credit Agreement, as amended hereby, constitutes a legal, valid and binding obligation of the Company, enforceable against the Company in accordance with its terms, except as the enforceability thereof may be limited by (i) applicable bankruptcy, insolvency, reorganization, moratorium, fraudulent transfer or similar laws of general application relating to or affecting the rights and remedies of creditors generally or (ii) general principles of equity;

(c) No authorizations, approvals or consents of, and no filings or registrations with, any Governmental Authority, or any securities exchange, are necessary for the execution, delivery or performance by the Company of this Amendment or for the legality, validity or enforceability hereof, other than authorizations, approvals, consents, and filings and registrations that have already been obtained prior to the date hereof;

(d) The representations and warranties set forth in Article III of the Credit Agreement shall be true and correct on the date hereof and as of the First Amendment Effective Date with the same effect as though made on and as of such date, except to the extent such representations and warranties expressly relate to an earlier date; and

(e) No Event of Default or Default shall have occurred and be continuing.

SECTION 4. <u>Ratification</u>. Except as amended hereby, the Credit Agreement and all other documents executed in connection therewith shall remain in full force and effect. The Credit Agreement, as amended hereby, and all rights and powers created thereby or thereunder and under such other documents are in all respects ratified and confirmed.

SECTION 5. <u>Consent Fee</u>. The Company agrees to pay to each Lender executing and delivering (including by facsimile transmission or electronic mail) this Amendment (such Lender, a "<u>Consenting Lender</u>"), on or before 5:00 P.M., New York time, August 26, 2013, a fee equal to 0.05% of the Commitment of such Lender (the "<u>Consent Fee</u>"). The Company shall pay the Consent Fee to the Administrative Agent (for the benefit of the Consenting Lenders) no later than the First Amendment Effective Date.

SECTION 6. <u>Conditions Precedent</u>. The effective date of this Amendment (the "<u>First Amendment Effective Date</u>") shall be that date when, to the satisfaction of the Administrative Agent, the following conditions shall been satisfied or waived:

(a) The Administrative Agent (or its counsel) shall have received from the Borrower and the Required Lenders either (i) a counterpart of this Amendment signed on behalf of such party or (ii) written evidence satisfactory to the Administrative Agent (which may include facsimile transmission or electronic mail transmission of a signed signature page of this Amendment) that such party has signed a counterpart of this Amendment;

(b) The representations and warranties set forth in Section 3 of this Amendment are true and correct on and as of the First Amendment Effective Date;

(c) No Event of Default or Default shall have occurred and be continuing on the First Amendment Effective Date;

(d) The Administrative Agent shall have received a certificate executed by an authorized officer of the Borrower dated the First Amendment Effective Date confirming compliance with the conditions precedent set forth in paragraphs (b) and (c) of this Section 6;

(e) The Administrative Agent shall have received (i) a certificate of the Secretary or Assistant Secretary of the Company dated the First Amendment Effective Date certifying (A) that attached thereto is a true and complete copy of resolutions adopted by the Board of Directors of the Company authorizing the execution, delivery and performance of this Amendment, and that such resolutions have not been modified, rescinded or amended and are in full force and effect and (B) as to the incumbency and specimen signature of each officer of the Company executing this Amendment or any other document delivered in connection herewith and (ii) a certificate of another officer of the Company as to the incumbency and signature of the Secretary or such Assistant Secretary of the Company executing the certificate pursuant to (i) above;

(f) The Administrative Agent shall have received payment (for the benefit of the Consenting Lenders) of the Consent Fee; and

(g) Each of the Administrative Agent and Citigroup Global Markets Inc., as arranger, shall have received payment or reimbursement of all fees payable to it by the Company on or prior to the First Amendment Effective Date and its reasonable outof-pocket expenses in connection with this Amendment including the reasonable fees, charges and disbursements of counsel for the Administrative Agent, in each case, for which invoices have been received by the Company on or prior to the First Amendment Effective Date.

SECTION 7. Miscellaneous.

(a) The Credit Agreement and this Amendment shall be read, taken and construed as one and the same instrument from and after the Effective Date.

(b) This Amendment shall be governed by, and construed in accordance with, the laws of the State of New York.

(c) Any references in the Credit Agreement to "this Agreement", "hereunder", "herein" or words of like import, and each reference in any other document executed in connection with the Agreement, to "the Agreement", "thereunder", "therein" or words of like import, shall, from and after the First Amendment Effective Date, mean and be a reference to the Agreement as amended hereby.

(d) The Company agrees to reimburse the Administrative Agent for its reasonable out-of-pocket expenses in connection with this Amendment including the reasonable fees, charges and disbursements of counsel for the Administrative Agent, whether or not the First Amendment Effective Date occurs.

(e) This Amendment shall be deemed to be a Loan Document.

(f) This Amendment may be executed in two or more counterparts, each of which shall constitute an original but all of which when taken together shall constitute but one contract, and shall become effective as provided in Section 6. Delivery of an executed signature page of this Amendment by facsimile transmission or electronic mail shall be as effective as delivery of a manually executed counterpart hereof.

[Remainder of page intentionally left blank; Signature pages follow]

IN WITNESS WHEREOF, the parties hereto have caused this Amendment to be executed by their respective officers thereunto duly authorized, as of the date first above written.

HARSCO CORPORATION, as Borrower

By <u>/s/ F. Nicholas Grasberger, III</u> Name: F. Nicholas Grasberger, III Title: Senior Vice President and Chief Financial Officer

CITIBANK, N.A., as Administrative Agent

By <u>/s/ Christopher Hartzell</u> Name: Christopher Hartzell Title: Vice President

CITIBANK, N.A., as Lender

By <u>/s/ Christopher Hartzell</u> Name: Christopher Hartzell Title: Vice President

THE ROYAL BANK OF SCOTLAND PLC, as Lender

By <u>/s/ L. Peter Yetman</u> Name: L. Peter Yetman Title: Director

HSBC Bank USA, N.A., as Lender

By <u>/s/ Tanya Dyke</u> Name: Tanya Dyke Title: VP, Global Relationship Manager

ING BANK N.V., DUBLIN BRANCH, as Lender

By <u>/s/ Pádraig Matthews</u> Name: Pádraig Matthews Title: Vice President

By <u>/s/ Sean Hasett</u> Name: Sean Hasett Title: Director

JPMorgan Chase Bank, N.A., as Lender

By <u>/s/ Deborah R. Winkler</u> Name: Deborah R. Winkler Title: Vice President

LLOYDS TSB BANK PLC, as Lender

By <u>/s/ Karen Weich</u> Name: Karen Weich Title: Vice President - W011

By <u>/s/ Dennis McClellan</u> Name: Dennis McClellan Title: Assistant Vice President - M040

The Bank of Tokyo-Mitsubishi UFJ, Ltd., as Lender

By <u>/s/ George Stoecklein</u> Name: George Stoecklein Title: Director

PNC BANK, NATIONAL ASSOCIATION, as Lender

By <u>/s/ Domenic L. D'Ginto</u> Name: Domenic L. D'Ginto, CFA Title: Senior Vice President

Commerzbank AG, New York Branch, as Lender

By <u>/s/ Patrick Hartweger</u> Name: Patrick Hartweger Title: Managing Director

By <u>/s/ Michael Ravelo</u> Name: Michael Ravelo Title: Director

U.S. Bank National Association, as Lender

By <u>/s/ Mark E. Irey</u> Name: Mark E. Irey Title: Assistant Vice President

Wells Fargo Bank, N.A., as Lender

By <u>/s/ James Travagline</u> Name: James Travagline Title: Director

Manufacturers and Traders Trust Company, as Lender

By <u>/s/ Derek Lynch</u> Name: Derek Lynch Title: Assistant Vice President

THE NORTHERN TRUST COMPANY, as Lender

By <u>/s/ Andrew D. Holtz</u> Name: Andrew D. Holtz Title: Senior Vice President

By <u>/s/ Maria Kolsrud</u> Name: Maria Kolsrud Title: Vice President

By <u>/s/ Anders Abelson</u> Name: Anders Abelson Title: Senior Vice President

HARSCO INFRASTRUCTURE EXECUTIVE RETENTION PLAN

RETENTION AND SEVERANCE AGREEMENT

THIS RETENTION AND SEVERANCE AGREEMENT (the "<u>Agreement</u>") made as of this 27th day of October, 2013, by and between Harsco Corporation (the "<u>Company</u>") and Mark Kimmel (the "<u>Executive</u>") sets forth the terms of the Executive's Retention Bonuses, Transaction Payments and Benefits, and Severance Amount as set forth in the Harsco Infrastructure Executive Retention Plan (the "<u>Plan</u>"), as well as certain other terms and conditions with respect to such benefits. For the purpose of this Agreement, unless otherwise indicated, all capitalized terms in this Agreement shall have the definition ascribed to them in the Plan.

- 1. <u>Retention Bonuses</u>. The Executive is entitled to a Retention Bonus Amount of \$550,000 (which is approximately 123% of the Executive's annual base salary), subject to the following terms and conditions.
 - a. <u>First Retention Bonus</u>. The Executive shall not be entitled to a First Retention Bonus.
 - b. <u>Second Retention Bonus</u>. Provided the Executive remains employed with the Company through the Transaction Date (as defined below), the Executive will receive a Second Retention Bonus of \$550,000, representing 100% of the Retention Bonus Amount, as follows: if the Transaction Date occurs on or before December 31, 2013, the Second Retention Bonus shall be paid to the Executive no later than March 31, 2014; and if the Transaction Date occurs after December 31, 2013, the Second Retention Bonus shall be paid to the Executive no later than March 31, 2014; and if the Transaction Date occurs after December 31, 2013, the Second Retention Bonus shall be paid to the Executive on the earlier of (i) 60 days after the Transaction Date and (ii) June 30, 2014. For purposes of this Agreement, the "<u>Transaction Date</u>" shall mean the date upon which the transaction (the "<u>Transaction</u>") contemplated by the Purchase Agreement, dated September 16, 2013, by and among the Company, certain subsidiaries of the Company, certain subsidiaries of Clayton, Dubilier & Rice, Bullseye Investors, Inc., and CD&R Bullseye Holdings, L.P., as may be amended, is consummated.
 - c. <u>Performance Transaction Bonus</u>. Provided the Executive remains employed with the Company through the Transaction Date, the Executive will receive an additional transaction bonus of \$100,000 on the sixtieth (60) day following the Transaction Date solely in the event that the goals pursuant to the Operating Business Plan for Harsco Infrastructure for 2013, as approved by the CEO, are achieved through the Transaction Date (in the event the Transaction is consummated prior to December 31, 2013) or for 2013 (in the event the Transaction is consummated on or after December 31, 2013). The Executive understands and agrees that the Executive will not be entitled to this additional transaction bonus in the event that either (1) the Transaction is not consummated or (2) the Operating Business Plan for Harsco Infrastructure goals are not achieved through the Transaction Date (in the event the Transaction is consummated prior to December 31, 2013) or for 2013 (in the event the Transaction Date or (2) the Operating Business Plan for Harsco Infrastructure goals are not achieved through the Transaction Date (in the event the Transaction is consummated or or after December 31, 2013).

2. <u>Transaction Payments and Benefits</u>.

If the Transaction is consummated and the Executive remains employed through the Transaction Date, notwithstanding anything to the contrary contained in the Plan or otherwise, the Executive shall be entitled to receive the following rights and benefits with respect to the Executive's incentive compensation and equity awards issued or issuable pursuant to the 1995 Plan (in lieu of any other payments or benefits under the 1995 Plan with respect to the period ending on the Transaction Date).

- a. <u>Annual Incentive Plan</u>.
 - 1. If the Transaction Date occurs prior to December 31, 2013, the Executive shall be entitled to receive in respect of his incentive award pursuant to the 2013 Annual Incentive Plan a payment equal to the Payment Amount for 2013. Any such payment shall be paid to the Executive at such time as awards are generally paid to participants under the 2013 Annual Incentive Plan.
 - 2. If the Transaction Date occurs on or after December 31, 2013, the Executive shall remain entitled to receive any payment that he would otherwise be entitled to receive pursuant to the terms of the 2013 Annual Incentive Plan, notwithstanding that the Executive may not remain employed with the Company after the Transaction Date. Any such payment shall be paid to the Executive at such time as awards are generally paid to participants under the 2013 Annual Incentive Plan.
 - 3. If the Transaction is consummated, the Executive shall not be entitled to any annual incentive payment with respect to any portion of 2014.
 - 4. Definitions.
 - (a) "<u>Modified AIP Bonus Opportunity</u>" shall mean the product achieved by multiplying (i) the Executive's Annual Incentive Plan bonus percentage for the applicable calendar year and (ii) a prorata portion of the Executive's annual base salary for such year, as determined by multiplying the Executive's annual base salary for the year by a fraction, the numerator of which is the number of days the Executive has been employed by the Company during the applicable calendar year through the Transaction Date, and the denominator of which is three hundred and sixty-five (365).
 - (b) "<u>Payment Amount</u>" shall mean, with respect to a calendar year, an amount equal to the product of (i) the Executive's Modified AIP Bonus Opportunity and (ii) the Relative Performance Factor.
 - (c) "<u>Relative Performance Factor</u>" shall be the performance factor achieved for the plan year through the Transaction Date determined by comparing the Harsco Infrastructure Performance Plan ("<u>Performance Plan</u>") operating income with the actual operating income achieved as of the Transaction Date. If actual operating income achieved as of the Transaction Date results, the Relative Performance Factor is the payout level that would have been achieved for achieving the Performance Plan results under the EVA-based plan as of December 31 of the applicable calendar year. Any absolute under- or over-performance of operating income as of the Transaction Date shall be extrapolated until the end of the year and incorporated into the existing EVA-based AIP plan calculation. All other components of the EVA calculation remain the same as per the AIP, such as Capital and Overhead Allocations. Thus, the Relative Performance Factor would be the payout level that would have been achieved based on the Plan Capital and Allocations and revised Operating Income achieved under the EVA-based plan as of December

31 of the applicable calendar year. For the avoidance of doubt, the Performance Plan operating income shall refer to the applicable plan associated with each Participant. An example of the relevant calculations is set forth below:

- Transaction Date: August 31, 2013
- Plan Operating Income YTD August: \$2,000
- Actual Operating Income YTD August: \$2,800
- Plan Operating Income Full Year: \$3,000
- Extrapolated Operating Income for Full Year: (\$2,800-\$2,000)/8*12+\$3,000=\$4,200
- AIP Bonus Calculation based on Plan Capital and Allocations and revised Operating Income of \$4,200.
- 5. <u>Stock Options</u>. Provided the Executive remains employed with the Company through the Transaction Date, a pro-rata portion of the Executive's outstanding stock options (including both nonqualified stock options and incentive stock options) shall vest upon the Transaction Date. With respect to each award of stock options outstanding as of the Transaction Date, the number of stock options to vest shall be determined as follows: the total number of stock options shall be multiplied by a fraction, the numerator of which is the number of days from the date of grant through the Transaction Date and the denominator of which is one thousand and ninety-six (1096). All unvested stock options will be forfeited upon the Transaction Date. In addition, the Executive shall have (i) three (3) months following the Transaction Date to exercise any vested incentive stock options held by the Executive; and (ii) twelve (12) months following the Transaction Date to exercise any vested nonqualified stock options held by the Executive.

b. <u>2011 LTIP Awards</u>.

- <u>Time-Based Awards</u>. Provided the Executive remains employed with the Company through the Transaction Date, the Executive's outstanding time-based awards with respect to the 2011 LTIP to the extent unvested as of the Transaction Date ("<u>2011 Time-Based Awards</u>") shall be forfeited and the Company shall pay to the Executive instead an amount in cash equal to the amount the Executive would have received had all of the 2011 Time-Based Awards vested as of December 31, 2013. Such cash amount shall be paid at such time as the 2011 Time Based Awards are generally paid to plan participants.
- 2. Performance-Based Awards. Provided the Executive remains employed with the Company through the Transaction Date, the Executive's outstanding performance-based award with respect to the 2011 LTIP to the extent unvested as of the Transaction Date ("2011 Performance-Based Award") shall be forfeited and the Company shall pay to the Executive instead an amount in cash equal to the amount, if any, the Executive would have received had a pro rata portion of the 2011 Performance-Based Award vested on the Transaction Date based on actual performance through December 31, 2013 to be determined as follows: the total 2011 Performance-Based Award that would otherwise be paid shall be multiplied by a fraction, the numerator of which is the number of days from January 1, 2011 through the Transaction Date, and the denominator of which is one thousand and ninety-six (1096). Such cash amount, if any, shall be paid in February 2014.
 - 3

2012 LTIP Awards.

c.

- 1. <u>Restricted Stock Units</u>. Provided the Executive remains employed with the Company through the Transaction Date, a pro-rata portion of the Executive's outstanding RSUs in respect of the 2012 LTIP ("<u>2012</u> <u>RSUs</u>") shall vest upon the Transaction Date as follows: the total number of outstanding 2012 RSUs shall be multiplied by a fraction, the numerator of which is the number of days from the grant date of such 2012 RSUs through the date of the Transaction Date, and the denominator of which is one thousand and ninety-five (1095). All other 2012 RSUs held by the Executive at the time of the Transaction Date shall be forfeited upon the Transaction Date.
- 2. <u>SARs</u>. Provided the Executive remains employed with the Company through the Transaction Date, a pro-rata portion of the Executive's outstanding SARs in respect of the 2012 LTIP (<u>"2012 SARs</u>") shall vest upon the Transaction Date as follows: the total number of 2012 SARs that are scheduled to vest at the next vesting date attributable to such award shall be multiplied by a fraction, the numerator of which is the number of days from the immediately preceding vesting date attributable to such award through the Transaction Date and the denominator of which is three hundred and sixty-five (365). All unvested 2012 SARs held by the Executive at the time of the Transaction Date will be forfeited as of the Transaction Date. In addition, the Executive shall have twelve (12) months following the Transaction Date to exercise any vested 2012 SARs held by the Executive.

d. <u>2013 LTIP Awards</u>.

- 1. <u>Restricted Stock Units</u>. Provided the Executive remains employed with the Company through the Transaction Date, a pro-rata portion of the Executive's outstanding RSUs in respect of the 2013 LTIP ("<u>2013</u> <u>RSUs</u>") shall vest upon the Transaction Date as follows: the total number of outstanding 2013 RSUs shall be multiplied by a fraction, the numerator of which is the number of days from the grant date of such 2013 RSUs through the date of the Transaction Date, and the denominator of which is one thousand and ninety-five (1095). All other 2013 RSUs held by the Executive at the time of the Transaction Date shall be forfeited upon the Transaction Date.
- 2. <u>SARs</u>. Provided the Executive remains employed with the Company through the Transaction Date, a pro-rata portion of the Executive's outstanding SARs in respect of the 2013 LTIP ("<u>2013 SARs</u>") shall vest upon the Transaction Date as follows: the total number of 2013 SARs that are scheduled to vest at the next vesting date attributable to such award shall be multiplied by a fraction, the numerator of which is the number of days from the immediately preceding vesting date attributable to such award through the Transaction Date and the denominator of which is three hundred and sixty-five (365). All unvested 2013 SARs held by the Executive at the time of the Transaction Date will be forfeited as of the Transaction Date. In addition, the Executive shall have twelve (12) months following the Transaction Date to exercise any vested 2013 SARs held by the Executive.

3. <u>Severance Amount</u>.

a. Notwithstanding anything in the Plan to the contrary, in the event that the Executive remains employed with the Company through the Transaction Date and the Executive is terminated by the Company without Cause or the Executive terminates the Executive's employment with the Company for Good Reason during the Protected Period, the Executive will be entitled to a series of installment payments to be paid over a period of eighteen (18) months in an aggregate amount equal to the sum of (x) 150% of the Executive's annual base salary in effect

immediately prior to the Transaction Date, and (y) 150% of the Executive's target annual incentive opportunity in effect immediately prior to the Transaction Date under the Annual Incentive Plan, (such amount (x) plus (y), the Executive's "<u>Severance Amount</u>"). The Executive's Severance Amount shall be paid in monthly installments, commencing sixty (60) days after the termination of the Executive's employment with the Company (the "*Employment Termination Date*"). Notwithstanding anything to the contrary herein the obligation to pay the Executive's Severance Amount shall cease as of the end of the month in which the Executive commences full time employment with an employer at a Comparable Position to which he is employed as of the date hereof; for the avoidance of doubt, in such instance, the aggregate amount of the Executive's Severance Amount as set forth herein shall be reduced accordingly. For purposes of this Section 3(a), "<u>Comparable Position</u>" shall mean any position with a target total direction compensation (base salary plus target annual bonus plus target long-term incentives) equal to at least 75% of Executive's target total direct compensation with the Company as of the date hereof.

- b. Notwithstanding anything in the Plan to the contrary, the Executive acknowledges and agrees that this provision only comes into effect following the Transaction Date, and if the Executive does not experience a termination of employment with the Company following the Transaction Date in accordance with the provisions of Section 3(a), the Executive shall not be entitled to any severance payment under the Plan or this Agreement.
- c. Executive hereby consents to notify the Company upon accepting employment with any employer during the period beginning with his termination date and ending on the date that he is no longer entitled to receive the Severance Amount in accordance with Section 3(a).
- d. Notwithstanding anything in the Plan to the contrary, the Protected Period shall be the twelve (12) month period beginning upon the Transaction Date.
- e. In the event that the Executive remains employed with the Company through the Transaction Date and the Executive becomes entitled to receive the Severance Amount accordance with Section 3(a), the Executive will also be entitled to receive (x) a lump sum cash payment equal to \$40,000 to be paid on the sixtieth (60th) day after the Employment Termination Date or shortly thereafter in accordance with normal payroll practices and (y) outplacement services following the Employment Termination Date in an amount not to exceed \$7,500.
- f. Provided the Transaction Date has occurred, the Company intends to terminate the Executive's employment without Cause on or prior to February 28, 2014.
- 4. <u>Release Agreement</u>.
 - a. Payment to the Executive of the payments and benefits as set forth in Section 1(b) and, if applicable, Section 2 shall be conditioned upon the Executive's execution, delivery and non-revocation of a valid and enforceable general release of claims against the Company Released Parties (as defined therein) in the form of release agreement substantially in the form attached hereto as Exhibit A taking into account applicable local law (the "<u>Release Agreement</u>"), within sixty (60) days after the date of the Transaction Date.
 - b. Payment to the Executive of the payments and benefits as set forth in Section 3 shall be conditioned upon the Executive's execution, delivery and non-revocation of a Release Agreement against the Company Released Parties within sixty (60) days after the date of the Executive's termination of employment with the Company.
- 5. <u>Successors; Binding Agreement</u>. The Plan and this Agreement shall be binding upon and shall inure to the benefit of the Company and its successors and assigns, and the Company shall have the right to assign its obligations under the Plan and this Agreement, in whole or in part, to any Successor Employer or its affiliates, in which case, the Company shall no further liability with respect to the

assigned obligations (but shall remain entitled to its rights or benefits) pursuant to the Plan and this Agreement.

- 6. <u>Non-exclusivity of Rights</u>. Nothing in this Agreement shall prevent or limit the Executive's continuing or future participation in any benefit, bonus, incentive or other plan or program provided by the Company (except for any severance or termination policies, plans, programs or practices during the Protected Period and as otherwise set forth in Section 2) and for which the Executive may qualify, nor shall anything herein limit or reduce such rights as the Executive may have under any other agreements with the Company (except for any severance or termination agreement during the Protected Period or with respect to the 1995 Plan as set forth in Section 2). Amounts which are vested benefits or which the Executive is otherwise entitled to receive under any plan or program of the Company shall be payable in accordance with such plan or program, except as explicitly modified by the Plan.
- 7. <u>No Guaranteed Employment</u>. The Executive and the Company acknowledge that, except as may otherwise be provided under any other written agreement between the Executive and the Company, the employment of the Executive by the Company or is "at will" and may be terminated by either the Executive or the Company at any time.
- 8. Unauthorized Disclosure. The Executive agrees and understands that in the Executive's position with the Company, the Executive has been and will be exposed to and has and will receive information relating to the confidential affairs of the Company and its affiliates, including, without limitation, technical information, intellectual property, business and marketing plans, strategies, customer information, software, other information concerning the products, promotions, development, financing, expansion plans, business policies and practices of the Company and its affiliates and other forms of information considered by the Company and its affiliates to be confidential or in the nature of trade secrets (including, without limitation, ideas, research and development, know-how, formulas, technical data, designs, drawings, specifications, customer and supplier lists, pricing and cost information and business and marketing plans and proposals) (collectively, the "<u>Confidential Information</u>"). Confidential Information shall not include information that is generally known to the public or within the relevant trade or industry other than due to the Executive's violation of this Section 8 or disclosure by a third party who is known by the Executive to owe the Company an obligation of confidentiality with respect to such information. The Executive agrees that at all times during the Executive's employment with the Company and thereafter, the Executive shall not disclose such Confidential Information, either directly or indirectly, to any Person without the prior written consent of the Company and shall not use or attempt to use any such information in any manner other than in connection with the Executive's employment with the Company, unless required by law to disclose such information, in which case the Executive shall provide the Company, with written notice of such requirement as far in advance of such anticipated disclosure as possible. This confidentiality covenant has no temporal, geographical or territorial restriction. Upon termination of the Executive's employment with the Company, the Executive shall promptly supply to the Company, all property, keys, notes, memoranda, writings, lists, files, reports, customer lists, correspondence, tapes, disks, cards, surveys, maps, logs, machines, technical data and any other tangible product or document which has been produced by, received by or otherwise submitted to the Executive during or prior to the Executive's employment with the Company, and any copies thereof in his (or capable of being reduced to his) possession.
- 9. <u>Non-Disparagement</u>. The Executive agrees that he will not make or publish any statement critical of the Company, its affiliates and its executive officers and directors, or in any way adversely affecting

or otherwise maligning the business or reputation of any member of the Company, its affiliates and its officers, directors and employees. The Company agrees that its executive officers and directors will not make or publish any statement critical of the Executive, or in any way adversely affecting or otherwise maligning the business or reputation of the Executive. This paragraph shall not prohibit the Executive nor the Company or any of its/his representatives from taking action to enforce their rights under this Agreement, making disclosures required by law or the rules and regulations of any securities exchange upon which the securities of the Company are listed or from cooperating with or responding to any governmental, administrative or judicial investigations or proceedings.

- 10. <u>Non-Solicitation of Employees</u>. During the Executive's employment with the Company and during a eighteen (18) month period following the Executive's termination of employment with the Company for any reason, the Executive shall not directly or indirectly solicit (or assist any Person to solicit) for employment any person who is, or within twelve (12) months prior to the date of such solicitation was, an employee of the Company or any of its affiliates.
- 11. <u>Compliance with Company Policies</u>. The Executive agrees to continue to comply with his obligations under any Company policy or agreement, including, but not limiting to, such restrictions relating to non-competition, non-disparagement, non-solicitation, confidentiality, hedging, clawback, ownership guideline policies and any other contractual provisions with the Company to which he is now or may become subject prior to the Employment Termination Date.
- 12. <u>Consent</u>. By executing this Agreement, the Executive hereby approves and consents to the terms of the Agreement and the treatment of the Executive's incentive and equity awards in accordance with Section 2 of this Agreement.
- 13. <u>Miscellaneous</u>. No provision of this Agreement may be modified, waived or discharged unless such waiver, modification or discharge is agreed to in writing and signed by the Executive and the Company. No waiver by either party hereto at any time of any breach by the other party hereto of, or compliance with, any condition or provision of this Agreement to be performed by such other party shall be deemed a waiver of similar or dissimilar provisions or conditions at the same or at any prior or subsequent time. No agreement or representations, oral or otherwise, express or implied, with respect to the subject matter hereof have been made by either party which are not expressly set forth in this Agreement.
- 14. <u>Section 409A Considerations</u>. All of the payments and benefits payable pursuant to this Agreement are intended to comply with, or be exempt from, Section 409A to the extent the requirements of Section 409A are applicable hereto, and the provisions of this Agreement shall be construed and administered in a manner consistent with that intention. Without limiting the foregoing, the terms "terminates" or "termination of employment" or similar terms used in the Plan shall be interpreted to mean to occur when a "separation of service" occurs as defined under Section 409A. For purposes of Section 409A, the Executive's right to receive any installment payments pursuant to this Agreement shall be treated as a right to receive a series of separate and distinct payments. Notwithstanding anything to contrary herein, if the Executive is a "specified employee" for purposes of Section 409A at the time of his termination of employment, any Severance Amounts required to be made pursuant to Section 3 which are considered "nonqualified deferred compensation" subject to Section 409A shall commence on the day after the first to occur of (i) the day which is six (6) months from the Employment Termination Date, or (ii) the date of the Executive's death. However, the Executive understands and agrees that the Company is not making any representations regarding any tax treatment of amounts paid pursuant to this Agreement.
 - 7

- 15. <u>Governing Law</u>. This Agreement shall be governed by and construed and enforced in accordance with the laws of the State of Delaware without giving effect to the conflict of laws principles thereof. Any action brought by any party to this Agreement shall be brought and maintained in a court of competent jurisdiction in the State of Delaware.
- 16. <u>Severability</u>. The provisions of this Agreement shall be deemed severable and the invalidity or unenforceability of any provision shall not affect the validity or enforceability of the other provisions hereof.
- 17. <u>Entire Agreement</u>. This Agreement (together with the Plan) constitutes the entire agreement between the parties hereto and supersedes all prior agreements, if any, understandings and arrangements, oral or written, between the parties hereto with respect to the subject matter hereof. In the event of a conflict between the Plan and this Agreement, the Plan shall control except as explicitly set forth herein.
- 18. <u>Counterparts</u>. This Agreement may be executed in any number of counterparts, each of which shall be deemed an original but all such counterparts shall together constitute one and the same instrument.

IN WITNESS WHEREOF, the Company has caused this Agreement to be executed by its duly authorized officer and the Executive has executed this Agreement as of the day and year first above written.

HARSCO CORPORATION

By: <u>/s/ A. Verona Dorch</u> Name: A. Verona Dorch Title: General Counsel

EXECUTIVE

<u>/s/ Mark E. Kimmel</u> Mark Kimmel

Release Agreement

This Release Agreement (this "<u>Release Agreement</u>") dated as of [], 201[] (the "<u>Effective Date</u>") is entered into by Mark Kimmel (the "<u>Executive</u>") for the benefit of Harsco Corporation, a Delaware corporation (the "<u>Company</u>").

In consideration of the payments described in Section[s] [1(b),][2,] [and 3]] of the Retention and Severance Agreement (the "<u>Retention and Severance Agreement</u>") dated as of October [], 2013 by and between Harsco Corporation and the Executive and other good and valuable consideration, which are given to the Executive specifically in exchange for this release as a result of negotiations between the Company and the Executive, the Executive, on behalf of himself, his heirs, successors and assigns, hereby releases and discharges the Company, its subsidiaries and affiliates, its and their employee benefit plans, its and their current or former directors, officers, executives, agents, insurers, attorneys, consultants, and auditors, and any and each of their successors and assigns and predecessors ("<u>Company Released Parties</u>"), from any and all claims, charges, causes of action and damages (including attorneys' fees and costs actually incurred) ("<u>Claims</u>"), known and unknown, including those Claims related in any way to the Executive's employment with the Company or any of its subsidiaries, or the termination of his employment relationship or positions as an officer of the Company, arising on or prior to the Effective Date. The waivers in this Release Agreement shall not waive the Executive's rights respecting (i) the Company's obligations under the Retention and Severance Agreement; (ii) the Executive's right to receive COBRA continuation coverage in accordance with applicable law; and (iii) claims for benefits under any health, disability, retirement, life insurance or other, similar employee benefit plan (within the meaning of Section 3(3) of the Employee Retirement Income Security Act of 1974, as amended ("<u>ERISA</u>")) of the Company.

For the purposes of implementing a full and complete release and discharge of the Company Released Parties, the Executive expressly acknowledges that this Release Agreement is intended to include in its affect, without limitation, all Claims which he does not know or suspect to exist in his favor at the time he signs this Release Agreement, and that this Release Agreement is intended to fully and finally resolve any such Claim or Claims.

This release specifically includes, but is not limited to, rights and claims under the local, state or federal laws prohibiting discrimination in employment, including the Americans with Disabilities Act, the Age Discrimination in Employment Act, the Family and Medical Leave Act, the Pennsylvania Human Relations Act, ERISA (except as otherwise stated herein), the Executive protection provisions of the Federal Deposit Insurance Act (12 U.S.C. § 1831j), Title VII of the Civil Rights Act of 1964, the Sarbanes- Oxley Act of 2002, as well as any other state or federal laws or common law theories relating to discrimination in employment, the termination of employment, or personal injury, including without limitation all claims for wrongful discharge, breach of contract, breach of an implied covenant of good faith and fair dealing, intentional infliction of emotional distress, tortious interference with contract or prospective economic advantage, defamation, loss of consortium, infliction of emotional distress; or any claim for any compensation, including, but not limited to additional compensation, back pay, front pay, or benefits (other than as provided for in the Retention and Severance Agreement), severance, reinstatement, or any other form of economic loss; and all claims for personal injury, including, but not limited to: mental anguish, emotional distress, pain and suffering, humiliation, and damage to name or reputation; and all claims for liquidated damages and punitive damages and all claims for counsel fees and costs.

The Executive represents that he has not filed any Claim that was released in this Release Agreement against the Company Released Parties with any court or government agency, and that he will not, to the extent allowed by applicable law, do so at any time in the future; provided, however, that the covenants contained in herein will not prevent the Executive from filing a claim to enforce the terms of the Retention and Severance Agreement or any other Claim not released hereunder. If any government agency brings any claim or conducts any investigation against the Company, nothing in this Release Agreement shall prevent the Executive from cooperating in such proceedings.

Executive represents that he has read carefully and fully understands the terms of this Release Agreement, and that Executive has been advised to consult with an attorney and has had the opportunity to consult with an attorney prior to signing this Release Agreement. Executive acknowledges that he is executing this Agreement voluntarily and knowingly and that he has not relied on any representations, promises or agreements of any kind made to Executive in connection with Executive's decision to accept the terms of this Agreement, other than those set forth in the Retention and Severance Agreement and this Release Agreement. Executive acknowledges that Executive has been given at least forty-five (45) days to consider whether Executive wants to sign this Release Agreement and that the Age Discrimination in Employment Act gives Executive the right to revoke this Release Agreement within seven (7) days after it is signed, and Executive understands that he will not receive any payments not otherwise due him under this Agreement until such seven (7) day revocation period (the "<u>Revocation Period</u>") has passed and then, only if Executive has not revoked this Agreement. Executive may revoke the Agreement during the Revocation Period by providing written notice of the revocation to Verona Dorch at the Company's corporate office at 350 Poplar Church Road, Camp Hill, PA 17011. Upon such revocation, this Agreement and Section[s] [1(b),][2,] [and 3]] the Retention and Severance Agreement shall be null and void and of no further force and effect. To the extent Executive has executed this Agreement within less than forty-five (45) days after its delivery to Executive, Executive hereby acknowledges that his decision to execute this Release Agreement prior to the expiration of such forty-five (45) day period was entirely voluntary.

The Executive understands and acknowledges the significance of this Release Agreement and acknowledges that this Release Agreement is voluntary and has not been given as a result of any coercion. The Executive also acknowledges that he has been given full opportunity to review and negotiate the Retention and Severance Agreement and this Release Agreement, that he has been specifically advised to consult with legal counsel prior to signing it, that he has in fact carefully reviewed it with his attorney before signing it, and that he executes this Release Agreement only after full reflection and analysis.

The following information is required by the Older Workers Benefit Protection Act:

[The class, unit, or group of individuals covered by the program includes all employees in the Harsco Infrastructure business. Certain employees in the Harsco Infrastructure business as of the Transaction Date (as defined in the Retention and Severance Agreement) as listed below are eligible to participate in the Harsco Infrastructure Executive Retention Plan.

The following is a listing of the ages and job titles of employees who were and were not selected for termination in connection with an employment transfer to the Successor Employer (as defined in the Retention and Severance Agreement) and offered consideration for signing the waiver. Except for those employees in the Harsco Infrastructure business, no other employee is eligible or offered consideration in exchange for signing the waiver:]

Other than as to rights, claims and causes of action arising under the ADEA, the release of claims set forth in this Release Agreement shall be immediately effective upon execution by the Executive.

The Executive acknowledges and agrees that he has not, with respect to any transaction or state of facts existing prior to the date hereof, filed any complaints, charges or lawsuits against any Company Released Party with any governmental agency, court or tribunal.

The Executive acknowledges that the separation payments he is receiving under Section[s] [1(b),][2,] [and 3]] under the Retention and Severance Agreement in connection with the release of claims set forth in this Release Agreement and his obligations under this Release Agreement are in addition to anything of value to which the Executive is entitled from the Company.

Each provision hereof is severable from this Release Agreement, and if one or more provisions hereof are declared invalid, the remaining provisions shall nevertheless remain in full force and effect. If any provision of this Release Agreement is so broad, in scope, or duration or otherwise, as to be unenforceable, such provision shall be interpreted to be only so broad as is enforceable.

This Release Agreement constitutes the complete agreement in respect of the subject matter hereof and shall supersede all prior agreements between the Company and the Executive in respect of the subject matter hereof except to the extent set forth herein.

The failure to enforce at any time any of the provisions of this Release Agreement or to require at any time performance by the Executive of any of the provisions hereof shall in no way be construed to be a waiver of such provisions or to affect the validity of this Release Agreement, or any part hereof, or the right of the Company [the Successor Employer] thereafter to enforce each and every such provision in accordance with the terms of this Release Agreement.

This Release Agreement shall be binding upon any and all successors and assigns of the Executive and the Company.

Except for issues or matters as to which federal law is applicable, this Release Agreement shall be governed by and construed and enforced in accordance with the laws of the State of Delaware without giving effect to the conflicts of law principles thereof.

IN WITNESS WHEREOF, the Executive has executed this Agreement as of the date set forth above.

Executive

HARSCO CORPORATION CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, Patrick K. Decker, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Harsco Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

November 7, 2013

/s/ Patrick K. Decker Patrick K. Decker Chief Executive Officer

HARSCO CORPORATION CERTIFICATION PURSUANT TO RULE 13a-14(a) OR 15d-14(a) AS ADOPTED PURSUANT TO SECTION 302 OF THE SARBANES-OXLEY ACT OF 2002

I, F. Nicholas Grasberger, III, certify that:

- 1. I have reviewed this quarterly report on Form 10-Q of Harsco Corporation;
- 2. Based on my knowledge, this report does not contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements made, in light of the circumstances under which such statements were made, not misleading with respect to the period covered by this report;
- 3. Based on my knowledge, the financial statements, and other financial information included in this report, fairly present in all material respects the financial condition, results of operations and cash flows of the registrant as of, and for, the periods presented in this report;
- 4. The registrant's other certifying officer(s) and I are responsible for establishing and maintaining disclosure controls and procedures (as defined in Exchange Act Rules 13a-15(e) and 15d-15(e)) and internal control over financial reporting (as defined in Exchange Act Rules 13a-15(f) and 15d-15(f)) for the registrant and have:
 - (a) Designed such disclosure controls and procedures, or caused such disclosure controls and procedures to be designed under our supervision, to ensure that material information relating to the registrant, including its consolidated subsidiaries, is made known to us by others within those entities, particularly during the period in which this report is being prepared;
 - (b) Designed such internal control over financial reporting, or caused such internal control over financial reporting to be designed under our supervision, to provide reasonable assurance regarding the reliability of financial reporting and the preparation of financial statements for external purposes in accordance with generally accepted accounting principles;
 - (c) Evaluated the effectiveness of the registrant's disclosure controls and procedures and presented in this report our conclusions about the effectiveness of the disclosure controls and procedures, as of the end of the period covered by this report based on such evaluation; and
 - (d) Disclosed in this report any change in the registrant's internal control over financial reporting that occurred during the registrant's most recent fiscal quarter (the registrant's fourth fiscal quarter in the case of an annual report) that has materially affected, or is reasonably likely to materially affect, the registrant's internal control over financial reporting; and
- 5. The registrant's other certifying officer(s) and I have disclosed, based on our most recent evaluation of internal control over financial reporting, to the registrant's auditors and the audit committee of the registrant's board of directors (or persons performing the equivalent functions):
 - (a) All significant deficiencies and material weaknesses in the design or operation of internal control over financial reporting which are reasonably likely to adversely affect the registrant's ability to record, process, summarize and report financial information; and
 - (b) Any fraud, whether or not material, that involves management or other employees who have a significant role in the registrant's internal control over financial reporting.

November 7, 2013

/s/ F. Nicholas Grasberger, III

F. Nicholas Grasberger, III Chief Financial Officer

HARSCO CORPORATION CERTIFICATIONS PURSUANT TO 18 U.S.C. SECTION 1350, AS ADOPTED PURSUANT TO SECTION 906 OF THE SARBANES-OXLEY ACT OF 2002

In connection with the Annual Report of Harsco Corporation (the "Company") on Form 10-K for the period ending September 30, 2013, as filed with the Securities and Exchange Commission on the date hereof (the "Report"), we certify, pursuant to 18 U.S.C. Section 1350, as adopted pursuant to Section 906 of the Sarbanes-Oxley Act of 2002, that, to the best of our knowledge:

(1) The Report fully complies with the requirements of Section 13(a) or 15(d) of the Securities Exchange Act of 1934; and

(2) The information contained in the Report fairly presents, in all material respects, the financial condition and results of operations of the Company.

November 7, 2013

/s/ PATRICK K. DECKER

Patrick K. Decker Chief Executive Officer

/s/ F. NICHOLAS GRASBERGER, III

F. Nicholas Grasberger, III Chief Financial Officer

A signed original of this written statement required by Section 906 has been provided to Harsco Corporation and will be retained by Harsco Corporation and furnished to the Securities and Exchange Commission or its staff upon request.